

Studies on Use of Bacteriophages in Treating Antibiotic-Resistant Infections of Diabetic Foot Ulcers

by

Legesse Garedew Kifelew (DVM, MSc)

Thesis

Submitted to Flinders University for the degree of

Doctor of Philosophy

College of Science and Engineering

June 2020

Table of contents

List of tables	viii
List of figures	ix
Abbreviations	x
Declaration	xii
Acknowledgments	xiii
Dedication	XV
Summary	xvi
Thesis structure	xviii
Chapter 1: General introduction	2
1. Introduction	2
1.1 Diabetic foot ulcer infection	2
1.2 <i>S. aureus</i>	4
1.3 P. aeruginosa	5
1.4 Bacterial biofilms	8
1.5 Treatment of DFU	9
1.6 Animal models for DFU infection and treatment	10
1.7 Bacteriophages	12
1.7.1 Lytic phages	14
1.7.2 Lysogenic phages	16
1.8 The role of phages on human microbial communities	16
1.9 The use of phages as therapeutic agents	
1.10 Effect of phages on biofilms	19
1.11 Features of phage therapy	21
1.12 Significance of the study	23

1.13 Objectives of the study	23
Chapter 2: Effect of Phage Cocktail AB-SA01 on Diabetic Foo Clinical <i>Staphylococcus aureus</i> Isolates and Biofilms	ot Ulcer 26
Abstract	
2.1 Introduction	
2.1.1 S. aureus in DFU	27
2.1.3 Identification of S. aureus	28
2.1.4 Antibiotic susceptibility testing of <i>S. aureus</i> isolates	28
2.1.5 Phages as an alternative treatment for <i>S. aureus</i> infection	29
2.1.6 Phage cocktail to treat <i>S. aureus</i> infection	30
2.2 Materials and methods	31
2.2.1 <i>S. aureus</i> isolates	31
2.2.2 Identification of <i>S. aureus</i> isolates	31
2.2.4 Antibiotic susceptibility of <i>S. aureus</i> isolates	32
2.2.5 Determination of phage titer	33
2.2.6 Phage efficacy and host range test	33
2.2.7 Measurement of the effect of phage cocktail on S. aureus biofilm	34
2.2.8 Measurement of the effect of phage cocktail on S. aureus isolates in biofil	ms36
2.2.9 Data management and statistical analysis	37
2.3 Results	
2.3.1 Isolation of <i>S. aureus</i> isolates using culture	
2.3.2 Identification of S. aureus isolates using MALDI-TOF MS	
2.3.3 Identification of <i>S. aureus</i> using latex agglutination test	39
2.3.4 Antibiotic susceptibility of <i>S. aureus</i> isolates using VITEK [®] 2 test	39
2.3.5 Determination of phage titer	41
2.3.6 Efficacy of phages on <i>S. aureus</i> isolates	42

2.3.7 Evaluation of the effect of AB-SA01 on S. aureus isolates biofilm	43
2.4 Discussion	47
Chapter 3: Effect of Phage Cocktail AB-PA01 on <i>Pseudomonas</i> Diabetic Foot Ulcer Clinical Isolates and Biofilms	aeruginosa 54
Abstract	54
3.1 Introduction	55
3.1.1 <i>P. aeruginosa</i> in DFUs	55
3.1.2 Identification of <i>P. aeruginosa</i>	56
3.1.3 Antimicrobial susceptibility of <i>P. aeruginosa</i>	56
3.1.4 Phages as an alternative treatment for <i>P. aeruginosa</i>	58
3.1.5 Phage cocktail to treat <i>P. aeruginosa</i> infections	58
3.2 Materials and methods	60
3.2.1 Identification of <i>P. aeruginosa</i> isolates	60
3.2.2 Antibiotic susceptibility profile of <i>P. aeruginosa</i> isolates	61
3.2.3 Phage titer	61
3.2.4 Phage efficacy and host range test	62
3.2.5 Effect of AB-PA01 on <i>P. aeruginosa</i> biofilms	63
3.2.6 Data management and statistical analysis	65
3.3 Results	66
3.3.1 Isolation of <i>P. aeruginosa</i> using culture	66
3.3.2 Identification of <i>P. aeruginosa</i> using MALDI-TOF MS	66
3.3.3 Antimicrobial susceptibility of <i>P. aeruginosa</i> using VITEK [®] 2	67
3.3.4 Titer of the phage cocktail AB-PA01 and its components	67
3.3.5 Lytic efficacy and host range of phages on spot test	68
3.3.6 Effect of AB-PA01 on biofilms of <i>P. aeruginosa</i> isolates	70
3.4 Discussion	73

Abstract	•••••••••••••••••••••••••••••••••••••••
4.1 Introduction	
4.1.1 Antibiotic-resistant bacterial infection	
4.1.2 Polymicrobial wound infections	
4.1.3 Polymicrobial nature of DFU infections	
4.1.4 Role of <i>S. aureus</i> in DFU infection	
4.1.5 Role of <i>P. aeruginosa</i> in DFU infection	
4.1.6 S. aureus and P. aeruginosa co-infection in DFU	
4.1.7 S. aureus and P. aeruginosa co-culture	
4.1.8 Effect of phages on mixed-species biofilms	
4.2 Materials and methods	
4.2.1 Bacteria	
4.2.2 Phage cocktails	
4.2.3 Bacterial typing and antibiotic susceptibility testing	
4.2.4 P. aeruginosa-S. aureus co-culture test	
4.2.5 Phage treatment of fluorescent mixed-species planktoni	ic culture
4.2.6 Phage treatment of clinical isolates mixed-species planl	ktonic culture
4.2.7 Phage treatment of <i>in vitro</i> mixed-species biofilm	
4.2.8 Colony count	
4.2.9 Data management and statistical analysis	
4.3 Results	
4.3.1 Colony count result of co-culture tests	
4.3.2 Phage cocktail treatment on fluorescent mixed-species	planktonic culture
4.3.3 Efficacy of phage cocktails on laboratory and clinical is planktonic culture	solates mixed-species

4.3.4 Efficacy of phage cocktails on mixed-species biofilms	103
4.4 Discussion	
Chapter 5: Efficacy of Phage Cocktail Therapy in Diabetic Infections Caused by Multidrug-Resistant <i>Staphylococcus au</i>	Mouse Wound breus115
Abstract	115
5.1 Introduction	116
5.1.1 Diabetic foot ulcers	116
5.1.2 Antibiotic-resistant S. aureus in DFU	116
5.1.3 Phage therapy in DFU	117
5.1.4 Mouse skin wound model	119
5.2 Materials and methods	120
5.2.1 Bacterial isolates	120
5.2.2 Phage cocktail	120
5.2.3 Mice management	120
5.2.4 Induction of diabetes in mice	121
5.2.5 Insulin treatment of diabetic mice	123
5.2.6 Selection of <i>S. aureus</i> isolates for mice skin wound infections	123
5.2.7 Preparation of bacteria for infection	123
5.2.8 Infection of mouse wound	124
5.2.9 Quantification of bacteria in wound swabs	125
5.2.10 Excisional wound infliction	125
5.2.11 Preparation of an isolate for mouse wound infection and treatme	ent experiment 126
5.2.12 Vancomycin treatment	127
5.2.13 Phage treatment	127
5.2.14 Assessment of treatment effect	128
5.2.15 Data management and statistical analysis	

5.3 Results
5.3.1 Maintenance and clinical monitoring of mice
5.3.2 Infection establishment of <i>S. aureus</i> isolates on mice wound129
5.3.3 Evaluation of bacterial load during infection establishment study130
5.3.4 Assessment of vancomycin toxicity in mice130
5.3.5 Induction of diabetes130
5.3.6 Effect of insulin treatment on hyperglycaemic condition and body weight of mice
5.3.7 Result of splint excisional wound infliction
5.3.8 Result of inoculation of MDR S. aureus into diabetic mice splint wound133
5.3.9 Effect of AB-SA01 on MDR S. aureus infection in diabetic mice wounds134
5.3.10 Effect of AB-SA01 on diabetic mice wound healing infected with MDR <i>S. aureus</i>
5.3.10 Clinical assessment of diabetic mice infected with MDR S. aureus and treated 138
5.3.11 Mice mortality and post-mortem findings138
5.4 Discussion
Chapter 6: General Discussion146
6.1 Discussion
6.2 Directions for future research157
7. Appendices161
8. References

List of tables

able 1. 1 ICTV classification of phages13	
able 4. 1 Bacteria cell count (log ₁₀ (CFU/ml)) of <i>S. aureus</i> and <i>P. aeruginosa</i> in mixed-species planktonic culture after 24 hours phage cocktails, gentamicin, or PBS treatment	
able 4. 2 Bacteria count (log ₁₀ (CFU/ml)) of <i>S. aureus</i> and <i>P. aeruginosa</i> in mixed-species biofilms after 24 hours phage cocktails, tetracycline, and PBS treatment103	
able 5. 1 Bacterial cell count of <i>S. aureus</i> from mice wound infection130	
able 5. 2 Effect of insulin on BGL, body weight, and mortality of mice	
able 5. 3 Bacterial loads (in log ₁₀ CFU/swab) in diabetic mice wounds infected with MDR <i>S. aureus</i> isolate	
SA63-2498 and treated with phage cocktail AB-SA01 or controls	

List of figures

Figure 1. 1 Pathogenesis of DFU infection	2
Figure 1. 2 Characteristics of <i>S. aureus</i> colonies and cells.	4
Figure 1. 3 Characteristics of <i>P. aeruginosa</i> colonies and cells	6
Figure 1. 4 Morphology of tailed bacteriophage	14
Figure 1. 5 Lytic and lysogenic life cycle of phages	15

Figure 2. 1 Tailed phages	30
Figure 2. 2 Yellow S. aureus colonies and zone around the colonies on MSA	38
Figure 2. 3 Positive latex agglutination test to identify <i>S. aureus</i>	39
Figure 2. 4 Antibiotic susceptibility profile of <i>S. aureus</i> isolate used in this study	40
Figure 2. 5 Category of S. aureus due to their antibiotic susceptibility profile	40
Figure 2. 6 Plaques of AB-SA01 on <i>S. aureus</i> RN4220	41
Figure 2. 7 Lytic efficacy and host range of AB-SA01 and its components on S. aureus isolates	42
Figure 2. 8 Lytic activity of AB-SA01 on <i>S. aureus</i> isolates	43
Figure 2. 9 Quantification of biofilm	44
Figure 2. 10 Effect of AB-SA01 on <i>S. aureus</i> isolates biofilm	45
Figure 2. 11 Comparison of biofilm biomass reduction between AB-SA01 and tetracycline (TTC) treatme	ents.
	46

Figure 3. 1 Antibiotic susceptibility of <i>P. aeruginosa</i> isolates collected from DFU patients	67
Figure 3. 2 PFU of AB-PA01 on P. aeruginosa PA01 during evaluation phage titer	68
Figure 3. 3 Lytic activity of AB-PA01 on <i>P. aeruginosa</i> isolates	69
Figure 3. 4 Phage susceptibility of <i>P. aeruginosa</i> isolates collected from DFU patients	70
Figure 3. 5 Biofilm formation capability of <i>P. aeruginosa</i> isolates from DFU patients	. 70
Figure 3. 6 Effect of AB-PA01 on P. aeruginosa isolates biofilm biomass reduction	71
Figure 3. 7 Comparison of the effect of AB-PA01 and tetracycline (TTC) treatment on biofilm biomass	
reduction of <i>P. aeruginosa</i> isolates	72
Figure 3. 8 Comparison of the effect of AB-PA01 and gentamicin treatment on P. aeruginosa isolates biofil	m
biomass reduction	72

Figure 4. 1 Fluorescence strain under confocal microscopy	96
Figure 4. 2 PBS-treated mCherry S. aureus KUB7 single-species culture	. 98

Figure 5. 1 Splint excisional wound infliction and wound splint silicone sheet application	133
Figure 5. 2 Effect of phage treatment on MDR S. aureus bacterial load compared to control and IP	
vancomycin treatment	134
Figure 5. 3 Diabetic mice wound appearance at day 10	136
Figure 5. 4 Effect of phage AB-SA01 on wound healing	137

Abbreviations

μg	Microgram		
μl	Microliter		
ATCC	American type culture collection		
Вар	Biofilm associated protein		
BGL	Blood glucose level		
BSA	Bovine serum albumin		
CFU	Colony forming unit		
cGMP	Current good manufacturing product		
CRISPR-cas	Clustered regularly interspaced short palindromic repeats associated system		
CV	Crystal violate		
DFU	Diabetic foot ulcer		
DNA	Deoxyribonucleic acid		
EPS	Extracellular polysaccharide substance		
GFP	Green fluorescence protein		
HCCA	Alpha-cyano-4-hydroxy cinnamic acid		
HQNO	2-heptyl-4-hydroxyquinoline N-oxide		
ICTV	International committee for taxonomy of viruses		
IP	Intraperitoneal		
IU	International unit		
L	Littre		
LB	Luria-Bertani		
m/z	Mass charge ratio		
MALDI-TOF	MS Matrix assisted laser desorption/ionization time of flight mass		
	spectrometry		
MDR	Multidrug resistant		
MIC	Minimum inhibitory concentration		
ml	milliliter		
mM	Millimolar		
mmol	Millimole		
MOI	Multiplicity of infection		
MRSA	Methicillin resistant S. aureus		

MSA	Mannitol salt agar
MSSA	Methicillin susceptible S. aureus
NCCLS	National committee for clinical laboratory standards
°C	Degree centigrade
OD	Optical density
P. aeruginosa	Pseudomonas aeruginosa
PBS	Phosphate buffered saline
PFU	Plaque forming unit
PVL	Panton-Valentine leucocidin
RFU	Relative fluorescence unit
RM	Restriction-modification
RNA	Ribonucleic acid
rpm	revolutions per minute
S. aureus	Staphylococcus aureus
SCV	Small colony variant
Sie	Super-infection exclusion
STZ	Streptozotocin
TSA	Tryptic soy agar
TSB	Tryptic soy broth
TTC	Tetracycline

Declaration

I certify that this thesis does not incorporate without acknowledgment any material previously submitted for a degree or diploma in any university; and to the best of my knowledge and belief it does not contain any material previously published or written by another person except where due reference is made in the text.

Legesse Garedew Kifelew

June 2020 Date

Acknowledgments

My journey into the phage therapy world started when my principal supervisor A/Prof. Peter Speck's passionate knowledge about bacteriophage considerably opened my curiosity and interest in bacteriophages that gave me essential tools and basis to have excellence allied to my work. Hence, I am deeply indebted to A/Prof. Peter Speck, for his insightful taught, care, encouragement, advice, guidance, and support all way through, allowing me to build up my independence with an enthusiastic attitude towards my work and life. I also wish to express my earnest gratitude to my associate supervisor, Professor James G. Mitchell (Jim), for his continuous teaching, guidance, encouragement, and invaluable suggestions during this work.

I would like to thank Dr. Nicky Thomas from the University of South Australia, School of Pharmacy and Medical Sciences, and the Basil Hetzel Institute for Translational Health Research and Prof. Melissa Brown from Flinders University, College of Science and Engineering, who gave me invaluable technical advice.

I am also very much grateful to Prof. Jim lab members: Dr. James Peterson, Dr. Lisa Dann, Dr. Sylvia Sapula, Dr. Jessica Carlson-Jones, Michelle Clanahan, Matthew Hisee, Ashleigh Hisee, Reuben Wheeler, and others for sharing me their knowledge, skill, and experience during the laboratory meetings and experiments.

A special acknowledgment goes to Dr. Lewis Vaughan, who taught and helped me a lot to complete the mouse experiment with great success. The study on the mouse wound infection and treatment model was made possible only through the total engagement and strong technical support of the Animal Facility staff, College of Medicine and Public Health, Flinders University. I gratefully acknowledge the Australian government for the prestigious scholarship, The Research Training Program (RTP), Flinders University for Hosting me, South Australia Pathology in helping collecting *S. aureus* and *P. aeruginosa* clinical isolates, and AmpliPhi Biosciences Corporation for generously providing AB-SA01 and AB-PA01 phages. I shall forever be indebted to St Paul's Hospital Millennium Medical College, Ethiopia, for giving me the study leave and all-round support throughout my study period. I also acknowledge Diabetes Australia for funding the study.

I appreciate A/Prof. Heather Jordan of Mississippi State University, USA, and Dr. Nicky Thomas of University of South Australia, Australia, for providing the fluorescing bacterial strains, and Prof. Catherine Abbott for allowing me to use her laboratory consumables. Thank you, Dr. Zelalem, Dr. Hailay, Dr. Fisaha, and Mr. Abel for positive influence and encouragement that sustained throughout this Ph.D. study time. Many individuals and institutions have contributed to the smooth accomplishment of this Ph.D. journey, though it is difficult to list them all, I recognize their help and positive influence.

Finally, my diction is too poor to translate my gratitude to my beloved wife, Alemwork Abate Kebede, who has always given me all the possible relentless support, wise advice, encouragement, and unwavering belief in me throughout. I extend my special thanks to my children Brook Legesse Garedew and Nahom Legesse Garedew, who are a powerful continuous source of inspiration.

Dedication

To my wife Alemwork Abate Kebede, who teach me selfless sacrifice,

and

To my children, Brook Legesse Garedew and Nahom Legesse Garedew, who give meaning to everything I do.

You have made me more fulfilled than I could have ever imagined.

Summary

This work is in response to two severe problems confronting global health systems: diabetes and antibiotic resistance. Diabetic foot ulcers (DFUs) are open lesions that fail to heal in the foot of a diabetic patient. About one in four people living with diabetes develop diabetic foot ulceration during their lifetime. DFUs causes close to 90% of limb amputations among persons with diabetes. The 5-year mortality rate following foot amputation due to DFUs is up 74%. Most DFUs become infected, and antibiotic-resistant Staphylococcus to aureus and Pseudomonas aeruginosa are often isolated from DFUs. The increasing failure of antibiotics prompts the development of bacteriophages ("phages") as alternative anti-infective agents.

An important aspect of bringing phages to the clinic is that they are made to high standards. We have access to phage cocktails AB-SA01, against *S. aureus*, and AB-PA01, against *P. aeruginosa*, both of which are made to current good manufacturing practice (cGMP) standard. This study sought to investigate the lytic efficacy of AB-SA01 and its three component phages against *S. aureus*, and AB-PA01 and its four component phages against *P. aeruginosa* clinical isolates. Because wound infections are often polymicrobial and contain bacteria in biofilm form, this work sought to examine the efficacy of phages in such a setting.

Findings in chapters 2 and 3 indicate that phage cocktails AB-SA01 and AB-PA01 and their components are strongly lytic on most clinical isolates in planktonic and biofilms states. Moreover, these phage cocktails produced significant biomass reduction in single-species biofilms similar or superior to antibiotics used as a positive control. The efficacy and host range of phage cocktails AB-SA01 and AB-PA01 and their components suggest the

enormous potential of phages in the treatment of antibiotic-resistant S. *aureus* and *P*. *aeruginosa* infections.

Real-time monitoring using spectrophotometry, supported by the colony count method, demonstrates that the fluorescence and population density of mCherry *S. aureus* KUB7 and GFP *P. aeruginosa* PAO1 significantly decreased when treated by their phage cocktail alone and the mixture of the two phage cocktails, similar to gentamicin treatment, as detailed in chapter 4. The findings in chapter 4 show that AB-SA01 and AB-PA01 effectively lysed their hosts in the presence of non-susceptible bacteria, both in planktonic and biofilm states. Moreover, the findings demonstrate that mCherry- and GFP-based mixed-species microplate assay using spectrophotometry combines reproducibility, rapidity, ease of management, and compatibility with high-throughput screening.

The results of multidrug-resistant *S. aureus* infected diabetic mice wound treatment show that AB-SA01 effectively decreased the bacterial load and significantly improved wound healing, similar to vancomycin treatment. In contrast, wounds of saline-treated mice showed no healing, but expanded and became inflamed, ulcerated, and suppurating. No adverse effect related to the application of the phage cocktail was observed. Hence, the results suggest that topical phage cocktail treatment may be useful in treating antibiotic-resistant *S. aureus* infections.

Overall, this thesis shows that the lytic efficacy, broad host range, and significant biofilm biomass reduction capability, in single-species and mixed-species biofilms, of the phages, makes them suitable candidates for therapeutic use in the DFU setting. Further investigation is required to evaluate the clinical application and off-target effect of phages and phage resistance development.

Thesis structure

This thesis is structured in manuscript format, with chapters 2 to 5 submitted or under preparation for submission on peer-reviewed journals. Chapters 4 and 5 are published and in press in Viruses and BMC Microbiology journals, respectively. Chapters 2 and 3 are under preparation to be submitted to Biofouling and Frontiers in Microbiology journals, respectively. Because the studies are continuations of others or related in many ways, redundancies may present mainly in the introductions and methodology sections. Chapter 2 describes the efficacy and host range of AB-SA01 and its component phages on laboratory strains and clinical isolates of diabetic foot ulcers on *in vitro* planktonic and biofilm states. Chapter 3 analyzes the lytic effect and host range of AB-PA01 and its components on laboratory strains and isolates collected from diabatic foot ulcer patients on in vitro planktonic and biofilm phases. Chapter 4 studies the lytic efficacy of phage cocktails AB-SA01 and AB-PA01 on their hosts in mixed-species planktonic and biofilm cultures. Chapter 5 examines the effect of AB-SA01 on diabetic mice wound infection caused by multidrugresistant (MDR) S. aureus. Chapter 6 discusses the general findings of chapters 2 to 5. All cited literature is provided in one collated reference list at the end of this thesis to avoid redundancy. Although the nominative form "we" is used in the manuscripts under preparation or submitted for journals, manuscripts of Chapters 2 - 5, the work presented was undertaken by myself under the supervision of the coauthors.

Chapter 1

General Introduction

Chapter 1: General introduction

1. Introduction

1.1 Diabetic foot ulcer infection

Diabetic foot ulcer (DFU) is defined as foot, in a diabetic patient, with an open sore caused by a break in the skin that fails to heal. It is a devastating complication of diabetes often associated with neuropathy and peripheral arterial disease of lower limbs (Alexiadou and Doupis, 2012), as illustrated in Figure 1.1. About one in four people living with diabetes develop diabetic foot ulceration during their lifetime (Armstrong *et al.*, 2017). DFU is one of the most common reasons for diabetic-related lower extremity amputation (Uçkay *et al.*, 2015). Over 50% of DFU become clinically infected (Lavery *et al.*, 2007). DFU infections are costly complications of the diabetic foot associated with high morbidity, mortality, treatment costs, risk of lower extremity amputation, and poor quality of life (Alexiadou and Doupis, 2012).



Figure 1. 1 Pathogenesis of DFU infection (adapted from (Turan et al., 2015)

DFU infections are usually polymicrobial, mainly bacterial agents such as *Staphylococcus aureus*, and *Pseudomonas aeruginosa* (Lee *et al.*, 2017; Malik *et al.*, 2013; Ramakant *et al.*, 2011). *S. aureus* and *P. aeruginosa* are the most common bacterial species isolated from DFUs, with up to 93.5% and 52.2% detection rate, respectively (Gjødsbøl *et al.*, 2006). They are frequently isolated together from chronic wound infections (Bessa *et al.*, 2015; Trivedi *et al.*, 2014; Körber *et al.*, 2010). Limited studies showed that *S. aureus-P. aeruginosa* mixed-species infections are more virulent than single-species infections (Pastar *et al.*, 2013; Hendricks *et al.*, 2001). Because of the formation of polymicrobial biofilms, expression of virulence factors, and synergistic interactions between *P. aeruginosa* and *S. aureus*, delayed wound healing is seen compared to single species infected wounds (Pastar *et al.*, 2013).

Due to the impaired immune defense around the necrotic tissue of DFU, low-virulence bacteria can also play a pathogenic role (Lipsky, 1999). Close to 70% - 85% of lower-limb amputations in patients with diabetes are caused by biofilm-related foot ulcer infections (Malik *et al.*, 2013; Dowd *et al.*, 2008). Bacteria residing within biofilms are resistant to many antibiotics (Uçkay *et al.*, 2016; Malik *et al.*, 2013; Ramakant *et al.*, 2011; Davis *et al.*, 2006). *P. aeruginosa* and *S. aureus* usually form biofilms *in vivo* that could contribute to their antibiotic resistance characteristic (Ramakant *et al.*, 2011; Dowd *et al.*, 2008). Difficulty in growing different bacterial species together *in vitro* has made the study of species interactions challenging; this is true for *S. aureus* and *P. aeruginosa* (Kahl, 2017; Armbruster *et al.*, 2016; Korgaonkar *et al.*, 2013; Mashburn *et al.*, 2005). Though they are commonly isolated together from mixed-species infections, reports indicate that *P. aeruginosa* often kills *S. aureus* both *in vitro* (Kumar and Ting, 2015; DeLeon *et al.*, 2014) and *in vivo* (Filkins *et al.*, 2015; Pernet *et al.*, 2014).

1.2 S. aureus

S. aureus is a non-motile and facultative anaerobic bacterium with 0.5 - 1.5 µm diameter. It is a Gram-positive coccus that typically grows in clusters; hence the name from Greek word *staphylé* to say, "a bunch of grapes" and *kokkos* means "berry" or "seed" (Wilson, 1987) and *aureus* from Latin word *aurum* meaning "gold" to describe its golden color colonies (Stryjewski and Corey, 2014), as shown in Figures 1.2A and 1.2B. *S. aureus* can grow at high salt concentration and temperature ranging from 18 °C to 40 °C (Tong *et al.*, 2015). It is also strongly resistant to hostile environmental conditions, including low water activity and high osmotic pressure (Gnanamani *et al.*, 2017). Most *S. aureus* strains express a polysaccharide capsule that protects the bacterium from phagocytosis (Vandenesch *et al.*, 1994).



Figure 1. 2 Characteristics of *S. aureus* colonies and cells: A - golden colonies of *S. aureus* on nutrient agar, B - yellow colonies of *S. aureus* on mannitol salt agar, and C - cluster formation of *S. aureus* in Gram-staining. (Adopted from (Murray *et al.*, 2015)).

S. *aureus* produces several toxins that contribute to its virulence including cytolytic toxins α , β , δ , γ , and Panton-Valentine leucocidin (PVL), and exfoliative toxins such as enterotoxins and toxic shock syndrome toxin-1 (Prevost *et al.*, 1995). It also contains enzymes that help its spread within the human body and manipulate immune responses. Some of the enzymes are (i) coagulase - converts fibrinogen to fibrin, (ii) catalase - converts hydrogen peroxide to water and oxygen, (iii) hyaluronidase - breaks down hyaluronic acid, and (iv) staphylokinase - inactivates the bactericidal α -defensins (Dunyach-Remy *et al.*, 2016; Dinges *et al.*, 2000).

These virulence factors help *S. aureus* to establish infection through evading host immune responses and facilitating tissue attachment and invasion (Gnanamani *et al.*, 2017).

S. aureus occurs as a human commensal, a common colonizer of the skin, anterior nares, oropharynx, gastrointestinal tract, and urogenital tract (Messad *et al.*, 2015). Approximately 30% and 70% of the human population carry *S. aureus* in their anterior nares permanently and periodically, respectively (Mulcahy and McLoughlin, 2016; Wertheim *et al.*, 2005; Kluytmans *et al.*, 1997). Wound infections, pneumonia, osteomyelitis, meningitis, endocarditis, and thrombophlebitis are some of the diseases caused by *S. aureus*. Moreover, it is one of the most prevalent causes of nosocomial infections (Tong *et al.*, 2015; Lowy, 1998).

S. aureus is one of the most commonly isolated bacteria from DFUs, including from soft tissue infections and osteomyelitis. Its toxins and virulence factors have an essential role in deepening and spreading the DFU infections through considerable proteolytic and toxic activity (Viquez-Molina *et al.*, 2018). Of all *S. aureus* infections, about 15% were caused by methicillin-resistant *S. aureus* (MRSA) strains. The prevalence of MRSA in DFUs is estimated at 30% (Lavery *et al.*, 2014). The high prevalence of *S. aureus* and the rapid rise of its antibiotic resistance forced the development of guidelines recommending empirical antibiotic therapy with anti-Staphylococcal coverage for all patients with DFU infections (Lipsky *et al.*, 2016). MRSA nasal colonization and infection with *S. aureus* strains resistant to various antibiotics are risk factors for higher MRSA infection prevalence in DFU patients (Lavery *et al.*, 2014).

1.3 P. aeruginosa

P. aeruginosa belongs to the genus Pseudomonas (from the Greek words "*pseudes*" and "*monas*" meaning false unit). It is a rod-shaped with 0.5 to 0.8 μm width, and 1.5 to 3.0 μm

lengths. *P. aeruginosa* is a Gram-negative, facultative aerobic, and motile bacteria (Sneath *et al.*, 1986). *P. aeruginosa* grows best at 37 °C (Anzai *et al.*, 2000). Its colonies appear mucoid because most strains produce a thick and viscous polysaccharide capsid (Stehling *et al.*, 2008), as shown in Figure 1.3. Most strains of *P. aeruginosa* produce fluorescent pigments, including pyocyanin (blue), pyorubin (red-brown), and pyoverdine (yellow-green). These pigments interfere with host cell respiration (pyocyanin) and iron transport (pyoverdine) and protect the bacteria from hydrogen peroxidase (pyorubin) (Orlandi *et al.*, 2015; Jayaseelan *et al.*, 2014; Rodríguez-Rojas *et al.*, 2009; Visca *et al.*, 2007; Lau *et al.*, 2004).



Figure 1. 3 Characteristics of *P. aeruginosa* colonies and cells: **A** - colonies of *P. aeruginosa* on nutrient agar, **B** - metallic sheen colonies of *P. aeruginosa* on Mac Conkey agar, and **C** - road-shaped *P. aeruginosa* in Gram-staining. (From (Murray *et al.*, 2015)).

P. aeruginosa is armed with several virulence factors, including toxins, enzymes, adhesins, and pigments that collectively help it to escape the immune system and propagate. Among the toxins, exotoxin A contributes to its pathogenesis (Jaffar-Bandjee *et al.*, 1995). Once exotoxin A is inside the host cell, it can disrupt host cell protein synthesis, resulting in tissue damage (Breidenstein *et al.*, 2011; Lister *et al.*, 2009; Hancock, 1998). Its regulatory mechanisms and virulence factors render *P. aeruginosa* metabolically versatile and capable of inhabiting humans, animals, plants, and environments (Breidenstein *et al.*, 2011; Frimmersdorf *et al.*, 2010).

P. aeruginosa is resistant to many commonly used antibiotics, weak antiseptics, and high concentrations of salts and dyes. It shows high intrinsic resistance to a wide variety of antibiotics, including β -lactams, fluoroquinolones, and aminoglycosides, mainly because of its membrane low permeability due to the absence of efficient transmembrane proteins (porins) that form water-filled pores in the hydrophobic part of the membrane (Nicas and Hancock, 1983). This characteristic facilitates increased efflux and enzymatic modification of antibiotics (Hancock, 1998; Nicas and Hancock, 1983). Moreover, *P. aeruginosa* can develop acquired antimicrobial resistance through horizontal genetic elements transfer and mutation. Conjugation, transformation, and transduction can transfer resistance islands, prophages, plasmids, integrons, and transposons that harbor antibiotic resistance genes to the next generation (Breidenstein *et al.*, 2011). The transfer of antibiotic-resistant genes may exacerbate the antimicrobial-resistance characteristics of the bacteria (Breidenstein *et al.*, 2011; Li *et al.*, 2000).

P. aeruginosa is a predominant pathogen causing severe diseases in immunocompromised individuals (Høgsberg *et al.*, 2011). *P. aeruginosa* is one of the causes of treatment failure due to its capability of rapidly acquiring resistance to a wide range of antibiotics (Serra *et al.*, 2015; Lister *et al.*, 2009). Antibiotic resistance of *P. aeruginosa* isolates has been significantly increased during the past two decades (Chatterjee *et al.*, 2016; Breidenstein *et al.*, 2011). *P. aeruginosa* infected wound is characterized by a significantly broader ulcer area coverage and a delayed or poor healing process (Gjødsbøl *et al.*, 2006). *P. aeruginosa* is among the most commonly isolated bacteria from DFU and often localized in the deepest region of the wound, usually as a biofilm (Fazli *et al.*, 2009).

1.4 Bacterial biofilms

Studies using advanced scanning techniques have demonstrated the micro-heterogeneity of biofilms with variable distribution of cells, matrix, and fluid-filled channels and pores (Wood *et al.*, 2000). Biofilm formation will begin by adherence to surfaces with the assistance of flagella and pili in Gram-negative (Bjarnsholt *et al.*, 2013) or surface proteins in Gram-positive bacteria (Cucarella *et al.*, 2001). Some bacterial species capable of producing proteins like biofilm-associated protein (*bap*) are found to be surface adherent and strong biofilm producers. *Bap* also involves in pathogenesis, causing a persistent infection (Cucarella *et al.*, 2001). Cell-to-cell signaling is required for biofilm formation (Macia *et al.*, 2014) and may depend on the concentration of diffusible signal molecules such as second messengers and small ribonucleic acids (RNAs) (Bjarnsholt *et al.*, 2013). Various microorganisms share a common milieu and coexist in niches forming multispecies biofilms (Thein *et al.*, 2007). The synergistic interactions of bacterial species may contribute to the delay in healing from infection and antibiotic tolerance by co-infecting bacteria (Dalton *et al.*, 2011).

Biofilm formation occurs in response to environmental changes that induce multiple regulatory mechanisms, which ultimately lead to spatial and temporal reorganization of the bacterial cells as a means of survival (O'Toole *et al.*, 2000). This phenomenon affects the expression of surface molecules and virulence factors and nutrient utilization of the bacteria (Kostakioti *et al.*, 2013). Biofilm can be formed on biotic and abiotic surfaces (O'Toole and Wong, 2016). Bacterial biofilms consist of aerobic and anaerobic growing subpopulations (Bjarnsholt *et al.*, 2013). Bacteria cells occupy 5% - 30% of the biofilm volume (Macia *et al.*, 2014). The remaining portion of the biofilm is EPS, composed mainly of exopolysaccharides, and small amounts of protein, DNA, bacterial lytic products, and compounds from the host. The nature, concentration, and arrangement of these components determine the density and

porosity of the biofilm (Flemming and Wingender, 2010). The EPS has the potential to modify the response to antimicrobial treatments through its action as a diffusion barrier and neutralizer (Gilbert *et al.*, 2002).

Bacteria residing in biofilms are more resistant to the chemical and physical effects of the environment than their planktonic counterparts (Steenackers *et al.*, 2016; Gilbert *et al.*, 2002). Bacterial cells present in biofilms are thought to be up to 1000 times more resistant to antibiotics and biocides than their planktonic cells (Høiby *et al.*, 2010; Ceri *et al.*, 1999). An estimated 65% - 80% of all human infections, especially chronic wounds such as DFU, are biofilm-related (Percival *et al.*, 2012; Coenye and Nelis, 2010). Biofilms are tolerant to physical and chemical effects due to the complex nature of their extracellular matrix and the presence of metabolically inactive persister cells (Harper *et al.*, 2014). Thus, the need for effective biofilm elimination strategies becomes imperative (Kostakioti *et al.*, 2013).

1.5 Treatment of DFU

The main aim of DFU management is to achieve wound healing and wound closure. The complete treatment of DFU includes the draining of invasive infections, debriding necrotic tissues, administering appropriate antibiotics, and performing vascular reconstruction (Bradbury *et al.*, 2005). The wound healing process is characterized by homeostasis, inflammation, and proliferation and remodelling of keratinocytes, fibroblasts, Langerhans cells, and endothelial cells (Wang *et al.*, 2012). One of the challenges in the treatment of DFU is co-infection by antibiotic-resistant bacteria such as *P. aeruginosa* and *S. aureus* (Fazli *et al.*, 2009; Gjødsbøl *et al.*, 2006). Often, *S. aureus* is localized in the top while *P. aeruginosa* is in the deepest and broader area of wounds (Serra *et al.*, 2015; DeLeon *et al.*, 2014; Gjødsbøl *et al.*, 2006). Studies showed that the antibiotic resistance of *P. aeruginosa*

and *S. aureus* increase when they grow together (DeLeon *et al.*, 2014) and exist as coinfection in wound ulcers that deter healing processes (Pastar *et al.*, 2013).

Biofilms mostly contain polymicrobial population adherent to surfaces using extracellular polymeric materials that have antiphagocytic properties. The extracellular polymeric material can inactivate the complement system and antibodies to cause tissue damage and chronic inflammation (Wolcott et al., 2013). Hyperglycaemia enhances oxidative stress and increased interleukin-8 secretion from keratinocytes that may contribute to the delayed healing of diabetic wounds (Lan et al., 2013). Removing biofilm through debridement supported by antibiotics could improve wound healing (Malik et al., 2013). Vancomycin plus either ciprofloxacin, meropenem, ertapenem, or piperacillin/tazobactam are antibiotics recommended through systemic route for S. aureus and P. aeruginosa co-infection (Lipsky et al., 2012b). Except for the experimental use of moxifloxacin (Bessa et al., 2015), no topical broad-spectrum antibiotic is available to treat DFU to date. Furthermore, no drug is designed against bacteria in biofilms.

1.6 Animal models for DFU infection and treatment

In vitro and *in vivo* validations are required before the application of therapeutic agents in clinical settings. As *in vitro* experiments fail to reproduce the complex physiology of the host organism and pathophysiology of the infected wound, the *in vivo* models play an important role in ensuring clinical relevance (Mendes *et al.*, 2012a). When the *in vitro* tests are showing promising findings, the next logical step is to study the agent in an animal model. Wound infection and treatment effect studies have been carried out in animal models, including mice (Turner *et al.*, 2017; de Mayo *et al.*, 2017), rats (Mendes *et al.*, 2012a), rabbits (Wills *et al.*, 2005), and pigs (Velander *et al.*, 2008). Different wound infection and treatment models, including incisional and excisional wounds in animals rendered diabetic through chemical

induction or genetic modification, are also documented (Turner *et al.*, 2017; Abendroth *et al.*, 2010; Galiano *et al.*, 2004).

Genetically diabetic animal models are limited in availability and expensive. Besides, they are often obese that may impede wound closure through physiological dysfunction and excess subcutaneous fat accumulation (Abedon, 2012; Chen et al., 2011; Davidson, 1998). Chemically induced diabetes model is recommended when testing therapeutic agents (Davidson, 1998); hence it is employed in this study. Injection of streptozotocin (STZ) is widely in use to induce diabetes as it is cost-effective, rapid, and known to induce type 2 diabetes mellitus in most species and strains of rodents (Deeds et al., 2011). STZ (N-nitroso derivative of glucosamine) is a broad-spectrum antibiotic extracted from Streptomyces acromogenes bacteria (Hayashi et al., 2006). STZ is used for the treatment of the human pancreas islet cell metastatic carcinoma (Deeds et al., 2011; Hayashi et al., 2006). It induces rapid and irreversible necrosis to insulin-producing β -cells of rodents' pancreatic islets (Lenzen, 2008). STZ is taken up through GLUT2 glucose transporter of the cell membrane in rodents and causes alkylation of DNA and β -cell death (Lenzen, 2008; Szkudelski, 2001). Sensitivity to STZ is highly variable in rodent species and strains (Deeds et al., 2011; Hayashi et al., 2006). Rodents also show gender difference to STZ toxic effect; males are profoundly affected by STZ toxicity (Cortright et al., 1996; Leiter, 1985).

An excisional wound model in mice has been used to evaluate the efficacy of topically applied antibacterial agents (Davidson *et al.*, 2013). Mouse wound models are commonly used because of low cost and ease for management (Park *et al.*, 2014). Unlike the re-epithelization and the granulation process of human wound healing, mouse wound healing is through skin constriction (Park *et al.*, 2014; Dunn *et al.*, 2013). A splint excisional wound model using a silicone sheet that prevents excessive skin contraction is used to mimic human

wound healing type in mice (Park *et al.*, 2014; Dunn *et al.*, 2013; Lindblad, 2008). Treatment in all these methods is by antibiotic agents. In this thesis, the use of bacteriophage treatment is studied.

1.7 Bacteriophages

Bacteriophages ("phages") are viruses that infect bacteria. The word "phage" derives from the Greek word "phagein" meaning, "to eat"; hence bacteriophages mean "bacteria eaters". Phages were first discovered independently by the British scientist Frederick William Twort (Twort, 1915) and the French-Canadian scientist Felix Hubert d'Herelle (d'Herelle, 1917) in 1915. However, d'Hérelle was the first scientist who studied phages to combat bacterial infections. d'Hérelle is also the author of the terms bacteriophages and plaque (the small clear areas seen in agar plates caused by bacterial killing by phages) (d'Hérelle and Smith, 1922; d'Herelle, 1917).

Commercialization of phages was started in the 1920s by large companies of the time (Pirnay *et al.*, 2011; Carlton, 1999). Following the discovery of penicillin in 1928 by Alexander Fleming, the treatment of infectious diseases relied almost exclusively on antibiotics. Since then, phage therapy was largely forgotten until recently in most parts of the globe except in the former Soviet Union and Eastern Europe (Chanishvili *et al.*, 2002; Alisky *et al.*, 1998). Phages are 20 - 200 nm in size (Ly-Chatain, 2014). To date, the International Committee for Taxonomy of Viruses (ICTV) recognizes 13 phage families, as listed in Table 1.1.

Family	Morphology	Nucleic acid	Examples
Myoviridae	Non-enveloped, contractile tail	Linear dsDNA	Tailed phage, T4
Siphoviridae	Non-enveloped, non-contractile tail	Linear dsDNA	Phage λ
Podoviridae	Non-enveloped, non-contractile tail	Linear dsDNA	Coliphage T7
Tectiviridae	Non-enveloped, isometric	Linear dsDNA	Phage PRDI
Corticoviridae	Non-enveloped, isometric	Circular dsDNA	PM2
Lipothrixviridae	Enveloped, rod-shaped	Linear dsDNA	Thermoproteus Φ phage I
Plasmaviridae	Enveloped, pleomorphic	Circular dsDNA	Acholeplasma phage
Rudiviridae	Non-enveloped, rod-shaped	Linear dsDNA	Rudivirus
Fuselloviridae	Non-enveloped, lemon-shaped	Circular dsDNA	SSV-I
Inoviridae	Non-enveloped, filamentous	Circular ssDNA	Coliphage fd, MS2, ΦX174
Microviridae	Non-enveloped, isometric	Circular ssDNA	Spiroplasma phages
Leviviridae	Non-enveloped, isometric	Linear ssRNA	Coliphage QB
Cystoviridae	Non-enveloped, spherical	Segmented dsDNA	Фб

Table 1.1 ICTV classification of phages (adapted from (Ackermann and Prangishvili, 2012)).

The order *Caudovirales* comprises the vast majority (96%) of known phages. Members of the *Caudovirales* have DNA genome, head, and tail in common, as illustrated in Figure 1.4. The head has an icosahedral or elongated morphology. The tail is helical and generally provided with fixation structures (baseplates, spikes, and fibres) and serves as a DNA injection apparatus during the host infection process (Ackermann and Prangishvili, 2012). The morphological difference of the tail of *Caudovirales* member phages defines the three families of the order: *Myoviridae*, *Siphoviridae*, and *Podoviridae*. While the tails of *Myoviridae* are long and contractile, the tails of *Podoviridae* are short and non-contractile (Ackermann, 2009).



Figure 1. 4 Morphology of tailed bacteriophage (from (Nobrega et al., 2018)).

Phages are the most abundant and diverse components of microbial communities (Kristensen *et al.*, 2013). It is estimated that about 10^{31} phages are present in the global biosphere, 10 times more genetic units than bacteria (Rohwer and Edwards, 2002). Phages can be found in places where bacteria or archaea are present, including humans, animals, foods, water, and soil (Kutter and Sulakvelidze, 2004). They are diverse in size, shape, structure, symmetry, and nucleic acid content (Ackermann, 2009).

Phages can be lytic (virulent) or lysogenic (temperate) depending on their life cycle, as shown in Figure 1.5. Lytic phages cause metabolism disruption and lysis of bacterial host cells soon after the initiation of infection, while lysogenic phages insert their genome into the bacterial genome. Lysogenic phages are mostly considered as horizontal gene transfer vectors, which may considerably contribute to the evolution of the bacterial host (Harper *et al.*, 2014; Gama *et al.*, 2013; Canchaya *et al.*, 2003).

1.7.1 Lytic phages

After locating the receptors, lytic phages penetrate the cell wall and inject their genetic material into the cytoplasm of the host cell. The genetic material of phages will be taken up by the host and divert cell biosynthetic machinery of the host for phage reproduction (Young,

1992a). Phage gene expression, genome replication, morphogenesis (formation of capsids and tails), and assembly and packaging of parts occur inside the host cell (Carlton, 1999). Once the production of phage virions in a good number or beyond the carrying capacity of host bacteria is completed, lysis of the host bacteria cell can be completed within few minutes using lytic enzymes such as a muralytic enzyme, or endolysin, and a holin (Paul *et al.*, 2011). The phage-encoded holins are transmembrane proteins that create lesions in the host cytoplasmic membrane through which endolysins gain access to the peptidoglycan layer (Young, 1992b). Additional membrane proteins called spanins are involved in the lysis process of the outer membrane in Gram-negative hosts by catalysing fusion of the inner and outer membranes (Young, 2014). Most phages use these enzymes to permeabilize the bacterial host membrane at a programmed time and thus control the length of the vegetative cycle. By contrast, single-stranded phages accomplish lysis by producing lysis protein without muralytic activity (Young *et al.*, 2000). Lysis of the host cell will result in the release of hundreds of fully matured virulent phage virions capable of infecting other host bacteria and initiate new phage lytic cycles (Richards, 2014).



Figure 1. 5 Lytic and lysogenic life cycle of phages (from (Vander Elst and Meyer, 2018)).

1.7.2 Lysogenic phages

Upon successful entry to the bacterial host the same way as lytic phages, lysogenic phages integrate their genome into the host chromosome, with their genome in this state termed prophage. As a result, the phage genome stays dormant and replicates as part of the host, called lysogeny, and the genome is transmitted to a daughter cell when a host cell division occurs. Phages following this type of replication are also called temperate phages, and the host cells carrying a prophage are known as lysogenic cells (Oppenheim and Adhya, 2007). When a host bacterium is lysogenised, it may become resistant to infection by other related phages that share the same immunity group profile or repressor specificity (Davies *et al.*, 2016; Labrie *et al.*, 2010; Kenny *et al.*, 2006). However, due to stimuli such as environmental stress that bring bacterial host DNA damage, the prophage can be activated and induce lytic infection (Erez *et al.*, 2017; Oppenheim *et al.*, 2005). Lysogenic phages are not used for therapeutic purposes because they are inefficient in killing the bacterial host, or there is an increased risk of horizontal gene transfer that may lead to the transfer of virulence factors to the host (Davies *et al.*, 2016; Boyd and Brüssow, 2002).

1.8 The role of phages on human microbial communities

Phages have been recovered from diverse sites of the human body, including the skin (Hannigan *et al.*, 2015; Weyrich *et al.*, 2015), digestive tract (Edwards *et al.*, 2019; Lepage *et al.*, 2013; Mills *et al.*, 2013), respiratory tract (Willner *et al.*, 2009), and oral cavity (Pride *et al.*, 2012). Metagenomic studies on the microbiota of humans indicate that many of the viruses associated with healthy human tissues are phages (Hannigan *et al.*, 2015; Wylie *et al.*, 2014). A recent global study on human gut viruses showed that the phages termed crAssphages are widespread viruses strongly correlated with different clades of bacteria and an abundant benign component of the healthy human gut microbiome. The study confirmed

that crAssphages could be the reliable human fecal contamination markers signifying the diagnostic importance of phages (Edwards *et al.*, 2019). The study also lends strong support for the earlier discovery of crAssphage, which is more abundant than all other known phages together documented in the publicly available metagenomic data (Dutilh *et al.*, 2014).

Gut phages are believed to be individual specific as a result of the rapid evolution of phages and the complex gut microbiome dynamics (Minot *et al.*, 2011). Nevertheless, some phages in the gut might be highly conserved in people all over the world (Dutilh *et al.*, 2014). Phages are likely to produce a strong influence on the diversity and structure of bacterial communities of the human body either through predation or by supplying a pool of new, possibly beneficial genes (Hannigan *et al.*, 2015; Stern and Sorek, 2011; Breitbart *et al.*, 2003). They can also influence microbial diversity and structure by acting directly as immunomodulation agents (Hannigan *et al.*, 2015) and creating dysbiosis (De Paepe *et al.*, 2014). The switching of phages between lytic and lysogenic pathways also results in the evolution of bacterial hosts (Erez *et al.*, 2017; Klimenko *et al.*, 2016; Levin *et al.*, 2013).

How virulent phages avoid driving their bacterial prey to extinction despite being efficient predators is not yet elucidated (Heilmann *et al.*, 2012; Kim *et al.*, 2011). Lytic phages may be constraining factors for microbial community evolution because the lytic infection will decrease bacterial speciation rates by more than one order (Klimenko *et al.*, 2016; Levin *et al.*, 2013). Lysogenic infection may speed up bacterial evolutionary processes by extracting and inserting genetic materials into new hosts, shuffling genetic diversity in the bacterial hosts (Armstrong *et al.*, 2015). Some prophages may provide beneficial roles to their bacterial host, for example, by increasing biofilm formation capability and promoting colonization (Messad *et al.*, 2015). Hence, phage-bacteria interaction may act as a stabilizing factor for the microbiome through slowing down speciation by infecting the bacterial host.

and creating a stable state of the microbiome community (Klimenko *et al.*, 2016; Maura and Debarbieux, 2012).

1.9 The use of phages as therapeutic agents

Since the discovery of phages as therapeutic agents in humans in 1919, the use of phage has been tested in plants, animals, and humans with various levels of success (Haq *et al.*, 2012). Phages have also been used for disease diagnosis, bacterial typing, and vaccine production (Sagona *et al.*, 2016; Tawil *et al.*, 2014). Studies have shown that phage therapy could be a potential alternative in combating antibiotic-resistant bacterial infections (Wittebole *et al.*, 2014; Górski and Weber-Dabrowska, 2005). Lytic phages can reach and lyse their host bacteria residing in biofilms (Olszak *et al.*, 2015; Saussereau *et al.*, 2014). Lytic phages could also be used as prophylactic agents against bacterial infections (Morello *et al.*, 2011). As phages are specific to their hosts, they show no or minimum off-target effects (Fu *et al.*, 2010). Phages are safe to apply in any route of administration (Pirnay *et al.*, 2015).

Some phages may have specificity for one or few host strains and others may infect different strains within a single species or multiple related bacteria species (Donlan, 2009; O'Flaherty *et al.*, 2005; Jensen *et al.*, 1998). The determining factor for the host range of phages is the specificity of receptors for the phage strain. A commonly described mechanism of bacterial resistance against phage attack is the prevention of phages from attaching to cell surface due to the absence or alteration of receptors (Hyman and Abedon, 2010). Preventing phage genome entry, degradation of phage genome by the restriction-modification (RM) and clustered regularly interspaced short palindromic repeats (CRISPR-cas) associated systems, preventing genome integration by super-infection exclusion systems (Sie), and blocking phage replication, transcription, translation, or virions assembly via the abortive-infection system are other bacterial resistance mechanisms to phage attack (Labrie *et al.*, 2010; Cairns
et al., 2009). The bacteria may also carry out apoptosis when infected with phage, which is part of an abortive infection (Labrie *et al.*, 2010; Chopin *et al.*, 2005).

Phage cocktails and engineered phages are preparations that could potentially improve the result of phage therapy (Bernasconi *et al.*, 2017; Alves *et al.*, 2016; Westwater *et al.*, 2003). High efficacy, bacterial host specificity, quick onset of action, and low risk of resistance, as in the case of phage cocktails, make phage therapy a feasible option for use in humans (Pastagia *et al.*, 2011). Phage cocktails have been found to exploit the synergistic effect of component phages, which could facilitate their adsorption and diffusion in multispecies biofilm infections (Chhibber *et al.*, 2015; Sillankorva *et al.*, 2010). When more than one types of phages from the phage cocktail attack the same bacterium, the phage cocktail may produce a better lytic effect than any single component phage due to synergy, or the mixture may be less effective than the best single phage because of phage interference (Schmerer *et al.*, 2014).

The advantages of phage cocktail over single phage are: a broad collective host range, preventing the development of phage-resistant bacterial mutant, and kill the bacterial population more rapidly or completely (Schmerer *et al.*, 2014). Hence, phage cocktail therapy is strongly recommended to effectively eliminate bacterial cells both in planktonic and biofilm states (Alves *et al.*, 2016).

1.10 Effect of phages on biofilms

Lytic phages have been found to produce enzymes that degrade bacterial capsule and biofilm extracellular polymeric substances (EPS) (Sutherland *et al.*, 2004). Phage-associated enzymes facilitate phage migration through mucoid biofilms (Sutherland *et al.*, 2004). With the help of these enzymes, phages move through biofilms and lyse bacterial host cell by endolysin

enzyme and holin and spanins proteins (Young, 2014). Then, phages eliminate their hosts via lytic activity (Zhang *et al.*, 2014b; Gilbert *et al.*, 2002). The weakening of the EPS and the capsule by phage-encoded enzymes facilitate phagocytosis or supports the effect of antibiotics (Morello *et al.*, 2011). These phage-associated enzymes are capable of degrading EPS of many genera (Pei and Lamas-Samanamud, 2014; Sutherland *et al.*, 2004). Phages with these depolymerase enzymes can readily reach the host cell surface by digesting their way and penetrate the bacterial cell membrane and cell wall by the help of endolysin, holin, and spanins or by contracting a sheath and injecting their genome into the cytoplasm of the host bacterium (Gonzalez *et al.*, 2017).

There is no consensus on whether phages maintain lytic efficacy in multispecies biofilms (Geredew Kifelew *et al.*, 2019). Some studies suggest that co-aggregation of bacterial cells and tight cell-to-cell binding in multispecies biofilms may occlude phage receptors (Kay *et al.*, 2011; Teplitski and Ritchie, 2009; Rickard *et al.*, 2003). As opposed to this, reports showed that most of the biofilms possess open architectures with water-filled channels that could allow phages to get access to the interior of the biofilms (Gutierrez *et al.*, 2015; Briandet *et al.*, 2008; Rickard *et al.*, 2003). It has been suggested that the presence of non-susceptible bacteria will not prevent phages from reaching and lysing their susceptible hosts in multispecies biofilms (Sillankorva *et al.*, 2010). Phage could be released either when the biofilms are sloughed-off because of aging, EPS degrading enzymes, hydrophobic and electrostatic interactions in membranes, or a combination of these factors (Sutherland *et al.*, 2004; Van Voorthuizen *et al.*, 2001). The high number of host bacteria within limited spaces in biofilms will facilitate phages movement from one host to the next (Harper *et al.*, 2014).

1.11 Features of phage therapy

Lytic phages could reduce the consequence of bacterial infection (Stratton, 2003). The process of phage infection and subsequent self-replication in bacteria offers advantages over antibiotics: phages amplify themselves at the infection site provided there are sensitive bacterial hosts (Wittebole *et al.*, 2014). The capability of phages to replicate in their bacterial host could reduce the number of repeated doses (Kelly *et al.*, 2012; Vieira *et al.*, 2012; Ryan *et al.*, 2011). Phages are safer as therapeutics than antibiotics because phages will only increase in density and remain in the body or environment as long as they get bacterial hosts (Mendes *et al.*, 2014; Kutter *et al.*, 2010; McVay *et al.*, 2007). Phages can evolve to overcome the resistance by previously susceptible bacteria (Levin *et al.*, 2013).

Limited studies indicated the importance of a phage cocktail in the treatment of multispecies biofilms that benefited from the synergistic effect of phages in facilitating phage adsorption and diffusion (Chhibber *et al.*, 2015; Sillankorva *et al.*, 2010; Schmerer *et al.*, 2014). The discovery of polyvalent phages affecting several species of bacteria shows the possibility of developing broader host range phages (Malki *et al.*, 2015; Kim *et al.*, 2012; Lin *et al.*, 2012; Sillankorva *et al.*, 2010; O'Flaherty *et al.*, 2005; Jensen *et al.*, 1998).

Phages do not cause the adverse side effects of antibiotics (Wittebole *et al.*, 2014). Some advantages of phages over antibiotics are (i) specificity to bacterial host that reduces damage to normal flora, (ii) self-limitation in the absence of bacterial host, (iii) equally effectiveness against antibiotic sensitive and resistant bacteria, (iv) better ability to disrupt bacterial biofilms, and (v) ease to isolate from the environment or select from a collection (Pirnay *et al.*, 2015; Golkar *et al.*, 2014; Wittebole *et al.*, 2014; Yosef *et al.*, 2014; Kutter *et al.*, 2010).

The human body is continually interacting with phages from the surrounding environment or human microbiome, and there is no evidence on phages' direct adverse effect (Edwards et al., 2019; Loc-Carrillo and Abedon, 2011; Gorski et al., 2003). There is evidence that phage therapy is safe to use in different routes of administration (Fong et al., 2019; Ooi et al., 2019; Drilling et al., 2017). Therapeutic phages should be obligatory lytic and free from antibiotic resistance or bacterial virulence genes. Besides, phage cocktails should be affective against broad host range of the target pathogen without side-effect on non-target microbes and capable of complementation, in which resistant strains or mutants to one phage are lysed by other component phages (Law et al., 2019; Casey et al., 2018; Ceyssens et al., 2009). The two-phage cocktails, AB-SA01 and AB-PA01, and their components used in this study satisfy these criteria and have entered clinical development. All components of each phage cocktail are well-characterized and produced under current good manufacturing practice (cGMP) standards (Lehman et al., 2019; Fong et al., 2019). Moreover, AB-SA01 and its components are approved by that complies with the U.S. Food and Drug Administration (FDA) and Australia's Therapeutic Goods Administration (TGA) to conduct clinical phase I trials and single-patient emergency treatment (Law et al., 2019).

AmpliPhi Biosciences Corporation generously provided AB-SA01 and AB-PA01 phages as cocktails and their components. AB-SA01 is a combination of three *Myoviridae* phages designated J-Sa 36, Sa 83, and Sa 87 (Lehman *et al.*, 2019). AB-PA01 is a combination of Pa 193 and Pa 204 from *Myoviridae* and Pa 222 and Pa 223 from *Podoviridae* (Fong et al., 2017).

1.12 Significance of the study

Elements for the conception of this study are:

- A rapid rise of antibiotic-resistant infection is a big problem in DFU infection. The difficulty in developing new antibiotics for the last decade has prompted the search for new agents with effective antibiotic activity, including phages.
- There is evidence that phages can reduce bacterial biofilm biomass, but their efficacy against their host, mainly when it is found in multispecies biofilms has not been well addressed.
- DFU infection is polymicrobial and often involves antibiotic-resistant microbes, including *S. aureus* and *P. aeruginosa*. It is a chronic disease usually complicated with bacterial biofilms. Biofilms display heightened resistance to antibiotics. The limited information about the presence or absence of such heightened resistance by the target bacteria to phages in multispecies biofilms is the driving force for phage therapy development.
- Absence of approved broad-spectrum topical antibiotics and poor outcome of systemically administered antibiotics due to the widespread presence of MDR bacteria in polymicrobial DFU infections are compelling factors to study topically applied phage cocktails.

1.13 Objectives of the study

This research project was initiated to study the effect of phages in treating antibiotic-resistant infections of DFU with the following objectives.

To evaluate the *in vitro* lytic efficacy and host range of phage cocktail AB-SA01 and its components against *S. aureus* clinical isolates in single-species planktonic and biofilm states.

- To examine the *in vitro* lytic efficacy and host range of phage cocktail AB-PA01 and its components on *P. aeruginosa* clinical isolates in single-species planktonic and biofilm cultures.
- To assess the efficacy of phage cocktails AB-SA01 and AB-PA01 against their host bacteria in mixed-species planktonic and biofilm cultures.
- To determine the effectiveness of topical AB-SA01 treatment in controlling antibioticresistant *S. aureus* infection and facilitating healing on a diabetic mice skin wound.

Chapter 2

Effect of Phage Cocktail AB-SA01 on Diabetic Foot Ulcer Clinical *Staphylococcus aureus* Isolates and Biofilms

Chapter 2: Effect of Phage Cocktail AB-SA01 on Diabetic Foot Ulcer Clinical *Staphylococcus aureus* Isolates and Biofilms

Abstract

The increasing incidence of infections caused by antibiotic-resistant bacteria such as methicillin-resistant and multidrug-resistant *Staphylococcus aureus*, a virulent pathogen infecting diabetic foot ulcer (DFU), requires alternative therapeutic agents such as phages with low cost, ready availability, minimum side effect, and high efficacy. This study examined the efficacy of three-phage cocktail AB-SA01, which is made to current good manufacturing practice (cGMP) standard, on *S. aureus* clinical isolates from DFUs. It was found that 94.3% of the isolates were susceptible to AB-SA01. AB-SA01 showed a stronger lytic efficacy than each of its components did, as measured with spot tests. AB-SA01 effectively decreased the biofilm biomass of *S. aureus* clinical isolates. Biofilms of isolates that were resistant during the spot test also showed reduced biofilm biomass reduction. A higher number of resistant biofilms and host cells in the biofilms are more resistant than in planktonic states for phage activity. The phage cocktail and its components were effective irrespective of the antibiotic susceptibility of the isolates. AB-SA01 is a potential treatment for antibiotic-resistant *S. aureus* DFU infections.

2.1 Introduction

2.1.1 S. aureus in DFU

Approximately one-in-three people living with diabetes will develop a foot ulcer, and most of these ulcers will become infected (Armstrong *et al.*, 2017; Neyra *et al.*, 2014; Zaine *et al.*, 2014). *S. aureus* remains a common and virulent cause of DFU infections (Richard *et al.*, 2011; Lipsky *et al.*, 2004). In some studies, close to 50% *S. aureus* isolates associated with DFU are MRSA (Tentolouris *et al.*, 2006; Zetola *et al.*, 2005). In most cases, *S. aureus* strains isolated from DFU have several virulence factors including toxins that play an important role in initiating inflammation and spreading the infection (Viquez-Molina *et al.*, 2018; Sotto *et al.*, 2008), exfoliatins which participate in skin invasion (Nhan *et al.*, 2011), and the *mec*A gene product that confers resistance to β -lactam antibiotics (Vu *et al.*, 2014). The effect of these virulence factors leads to deeper tissue infection, including bones and antibiotic treatment failure (Viquez-Molina *et al.*, 2018).

A quarter of DFU patients will undergo lower limb amputation due to antibiotic-resistant, often MDR, bacterial infections that result in treatment failure (Soge *et al.*, 2016; Brem *et al.*, 2004). Because clinical signs of systemic infections for example leukocytosis and fever are usually not manifested even in a severe cases (Armstrong *et al.*, 2017) and local symptoms and clinical signs of infection diminish due to concomitant peripheral neuropathy and ischemia (Ertugrul *et al.*, 2017), diagnosis and treatment of the DFU infection can be difficult. Therefore, amputation is considered a better outcome measure (Pickwell *et al.*, 2015). After the first lower limb amputation, up to 50% of patients require another amputation within three to five years (Armstrong *et al.*, 2007). Most patients with DFU suffer from a severe infection, an independent predictor of amputation, according to the

International Working Group on the Diabetic Foot (Bakker *et al.*, 2016; Pickwell *et al.*, 2015).

The prevalence of MDR infections in DFU is primarily associated with *S. aureus* (Aysert Yildiz *et al.*, 2018; Trivedi *et al.*, 2014). The failure of antibiotic therapy in some strains of *S. aureus*, such as small colony variants, could be because of the intracellular location and biofilm production capability (Cervantes-Garcia *et al.*, 2015). Identification of causative agents to the species level and the use of effective therapeutic agents after *in vitro* susceptibility tests are important for successful treatment of DFU (Abouhmad *et al.*, 2016; Lipsky and Hoey, 2009).

2.1.3 Identification of S. aureus

The commonly used biochemical identification methods of bacterial pathogens are timeconsuming and often produce incomplete diagnosis (van Belkum *et al.*, 2013). The recently invented bacterial identification tool, matrix-assisted laser desorption/ionization-time of flight mass spectrometry (MALDI-TOF MS), is a robust, reliable, and simple compared to the traditional methods (Fondrie *et al.*, 2018; Perez-Sancho *et al.*, 2018; Harris *et al.*, 2010). MALDI-TOF MS uses soft laser ionization on bacteria or bacterial extracts to detect peptide and protein ions on the cell surfaces based on their relative masses and charges (Edwards-Jones *et al.*, 2000). Several studies have demonstrated the accuracy of MALDI-TOF MS for the identification of *S. aureus* (Perez-Sancho *et al.*, 2018; Elbehiry *et al.*, 2016; Spanu *et al.*, 2011; Harris *et al.*, 2010); hence used in this study.

2.1.4 Antibiotic susceptibility testing of S. aureus isolates

A short turn-around time of antibiotic susceptibility testing is important in improving clinical outcomes of bacterial infectious disease patients through the provision of appropriate and

timely antibiotic treatment (Ha *et al.*, 2018). Modern laboratories worldwide use an automated system for rapid antibiotic susceptibility testing such as the VITEK[®] 2 to test bacteria, including *S. aureus* (Ligozzi *et al.*, 2002; Horstkotte *et al.*, 2002). VITEK[®] 2 uses analysis of minimum inhibitory concentration (MIC) patterns using Advanced Expert System (AES) software. The VITEK[®] 2 system demonstrated acceptable accuracy for a range of antibiotics against *S. aureus* isolates (Nonhoff *et al.*, 2005). According to the National Committee for Clinical Laboratory Standards (NCCLS) breakpoints, VITEK[®] 2 can provide up to 94 - 100% correct category agreement for *S. aureus* (Bobenchik *et al.*, 2014; Ligozzi *et al.*, 2002).

2.1.5 Phages as an alternative treatment for S. aureus infection

The prevalence of MDR *S. aureus* infections (Gupta and Prasad, 2011) and the absence of new antibiotics necessitate the discovery and development of new antimicrobial agents such as phages (Sotto *et al.*, 2008). Lytic phages are potential alternatives to antibiotics because they can kill host bacteria irrespective of their antibiotic susceptibility (Gu *et al.*, 2011). As reviews on the efficacy of lytic phages to control bacterial infections as a replacement or supplement of antibiotics showed (Golkar *et al.*, 2014; Maura and Debarbieux, 2011; Kutateladze and Adamia, 2010; O'Flaherty *et al.*, 2009), development of resistance by bacterial host against phages can be prevented or delayed by using lytic phage cocktails that target diverse bacterial cell surface receptors (Gutierrez *et al.*, 2018; Estrella *et al.*, 2016). Staphylococcal lytic phages have the potential as a therapeutic agent of human infections (Ceyssens *et al.*, 2009).

2.1.6 Phage cocktail to treat S. aureus infection

Most of the phages used for therapy are obligatory lytic, possess a proteinaceous tail, as illustrated in Figure 2.1, and are from *Myoviridae*, *Siphoviridae*, or *Podoviridae* (Gill and Hyman, 2010). A phage or a mixture of phages must have the capability of infecting a wide variety of bacterial host strains (Alves *et al.*, 2014; Synnott *et al.*, 2009; O'Flaherty *et al.*, 2005). Development of phage cocktails is one way to broaden the host range of phages (Manohar *et al.*, 2019) and delay phage resistance development (Alves *et al.*, 2016; Born *et al.*, 2011). Phage cocktail AB-SA01, used in the current study, is a fixed-composition of three naturally occurring and obligately lytic myoviruses related to Staphylococcus phage K. AB-SA01 is intended to treat *S. aureus* infections. It is produced under current good manufacturing practices, and characteristics of the component phages meet regulatory and generally accepted criteria for human use (Law *et al.*, 2019).



Figure 2. 1 Tailed phages (from (Harper and Enright, 2011)).

2.2 Materials and methods

2.2.1 S. aureus isolates

Bacterial isolates presumed to be *S. aureus* were collected from DFU patients seen in diabetic clinics in various hospitals in the Adelaide metropolitan area, South Australia. All specimens were identified using a previously described protocol (Mendes *et al.*, 2013; Blair *et al.*, 1967; Chapman, 1945). South Australia Pathology generously provided the isolates on Muller Hinton agar plates. A colony of each isolate was taken and platted on selective agar for *S. aureus*, mannitol salt agar (MSA) (Thermo Fisher Scientific, Victoria, Australia). After 24 hours of incubation aerobically at 37 °C, yellow or golden colonies surrounded by yellow zones were considered *S. aureus*. After 24 hours incubation at 37 °C, cultures on MSA plates were stored at 4 °C until further analysis.

2.2.2 Identification of S. aureus isolates

At about 18 hours of growth of the presumed *S. aureus* isolates on MSA, an isolated colony or a portion of it was smeared onto the sample spot of a stainless steel MALDI-TOF MS target. A thin film was formed by spreading the bacterial cells evenly using a sterile inoculation stick. One microliter of 70% formic acid was applied to the samples and left to dry at room temperature. Samples were then overlaid with 1 μ l of the alpha-cyano-4hydroxycinnamic acid (HCCA) matrix solution and allowed to dry in a fume hood before putting the sample into the Bruker Daltonik MALDI-TOF MS Biotyper (BRUKER Pty. Ltd., Victoria, Australia) to identify bacterial species as described (Lee *et al.*, 2015; Schumann and Maier, 2014; Lasch *et al.*, 2014; Harris *et al.*, 2010). The results were analyzed using Biotyper 3.0 software. Isolates were considered *S. aureus* when both the first- and secondbest match organism scores were ≥ 2.00 . Reference strains *S. aureus* subsp. *aureus* Rosenbach (ATCC 6538) and *S. aureus* subsp. *aureus* RN4220 were used for quality control. Isolates were stored in 25% glycerol nutrient broth at -150 ^oC until further analysis.

To quickly confirm any doubtful isolates, the latex agglutination test using a Staphaurex* (Thermo Fisher Scientific, South Australia, Australia) was done following the manufacturer's instructions. In summary: a drop of Staphaurex* reagent was dispensed onto a reaction card, a colony was picked and mixed with the Staphaurex*, and any resulting agglutination was examined after rotating the card gently for about 20 seconds. The development of agglutination was considered a positive result, whereas no agglutination was taken as a negative result.

2.2.4 Antibiotic susceptibility of S. aureus isolates

Following the manufacturer's instructions, *S. aureus* isolates were diluted in 0.45% sterile saline solution and standardized between 0.5 and 0.63 McFarland turbidity standards. Next, 280 µl of the bacterial suspension was transferred into 3 ml of a 0.45% saline solution tube, and the tube was placed in the cassette with a susceptibility card in it. The antimicrobial susceptibility test card loaded tube was incubated in the VITEK® 2 machine (bioMérieux Australia Pty Ltd, New South Wales, Australia) for overnight analysis as described earlier (Weber *et al.*, 2017; Gardiner *et al.*, 2013; Cartwright *et al.*, 2013; Ligozzi *et al.*, 2002). A loopful of the suspension was also cultured on blood agar overnight at 37 °C to check contamination during sample handling. Antibiotics included in the susceptibility test card (VITEK[®] 2 AST-P656) were benzylpenicillin, oxacillin, gentamicin, ciprofloxacin, erythromycin, clindamycin, linezolid, daptomycin, teicoplanin, vancomycin, tetracycline, nitrofurantoin, fusidic acid, mupirocin, rifampicin, and trimethoprim-sulfamethoxazole

(TSM). The VITEK[®] 2 system software 8.01 was used to produce antibiotic susceptibility profile data.

2.2.5 Determination of phage titer

Phage cocktail AB-SA01 and each of its phage constituents (J-Sa 36, Sa 83, and Sa 87) were provided by AmpliPhi Biosciences Corporation. Phage solutions were transported and stored under +4 ^oC temperature. All phages were from the *Myoviridae* family, and they were characterized by whole-genome sequence analysis and proved free from bacterial virulence or drug-resistance genes. Moreover, AB-SA01 complied with the U.S. Food and Drug Administration (FDA) and Australia's Therapeutic Goods Administration (TGA) requirements to conduct clinical trials and single-patient emergency treatment (Lehman *et al.*, 2019).

Phage titer was determined using double agar overlay plaque assays as described previously (Mirzaei and Nilsson, 2015; Merabishvili *et al.*, 2009) on susceptible two laboratory strains and two clinical isolates of *S. aureus*. Each phage and the phage cocktail titers were examined in triplicate. For estimating the initial phage concentration, the average number of plaques forming units (PFU) was taken, and the phage titer in terms of PFU in 1 ml phage

suspension was calculated using the equation: $pfu/ml = \frac{Number of plaques}{d \cdot V}$ where d stands for dilution factor and V for the volume of inoculum. The mean was then calculated for the triplicate plates.

2.2.6 Phage efficacy and host range test

To evaluate the lytic efficacy and host range of phages, spot test was performed based on the previously described procedure (Mirzaei and Nilsson, 2015; Alves *et al.*, 2014). Briefly, 100

µl of an overnight culture of each bacterial isolate was mixed with 3 ml of tryptic soy soft agar that was warmed at 42 °C and poured onto pre-warmed at 37 °C tryptic soy agar (TSA) plates. Plates were rotated to ensure even distribution of the soft agar and left to dry at room temperature for 30 minutes. Ten microliters of each phage solution standardized to a titer of 9 log₁₀ (PFU/ml) was serially diluted with tryptic soy broth (TSB). Ten microliters of 10⁻¹, 10⁻³, 10⁻⁵, and 10⁻⁷ dilutions were spotted onto the bacterial lawns in triplicate. Sterile PBS was used as a negative control. Plates were incubated inverted at 37 °C aerobically overnight and examined the following day. Based on the AmpliPhi Biosciences Corporation instruction, the criteria used to assess susceptibility were: (i) resistance if a very week or no activity is seen, (ii) intermediate when faint and turbid spots or partial activity is observed, and (iii) susceptible when clear spot or full activity with no bacterial growth within the spot is observed.

2.2.7 Measurement of the effect of phage cocktail on S. aureus biofilm

Biofilms of two laboratory strains and 52 clinical isolates of *S. aureus* were established in flat-bottom 96-well microtiter plates after 48 hours incubation at 37 °C following the established model and measured using optical density at 600_{nm} (Avila-Novoa *et al.*, 2018; Kouidhi *et al.*, 2010; Rohde *et al.*, 2007). Isolates were assigned based on recommended criteria as: (i) optical density (OD)₆₀₀ < 0.1 non-, (ii) $0.1 \le OD_{600} < 1$ weak-, and (iii) $OD_{600} \ge$ 1 strong-biofilm producers. Phage cocktail treatment was applied as described (Gonzalez *et al.*, 2017; Thomas *et al.*, 2015; Mendes *et al.*, 2014) with the following modifications. *S. aureus* isolates were cultured overnight on MSA at 37 °C. A single colony from each isolate was transferred into a sterile glass tube of 0.45% sterile saline and adjusted to 1.0 McFarland turbidity standard. The suspension was diluted at 1:100 into 1% glucose-TSB, and 150 µl of the suspension was transferred to a sterile flat-bottom 96-well Greiner CELLSTAR[®] polystyrene tissue culture plate (Sigma-Aldrich, New South Wales, Australia). Plates were incubated for 48 hours on a gyratory mixer of 70 revolutions per minute (rpm) at 37 °C.

Following incubation, the liquid part of the culture was removed, and the plates were washed three times with sterile deionized water through gentle pipetting. After the biofilms are airdried, 180 μ l of phage, tetracycline (16 μ g/ml, positive control), or sterile PBS (negative control) solutions in broth were applied. The treatment broths were: (i) 18 μ l phage solution (at 9 log₁₀ (PFU/ml)) mixed with 162 μ l broth, (ii) 3 μ g tetracycline in 180 μ l broth, and (iii) 18 μ l PBS in 162 μ l broth. After 12 hours of incubation in a static condition, the fluid portion of the culture was decanted, and the biofilm was washed twice through gently immersing into distilled water and inverting onto a paper towel. The biofilms were then fixed using 95% methanol for 30 minutes. The methanol was decanted, and the plates were washed once with sterile distilled water and air-dried.

The dried biofilms were stained with 190 μ l of 0.2% crystal violet (CV) per well for 60 minutes. The excess stain was removed by gentle washing twice with distilled water, and the stained biofilm was left in a dark room overnight to dry. The stained biofilm was eluted in 200 μ l of 30% acetic acid for 30 minutes at room temperature. The eluted suspension was transferred into a new micro-plate using 30% acetic acid as a negative control, and OD measurement was performed at 600_{nm} using a FLUOstar® Omega multi-mode microplate reader (BMG LABTECH Pty. Ltd. Victoria, Australia). The OD reading data was analyzed and compared with the mean of the negative controls. The percentage of biofilm biomass reduction designated as %BK was calculated from the absorbance of background-corrected untreated controls (I_C, PBS treated) and the absorbance of the treatments (I_T) as indicated

$$\% \mathbf{BK} = \frac{I_{\rm C} - I_{\rm T}}{I_{\rm C}} \times 100\%$$

previously (Thomas et al., 2015):

2.2.8 Measurement of the effect of phage cocktail on S. aureus isolates in biofilms

Measurement of the effect of phage cocktail treatment on *S. aureus* isolates in biofilms was conducted through colony count following the established procedure (O'Toole, 2016; Gutierrez *et al.*, 2015; Mendes *et al.*, 2013; Sillankorva *et al.*, 2010). On the 48 hours biofilm prepared as above, phage cocktail and control treatments were applied for 12 hours. Next, the treated biofilms were washed twice through gentle immersing into distilled water, and the washing water was removed by inverting onto a paper towel for a few minutes. The biofilm-associated cells were collected with 190 μ l TSB using pipette after scrapping the bottom and wall of the wells with a loop. For the complete disruption of the biofilm and detachment of the cells, the TSB suspension was pipetted up and down 10 times and homogenized with a vortex mixer.

The phage-treated wells cell suspensions were serially diluted in a virucide solution, ferrous ammonium sulfate (FAS) (McNerney *et al.*, 1998), to inactivate free phages. FAS solution was prepared at 10 mM concentration immediately before use and filter sterilized using 0.22 μ m pore size. After 15 minutes incubation at room temperature, 100 μ l of serially diluted solutions were mixed with 3 ml of tryptic soy soft agar warmed at 42 °C and poured onto prewarmed at 37 °C TSA plates. Treatment control wells also passed through a similar process except for FAS treatment. After 24 hours of incubation at 37 °C, colony count was performed, and the number of bacterial cells calculated. Three independent experiments were performed in triplicate. This was conducted on randomly selected one laboratory strain and six clinical isolates of *S. aureus*. Plates with an estimated 30 - 300 colonies were selected for colony count. The mean colony-forming unit (CFU) of the three plates of the same dilution was taken as final. The number of bacterial cells was calculated using the formula B = N/d where B = number of bacteria; N = average number of colonies; and d = dilution factor as previously established (Mendes *et al.*, 2013).

2.2.9 Data management and statistical analysis

Data were double entered, encoded, and stored using Microsoft Excel Spreadsheet. STATA version 16 software was used for statistical analysis. All results are reported as the mean \pm standard error of the mean and are compiled from at least three separate experiments and are expressed as logarithm-transformed values log_{10} (CFU /ml) over time. A comparison of experimental groups was performed using a one-way analysis of variance (two-tailed) or paired 't-test'. A p < 0.05 value was considered statistically significant.

2.3 Results

2.3.1 Isolation of S. aureus isolates using culture

The presumed *S. aureus* isolates were subcultured on MSA. Large yellow colonies surrounded by yellow zones on MSA, as shown in Figure 2.2, which confirmed mannitol fermentation after 24 hours of aerobic incubation at 37 °C, were considered *S. aureus* colonies. A total of 54 clinical *S. aureus* isolates were isolated and stored at -150 °C in 25% glycerol until further use.



Figure 2. 2 Yellow S. aureus colonies and zone around the colonies on MSA (A and B)

2.3.2 Identification of S. aureus isolates using MALDI-TOF MS

All the 54 *S. aureus* isolates sorted by colony characteristics on MSA were examined using MALDI-TOF MS as above, and the 52 isolates were identified as *S. aureus*. The remaining two isolates were *Staphylococcus epidermidis* and *Staphylococcus lugdunensis*. Quality control was performed using *S. aureus* subsp. *aureus* Rosenbach (ATCC 6538) and *S. aureus* subsp. *aureus* RN4220 reference strains.

2.3.3 Identification of S. aureus using latex agglutination test

Isolates that showed different colony characteristics on MSA during subsequent cultures after identified as *S. aureus* through MALDI-TOF MS were also confirmed using the latex agglutination test, as shown in Figure 2.3. The five clinical isolates with doubtful colony characteristics on MSA were confirmed *S. aureus*.



Figure 2. 3 Positive latex agglutination test to identify S. aureus.

2.3.4 Antibiotic susceptibility of S. aureus isolates using VITEK® 2 test

VITEK[®] 2 was used to read MICs through the analysis of bacterial growth with antibiotics in test cards. The result indicated that all the 54 tested *S. aureus* isolates, including laboratory strains, are susceptible to gentamicin, nitrofurantoin, fusidic acid, mupirocin, and rifampicin. Besides, 98.2% (n = 53) of the isolates were susceptible to linezolid, tetracycline, and trimethoprim-sulfamethoxazole (TSM), as shown in Figure 2.4. The highest antibiotic resistance was seen against benzylpenicillin followed by erythromycin, oxacillin, and clindamycin with 77.8% (n = 42), 20.4% (n = 11), 18.5% (n = 10), and 18.5% (n = 10) resistant isolates, respectively.



Figure 2. 4 Antibiotic susceptibility profile of *S. aureus* isolate used in this study The antibiotic susceptibility test also showed that 6.7% (n = 9) of the isolates were MRSA, as shown in Figure 2.5. Worryingly, 13% (n = 7) of the isolates were MDR, resistant to three or more antimicrobial classes.



Figure 2. 5 Category of *S. aureus* due to their antibiotic susceptibility profile: MSSA – methicillin-susceptible *S. aureus*, MRSA – methicillin-resistant *S. aureus*, MDR – multidrug-resistant *S. aureus*, and MDR-MRSA – multidrug- and methicillin-resistant *S. aureus*.

2.3.5 Determination of phage titer

The phage titer test was conducted on susceptible two laboratory strains (RN4220 and SA6538) and two clinical isolates of *S. aureus* (63-2498 and 63-6378), as illustrated in Figure 2.6. The mean titer, presented in plaque-forming unit/millilitre - PFU/ml, was 9.3 log₁₀ (PFU/ml) for J-Sa-36 and Sa-83, 9.0 log₁₀ (PFU/ml) for Sa-87, and 9.1 log₁₀ (PFU/ml) for the combined product AB-SA01. The result showed a mean of 9.2 log₁₀ (PFU/ml) for all the three *S. aureus* phages (J-Sa 36, Sa 83, and Sa 87) used in this study. This phage titer was similar to the AmpliPhi Bioscience Corporation evaluation, which was 9.3 log₁₀ (PFU/ml) for J-Sa 36 and Sa 83, and 9.0 log₁₀ (PFU/ml) for Sa 87.



Figure 2. 6 Plaques of AB-SA01 on S. aureus RN4220

2.3.6 Efficacy of phages on S. aureus isolates

In evaluating the lytic efficacy and the host range of phage cocktail AB-SA01 and its components, a total of 53 *S. aureus*, 51 clinical isolates, and two laboratory strains were spot tested. The result showed that J-Sa 36, Sa 83, Sa 87, and AB-SA01 effectively lysed 75.5% (n = 40), 80% (n = 44), 94.3% (n = 50), and 88.7% (n = 47), respectively, of the isolates with full or partial activity as illustrated in Figure 2.7. Although Sa 87 lysed a greater number of isolates than AB-SA01, the difference was not statistically significant (p > 0.05; 94.3% vs 88.7%).



Figure 2. 7 Lytic efficacy and host range of AB-SA01 and its components on *S. aureus* isolates.

Clear full lysis was observed in 95.8% (n = 45/47), 88% (n = 44/50), 85% (n = 34/40), and 81.8% (n = 36/44) of the susceptible isolates to AB-SA01, Sa 87, J-Sa 36, and Sa 83, respectively, as shown in Figure 2.8. Two *S. aureus* clinical isolates were resistant for all the phages, including the phage cocktail AB-SA01. All the isolates that were resistant against AB-SA01 were either resistant or intermediately susceptible (partial lysis) to all component

phages. All MDR, MRSA, and MDR MRSA isolates, except two MDR MRSA, were susceptible to AB-SA01. There was no statistically significant difference in phage susceptibility (p > 0.05) among antibiotic-susceptible (89.5% AB-SA01 susceptible) and antibiotic-resistant (85.7% AB-SA01 susceptible) isolates.



Figure 2. 8 Lytic activity of AB-SA01 on *S. aureus* isolates: A complete, B intermediate, and C no lysis.

2.3.7 Evaluation of the effect of AB-SA01 on S. aureus isolates biofilm

The effect of AB-SA01 on established biofilm biomass was studied and quantified using the CV staining method, as shown in Figure 2.9. Based on the criteria set in the methods section, 92.5% (n = 49) and 7.5% (n = 4) isolates were weak- and strong-biofilm producers, respectively. One isolate failed to produce a measurable biofilm.



Figure 2. 9 Quantification of biofilm: **A** - CV stained biofilms (T, P, and S represent tetracycline, phage, and PBS treated group, respectively), and **B** - eluted solution of the stained biofilms for OD_{600nm} reading.

The established biofilms were treated with AB-SA01 suspension at one multiplication of infection (MOI) based on the result of a pilot test to determine bacterial density in 48 hours biofilm. After 12 hours of treatment, the biofilm biomass reduction due to AB-SA01 treatment was compared with PBS and tetracycline (TTC) treatment on seven randomly selected isolates biofilms. The biofilm biomass reduction was categorized: (i) insignificant; < 20%, (ii) weak; 20.1- 40.0%, (iii) moderate; 40.1- 60.0%, (iv) strong; 60.1- 80.0 %, and (v) very strong > 80 %. Based on these criteria, 20.8% (n = 11), 15.0% (n = 8), 17.0% (n = 9), 41.5% (n = 22), and 5.7% (n = 3) of the isolates' biofilms showed insignificant, weak, moderate, strong, and very strong biomass reduction, respectively, as shown in Figure 2.10. Biofilms developed from five susceptible *S. aureus* isolates did not show significant biomass reduction because of AB-SA01 treatment.



Figure 2. 10 Effect of AB-SA01 on S. aureus isolates biofilm.

Eighty percent of the biofilms showed a statistically significant reduction in biomass (p < 0.05; 0.13 vs 0.34) as a result of AB-SA01 treatment regardless of the antibiotic susceptibility profile of the isolates. There was no statistical significance difference in biofilm biomass reduction (p > 0.05; 0.18 vs 0.16) between antibiotic-resistant and antibiotic-susceptible *S. aureus* isolates. The maximum biomass reduction due to AB-SA01 treatment was 91.7% in laboratory strains and 81.3% in clinical isolates. The tetracycline treated biofilms also showed statistically significant biomass reduction (p < 0.05; 0.09 vs 0.26). In this study, there was no statistically significant difference (p > 0.05; 0.07 vs 0.09) between AB-SA01 and tetracycline treatment in reducing the biofilm biomass of *S. aureus* isolates, as illustrated in Figure 2.11.



Figure 2. 11 Comparison of biofilm biomass reduction between AB-SA01 and tetracycline (TTC) treatments.

2.4 Discussion

S. aureus abundantly grows on MSA with large colonies surrounded by yellow zones, but other staphylococci produce small pink to red colonies surrounded by red or purple zones while other bacteria are inhibited (Blair *et al.*, 1967; Chapman, 1945). It was noted that *S. aureus* isolates tested in this study showed the colony characteristics of *S. aureus* discussed above on MSA. A few isolates that showed small yellow colonies surrounded by weak yellow zones were further confirmed by latex test based on the criteria for the identification of *S. aureus*. The finding of *S. aureus* identification using latex test in the current study agreed with previous reports (Smole *et al.*, 1998; Fournier *et al.*, 1993; Qadri *et al.*, 1991).

Diagnostic laboratories have widely used MALDI-TOF MS for bacterial identification. MALDI-TOF MS is a rapid, high throughput, reliable, and direct colony method (Tracz *et al.*, 2018; Dingle and Butler-Wu, 2013; Harris *et al.*, 2010; Lay, 2001). Using MALDI-TOF MS, 52 *S. aureus* clinical isolates were identified in the current study from 54 presumed *S. aureus* isolates. Findings similar to this study results are available in studies conducted to confirm the suitability of MALDI-TOF MS in identifying *S. aureus* at the species level (Perez-Sancho *et al.*, 2018; Camoez *et al.*, 2016; Lasch *et al.*, 2014; Harris *et al.*, 2010). A limitation to the use of MALDI-TOF MS is that it cannot identify bacteria to strain level. Hence, advanced techniques such as sequencing are recommended to strain level identification.

Although it is difficult to compare antibiotic-resistance data due to differences in sample size and study designs, a recent report of a high number of antibiotic-resistant *S. aureus* isolates circulating in the community is worrying (Haddad Kashani *et al.*, 2018). In this study, a significant proportion, 16.7%, of *S. aureus* isolates were MRSA, which agreed with previous reports (Ashong *et al.*, 2017; Bessa *et al.*, 2015). However, this finding is lower compared with 50% - 60% MRSA reports (Reveles *et al.*, 2016; Trivedi *et al.*, 2014; Tentolouris *et al.*, 2006). Recent reports show that MRSA isolates are quickly becoming MDR *S. aureus* (Soge *et al.*, 2016; Kim and Jeon, 2016). In this study, more than half of the MRSA isolates were MDR *S. aureus*. This observation was consistent with the previous study findings on wound infections (Bessa *et al.*, 2015; Trivedi *et al.*, 2014; Galkowska *et al.*, 2009).

In line with earlier reports (Uçkay *et al.*, 2015; Diemond-Hernandez *et al.*, 2010), the highest number of resistant isolates were against benzylpenicillin, erythromycin, and oxacillin. The resistance against these antibiotics in the current finding is similar to earlier reports (Lebowitz *et al.*, 2017; Neyra *et al.*, 2014). Vigilant use of the latest generation antibiotics is mandatory as a significant number of resistant isolates against ciprofloxacin, clindamycin, and erythromycin were identified in this study. Resistance against those latest generation antibiotics are available in the literature (Hiramatsu *et al.*, 2014; Sekhar *et al.*, 2014). The repeated antimicrobial treatment may partly explain the overall high rates of MDR *S. aureus* and MRSA in DFU patients due to the chronic and polymicrobial nature of the infection (Bertesteanu *et al.*, 2014; Trivedi *et al.*, 2014; Dalton *et al.*, 2011; Demling and Waterhouse, 2007).

The increase in the prevalence of antibiotic-resistant pathogenic bacteria, including *S. aureus*, has led to an increased search for alternative therapeutic agents with low cost, easy availability, minimum side effects, and high efficacy. Phages may fulfill some of these features and kill bacterial hosts irrespective of their antibiotic susceptibility (Kelly *et al.*, 2011). Hence, phages are increasingly becoming a focus of research (Chan *et al.*, 2013; O'Flaherty *et al.*, 2009). This study shows that at least 75% and 85% of *S. aureus* isolates are susceptible to each component phage and phage cocktail AB-SA01, respectively. Similar efficacy and host range were reported for Sa 83, and Sa 87 phages on antibiotic-resistant and

antibiotic-susceptible *S. aureus* isolates collected from chronic rhinosinusitis patients (Zhang *et al.*, 2018).

Although it is usually envisaged that phage cocktails are better effective (synergy) than single phage treatment to a particular bacterial host (Schmerer *et al.*, 2014), there is a rare possibility that coinfection of component phages of the phage cocktail to a bacterium negatively affect the outcome of phage infection (Abedon, 1990). There may be repeated secondary phage adsorptions resulting in the continuous induction of lysis inhibition (Abedon, 1992). This lysis inhibition occurs when the second phage adsorption extends the latent period of the first phage (Cairns and Payne, 2008; Abedon, 1992; Abedon, 1990). This phenomenon could be suggested as one of the reasons for the higher host range observed in one of the phage components than the phage cocktail used in this study. As a limitation to this study, specific receptors were not identified for each component phage of AB-SA01.

In the present study, the phage cocktail AB-SA01 produced a better quality of lysis (clear and full lysis) than any of its components. Clear and full lysis by the phage cocktail was observed on more than 95% of the susceptible isolates, including those resistant to or partially lysed by the component phages. The better efficacy of the phage cocktail shows that another component phage could lyse resistant isolate to one component phage in the mixture (Lehman *et al.*, 2019). The use of a phage mixture is mostly preferred over single phage as it results in a decreased rate of resistant mutant development (Gu *et al.*, 2012; Tanji *et al.*, 2005). The current findings were consistent with previous studies on the efficacy of various phage cocktails and component phages of *S. aureus* (Leskinen *et al.*, 2017; Mendes *et al.*, 2014; Kelly *et al.*, 2011). The findings of this study indicate that AB-SA01 and its components are effective on antibiotic-resistant and antibiotic-susceptible *S. aureus* isolates.

Drug Administration (FDA) and Australia's Therapeutic Goods Administration (TGA) requirements to conduct clinical trials and single-patient emergency treatment (Lehman *et al.*, 2019).

Most of the *S. aureus* isolates used in this study produced low-grade biofilm. This finding agrees with reports on *S. aureus* strains isolated from various sources (Avila-Novoa *et al.*, 2018; Szczuka *et al.*, 2013; Tang *et al.*, 2013; Gutierrez *et al.*, 2012). The present study result also lends strong support to the understanding that 99% of microbes present in ecosystems may grow as biofilms (Parasion *et al.*, 2014). The high rate of MRSA, MDR *S. aureus*, and MDR MRSA is also consistent with the fact that the ability to form biofilm increases resistance to antibiotics (Cervantes-Garcia *et al.*, 2015; Percival *et al.*, 2012).

S. aureus is known for producing biofilm in wounds that may be one cause of persistent and relapsing antibiotic-resistant infections in DFU patients (Shettigar *et al.*, 2016; Cervantes-Garcia *et al.*, 2015; Bjarnsholt *et al.*, 2008). Limited studies suggest the significant potential of phages to reduce or eliminate biofilms (Drilling *et al.*, 2014; Alves *et al.*, 2014; Kelly *et al.*, 2012; Cerca *et al.*, 2007). AB-SA01 showed a significant decrease in the biofilm biomass on the laboratory strains and clinical isolates of *S. aureus*. All isolates that were resistant and intermediately susceptible to AB-SA01 in the planktonic assays showed little (< 10%) to no biofilm biomass reduction. Biofilms developed from five *S. aureus* isolates, which were strongly susceptible to AB-SA01 in spot test, also did not show significant biomass reduction. The insignificant biofilm biomass reduction in about 20% of the isolates during phage cocktail treatment in this study was not unexpected. It is because bacteria in the biofilm state are more resistant to phage attacks than their counterparts in the planktonic phase (Fu *et al.*, 2010).

Studies have shown the promising efficacy of phage mixtures on the reduction of biofilm biomass of nonhuman and human clinical isolates of *S. aureus* (Alves *et al.*, 2014; Kelly *et al.*, 2011). Here, the biofilm biomass reduction capability of phage cocktail AB-SA01 on biofilms produced by *S. aureus* isolates collected from DFU patients was examined. This study suggests that AB-SA01, a phage cocktail that has undergone two Phase I trials and met the cGMP standard (Lehman *et al.*, 2019), therapy could provide a practical alternative to treat multidrug-resistant and biofilm-associated *S. aureus* infections. The present study shows the effectiveness of a phage cocktail on MDR MRSA and MDR *S. aureus* clinical isolates collected from DFU patients and their biofilms.

After trialing disinfectants (triton x-100, 2.5% cetylpyridinium chloride (CPC)) and tetracycline as a positive control (data provided under appendices), tetracycline 16 μ g/ml was found to be better in reducing biomass on 48 hours biofilms than the disinfectants tested. Besides, tetracycline was selected as a positive control since it was effective on both *S. aureus* and *P. aeruginosa* isolates selected for mixed-species planktonic and biofilm cultures treatment experiments (discussed in chapter 4). Hence, tetracycline was found to be an appropriate positive control that substantially decreased the biofilm biomass in most isolates tested. There was no statistically significant difference in biofilm biomass reduction (p < 0.05) between AB-SA01 and tetracycline treatment. In this study, a higher concentration of tetracycline (16 μ g/ml) was used because the MIC for all isolates obtained during the VITEK[®] 2 test was < 8 μ g/ml. The reasons for biofilm's resistance to disinfectants and antibiotics are usually multifaceted but mainly due to intrinsic resistance mechanisms and adaptive stress responses (Stewart, 2002).

Some reports indicated that biofilms have water-filled tubules that facilitate phage movement within the biofilms (Donlan, 2009; Sutherland *et al.*, 2004; Doolittle *et al.*, 1996). However,

other reports suggested that the presence of bacterial proteolytic enzymes that can inactivate phages, and the dormant metabolic state of the bacterial cells in the biofilm could hamper phage lytic activity (Gutierrez *et al.*, 2015; Cerca *et al.*, 2007; Doolittle *et al.*, 1996). These deterring factors could be the possible explanations for the lesser efficacy of AB-SA01 in the biofilm than planktonic states in this study.

In summary, this study describes the efficacy of phage cocktail AB-SA01 and its phage constituents against pathogenic and antibiotic-resistant *S. aureus* from DFU patients. Each phage and their combination showed a broad host range of infectivity on *S. aureus* isolates. Furthermore, the phage cocktail demonstrated efficacy in biofilm biomass reduction superior or similar to tetracycline, suggesting AB-SA01 is the right candidate for further development to treat chronic wound infections caused by *S. aureus*. There was no statistically significant difference in phage sensitivity between antibiotic-resistant and antibiotic-sensitive *S. aureus* isolates, shows the promising prospect of AB-SA01 and its components as alternatives to antibiotics.

Chapter 3

Effect of Phage Cocktail AB-PA01 on *Pseudomonas aeruginosa* Diabetic Foot Ulcer Clinical Isolates and Biofilms

Chapter 3: Effect of Phage Cocktail AB-PA01 on *Pseudomonas aeruginosa* Diabetic Foot Ulcer Clinical Isolates and Biofilms

Abstract

Pseudomonas aeruginosa infections are common amongst diabetic foot ulcer (DFU) patients. Most P. aeruginosa strains are antibiotic-resistant and form a biofilm, both of which may lead to treatment failure. This study describes the effect of a cGMP-quality phage cocktail AB-PA01 and four of its components on P. aeruginosa isolates from DFU infection patients, with phage applied to bacteria in planktonic and biofilm states. The objective of this study was to assess the activity of the four P. aeruginosa phages and their mixture, AB-PA01, against P. aeruginosa cells and biofilms in vitro. Bacterial isolates grown in the planktonic state were treated with AB-PA01 and its components and their biofilms with AB-PA01. The effect of phages on planktonic cells was assessed using a spot test. Biofilm biomass was measured using a crystal violet (CV) assay. Of 41 isolates tested, nearly 93% were strongly susceptible to AB-PA01. Moreover, 66% - 88% of isolates were susceptible to each of the four phages. AB-PA01 treatment significantly reduced biofilm biomass (p < 0.001), regardless of the antibiotic-resistant characteristics of the isolates. Furthermore, a single dose of a phage cocktail is capable of significantly reducing biofilms formed *in vitro* by a range of P. aeruginosa isolates from DFU patients. The lytic efficacy and broad host range of AB-PA01 and its components suggest the enormous potential of phages in the treatment of P. aeruginosa infections.
3.1 Introduction

3.1.1 P. aeruginosa in DFUs

DFUs cause substantial morbidity, impaired quality of life, and prolonged antibiotic treatment and hospitalization; they may require surgical interventions, and impose high health care cost (Wu *et al.*, 2017; Petrakis *et al.*, 2017; Lipsky *et al.*, 2016; Lipsky *et al.*, 2012a; Lavery *et al.*, 2006; Lavery *et al.*, 2003). Nearly 25% of people living with diabetes will be affected by DFUs in their lifetime, and 80% of diabetic-related lower-extremity amputations are due to DFUs (Hingorani *et al.*, 2016; Boulton *et al.*, 2005). Most DFU cases become infected (Lavery *et al.*, 2003). Often, DFU infections are polymicrobial (Mottola *et al.*, 2016; Dowd *et al.*, 2008) and associated with MDR microorganisms (Murali *et al.*, 2014). The common causative bacteria in DFUs differ by geographical regions; for instance, *P. aeruginosa* is common in warm and humid climates found in parts of Asia and Africa (Wu *et al.*, 2017; Hatipoglu *et al.*, 2014).

P. aeruginosa is a common pathogen in DFU infections, which associates with prolonged duration of antibiotic therapy and poorer outcomes (Commons *et al.*, 2015; Illgner *et al.*, 2013; Lipsky *et al.*, 2010). *P. aeruginosa* can establish an active and persistent infection in diabetic wounds (Goldufsky *et al.*, 2015). It often resides deep in tissues and is protected from antibiotics and the immune system due to biofilm formation. Another reason for the seriousness of DFU infections with *P. aeruginosa* is the ability of the bacteria to express virulence factors such as flagella, pili, and lipopolysaccharide. Moreover, toxin production and quorum sensing of *P. aeruginosa* significantly contribute to its pathogenesis and poor treatment outcome in DFUs (Ertugrul *et al.*, 2018; Høgsberg *et al.*, 2011). These factors lead to higher rates of unsuccessful treatment and often to lower extremity amputations (Petrakis *et al.*, 2017). DFU infections impose a significant health and financial problem on patients

and substantial economic pressure on health care institutions all over the globe (Petrakis *et al.*, 2017).

3.1.2 Identification of P. aeruginosa

P. aeruginosa can be challenging to identify using routine techniques as it is inert for some biochemical activities and difficult to interpret some of the phenotypic characteristics (Kacaniova *et al.*, 2019; Marko *et al.*, 2012). MALDI-TOF MS, a simple, cost-effective, rapid, discriminatory, and reliable technique, effectively identifies *P. aeruginosa* colonies with high sensitivity and specificity (Kacaniova *et al.*, 2019; Gautam *et al.*, 2017; Barnini *et al.*, 2015; Marko *et al.*, 2012). MALDI-TOF MS is used for species identification in many diagnostic microbiological laboratories (Angeletti, 2017; Schaumann *et al.*, 2012). It uses soft laser ionization to detect peptide and protein ions based on their relative masses and charges for identification since each bacterium has different spectra based on their mass charge ratio (Harris *et al.*, 2010; Bernardo *et al.*, 2002; Edwards-Jones *et al.*, 2000).

MALDI-TOF MS can detect and distinguish hundreds of proteins and peptides in seconds through a comparison of the spectra with those in a reference database (Angeletti, 2017; Barreiro *et al.*, 2017; Gagnaire *et al.*, 2012). In most cases, a mass-to-charge ratio of \geq 2.00 is recommended for species-level identification (Camoez *et al.*, 2016; Elbehiry *et al.*, 2016; Harris *et al.*, 2010). These principles were employed to identify *P. aeruginosa* clinical isolates collected from DFU patients who visited various hospitals in the Adelaide area during the study period.

3.1.3 Antimicrobial susceptibility of P. aeruginosa

Antibiotic therapy with proper wound dressing, including surgical intervention, where appropriate, is the primary treatment option for DFU infection (Ertugrul *et al.*, 2017). Prompt

identification of causative agents of DFU infection could limit indiscriminate use of broadspectrum antibiotics and improves antibiotic stewardship (Lebowitz *et al.*, 2017). *P. aeruginosa* has become resistant to many commonly prescribed antibiotics due to resistance mechanisms, including efflux pumps, antibiotic degrading enzymes, low membrane permeability, and strong biofilm formation capability (Mah *et al.*, 2003; Drenkard and Ausubel, 2002; Stewart, 2002). Broad-spectrum antibiotic treatment carries a risk of selection of resistant microorganisms and affecting normal flora of the skin, hence selected antibiotics with known efficacy on the specified pathogen should be used (Barwell *et al.*, 2017).

Better treatment outcomes are often associated with timely and appropriate antibiotic treatment based on the identified pathogen susceptibility pattern (Wu et al., 2017). The sensitivity and specificity of antimicrobial susceptibility tests for P. aeruginosa have a longstanding concern and better explained through inter-methods comparisons of automated systems (Pfaller *et al.*, 2014). Reports showed that VITEK[®] 2 performance is satisfactory for antimicrobial susceptibility testing of Gram-negative bacteria, including P. aeruginosa (Bobenchik *et al.*, 2017; Juretschko *et al.*, 2007). VITEK[®] 2 comprises a database of MICs for various antibiotics and the prevalent resistance mechanisms in different species, together with a series of algorithms (Bobenchik et al., 2015; Lavallee et al., 2010; Saegeman et al., 2005). The VITEK[®] 2 system automatically measures a turbidity signal for each test well containing an antibiotic, every 15 minutes for up to 18 hours. These data are used to generate growth curves, and MIC of each antibiotic is estimated by comparison with a control (Saegeman et al., 2005; Sanders et al., 2000; Doern et al., 1997). All P. aeruginosa species examined in the current study were evaluated for antibiotic susceptibility profile using VITEK[®] 2.

3.1.4 Phages as an alternative treatment for P. aeruginosa

The growing evidence of an increasing number of MDR *P. aeruginosa* strains has prompted the search for new therapeutic agents. Potential alternatives are lytic phages, which have the self-replicating ability, high specificity, and abundance in the environment (El Didamony *et al.*, 2015; Sulakvelidze, 2013). Phages are potentially effective against MDR bacterial infections because the mechanism by which they produce their lytic effect is not related to antibiotic resistance mechanisms (Shivshetty *et al.*, 2014). Phages can reportedly lyse their host bacteria even when inside biofilms (Holguin *et al.*, 2015; Krylov *et al.*, 2013; Kumari *et al.*, 2009; McVay *et al.*, 2007). Information about phage susceptibility of chronic DFU infections causing *P. aeruginosa* isolates and availability of effective phages is limited.

Exclusively lytic *P. aeruginosa* phages are diverse and distributed in at least seven genera in which different host ranges are observed (Essoh *et al.*, 2013). Narrow host range and emergence of resistant host strains are frequently listed among the shortcomings of phage therapy (Kwiatek *et al.*, 2015). Phage cocktail application is recommended to alleviate phage resistant problem and broaden the host range (Bernasconi *et al.*, 2017; Alves *et al.*, 2016; Schmerer *et al.*, 2014; Chan *et al.*, 2013; Gu *et al.*, 2012; Kelly *et al.*, 2011).

3.1.5 Phage cocktail to treat P. aeruginosa infections

It is possible to broaden the host range of phage treatment by combining various lytic phage types possessing different host ranges. Hence, phage cocktails may be more effective than single phage to treat bacterial diseases (Alves *et al.*, 2016; Born *et al.*, 2011). Furthermore, phages in the cocktail should be chosen to complement each other so that resistant mutants to one phage are susceptible to another (Lehman *et al.*, 2019; Mendes *et al.*, 2014). There are reports of effective therapeutic phage cocktail formulations to treat diseases ranging from

skin wound to systemic infections (Manohar *et al.*, 2019; Seo *et al.*, 2018; Ramirez *et al.*, 2018; Schooley *et al.*, 2017; Leskinen *et al.*, 2017; Bernasconi *et al.*, 2017; Gundogdu *et al.*, 2016; Alves *et al.*, 2016; Abdulamir *et al.*, 2015; Mendes *et al.*, 2014; Hooton *et al.*, 2011; Fu *et al.*, 2010; Merabishvili *et al.*, 2009; O'Flynn *et al.*, 2004). The phage cocktail, AB-PA01, evaluated in this study, is a constant-composition of naturally occurring and obligately lytic two *Myoviridae* and two *Podoviridae*. AB-PA01 is produced to treat *P. aeruginosa* infections, and none of its components has any known bacterial virulence or antibiotic resistance genes (Fong *et al.*, 2017).

3.2 Materials and methods

3.2.1 Identification of P. aeruginosa isolates

P. aeruginosa clinical isolates were collected from infected DFU patients from diabetic foot clinics in Adelaide, South Australia, during the study period and cultured on Muller Hinton agar. Single colonies of the presumed *P. aeruginosa* isolates from Muller Hinton agar plates that were generously provided by South Australia Pathology were subcultured onto vancomycin-supplemented MacConkey agar (Thermo Fisher, South Australia) following the established protocol (Eckmanns *et al.*, 2008; Kirisits *et al.*, 2005; Wuthiekanun *et al.*, 1990; Brodsky and Nixon, 1973). After 24 hours of incubation at 37 ^oC, MacConkey plate cultures were stored under 4 ^oC until further analysis. *P. aeruginosa* laboratory strains ATCC 15692 (PAO1) and ATCC 10145, generously provided by Dr. Nicky Thomas from the University of South Australia and Professor Melissa Brown from Flinders University, respectively, were used as quality control references.

A colony of each *P. aeruginosa* isolates showing typical characteristics on 18 hours vancomycin-supplemented MacConkey agar culture was applied to a MALDI-TOF MS sample plate using a sterile inoculation stick. Vancomycin was applied on MacConkey agar as a supplement to hinder Gram-positive bacteria growth. Vancomycin is one of the commonly prescribed antibiotics to treat most Gram-positive bacterial infections and was effective on all *S. aureus* isolates tested in this study. One microliter of 70% formic acid and 1 μ l of the alpha-cyano-4-hydroxycinnamic acid (HCCA) (Sigma-Aldrich Pty Ltd, New South Wells, Australia) matrix was applied following the manufacturer's instructions. Identification of the bacterial species was performed and analyzed using Bruker Daltonik MALDI Biotyper 3.0 (Bruker Pyt Ltd, Victoria, Australia), as previously described

(Kacaniova *et al.*, 2019; Hoyos-Mallecot *et al.*, 2014). The isolates were considered *P. aeruginosa* when the scores are ≥ 2.00 as both first- and the second-best match organism. After identification, all isolates were stored in 25% glycerol nutrient broth at -150 °C until further analysis.

3.2.2 Antibiotic susceptibility profile of *P. aeruginosa* isolates

P. aeruginosa isolates identified as above by MALDI-TOF MS were cultured on vancomycin-supplemented MacConkey agar for 18 - 24 hours at 37 °C aerobically. Three colonies were mixed with 3 ml of 0.45% saline and adjusted to the turbidity of a 0.5 - 0.63 McFarland standard following the manufacturer's instruction. Then, 145 μ l of the bacterial suspension was mixed with another 3 ml of 0.45% saline solution and loaded with antimicrobial susceptibility test cards and incubated in the VITEK[®] 2 machine (bioMérieux Australia Pty Ltd, New South Wales, Australia) for overnight analysis as described earlier (Juretschko *et al.*, 2007; Sader *et al.*, 2006; Bruins *et al.*, 2004; Sanders *et al.*, 2000; Doern *et al.*, 1997). A loopful of the suspension was also cultured on blood agar overnight at 37 °C to check the presence or absence of contamination during sample handling. The susceptibility test card (VITEK[®] 2 AST-N246) contains the following antibiotics: ticarcillin/clavulanic acid, piperacillin/tazobactam, ceftazidime, cefepime, meropenem, amikacin, gentamicin, tobramycin, ciprofloxacin, and norfloxacin.

3.2.3 Phage titer

Phage cocktail AB-PA01 and its components Pa 193, Pa 204, Pa 222, and Pa 223 donated by AmpliPhi Biosciences Corporation were used in this study. Pa 193 and Pa 204 are *Myoviridae*, and Pa 222 and Pa 223 are *Podoviridae*. The titer of the phage cocktail and component phages was determined using double agar overlay plaque assay, as indicated

previously (Mirzaei and Nilsson, 2015; Merabishvili *et al.*, 2009) on known susceptible *P*. *aeruginosa* laboratory strains PAO1 and PA10145 and two clinical isolates.

Dilution of phage solutions up to 10^{-8} dilution was performed in TSB after filter sterilization. One-hundred microliters of each phage and phage cocktail solution from $10^{-4} - 10^{-8}$ dilutions and 100 µl of 18 hours broth culture of each strain or isolate were mixed with 3 ml trypticase soy soft agar maintained at 42 °C, uniformly dispensed over pre-warmed at 37 °C TSA plates, and incubated overnight at 37 °C. Plaques were counted on the following day, and the phage titer was expressed as log-transformed plaque-forming units per milliliter (PFU/ml).

Each phage and phage cocktail solution was tested three times for each dilution. PFU counting was conducted on plates estimated to contain 30 - 300 distinguishable homogenous PFU from each dilution. The average number of PFU was taken and calculated by the following equation to estimate the original stock phage solution titer,

$$pfu/ml = \frac{Number of plaques}{d \cdot V}$$
, where d stands for dilution factor and V for a volume of inoculum,

the mean was then calculated for the three plates.

3.2.4 Phage efficacy and host range test

Evaluation of lytic efficacy of phage cocktail AB-PA01 and its components using a spot test was performed based on the established procedure (Mirzaei and Nilsson, 2015; Alves *et al.*, 2014). Briefly, 100 μ l of 16 - 18 hours culture of *P. aeruginosa* isolates was mixed with 3 ml of trypticase soy soft agar and poured onto pre-warmed at 37 °C TSA plates. After ensuring the even distribution of the soft agar suspension, plates were left to dry for 30 minutes at room temperature. Ten microliters of each phage and phage cocktail solutions adjusted to 9.0 \log_{10} (PFU/ml) titer were serially diluted with TSB, and 10 μ l of 10⁻¹, 10⁻³, 10⁻⁵, and 10⁻⁷

dilutions were spotted onto the bacterial lawns in triplicate. Ten microliters of sterile PBS were used as a positive control.

The plates were kept at room temperature for 30 minutes and incubated inverted at 37 ^oC overnight. This assay was performed in triplicate. The susceptibility patterns of the isolates were examined the following day. The observation was recorded according to AmpliPhi Biosciences Corporations' recommendation: (i) resistant if very weak to no activity is seen, (ii) intermediate when faint and turbid spots or partial activity is observed, and (iii) susceptible when a clear spot or full activity with no bacterial growth is observed within the spots. The results taken as final were the mean of the triplicate tests.

3.2.5 Effect of AB-PA01 on P. aeruginosa biofilms

The biofilm development on a flat-bottom transparent CELLSTAR polystyrene 96-well microplate (Sigma-Aldrich, NSW, Australia) was carried out as detailed earlier (Alves *et al.*, 2016; Pires *et al.*, 2011; Spoering and Lewis, 2001). *P. aeruginosa* isolates were cultured overnight on vancomycin-supplemented MacConkey agar under the aerobic condition at 37 ^oC to yield isolated colonies. After 18 hours of incubation, a single colony was transferred into a sterile glass test tube of 0.45% saline and adjusted to 1.0 McFarland turbidity standard.

The bacterial suspension in saline was diluted by a 1:100 ratio in 1% glucose Luria-Bertani (LB) broth, and 150 μ l of the suspension was transferred to the sterile wells of a microplate. After 48 hours of incubation at 37 °C with agitation by a gyratory mixer at 70 rpm, the liquid part of the culture was removed through gentle pipetting. The plates were washed with sterile deionized water three times by careful pipetting not to disrupt the biofilm. Plates were left at room temperature in a biosafety cabinet to dry in the air, and then 180 μ l phage-, gentamicin(positive control), tetracycline- (positive control), and sterile PBS-broth (negative control) solutions were applied.

The treatment-broth solutions were constituted: (i) 18 µl phage solution (at 9 \log_{10} (PFU/ml))) mixed with 162 µl LB broth, (ii) 23 µg tetracycline in 180 µl LB broth, (iii) 16 µg gentamicin in 180 µl broth, and (iv) 18 µl PBS in 162 µl LB broth. The concentration of tetracycline was 128 µg/ml because *P. aeruginosa* isolates are susceptible for a higher concentration of tetracycline (Zheng *et al.*, 2017; Lister *et al.*, 2009; Pai *et al.*, 2001; Li *et al.*, 1994; Petersen *et al.*, 1999). The higher MIC of gentamicin than from VITEK[®] 2 test results was used for biofilm treatment.

The treated biofilms were incubated at 37 $^{\circ}$ C under aerobic and static conditions. After 12 hours of incubation, the fluid portion of the culture was gently removed, biofilms were washed twice through gentle immersing of the whole microplate into distilled water and dry the microplates gently onto a paper towel. Then, the biofilms were fixed using 95% methanol for 30 minutes. Methanol was decanted, and plates were washed once with sterile distilled water and air-dried. The fixed biofilms were then stained with 190 µl per well of 0.2% CV. After staining for 60 minutes, excess CV stain was removed by gentle washing twice using distilled water and left in a dark room overnight to dry. The stained biofilm was eluted at room temperature for 30 minutes using 200 µl of 30% acetic acid.

The eluted suspension was transferred into a new microplate, and the optical density of the suspension was measured at 600 nanometers (600_{nm}) after including 30% acetic acid as a negative control. A FLUOstar[®] Omega multi-mode microplate reader (BMG LABTECH Pty. Ltd., Victoria, Australia) was employed to read the optical density following the manufacturer's instruction. The OD reading data was analyzed and compared with the blank-corrected mean of the three replicates. The percentage of biofilm biomass reduction (%BK)

was calculated from the absorbance of background-corrected untreated controls (I_C , PBS treated) and the absorbance of the treatments (I_T) (Thomas *et al.*, 2015) as follows:

$$\% \mathrm{BK} = \frac{I_\mathrm{C} - I_\mathrm{T}}{I_\mathrm{C}} \times 100\%$$

Based on the biofilm biomass they produced, the isolates were categorized as (i) non-biofilm producer if $OD_{600} < 0.1$, (ii) weak biofilm producer when $0.1 \le OD_{600} < 1$, and (iii) strong biofilm producer if $OD_{600} \ge 1$ based on literature (Avila-Novoa *et al.*, 2018; Rohde *et al.*, 2007). The biofilm biomass reduction as a result of treatment effect was categorized: (i) insignificant $\le 20\%$, (ii) weak 20.1- 40.0%, (iii) moderate 40.1- 60.0%, (iv) strong 60.1- 80.0%, and (v) very strong $\ge 80\%$ reduction.

3.2.6 Data management and statistical analysis

Microsoft Excel Spreadsheet was used to double enter, encode, and store the data. STATA version 16 software was used for statistical analysis. Data are reported as the mean \pm standard error of the mean and are compiled from at least three separate experiments and are expressed as logarithm-transformed values \log_{10} (CFU /ml) over time. A comparison of experimental groups was performed using a one-way analysis of variance (two-tailed) or paired 't-test'. A statistical significance was considered at p < 0.05.

3.3 Results

3.3.1 Isolation of P. aeruginosa using culture

Swab samples from wounds of DFU patients who visited various diabetic clinics in Adelaide were collected and cultured aerobically at 37 °C onto Muller Hinton agar plates by South Australia Pathology. Colonies presumed to be *P. aeruginosa* isolates were subcultured onto another Muller Hinton agar and sent to our laboratory for further identification, antimicrobial susceptibility, and phage treatment experiments. Mucoid and pigmented or sometimes metallic sheen, sticky (hyper-adherent), and aggregative colonies on Muller Hinton agar were considered *P. aeruginosa* isolates. Isolated colonies were taken and plated on 1.2% MacConkey agars supplemented with vancomycin to suppress the growth of Gram-positive bacteria. Based on colony characteristics on vancomycin-supplemented MacConkey agar, 44 *P. aeruginosa* clinical isolates were isolated. These were stored in 25% glycerol nutrient broth at -150 °C until further analysis.

3.3.2 Identification of P. aeruginosa using MALDI-TOF MS

All presumed *P. aeruginosa* isolates were typed using MALDI-TOF MS, and results were analyzed with Biotyper 3.0. Isolates with ≥ 2.00 score for both the first- and second-best matches were identified as *P. aeruginosa*. Of 44 clinical isolates tested, 88.6% (n = 39) were confirmed *P. aeruginosa*. The remaining isolates were *Proteus mirabilis* (2), *Serratia marcescens* (1), *Enterococcus faecalis* (1), and *Klebsiella pneumonia* (1). Reference *P. aeruginosa* strains PAO1 and PA10145 were also confirmed with good scores. All confirmed isolates were tested for antibiotic susceptibility against commonly prescribed drugs using the VITEK[®] 2 system.

3.3.3 Antimicrobial susceptibility of *P. aeruginosa* using VITEK[®] 2

All 41 *P. aeruginosa* isolates were susceptible to meropenem, amikacin, and tobramycin, as shown in Figure 3.1. Nearly 98% (n = 40) and 95% (n = 39) isolates were also susceptible to norfloxacin and gentamicin, respectively. The highest proportion of resistant isolates, 48.8% (n = 20), was found against ticarcillin/clavulanic acid. The next higher number of resistant isolates, 14.6% (n = 6), was observed against piperacillin/tazobactam. One isolate was resistant to half of the antibiotics (n = 5) it was exposed to and was considered MDR.



Figure 3. 1 Antibiotic susceptibility of *P. aeruginosa* isolates collected from DFU patients.

3.3.4 Titer of the phage cocktail AB-PA01 and its components

Plates with clear isolated plaques, as shown in Figure 3.2, were selected for each phage and phage cocktail PFU count. The finding was $10.4 - 10.5 \log_{10}$ (PFU/ml) mean titer for Pa 193, Pa 204, and Pa 222, and 9.5 \log_{10} (PFU/ml) mean titer for Pa 223. The mean titer of the phage cocktail AB-PA01, the combination of these four phages, was 10.3 \log_{10} (PFU/ml).

Consistent phage titers were obtained for each phage and the phage cocktail on the two clinical isolates and two laboratory strains.



Figure 3. 2 PFU of AB-PA01 on P. aeruginosa PA01 during evaluation phage titer.

3.3.5 Lytic efficacy and host range of phages on spot test

P. aeruginosa isolates were considered susceptible to phages when partial or full activity was observed on the spot test, as shown in Figure 3.3. Of the 41 isolates tested, 87.8% (n = 36), 73.1% (n = 30), and 65.9% (n = 27) were lysed by Pa 193 and Pa 222, Pa 204, and Pa 223, respectively, as illustrated in Figure 3.4. Five percent (n = 2) isolates were resistant against all component phages separately tested in this study. One isolate was resistant to all phages and the phage cocktail.



Figure 3. 3 Lytic activity of AB-PA01 on *P. aeruginosa* isolates: A complete lysis, **B** intermediate lysis, and **C** no lysis.

The phage cocktail, AB-PA01, lysed 92.7% (n = 38) of the isolates, and 89.5% (n = 34) of the lysis was complete, which was large and clear plaque without visible colony inside the spotted area. Isolates resistant against AB-PA01 were also resistant to at least two of the component phages. One *P. aeruginosa* isolate, which was MDR during the VITEK[®] 2 test, was susceptible to all component phages and the phage cocktail. In most of the cases, isolates that showed resistance to one phage also showed resistance to at least one other phage tested. Two *P. aeruginosa* isolates were resistant to both *Myoviridae* phages, and the other two isolates were resistant to *Podoviridae* phages only.



Figure 3. 4 Phage susceptibility of *P. aeruginosa* isolates collected from DFU patients.

3.3.6 Effect of AB-PA01 on biofilms of P. aeruginosa isolates

Of the 41 *P. aeruginosa* isolates cultured for biofilm development on 96-well as described in the material and method section, 97.6% (n = 40) isolates produced biofilm. The OD reading of the biofilm at 600_{nm} ranged from 0.17 to 3.14. Accordingly, more than 73% (n = 30) of the isolates were strong biofilm producers, as shown in Figure 3.5.



Figure 3. 5 Biofilm formation capability of *P. aeruginosa* isolates from DFU patients.

After treating the 48 hours biofilms with AB-PA01 at MOI 1 for 12 hours and staining with CV, the biofilm biomass was measured using the FLUOstar[®] spectrophotometer at OD_{600nm} . The biofilm biomass reduction due to AB-PA01 treatment was compared with PBS treatment. Tetracycline and gentamicin were used as positive controls. Based on the criteria described above, 17.5% (n = 7), 22.5% (n = 9), 30% (n = 12), and 12.5% (n = 5) of the isolates showed weak, moderate, strong, and very strong biofilm biomass reduction, respectively, due to AB-PA01 treatment effect as shown in Figure 3.6.

In summary, 82.5% of isolates biofilms showed significant biomass reduction because of AB-PA01 treatment. Of the remaining 17.5% (n = 7) isolates, four isolates did not show significant biofilm biomass reduction (< 20%), whereas the other three showed 10.6 - 26.3% biofilm biomass increase after phage cocktail treatment. The maximum biomass reduction observed in this study was 88.2%. Overall, statistically significant (p < 0.001; 0.82 vs 1.60) biofilm biomass reduction was observed because of AB-PA01 treatment compared to the negative treatment (PBS) control group.



Figure 3. 6 Effect of AB-PA01 on *P. aeruginosa* isolates biofilm biomass reduction.

There was no statistically significant difference (p > 0.05; 0.60 vs 0.58) between AB-PA01 and tetracycline treated group in biofilm biomass reduction, as shown in Figure 3.7. The OD_{600nm} reading of AB-PA01- and tetracycline-treated biofilms were substantially reduced compared with PBS treated (negative control) biofilms. Generally, the biofilm biomass reduction because of phage cocktail AB-PA01 and tetracycline treatment was strongly statistically significant (p < 0.001; 0.21 vs 0.81 and 0.23 vs 0.81) compared with the negative control.



Figure 3. 7 Comparison of the effect of AB-PA01 and tetracycline (TTC) treatment on biofilm biomass reduction of *P. aeruginosa* isolates.

Treating biofilms with gentamicin did not produce statistically significant biofilm biomass reduction (p > 0.05; 1.4 vs 1.5) compared to PBS treatment, as shown in Figure 3.8.



Figure 3. 8 Comparison of the effect of AB-PA01 and gentamicin treatment on *P. aeruginosa* isolates biofilm biomass reduction.

3.4 Discussion

P. aeruginosa is a leading causative agent of DFU infection among Gram-negative bacteria (Commons *et al.*, 2015; Lipsky *et al.*, 2010). In the warm and humid Asian and African countries, *P. aeruginosa* has been reported frequently to be a common causative microorganism of DFU infection (Wu *et al.*, 2017; Hatipoglu *et al.*, 2016). *P. aeruginosa* is also known for its persistent and aggressive infection (Lipsky *et al.*, 2010). Hence, a reliable, discriminatory, and quick identification method should be employed as *P. aeruginosa* isolates are inert for some biochemical tests and do not show unique phenotypic characteristics in the routine diagnostic tests (Kacaniova *et al.*, 2019). In this study, no Gram-positive bacterium was identified from the samples cultured on vancomycin-supplemented MacConkey agar, which agreed with previous reports (Myles *et al.*, 2016). In the literature, there is no consensus on the concentration of vancomycin to be included in the MacConkey medium to inhibit Gram-positive bacteria. However, from 4 to 64 mg/L vancomycin have been commonly used (Myles *et al.*, 2016; Brown and Walpole, 2003). In this study, the concentration of vancomycin added on the MacConkey agar was higher than the MICs found during VITEK[®] 2 tests to make the agar more selective.

MALDI-TOF MS analysis has been used extensively for microbial identification (Angeletti, 2017; Schumann and Maier, 2014). In this study, close to 90% of the suspected isolates were identified as *P. aeruginosa* species by MALDI-TOF MS. The identification score was \geq 2.00 for all isolates that demonstrate high confidence of identification to the species level (Kacaniova *et al.*, 2019).

P. aeruginosa isolates show a significant degree of inherent resistance to many antibiotics such as β -lactams, tetracycline, aminoglycosides, fluoroquinolones, and sulfonamide mainly because of the low permeability of its outer membrane (Breidenstein *et al.*, 2011; Li *et al.*,

1994). Intrinsic antibiotic resistance mechanisms are the integral parts of the *P. aeruginosa* genome that require high MIC (Lister *et al.*, 2009). About 50% of the isolates tested in the present study were resistant to ticarcillin/clavulanic acid. Moreover, 10% - 15% of the isolates showed resistance against ciprofloxacin, cefepime, ceftazidime, and piperacillin/tazobactam. These antibiotic-resistant patterns of *P. aeruginosa* isolates are consistent with a previous report (Otto-Karg *et al.*, 2009).

This study was unable to confirm reports that showed significant proportions of *P. aeruginosa* clinical isolates are MDR (Sivanmaliappan and Sevanan, 2011; Falagas *et al.*, 2008) because only one MDR isolate was identified. In contrast to this study finding, a survey conducted on *P. aeruginosa* isolates collected from United Kingdom, Belgium, and Germany showed that 60% of the isolates were MDR (Mustafa *et al.*, 2016). Compared to *P. aeruginosa* isolates sampled from DFU patients that showed 50% – 100% resistance for all antibiotics they were exposed to in a study in India (Sivanmaliappan and Sevanan, 2011), the majority of the isolates tested in this study were susceptible for most of the antibiotics tested except ticarcillin/clavulanic acid.

Rate of antibiotic-resistant isolates was lower in this research, except for ticarcillin/clavulanic acid, than previous studies with cystic fibrosis from 3 European countries (Mustafa *et al.*, 2016), DFU from India (Sivanmaliappan and Sevanan, 2011), ventilator-associated pneumonia from France (Fihman *et al.*, 2015), nosocomial pneumonia from northern Europe (Riou *et al.*, 2010), and chronic rhinosinusitis from Australia (Fong *et al.*, 2017). In the present study, it was observed that amikacin, meropenem, tobramycin, norfloxacin, and gentamicin demonstrated a high level of antipseudomonal activity, and these antibiotics are better choices to treat DFU caused by *P. aeruginosa*. The rising prevalence of antibiotic-resistant pathogenic bacteria has prompted the search for alternatives such as lytic phages

(Golkar *et al.*, 2014). Effective phage therapy requires careful design through a multistep evaluation. Combining different phages in a formulation frequently results in a broader spectrum of antibacterial activity (Chan & Abedon, 2012).

In this study, both the *Myoviridae* and *Podoviridae* phages exhibited lytic activity over a broad range of clinical isolates, suggesting these phages are good candidates for therapy. Phage cocktail AB-PA01 possessed a superior host range compared with the most active single phage treatment in spot tests. As in the current study, the use of high phage titers $(8 - 9 \log_{10} (PFU/ml))$ for host-range analysis is routine (Mendes *et al.*, 2014; Kutter, 2009). These individual phages and their cocktail used in the current study were effective against the higher bacterial inoculum (8 $\log_{10} (CFU/ml)$) used to mimic heavily infected wounds (Mendes *et al.*, 2014). AB-PA01 and its components produces this effect at MOI 1 which is lower than the 'multiplicity of 10 rule', which is a guideline stating that if the goal is a significant reduction in bacterial density, it is better to use in the order of 10 phages to a bacterial host (Abedon, 2009; Kasman *et al.*, 2002). The broad host range and efficacy of the phage cocktail observed here is consistent with reports that the use of phage cocktails rather than individual phages improves phages activity by expanding the host range (Leskinen *et al.*, 2017; Mendes *et al.*, 2014; Chan *et al.*, 2013; Hall *et al.*, 2012).

During spot tests in the present study, it was found that the plaques of each phage and their combination were surrounded by opaque halo zones, which could be an indication of the presence of a phage-associated depolymerase as reported earlier (Cornelissen *et al.*, 2011). Halos are indications of enzymatic molecules diffusion and phages capability to disrupt extracellular polymerase substance of biofilms enzymatically (Gutierrez *et al.*, 2016; Sutherland *et al.*, 2004). Phage-associated depolymerase and related enzymes have been reported to enhance biofilm-reduction activity compared with non-depolymerase-inducing

phages (Hughes *et al.*, 1998). Although evaluation of phage lytic activity on a bacterial planktonic form is useful, biofilm studies should also be conducted because most infections involve biofilms (James *et al.*, 2008). Previous studies on *P. aeruginosa* suggest the potential of phages to reduce or eliminate biofilms (Alves *et al.*, 2016; Pires *et al.*, 2011) and prevent biofilm formation (Alves *et al.*, 2016; Fu *et al.*, 2010).

The effort to use gentamicin as a positive control in the present study was not successful as it produced no significant biofilm biomass reduction compared to the negative controls. Some gentamicin-treated *P. aeruginosa* isolates produced more biofilm biomass than control groups. This ineffectiveness of gentamicin in biofilm biomass reduction might be attributed to the fact that aminoglycosides often exhibit poor penetration through biofilms, resulting in reduced efficacy of gentamicin in biofilms (Ronan *et al.*, 2016; Stewart, 2002). The finding of this study was supported by a study that found the exposure of *P. aeruginosa* PAO1 biofilm to gentamicin alone did not result in any qualitative changes (Ronan *et al.*, 2016). The biofilm biomass reduction due to tetracycline treatment was similar to the AB-PA01 treatment effect.

In this study, phage cocktail AB-PA01 displayed a strong anti-biofilm activity on *P. aeruginosa* isolates biofilms *in vitro*. Over 60% of the isolates showed \geq 50% biofilm biomass reduction compared to the negative control group, and of these, about 20% of the isolates showed \geq 75% biofilm biomass reduction. This effect may be expected to increase with higher MOI because higher MOI produces a more significant biofilm reduction effect (Alves *et al.*, 2016). The lytic effect of AB-PA01 on bacterial host planktonic state translated well to a reduction in biofilms, consistent with earlier reports (Gutierrez *et al.*, 2016; Fu *et al.*, 2010). The biofilm reduction capability of AB-PA01 was not related to the antibiotic-resistance characteristics of the isolates. In general, the findings of the present study suggest

that phage cocktails could be feasible alternatives to antibiotic treatments in combating serious *P. aeruginosa* biofilm-related DFU infections.

In conclusion, *P. aeruginosa* isolates were identified at the species level by MALDI-TOF MS with high confidence of identification. The antibiotic susceptibility test conducted using VITEK[®] 2 demonstrated that *P. aeruginosa* isolates circulating in DFU patients of the current study settings had a good susceptibility pattern for most of the antibiotics tested except for ticarcillin/clavulanic acid. The isolates also displayed high susceptibility for each phage and the phage cocktail they were exposed to during spot test. The use of the phage cocktail broadened the host range and lysed most of the isolates resistant to some of its components. The strong lytic potential of AB-PA01 in the planktonic state is translated well to the biofilm biomass reduction since it exhibited significant anti-biofilm activity against *P. aeruginosa* isolates *in vitro*. Importantly, the lytic efficacy of AB-PA01 on host bacteria in planktonic state and anti-biofilm activity in biofilm phases was not related to the antibiotic-resistant pattern of *P. aeruginosa* isolates. Hence, AB-PA01 can be formulated as an alternative anti-infective therapeutic agent that may be useful in the treatment of biofilm-related DFU infection caused by *P. aeruginosa*.

Chapter 4

Efficacy of Phage Cocktails on *Staphylococcus aureus* and *Pseudomonas aeruginosa* in mixed-species culture

Chapter 4: Efficacy of Phage Cocktails on *Staphylococcus aureus* and *Pseudomonas aeruginosa* in mixed-species culture

Abstract

Antibiotic-resistant Staphylococcus aureus and Pseudomonas aeruginosa are the commonly isolated bacteria from infected diabetic foot ulcers. Effective and easily available alternative therapeutic agents such as phages are needed. The efficacy of phages in mixed-species infections, especially in biofilms form, has been poorly examined. The in vitro lytic efficacy of phage cocktails AB-SA01, AB-PA01, and their combination was examined in S. aureus and P. aeruginosa mixed-species planktonic and biofilm cultures. Green fluorescent protein (GFP)-labelled P. aeruginosa PAO1, and mCherry-labelled S. aureus KUB7 laboratory strains and clinical isolates were used as target bacteria. During real-time monitoring of treatment effect using spectrophotometry, the density of mCherry S. aureus KUB7 and GFP P. aeruginosa PAO1 significantly decreased when treated by their respective phage cocktail, a mixture of phage cocktails, and gentamicin. The decrease in bacterial population measured by the reduction in fluorescence associated with the decrease in bacterial cell counts on selective agars. This mCherry- and GFP-based mixed-species microplate assay monitored through spectrophotometry combines reproducibility, rapidity, and ease of management. It is amenable to high-throughput screening for phage cocktail efficacy evaluation. Each phage cocktail, the combination of the two phage cocktails, and tetracycline produced significant biofilm biomass reduction in mixed-species biofilms. This study result shows that AB-SA01 and AB-PA01 lyse their hosts in the presence of non-susceptible bacteria, both in planktonic and biofilm states. The effect of each phage cocktail alone or in combination was more significant in planktonic cultures than in biofilms irrespective of host antibiotic resistance profile. These data support the use of phage therapy in polymicrobial infections.

4.1 Introduction

4.1.1 Antibiotic-resistant bacterial infection

Infections from antibiotic-resistant bacteria continue to threaten standard therapies and causing a global health crisis (Blair *et al.*, 2015). The proportion and number of MDR bacterial pathogens have increased in the last decade, and MDR bacterial infections are considered as emergent global diseases (Roca *et al.*, 2015). The direct impact of antibiotic-resistant bacterial infections includes a long duration of illness and increased mortality (Sowole *et al.*, 2018). It is estimated that in Europe, 25,000, and the United States, 23,000 people die each year as a result of MDR bacterial infections (Blair *et al.*, 2015). MDR *S. aureus* and *P. aeruginosa* are among the therapeutically challenging bacteria (Lebowitz *et al.*, 2017; Golkar *et al.*, 2014). DFU infections are usually treated with broad-spectrum antibiotics for prolonged durations (Uçkay *et al.*, 2015). The use of antibiotics for an extended time is one of the significant risk factors for antibiotics for the treatment of DFU has been difficult due to the emergence of antibiotic-resistant pathogens (Lipsky *et al.*, 2016). This might require the identification of causative bacterial species and the development of specific treatments without the off-target effects.

4.1.2 Polymicrobial wound infections

The human body is a dynamic ecosystem naturally inhabited by a wide diversity of microorganisms that are collectively referred to as "microbiota" (Scalise *et al.*, 2015). The host immune system maintains the balance of these microorganisms under normal physiological conditions (Wong *et al.*, 2013). Skin, as part of the immune system, protects underlying tissues from harmful microorganisms. As an exposed organ, skin can be damaged

through injury, and healthy skin microflora may proliferate and cause inflammation (Scalise *et al.*, 2015). Damaged and exposed subcutaneous tissue may provide an environment conducive to microbial proliferation. The microflora associated with a small, clean, and well-perfused wound in an immunocompetent host may produce minimal effects, whereas the microflora associated with the presence of foreign material and devitalized tissue may facilitate microbial proliferation (Robson, 1997).

Wound infections, particularly chronic wound infections, are often polymicrobial. Multiple microorganisms in a single community can share genetic material, metabolic cooperation, quorum sensing systems, passive resistance, niche optimization, host immune modulation, and virulence induction that together facilitate microbial proliferation (Tay *et al.*, 2016; Shettigar *et al.*, 2016; Wolcott *et al.*, 2013). Bacteria such as *Enterococcus faecalis* modulate the polymicrobial niche by supplying nutritional elements to the co-infecting bacteria (Keogh *et al.*, 2016). Stable complex microbial communities are associated with extended wound healing period (Loesche *et al.*, 2017). The majority of polymicrobial chronic wound infections involve biofilms that further complicate treatment outcomes because bacteria in biofilm forms exhibit increased tolerance to antibiotic treatment (Pastar *et al.*, 2013; Dalton *et al.*, 2011; Bowler *et al.*, 2001).

Wound-infecting bacteria could originate from the environment, surrounding skin microflora, or endogenous sources (Bowler *et al.*, 2001). There is a controversy about the precise mechanism used by microorganisms to produce infection and the significance of their contribution towards non-healing wounds (Bertesteanu *et al.*, 2014). It appears that microbial load is an important factor in determining whether a wound heals (Robson, 1999; Raahave *et al.*, 1986). Polymicrobial infections are often linked with severe infection and poor treatment outcomes (Duplantier and van Hoek, 2013). Through acting synergistically, co-infecting

microbes can potentially increase their virulence (Tay *et al.*, 2016). Hence, polymicrobial infections may require new treatment strategies (Wolcott *et al.*, 2013).

4.1.3 Polymicrobial nature of DFU infections

Diabetes mellitus is a severe public health problem globally, and about 80% of diabetesrelated lower extremity amputations are preceded by a foot ulcer (Hingorani *et al.*, 2016; Boulton, 2008). Most of the DFUs are infected, and the microbial burden is believed to contribute to poor healing (Prompers *et al.*, 2007). About one-third or more of DFU infections are found to be polymicrobial (Ramakant *et al.*, 2011). A recent study that investigated approximately 3,000 wounds found that only 7% of the wounds had singlespecies infections (Wolcott *et al.*, 2013). The polymicrobial bacterial communities in biofilms may contribute to delayed healing in DFUs (Mottola *et al.*, 2016). From polymicrobial DFUs infections, *S. aureus* and *P. aeruginosa* are commonly isolated microorganisms (Gjødsbøl *et al.*, 2006). DFU caused by *S. aureus* and *P. aeruginosa* is associated with increased wound severity and healthcare costs (Gjødsbøl *et al.*, 2006; Madsen *et al.*, 1996). A study by Mastropaolo and co-workers (2005), utilizing a diabetic mouse infection model, found that certain bacterial species significantly increased at the site of inoculation when co-infected with a second bacterium compared to a mono-species bacterial inoculation, indicating interspecies synergy (Mastropaolo *et al.*, 2005).

4.1.4 Role of S. aureus in DFU infection

It is estimated that 25% - 30% of adults are colonised with *S. aureus* (Schechter-Perkins *et al.*, 2011). In immunocompromised patients, or patients with potentially impaired wound healing, such as diabetes mellitus patients, *S. aureus* carriage can convert into a clinically evident infection (Cervantes-Garcia *et al.*, 2015; Lipsky *et al.*, 2011). The prevalence of *S.*

aureus in DFU patients varies in the range of 28–76% among Gram-positive bacteria (Eleftheriadou *et al.*, 2010).

S. aureus is found in about half of DFUs (Reveles *et al.*, 2016). Besides its abundance and antimicrobial resistance, *S. aureus* is virulent in DFUs (Richard *et al.*, 2011). Most of the *S. aureus* strains form a biofilm, which increases pathogenicity and contributes to treatment failure because it protects the pathogen from the effect of antibiotics and possibly impairs access to innate and adaptive immunity (Shettigar *et al.*, 2016; Vidlak and Kielian, 2016). *S. aureus* secretes virulence factors, including haemolysins, proteases, collagenases, and hyaluronidases, which help to necrotize host tissues and increase pathogenicity (Pfirman and Haile, 2018; Cosgrove *et al.*, 2003). Some *S. aureus* strains encode genes responsible for toxins production such as enterotoxins, toxic shock syndrome toxin-1 (TSST-1), exfoliative toxins A and B (*et*A and *et*B), and Panton-Valentine leucocidin, that are associated with high-grade infected diabetic ulcers and poor treatment outcomes (Viquez-Molina *et al.*, 2018; Sotto *et al.*, 2008).

4.1.5 Role of *P. aeruginosa* in DFU infection

Infection with MDR *P. aeruginosa* is increasing globally, primarily through nosocomial infections (Mesaros *et al.*, 2007). *P. aeruginosa* is the second most common pathogen found in DFUs after *S. aureus* (Sekhar *et al.*, 2018; Saltoglu *et al.*, 2018; Kateel *et al.*, 2018; Al Ayed *et al.*, 2018; Goldufsky *et al.*, 2015; Hatipoglu *et al.*, 2014). Due to its resistance to a wide variety of antibiotics via numerous antibiotic-resistance mechanisms, and the complexity of DFU, *P. aeruginosa* infection can be difficult to control through antibiotic treatment (Zhang *et al.*, 2014a; Mesaros *et al.*, 2007).

The presence of *P. aeruginosa* in wounds correlates with poor prognosis (Sivanmaliappan and Sevanan, 2011). Using its Type III Secretion System (T3SS), *P. aeruginosa* can cause severe and persistent infection in diabetic wounds (Goldufsky *et al.*, 2015). T3SS helps *P. aeruginosa* to release toxins, including ExoS, ExoT, ExoU, and ExoY, which can cause tissue damage, immune suppression, the spread of the bacteria, and inducing apoptosis of neutrophils (Goldufsky *et al.*, 2015; Zhang *et al.*, 2014a; Jia *et al.*, 2003). In studies with diabetic mouse wounds, infection with *P. aeruginosa* significantly increased the production of M1-phenotype macrophages which associates with the expression of pro-inflammatory and cytotoxic factors including tumour necrosis factor-a (TNF-a), interleukin-1 (IL-1), and interleukin-6 (IL-6) that may impair diabetic wound healing (Chen *et al.*, 2018).

4.1.6 S. aureus and P. aeruginosa co-infection in DFU

S. aureus and *P. aeruginosa* are commonly associated with chronic wound infections, as seen in DFUs (Trivedi *et al.*, 2014; Fazli *et al.*, 2009). The two pathogens can be found together in a non-random distribution in which *P. aeruginosa* occupies deeper regions of chronic wounds while *S. aureus* is found in the upper regions (Fazli *et al.*, 2009). *S. aureus* and *P. aeruginosa* form *in vivo* biofilms that contribute to antibiotic resistance of these bacterial species (DeLeon *et al.*, 2014). Difficulties in growing different bacterial species together *in vitro* make the study of co-infections challenging, and this is true in the case of *S. aureus* and *P. aeruginosa*. *P. aeruginosa* has been seen to kill *S. aureus* both in *in vitro* co-culture (Yang *et al.*, 2011; Machan *et al.*, 1991) and *in vivo* co-infection (Pernet *et al.*, 2014). This killing has been linked to various secretions of *P. aeruginosa* including LasA protease (Carnicero *et al.*, 1990), 4-hydroxy-2-heptylquinoline-N-oxide (HQNO) (Hoffman *et al.*, 2006), the pel and psl products (Qin *et al.*, 2009), and pyocyanin (Dietrich *et al.*, 2006). Conversely, there are reports that *S. aureus* was not affected by *P. aeruginosa*, and the growth activity of *P. aeruginosa* was enhanced in the presence of *S. aureus* (Michelsen *et al.*, 2014). It is reported that *P. aeruginosa* was protected from autolysis, antibiotic effect, and host immunity by extracellular *S. aureus* proteins (Armbruster *et al.*, 2016). A commensal interaction between *S. aureus* and *P. aeruginosa* may occur in co-infection (Armbruster *et al.*, 2016; Michelsen *et al.*, 2014; DeLeon *et al.*, 2014). It has been shown that *P. aeruginosa* and *S. aureus* can stably exist in the *in vitro* wound environment and mutually benefit from the co-infection (DeLeon *et al.*, 2014).

4.1.7 S. aureus and P. aeruginosa co-culture

Studies suggest that *P. aeruginosa* forces the conversion of *S. aureus* to small colony variant (SCV) form with decreased metabolic activity in co-cultures (Filkins *et al.*, 2015; Mitchell *et al.*, 2010; Biswas *et al.*, 2009; Hoffman *et al.*, 2006). It is indicated that *S. aureus* survival in co-culture with *P. aeruginosa* is dependent on its ability to convert to the SCV phenotype (Hoffman *et al.*, 2006). The converted SCV *S. aureus* is often associated with increased tolerance to environmental stress (Moisan *et al.*, 2006). Furthermore, the presence of *P. aeruginosa* could activate staphyloxanthin producing mechanism of *S. aureus* white phenotypic variants, and it induces catalase that renders *S. aureus* more virulent and antibiotic-resistant (Antonic *et al.*, 2013). A recent study showed that timing and bacterial concentration of *S. aureus* and *P. aeruginosa* might affect the expression of quorum sensing genes and final bacterial ratio during mixed-species biofilm development (Woods *et al.*, 2018). Medium containing bovine serum albumin (BSA) is recommended to allow better growth of *S. aureus* in the presence of *P. aeruginosa* (Radlinski *et al.*, 2017; Kart *et al.*, 2014; Pernet *et al.*, 2014).

4.1.8 Effect of phages on mixed-species biofilms

Of all human infections, 65% - 80% are biofilm-related, and biofilm formation is often considered an underlying reason for antibiotic treatment failure (Coenye and Nelis, 2010). Bacteria residing in biofilms are more tolerant to physical and chemical disruptions than their planktonic counterparts (Steenackers *et al.*, 2016; Gilbert *et al.*, 2002). Studies have also shown that bacteria in biofilms are 100 -1000 times more resistant to antibiotics than in the planktonic phase (Burmolle *et al.*, 2006; Ceri *et al.*, 1999). Bacterial resistance to antibiotics in biofilms is mainly due to the complex nature of the EPS that gives mechanical stability to the biofilm and serves as an external digestive system that keeps extracellular enzymes near the bacteria (Flemming and Wingender, 2010). The EPS also protects bacteria since it reduces the access of solutes to the bacteria through the combination of ionic interaction and molecular sieving (Allison, 2003). The presence of metabolically inactive persister cells also contributes to the antibiotic-resistance of bacteria in biofilms (Harper *et al.*, 2014). Natural biofilms are usually polymicrobial sharing a common milieu and coexisting in niches by forming multispecies biofilms (Zhao *et al.*, 2013; Thein *et al.*, 2007).

Although many studies have examined the efficacy of phages on single-species biofilms, few have evaluated it on multispecies biofilms as discussed in a review (Geredew Kifelew *et al.*, 2019). There is a disparity among studies that evaluated the effect of phages on multispecies biofilms. Some reports showed that phages achieved a higher titre and depleted their bacterial host population effectively (Gonzalez *et al.*, 2017; Gutierrez *et al.*, 2015; Kay *et al.*, 2011; Sillankorva *et al.*, 2010). On the contrary, some reports indicated the inefficiency of phages within multispecies biofilms (Burmolle *et al.*, 2006; Tait *et al.*, 2002). This chapter aimed to examine the effect of *S. aureus* and *P. aeruginosa* phage cocktails, AB-SA01 and AB-PA01, on mixed-species planktonic and biofilm states.

4.2 Materials and methods

4.2.1 Bacteria

The bacteria used in this study were randomly selected four *S. aureus* and four *P. aeruginosa* isolates. From each species, one was a laboratory strain, and three were clinical isolates. The clinical isolates were collected from infected DFU patients at diabetic foot clinics in Adelaide, South Australia, during the study period by South Australia Pathology and generously provided to our laboratory on Muller Hinton agar culture. GFP-labelled *P. aeruginosa* PAO1 (PAO1) and mCherry-labelled *S. aureus* (*S. aureus* KUB7) (Burcham *et al.*, 2016) laboratory strains were generously donated by Dr. Nicky Thomas of University of South Australia, and A/Prof. Heather Jordan of Mississippi State University, USA, respectively.

The two fluorescent proteins are driven by constitutive promoters and do not need antibiotics to maintain expression. These were among the isolates that showed strong susceptibility in spot test and 73% - 88% biofilm biomass reduction on single-species biofilm experiments following treatment with the phage cocktail and its components, as discussed in chapter 2 and 3. A single colony of *P. aeruginosa* from a Muller Hinton agar plate was sub-cultured onto vancomycin-supplemented MacConkey agar (Thermo Fisher, South Australia), and *S. aureus* on MSA (Thermo Fisher, South Australia) as described (Mendes *et al.*, 2013; Kirisits *et al.*, 2005; Chapman, 1945). After 18 hours of incubation aerobically at 37 °C, the isolates were processed for identification and antibiotic susceptibility tests.

4.2.2 Phage cocktails

Phage cocktails AB-SA01 and AB-PA01 were donated by AmpliPhi Biosciences Corporation. AB-SA01 is a combination of J-Sa 36, Sa 83, and Sa 873 phages that belong to the *Myoviridae*. AB-PA01 is a mixture of Pa 193 and Pa 204 from *Myoviridae*, and Pa 222 and Pa 223 from *Podoviridae*. None of these phage components encodes any known bacterial virulence or antibiotic resistance genes, and all phages were considered to be strictly lytic (Fong *et al.*, 2017; Lehman *et al.*, 2019). The phages are produced under cGMP standards and approved by the US Food and Drug Administration as an investigational new drug (Lehman *et al.*, 2019; Law *et al.*, 2019).

The phage titer determination was conducted on susceptible two laboratory strains and two clinical isolates for each phage cocktail using plaque assay following the established procedure (Mirzaei and Nilsson, 2015; Merabishvili *et al.*, 2009). The laboratory strains were *S. aureus* RN4220 and SA6538 for AB-SA01 and *P. aeruginosa* PAO1 and PA10145 for AB-PA01. The mean titers of AB-SA01 and AB-PA01 were 9.1 log₁₀ (PFU/ml) and 10.3 log₁₀ (PFU/ml), respectively.

4.2.3 Bacterial typing and antibiotic susceptibility testing

The bacterial isolates were identified using standard microbiology methods and MALDI-TOF MS biotyping (Bruker Daltonics Biotyper, Bruker Pty. Ltd., Victoria, Australia) as previously described (Lee *et al.*, 2015; Schumann and Maier, 2014; Lasch *et al.*, 2014; Harris *et al.*, 2010). The selective media MSA and vancomycin-supplemented MacConkey agar were used for *S. aureus*, and *P. aeruginosa* isolates, respectively. Antibiotic susceptibility pattern of the isolates was determined by VITEK[®] 2 (bioMérieux Australia Pty Ltd, New South Wales, Australia) as described earlier (Weber *et al.*, 2017; Gardiner *et al.*, 2013; Cartwright *et al.*, 2013; Ligozzi *et al.*, 2002).

4.2.4 P. aeruginosa-S. aureus co-culture test

Using the lab strains of *P. aeruginosa* (PAO1) and *S. aureus* (*S. aureus* KUB7), a pilot coculture test was conducted both in planktonic and biofilm phases following established protocol with minor modifications (Gonzalez *et al.*, 2017; Kay *et al.*, 2011; Sillankorva *et al.*, 2010). For planktonic culture, each bacterial species was streaked on a nutrient agar plate and grown for 24 hours at 37 °C. From each bacterial species, 2 - 3 colonies were suspended in sterile physiological saline and adjusted to 1.0 McFarland turbidity standard (approximately 8.5 log₁₀ cells). Each bacterial species' suspension was diluted in nutrient broth at 1:10 suspension-to-broth ratio. The two bacterial species cell suspensions were mixed in separate sterile test tubes at 1:1, 1:3, 1:5, and 1:7 *P. aeruginosa-to-S. aureus* ratio. Next, 50 µl of the mixture from each ratio was transferred to a culture tube containing 2.5 ml nutrient broth under sterile conditions.

The cultures in the tubes were grown in a shaker incubator at 37 °C and 160 rpm. After 18 hours incubation, 100 μ l of the culture was mixed with nutrient soft agar pre-warmed to 42 °C and plated onto MSA and vancomycin-supplemented MacConkey agar plates. The plates were incubated at 37 °C for 24 - 48 hours, followed by manual counting of the CFU as described (Kumar and Ting, 2015). MacConkey agar was made more selective against *S. aureus* by adding vancomycin at 3 μ l/ml of culture media based on VITEK[®] 2 antibiotic susceptibility test results. All *S. aureus* isolates used in this study were strongly susceptible to vancomycin.

Evaluation of the co-existence of *S. aureus* and *P. aeruginosa* in biofilms was conducted as described (Gutierrez *et al.*, 2015; Sillankorva *et al.*, 2010). Briefly: 200 μ l of mixed broth culture from each ratio was transferred into wells of a 96-well flat-bottom microplate. The microplate was covered with aluminium foil and incubated at 37 °C in a static condition.

After 48 hours of incubation, the liquid culture was removed through gentle pipetting. The biofilm was then disrupted by aggressive scrapping of the bottom and wall of the wells using a sterile wire loop and collected using 200 µl nutrient broth. One hundred microliters of the suspension were mixed with nutrient soft agar pre-warmed to 42 °C and evenly plated onto MSA and vancomycin-supplemented MacConkey agar plates. The plates were incubated at 37 °C for 24 - 48 hours, followed by manual counting of the CFU. The experiment was conducted in triplicate on three different days.

4.2.5 Phage treatment of fluorescent mixed-species planktonic culture

Phage treatment on mixed-species bacterial planktonic cultures was carried out following an established protocol with minor modifications (De Jong *et al.*, 2017; Kumar and Ting, 2015). Two bacterial constructs, namely mCherry protein-labelled *S. aureus* KUB7 and green fluorescent protein-labelled *P. aeruginosa* PA01, were used. From the 18 hours of selective agar culture plates, 2 - 3 colonies were suspended in sterile PBS and adjusted to 1.0 McFarland equivalence turbidity standard, containing approximately 8.5 log₁₀ (CFU/ml). Each bacterial species suspension was diluted at a ratio of 1:100 v/v with nutrient broth (Sigma-Aldrich, New South Wales, Australia) supplemented with 5% BSA) (Sigma-Aldrich, New South Wales, Australia) and incubated for 2 hours at 37 °C.

The two bacterial suspensions were then mixed at the ratio of 1:3 v/v of GFP PAO1 to mCherry *S. aureus* KUB7 in a sterile 10 ml test tube. This ratio of the two bacterial species was taken because it was from this ratio that equivalent bacterial density recovered during the pilot co-culture experiment. Two hundred microlitre of each mixture was transferred to a clear 96-well flat-bottom Greiner CELLSTAR[®] polystyrene tissue culture plate in triplicate (Sigma-Aldrich, New South Wales, Australia) and respective treatments were applied.
Mixed-species planktonic cell treatment were performed using five treatments: (i) a cocktail of *S. aureus* phage (AB-SA01), (ii) a cocktail of *P. aeruginosa* phage (AB-PA01), (iii) a mixture of *S. aureus* and *P. aeruginosa* phage cocktails (AB-SA01+AB-PA01), (iv) gentamicin, and (v) PBS treatment.

AB-SA01, AB-PAO1, or AB-SA01+AB-PAO1 phage cocktails were applied at the MOI of one to fluorescently labelled *S. aureus-P. aeruginosa* mixed-species culture. Gentamicin was used as a positive control at 16 µg/ml because its MIC for these isolates was ≤ 8 µg/ml during VITEK[®] 2 antimicrobial susceptibility test. An equal volume of PBS to phage solutions was applied to negative control groups. A plate cover was applied, and the plate was wrapped with aluminium foil from the top and sides. The plate was incubated in a CLARIOstar[®] Omega plate reader (BMG LABTECH Pty. Ltd., Victoria, Australia) for 24 hours at 37 °C with 100 rpm constant double-orbital shaking between measurements as described (Alves *et al.*, 2016).

The excitation and emission wavelengths were set at 570-15 and 620-20 nm for mCherry, and 470-15 and 515-20 nm for GFP detection, respectively. Fluorescence of mCherry and GFP were measured (in relative fluorescence unit, RFU) every 30 minutes in each well. Signals from triplicate wells were averaged and corrected for blank wells containing only nutrient broth, and the graph of fluorescence intensity was plotted. After 24 hours of incubation, bacterial colony counts were performed using serial dilution on selective agars for each species. The test was repeated three times using the same procedure on different days, and the mean was taken as final.

4.2.6 Phage treatment of clinical isolates mixed-species planktonic culture

A similar protocol with fluorescent mixed-species planktonic culture as above was followed for clinical isolates, except that incubation was in a standard incubator and no fluorescent intensity measurement. The effect of the treatment was evaluated using bacterial density reduction by colony count method.

4.2.7 Phage treatment of in vitro mixed-species biofilm

The mixed-species biofilm development and treatment were conducted as described (Gonzalez *et al.*, 2017; Mendes *et al.*, 2014) with few modifications. Briefly, 2 - 3 colonies of 18 hours culture of each isolate were independently suspended in sterile physiological saline and adjusted to 1.0 McFarland turbidity standard. These suspensions were pooled at a 1:3 v/v ratio of *P. aeruginosa*-to-*S. aureus*, and 100 μ l of the pooled suspension was transferred to 10 ml 5% BSA nutrient broth. The final suspension was supplemented with 1% sterile glucose to facilitate biofilm development. Two hundred microlitre of the suspension was transferred into a tissue culture plate in triplicate and incubated for 48 hours at 37 ^oC with 70 rpm agitation.

After 48 hours of incubation, the liquid culture was removed, and plates were washed gently twice using sterile deionized water. Then, 225 μl of AB-SA01, AB-PAO1, or AB-SA01+AB-PAO1 suspension in the nutrient broth was applied to the respective treatment group biofilms. The treatment categories of mixed-species biofilms were: (i) *S. aureus* phage cocktail AB-SA01, (ii) *P. aeruginosa* phage cocktail AB-PAO1, (iii) a mixture of the two phage cocktails, AB-SA01+AB-PAO1, (iv) tetracycline (positive control), and (v) PBS (negative control).

The effect of each phage cocktail and a combination of the two phage cocktails was compared to tetracycline- and PBS-treated groups. Tetracycline was used as a positive control in mixed-species biofilm treatment as it was strongly effective (p < 0.001), compared

with PBS treatment, in biofilm biomass reduction on single-species biofilm treatment of both *S. aureus* and *P. aeruginosa* isolates as discussed in chapters 2 and 3 of this thesis. Gentamicin did not produce significant biofilm biomass reduction (p > 0.05). Tetracycline was not used as a positive control in mixed-species planktonic culture treatment experiments to avoid exaggerated fluorescence detection because of its color and fluorescence nature (Ni *et al.*, 2010).

MOI of each phage cocktails or their mixture was one based on each bacterial species density recovered during the co-culture pilot experiment. An equivalent volume of tetracycline and PBS to phage solution in nutrient broth were also applied as controls. The concentration of tetracycline was 128 µg/ml because *P. aeruginosa* isolates are susceptible to a higher concentration of tetracycline (Zheng *et al.*, 2017; Pai *et al.*, 2001). The MIC of tetracycline for *S. aureus* was ≤ 8 µg/ml during VITEK[®] 2 antimicrobial susceptibility test. Treated biofilms were incubated for 12 hours at 37 °C at static conditions. The biofilm was washed twice using 250 µl sterile PBS through careful pipetting. The biofilm-associated cells attached to the well surface were collected with 225 µl nutrient broth through pipetting after scraping the wall and bottom of the wells with a loop as described (Gutierrez *et al.*, 2015; Sillankorva *et al.*, 2010). After homogenization with a vortex mixer, the cell suspension was serially diluted, $10^{-1} - 10^{-8}$, in filter-sterilized 10 mM ferrous ammonium sulphate (FAS) supplemented nutrient broth to inactivate free phages (McNerney *et al.*, 1998). The diluted suspension was incubated at room temperature for 15 minutes.

4.2.8 Colony count

Bacterial quantification was performed using the 10-fold serial dilution method (O'Toole, 2016; Mendes *et al.*, 2013). One hundred microlitres of the bacterial suspension from each serial dilution were mixed with nutrient soft agar warmed at 42 $^{\circ}$ C and dispensed over 37 $^{\circ}$ C

pre-warmed MSA and vancomycin-supplemented MacConkey agar in triplicate and incubated at 37 $^{\circ}$ C for 24 hours. Plates with approximately 30-300 colonies were taken from one of the dilutions, and colony count was carried out as described (O'Toole, 2016). The bacterial cell was calculated using the formula B = N/d, where B = number of bacteria, N = average number of colonies counted on the three plates, and d = dilution factor as described earlier (Mendes *et al.*, 2013). The results are expressed as logarithm-transformed values (log₁₀ (CFU/ml)).

4.2.9 Data management and statistical analysis

Data were double entered, encoded, and stored using Microsoft Excel Spreadsheet. STATA version 16 software was used for statistical analysis. Data are reported in terms of mean and are expressed as logarithm-transformed values log_{10} (CFU /ml) over time. A comparison of experimental groups was performed using a one-way analysis of variance (two-tailed) or paired 't-*test*'. A p < 0.05 value was considered statistically significant.

4.3 Results

4.3.1 Colony count result of co-culture tests

The capability of *S. aureus* KUB7 and *P. aeruginosa* PA01 to grow in planktonic culture and establish biofilm in the presence of each other was examined. The outcome of the co-culture study was evaluated by taking samples from mixed-species cultures and plating onto selective agars and performing colony counts. The results confirmed that *S. aureus* KUB7 and PA01 co-exist in planktonic and biofilm cultures. The mean bacterial population in planktonic and biofilm phases at 1:3 volume-to-volume PA01-to-*S. aureus* KUB7 ratio was 7 log₁₀ (CFU/ml) for each species. At 1:1 ratio, the mean CFU for *S. aureus* KUB7 was 5 log₁₀ (CFU/ml) while PA01 was 7 log₁₀ (CFU/ml). When *S. aureus* KUB7 concentration increase beyond 1:3 PA01-to-*S. aureus* KUB7 ratio, CFU of *S. aureus* KUB7 increased, but the CFU of PA01 decreased. Moreover, *S. aureus* KUB7 and PA01 mixed planktonic culture monitored by CLARIOstar spectrophotometer for 24 hours showed similar fluorescence detection and CFU results during colony count. Hence, a 1:3 *P. aeruginosa*-to-*S. aureus* ratio was selected for the subsequent mixed-species culture treatment experiments.

4.3.2 Phage cocktail treatment on fluorescent mixed-species planktonic culture

The effects of the treatments were evaluated by measuring fluorescence, and bacterial cell count of each species on selective agar medium. The fluorescence of each species alone or together was confirmed under a confocal microscope, as shown in Figures $4.1\mathbf{A} - 4.1\mathbf{C}$. No growth differences were observed between the mCherry *S. aureus* KUB7, GFP PAO1, and clinical isolates using the OD reading and CFU count in biofilm and CFU count in plate cultures. The magnitude of the fluorescence detected (in terms of relative fluorescence unit

(RFU)) at each time point of measurement for 24 hours demonstrated the efficacy of the treatments.



Figure 4. 1 Fluorescence strain under confocal microscopy: **A** - mCherry labelled *S. aureus* KUB7 (red data points) alone, **B** - GFP labelled *P. aeruginosa* PAO1 (green data points) alone, and **C** - mCherry labelled *S. aureus* KUB7 mixed with GFP labelled *P. aeruginosa* PAO1.

In single-species cultures treated with PBS, there was a marked increase in fluorescence of mCherry SA KUB7 and GFP PAO1, as shown in Figures 4.2A and 4.2B, indicating bacterial growth. In the mixed-species cultures without phages, the magnitude of fluorescence slowly increased with time (Figure 4.2C). However, the maximum red and green fluorescence obtained was much lower than the fluorescence detected during single-species PBS-treated cultures (Figures 4.2A and 4.2B), suggesting that the mixed-species exhibited co-inhibitory

effects. In mixed-species cultures treated with a single phage cocktail, the fluorescence of the target host was almost eliminated, while the non-target host was unaffected, as shown in Figures 4.2**D** and 4.2**E**, with fluorescence similar to PBS-treated single-species control.

When both phage cocktails were added to the mixed-species cultures, there was low fluorescence of both bacterial species, as shown in Figure 4.2**F**, similar to the inhibitory effect of gentamicin (Figure 4.2**G**), indicating that phage efficacy is not affected by the presence of non-host bacteria or other phages. The highest magnitude of red fluorescence in mCherry SA KUB7 was detected from the untreated single-species culture (Figure 4.2**A**), but in the case of GFP PAO1, the highest green fluorescence was detected from mixed-species culture treated with *S. aureus* phage cocktail, AB-SA01 (Figure 4.2**D**). The lowest magnitude of fluorescence from the target host in the mixed-species culture was observed when treated with each phage cocktail, as shown in Figures 4.2**D** and 4.2**E**. As expected, AB-SA01 and AB-PA01 exhibited no lytic effect on non-susceptible hosts (Figures 4.2**D** and 4.2**E**). While the fluorescence detected from a non-susceptible host showed an increase through time, the fluorescence obtained from the susceptible host remained low.

The decreases in fluorescence from each phage cocktail-, combinations of the two phage cocktails-, and gentamicin-treated groups were significantly lower compared to the PBS-treated group (Figures 4.2A - 4.2G). The corresponding colony count results for each treatment group after 24 hours are shown in Table 1 and confirm the results obtained with the fluorescence detection methodology.



Figure 4. 2 PBS-treated mCherry *S. aureus* KUB7 single-species culture. The green graph is due to background detection since the machine was set for mCherry and GFP detection.



Figure 4. 2B PBS-treated GFP PAO1 single-species culture. The red graph is due to background detection since the machine was set for mCherry and GFP detection.



Figure 4. 2C PBS-treated mCherry S. aureus KUB7 and GFP PAO1 mixed-species culture.



Figure 4. 2D AB-SA01-treated mCherry *S. aureus* KUB7 and GFP PAO1 mixed-species culture.



Figure 4. 2E AB-PA01-treated mCherry *S. aureus* KUB7 and GFP PAO1 mixed-species culture.



Figure 4. 2F AB-PA01+AB-SA01-treated mCherry *S. aureus* KUB7 and GFP PAO1 mixed-species culture.



Figure 4. 2G Gentamicin-treated mCherry *S. aureus* KUB7 and GFP PAO1 mixed-species culture.

Figure 2A - 2G: Effect of phage cocktails on mCherry S. aureus KUB7 (red data points) and GFP P. aeruginosa PAO1 (green data points) single- and mixed-species planktonic cultures. RFU represents relative fluorescence unit.

4.3.3 Efficacy of phage cocktails on laboratory and clinical isolates mixed-species planktonic culture

The population of each bacterial species in mixed-species planktonic cultures at the end of 24 hours of treatment was assessed. Compared to PBS-treated samples, AB-SA01- and AB-PA01-treated samples yielded a significantly lower bacterial population (p < 0.001), which is 3.3 log₁₀ (CFU/ml) and 5.1 log₁₀ (CFU/ml) reduction on their hosts, respectively, as shown in Table 4.1. When the same samples were treated by the combination of the two phage cocktails, AB-SA01+AB-PA01, the mean cell count of *S. aureus* and *P. aeruginosa* reduced by 4.7 log₁₀ (CFU/ml) and 3.8 log₁₀ (CFU/ml), respectively. The cell counts of one bacterial species showed an increase when the culture was treated with only a phage cocktail of the

other species in the mixed-species culture. All planktonic cultures treated with gentamicin yielded no visible bacterial cells.

		Bacterial cell counts after treatment						
Isolates			AB-	AB-	AB-SA01+			
Combination	Evaluated isolate	PBS	SA01	PA01	AB-PA01	Gentamicin		
S. aureus KUB7 ^S	S. aureus KUB7 ^S	5.5	3.6	8.7	3.0	0		
and PA01 GFP ^P	PA01 GFP ^P	7.9	8.9	0	3.7	0		
63-6538 ^s and 63-	63-6538 ^s	5.6	1.5	7.9	0	0		
6598 ^P	63-6598 ^P	5.3	6.0	0	0	0		
63-2498 ^s and 63-	63-2498 ^s	6.2	0	6.6	0	0		
5497 ^P	63-5497 ^P	7.5	8.0	3.5	4.0	0		
63-5656 ^s and 63-	63-5656 ^s	4.8	3.6	6.1	0	0		
6036 ^P	63-6036 ^P	6.6	8.1	3.3	4.2	0		
Summarized treatment effect								
	S. aureus mean	5.5	2.2	7.3	0.8	0		
	S. aureus reduction		3.3	+ 1.8	4.7	5.5		
	P. aeruginosa mean	6.8	7.8	1.7	3.0	0		
	P. aeruginosa reduction		+ 1.0	5.1	3.8	6.8		

Table 4. 1 Bacteria cell count (\log_{10} (CFU/ml)) of *S. aureus* and *P. aeruginosa* in mixed-species planktonic culture after 24 hours phage cocktails, gentamicin, or PBS treatment.

Key: ^S*S. aureus* isolates, ^P*P. aeruginosa* isolates, + indicates an increase in bacterial count compared to PBS treatment.

When the two phage cocktails were applied together, the mean *S. aureus* CFU reduction was $1.4 \log_{10}$ (CFU/ml) more than the reduction caused by AB-SA01 alone. On the contrary, AB-PA01 became less effective when applied in combination with AB-SA01 than alone. During mixed phage cocktails treatment, AB-PA01 produced 1.3 \log_{10} (CFU/ml) less effect compared to AB-PA01 only treatment. The increases of *S. aureus* and the decrease of *P. aeruginosa* during AB-SA01+AB-PA01 treatment was consistent across each bacterial species isolate tested. Each bacterial species was seen to grow better when the other species was removed from mixed-species culture by the phage of the other bacteria. When all the

eight bacterial isolates (four *S. aureus* and four *P. aeruginosa*) culture treated with gentamicin, no bacterial cell was detected from any of the isolates.

4.3.4 Efficacy of phage cocktails on mixed-species biofilms

The findings of this study demonstrate that phage cocktails AB-SA01, AB-PA01, and a mixture of AB-SA01 and AB-PA01 successfully lysed their hosts in the presence of biofilms of non-susceptible species. These phage cocktails applied to *S. aureus* and *P. aeruginosa* mixed-species biofilms caused a statistically significant reduction in the host cell population compared to the PBS-treated group (p < 0.05; Table 4.2). However, the reduction of the cell population in *S. aureus* and *P. aeruginosa* was less by half than the decrease observed in planktonic culture treatment. Most of the tetracycline-treated cultures produced no viable bacterial cells of the clinical isolates and low bacterial cell density of the laboratory strains.

		Bacterial cell counts after treatment					
Isolates		PBS	AB-	AB-	AB-SA01+		
Combination	Evaluated isolate		SA01	PA01	AB-PA01	Tetracycline	
<i>S. aureus</i> KUB7 ^s	<i>S. aureus</i> KUB7 ^s	6.2	4.5	7.4	5.5	3.8	
and PA 01 GFP^{P}	PA01 GFP ^P	6.4	6.3	3.8	4.0	3.9	
63-6538 ^s and 63-	63-6538 ^s	5.2	4.4	4.7	3.6	3.0	
6598 ^P	63-6598 ^P	5.5	5.6	3.6	3.7	0	
63-2498 ^s and 63-	63-2498 ^s	6.3	4.4	5.2	5.5	0	
5497 ^P	63-5497 ^P	7.1	6.2	5.5	5.2	0	
63-5656 ^s and 63-	63-5656 ^s	7	4.9	5.9	5.3	0	
6036 ^P	63-6036 ^P	8.5	8.9	4.7	6.3	0	
Summarized treatment effect							
	S. aureus mean	6.2	4.6	5.8	5.0	1.7	
	S. aureus reduction		1.6	0.4	1.2	4.5	
	P. aeruginosa mean	6.9	6.8	4.4	4.8	1.0	
	P. aeruginosa reduction		0.1	2.5	2.1	5.9	

Table 4. 2 Bacteria count (log₁₀ (CFU/ml)) of *S. aureus* and *P. aeruginosa* in mixed-species biofilms after 24 hours phage cocktails, tetracycline, and PBS treatment.

Key: ^S S. aureus isolates, ^P P. aeruginosa isolates.

Compared to PBS treatment, the application of AB-SA01 or AB-PA01 alone did not produce a statistically significant effect on the cell count of the non-host bacterial species population (p > 0.05; 6.9 vs 6.8 for AB-SA01 and 6.2 vs 5.8 for AB-PA01, Table 4.2). The mean bacterial cell population of each species remained unaffected when treated with the other species' phage cocktail alone. Treatment of mixed-species biofilms using the mixture of the two phage cocktails, AB-SA01+AB-PA01, produced similar cell reduction on both *S. aureus* and *P. aeruginosa* isolates as each phage cocktail treatment. The effect of tetracycline treatment caused a significant reduction in the population of both bacterial species.

4.4 Discussion

The two antibiotic-resistant bacterial pathogens commonly isolated from diabetic wound infections are *S. aureus* and *P. aeruginosa* (Ibberson *et al.*, 2017; Trivedi *et al.*, 2014; Fazli *et al.*, 2009). The rationale to examine the effect of phage cocktails treatment on mixed-species planktonic and biofilm cultures in the present study was that mixed-species biofilms cause many wound infections (Wolcott *et al.*, 2013; Ramakant *et al.*, 2011), and the efficacy of phages in such settings has been poorly examined. In polymicrobial infections, interspecies interactions range from antagonism to cooperation that can significantly impact the pathogenicity of microbes and clinical outcomes of the infection (Antonic *et al.*, 2013; Hoffman *et al.*, 2006).

In *P. aeruginosa* and *S. aureus* co-infections, HQNO produced by *P. aeruginosa* impairs *S. aureus* growth (Hoffman *et al.*, 2006) and facilitates the use of important *S. aureus* metabolites by *P. aeruginosa* (Mashburn *et al.*, 2005). HQNO is a molecule that inhibits Gram-positive bacteria respiration (Kharel *et al.*, 2004). Conversely, it was reported that HQNO selects for aminoglycoside-resistant *S. aureus* SCVs (Hoffman *et al.*, 2006) and protects *S. aureus* from stressful conditions such as oxidative stress in cystic fibrosis co-infection and co-culture with HQNO-producing *P. aeruginosa* (Antonic *et al.*, 2013). Thus, in *P. aeruginosa-S. aureus* co-infections, SCVs may be missed by clinical laboratories because of their slow growth and atypical phenotype (Hoffman *et al.*, 2006) and are capable of intracellular growth to cause chronic infections (Proctor *et al.*, 2006).

A co-infecting *P. aeruginosa* can activate some *S. aureus* virulence factors such as staphyloxanthin and catalase via inter-species communication (Antonic *et al.*, 2013). In this study, a mixed-species culture of mCherry-labelled *S. aureus* and GFP-labelled *P. aeruginosa* laboratory strains was used to examine the phage treatment effect by measuring the

magnitude of fluorescence of the two species within fixed time intervals for 24 hours by fluorescence spectrophotometry, with the results discussed below.

The death of a bacterial cell has been defined as the inability of a cell to grow to a detectable size or number on culture media (Berney *et al.*, 2007). Monitoring of the bacterial growth pattern using fluorescence detection at regular intervals, biofilm measurement using absorbance at OD_{600} nm, and CFU count results in the current study suggest that expression of fluorescent proteins does not negatively affect the growth fitness and viability of the fluorescently labelled strains (Burcham *et al.*, 2016; Barbier and Damron, 2016; Pereira *et al.*, 2010).

While light, fluorescence, confocal, and electron microscopy have been used to visualize cells in planktonic and biofilm cultures, microplate reader spectrophotometry allows the ready detection of fluorescently marked cells (Johnson *et al.*, 2015). The use of fluorescent proteins is common in microscopy and flow cytometry analysis (Pereira *et al.*, 2010; Briandet *et al.*, 2008; Hoerr *et al.*, 2007; Armstrong and He, 2001; Nebe-von-Caron *et al.*, 2000). However, the use of fluorescing proteins, particularly multiple fluorescent proteins simultaneously, in plate reader spectrophotometry, is limited.

Live cell examination techniques are essential tools to better understand microbial organization and function *in vitro* and *in vivo*. Multiple fluorescent proteins can be simultaneously applied to examine different microbial population interactions and to evaluate antimicrobial treatment effects in real-time (Lagendijk *et al.*, 2010; Chudakov *et al.*, 2005). The use of red fluorescent protein (mCherry) in combination with a green fluorescent protein (GFP) is suitable as the excitation and emission spectra of these proteins are well separated (Lagendijk *et al.*, 2010; Malone *et al.*, 2009; Matz *et al.*, 1999). The advantages of using mCherry and GFP as markers include ease of detection, no exogenous substrate is required

106

that may perturb biological samples, no need for cell processing to enable visualization, and the appropriateness for real-time monitoring of cells growth in terms of fluorescence intensity in culture (Vickerman *et al.*, 2015; Pereira *et al.*, 2010).

In this study, mCherry *S. aureus* KUB7 and GFP PAO1 were used to distinguish them in mixed-species culture. Loss of mCherry or GFP fluorescence from the respective bacterial population was considered an indicator of bacterial cell death due to the treatment effect. Previous studies demonstrated that the decrease in fluorescence is because of the cell death, not due to the degradation of the proteins, showing this fluorescence detection technique to be an early and sensitive marker of cell death (Steff *et al.*, 2001; Webb *et al.*, 2001). The magnitude of fluorescence detected, and hence the bacterial cell population recovered during single species culture without treatment, showed the growth and fitness capability of fluorescent proteins does not negatively affect the growth fitness and viability of the two fluorescently labelled species, consistent with published reports (Burcham *et al.*, 2016; Vickerman *et al.*, 2015; Pereira *et al.*, 2010; Barnes *et al.*, 2008; Maksimow *et al.*, 2002; Webb *et al.*, 2001).

The use of fluorescent spectrophotometry allowed real-time observation of the phagetreatment interaction. The findings of this study show that phage treatment effects can also be evaluated through fluorescence detection. There was a strong correspondence between the fluorescence detection and the bacterial population count result. For both the mCherry *S. aureus* KUB7 and GFP PAO1, the fluorescence signals continued to accumulate for no treatment, unaffected by the treatment, or negative control groups. Fluorescence signal magnitude of mCherry or GFP is reduced significantly when the mixed-species culture is treated with each phage cocktail alone. The change in the fluorescence signal in non-target species of phage cocktail treated wells resembles the fluorescence signal pattern observed during single-species culture without treatment. The higher green fluorescent detected in *S. aureus* phage-treated mixed-species culture than in GFP PAO1 single-species PBS-treated culture might be attributed to the enhanced growth *P. aeruginosa* population because of getting more nutrient such as iron supply from the dead *S. aureus* cells (Mashburn *et al.*, 2005). The fluorescence magnitude and bacterial cell population obtained during AB-SA01+AB-PAO1 and gentamicin treatment were similar; the fluorescence records are low, the fluorescence graphs are overlapped and remained at their lower levels throughout the experiment period, and bacterial population recovered at the end of the experiment is minimal or none.

The magnitude of fluorescence detected during untreated mixed-species culture was lower compared to untreated single-species cultures, which might be attributable to competition between the two bacterial species (Armbruster *et al.*, 2016; Baldan *et al.*, 2014; Michelsen *et al.*, 2014). All fluorescence detection results strongly associate with the respective colony count results. Continuous decrease or increase in fluorescence intensity with an associated decrease or increase in CFU, respectively, was consistently observed throughout the experiment. The reductions in bacterial cell count were associated with the susceptibility of each bacterial isolates to the specific phage cocktail and its component phages during spot test as it can be seen in Table 11 in the appendices. The bacterial cell density observed during untreated fluorescent mixed-species culture was similar to the result obtained during the co-culture study of the two species, including clinical isolates. This finding is supported by previous reports that show the magnitude of fluorescence, absorbance, and colony count results are supplementary to one another in showing bacterial growth (Kong *et al.*, 2016; Russo *et al.*, 2015).

The magnitudes of the fluorescence of mixed-species cultures treated with AB-SA01+AB-PAO1 and gentamicin were low but still higher than only AB-SA01 or AB-PA01 treated samples for each bacterial species. Nevertheless, this fluorescence magnitude difference was not significant. The time-dependent increase in the magnitude of fluorescence in untreated cultures and cultures treated with the phage cocktail of the other species is assumed to be consistent with the increase in that unaffected bacterial species population. While the fluorescence values of the untreated samples increase with time, it decreases for treated samples, which is consistent with efficient bacterial cell lysis by the host-specific phage cocktail, leading to a reduction in fluorescence. In this study, the decrease or loss of fluorescence corresponded with a decrease in the number of cells of each species, which is consistent with previous data comparing fluorescence detection with a colony count result (Webb *et al.*, 2001).

In mixed-species planktonic culture, infection with each phage cocktail alone caused a statistically significant decline of each bacterial host population (p < 0.001). Infection with the combination of the two phage cocktails (AB-SA01+AB-PA01) also resulted in a statistically significant decrease in cell density of both bacterial species (p < 0.001). These findings are similar to a study that showed planktonic *E. coli* grown in co-culture with *Salmonella* enterica did not survive because of the combined effect of infection from the lytic *E. coli* phage and competition from *Salmonella* enterica (Harcombe and Bull, 2005).

The effect of both phage cocktails separately and in combination was higher in planktonic culture than in biofilms, which agrees with previous reports (Fong *et al.*, 2017; Alves *et al.*, 2016; Mendes *et al.*, 2014; Kay *et al.*, 2011). Possible explanations include: the complex EPS may reduce efficacy of phages because of entrapment of phage particles in the biofilm matrix (Kay *et al.*, 2011), reduced multiplication of phages due to the large proportion of

metabolically inactive host cells (Harper *et al.*, 2014), and shedding of phage receptors from the host bacteria (Labrie *et al.*, 2010; Teplitski and Ritchie, 2009). This observation is consistent with the previous report that showed the application of phages phiIPLA-RODI, phiIPLA-C1C, and the combination of the two phages in the planktonic phase is more efficient than that in biofilm phase during *S. aureus* IPLA16 and *S. epidermidis* LO5081 mixed-species cultures (Gutierrez *et al.*, 2015). Tetracycline treatment in biofilms shows superior bacterial population reduction, on both bacterial species, compared to the phage cocktail formulations used in this study. Overall, the reduction of bacterial cell population during planktonic culture and biofilm treatment due to a specific phage cocktail or the combination of the two phage cocktails was significant.

Despite the current increasing interest in the role of biofilm in DFU infections, interspecies interactions within mixed-species biofilms are still poorly elucidated (Hoffman *et al.*, 2006). Our results are consistent with findings that demonstrate *S. aureus* growth and biofilm formation is affected in the presence of *P. aeruginosa* (Qazi *et al.*, 2006). When we compare the population of the two bacterial species under no treatment conditions, the population of *S. aureus* bacteria as measured by colony count was less by 0.7 log₁₀ (CFU/ml) and 1.3 log₁₀ (CFU/ml) in planktonic and biofilm states, respectively, compared to *P. aeruginosa* population. Suppression of *S. aureus* growth by *P. aeruginosa* might have occurred due to nutrient depletion, or production of specific metabolic inhibitors, or lytic enzymes by the latter bacteria (Machan *et al.*, 1991).

Several studies demonstrated an inhibitory effect of *P. aeruginosa* on *S. aureus* strains, including the highly virulent ones *in vitro* (Baldan *et al.*, 2014; Pastar *et al.*, 2013; Yang *et al.*, 2011; Qin *et al.*, 2009). In apparent contradiction to our results, a study that applied *P. aeruginosa* onto 5 day-old *S. aureus* biofilm, with 1:250 *P. aeruginosa* to *S. aureus* ratio,

showed stable co-existence between the two species (Woods *et al.*, 2018). The contradiction between the two co-existence study findings might be due to differences in mixed-species biofilm development methods, bacterial strain differences, age of the biofilms, and bacteria population ratio used.

In this study, phage control of mixed-species biofilms was investigated by two approaches, using phage cocktails specific for either of the bacterial species and a combination of the two phage cocktails, to determine the extent of population reduction. In the presence of 5% BSA and with 1:3 initial concentration of *P. aeruginosa*-to-*S. aureus*, both species grew well in biofilms. In mixed-species biofilms, treatment with a combination of AB-SA01 and AB-PA01 phage cocktails decreased the cell populations of both species. Compared to planktonic culture, the mixed-species biofilm cultures maintained higher numbers of each species in the presence of their specific phage cocktail or the combination of the two phage cocktails. Phage predation and bacterial competition might have a deleterious effect on planktonic populations, but the biofilm might have provided a sheltering mechanism whereby some bacteria can better persist despite the effects of phages and competition (Kay *et al.*, 2011). Studies have shown that phages can effectively reduce bacterial population in biofilms (Gonzalez *et al.*, 2017; Motlagh *et al.*, 2016; Gutierrez *et al.*, 2015; Parasion *et al.*, 2014; Sillankorva *et al.*, 2010).

When the mixed-species biofilms were treated with AB-SA01, AB-PA01, or AB-SA01+AB-PA01, the population of *S. aureus* and *P. aeruginosa* were lower than those in the non-treated control groups. These findings agree with the study findings that when *E. co*li-specific phages were added to an *E. coli-S.* Typhimurium mixed culture, *E. coli* became extinct or was maintained at a lower population density than without phage application (Harcombe and Bull, 2005). The present study provides evidence that a phage cocktail applied to mixed-species

biofilms can effectively reach and reduce the host cell population. The current study results differ from a previous observation (Tait *et al.*, 2002) in that it did not show that the presence of non-susceptible bacteria in mixed-species biofilms protected susceptible bacteria from being attacked by the specific phage. Because significant host bacterial cell reduction was obtained in this study not only with the combination of the two phage cocktails but also with each phage cocktail alone, the presence of bacterial protection from the non-susceptible host cannot be assumed as indicated by a report (Sillankorva *et al.*, 2010).

Mixed-species biofilms are complex microbial and extracellular matrix combinations in which the physiological state of bacterial host cells and the configuration of phage receptors play vital roles in the efficacy of phages (Gutierrez *et al.*, 2015; Sutherland *et al.*, 2004). This study shows that phage cocktails can potentially control mixed-species biofilms. Moreover, it demonstrates that despite the presence of non-susceptible hosts, phage cocktails can successfully lyse their host bacterium in a multispecies biofilm.

In conclusion, these findings suggest that the use of phage cocktails in mixed-species infection, with bacteria in planktonic or biofilm state, could provide practical alternative or adjunct to antibiotics in the settings of DFU infections. During the evaluation of phage cocktails treatment effect on fluorescent mixed-species culture, the fluorescence values increased for untreated samples, and decreased for treated samples, over time. This finding is consistent with efficient bacterial cell lysis by the phage cocktails in mixed-species culture - the decrease or loss of fluorescence associated with the decrease or absence of bacterial cells of each sample. The lytic efficacy of phage cocktails, both independently and in combination, was higher in planktonic cultures than in biofilms. The less efficacy of phage cocktails in biofilms could be because dead cells resulting from phage attack might facilitate the survival of remaining cells, or the reduced metabolism of bacterial cells in biofilms might poorly

support the proliferation of phages. This study confirms the lytic efficacy of AB-SA01, AB-PA01, and their combination on their hosts in mixed-species planktonic and biofilm states.

Chapter 5

Efficacy of Phage Cocktail Therapy in Diabetic Mouse Wound Infections Caused by Multidrug Resistant *Staphylococcus aureus*

Chapter 5: Efficacy of Phage Cocktail Therapy in Diabetic Mouse Wound Infections Caused by Multidrug-Resistant *Staphylococcus aureus*

Abstract

The diabetic foot ulcer (DFU) is a severe complication of diabetes mellitus. Antibioticresistant Staphylococcus aureus is frequently isolated from DFU infections. Phages represent an alternative or adjunct to antibiotic therapy. Here, the efficacy of AB-SA01, a cocktail of three S. aureus Myoviridae phages, made to cGMP standards and which has undergone two phase I trials, in the treatment of antibiotic-resistant S. aureus infections using a diabetic mouse model of deep skin wounds is described. Bilateral 6 mm excisional wounds, inflicted on the dorsum of Balb/c mice that were rendered diabetic by injecting streptozotocin, were infected with 6.7 log₁₀ CFU of multidrug-resistant (MDR) S. aureus. For 5 days, infections were treated topically with AB-SA01, and control groups received topical saline or topical saline plus intraperitoneal (IP) vancomycin. Bacterial load and wound healing parameters were used to assess treatment efficacy. Wounds of saline-treated mice showed no healing, but expanded and became inflamed, ulcerated, and suppurating. In contrast, AB-SA01 treatment decreased the bacterial load with efficacy similar or superior to vancomycin treatment. In phage-treated mice, wound healing was similar or superior to vancomycin treatment. No adverse effect related to the application of phages was observed. The findings of this study suggest that topical phage treatment may be useful in treating antibiotic-resistant S. aureus DFU infections.

5.1 Introduction

5.1.1 Diabetic foot ulcers

DFU is a frequently encountered complication, and represent the leading cause of health facility admission, among diabetic patients (Thurber *et al.*, 2017). Complications of DFUs are multifactorial in origin and costly (Thurber *et al.*, 2017). The global burden of DFU is rising, affecting up to 26.1 million people each year (Armstrong *et al.*, 2017). DFU is the precipitating cause for nearly 90% of limb amputations among persons with diabetes (Lavery *et al.*, 2007). Worldwide, 6.3% of persons living with diabetes are affected by DFU, and the lifetime incidence of a foot ulcer among persons with diabetes is estimated at 19% to 34% (Armstrong *et al.*, 2017). The 5-year mortality rate following foot amputation due to DFUs is up to 74% (Robbins *et al.*, 2008).

DFU management is costly because it may involve imaging studies, vascular surgery, wound dressing for lengthy periods, debridement, antibiotic therapy, and management of metabolic abnormalities (Richard *et al.*, 2011; Lipsky *et al.*, 2004). In the USA, the annual cost of DFU management is projected at an additional \$9-13 billion over the cost of diabetes itself (Rice *et al.*, 2013). In England, the cost of DFU management in 2014-2015 was estimated at between £837 million and £962 million (Kerr *et al.*, 2019).

5.1.2 Antibiotic-resistant S. aureus in DFU

Most DFU infections are polymicrobial with bacterial species ranging from aerobic Grampositive cocci to anaerobic Gram-negative bacilli (Mendes *et al.*, 2012b). *S. aureus* colonizes the skin and persists in the nares; nasal carriage is associated with an increased risk of clinical infection (Soge *et al.*, 2016). It is a virulent pathogen frequently isolated from DFU (Reveles *et al.*, 2016; Messad *et al.*, 2015; Rich and Lee, 2005; Lipsky *et al.*, 2004). Between 40% - 50% of *S. aureus* isolates identified from DFUs are MRSA, for which there are limited antimicrobial treatment options (Ding *et al.*, 2012). A recent Australian study found that 43.5% of DFUs were infected with *S. aureus*, and nearly half were MRSA (Commons *et al.*, 2015). *S. aureus* is a serious concern since it is becoming MDR to the most commonly used antibiotics (Gupta and Prasad, 2011).

The empirical antibiotic treatment for DFU infection includes an antibiotic against *S. aureus*. Thus, there is a high risk of emerging resistance strains (Barwell *et al.*, 2017; Bader, 2008). Hence, antibiotic-resistance is a major obstacle in treating infections caused by *S. aureus* (Uçkay *et al.*, 2015; Lipsky *et al.*, 2004). Furthermore, treatment may be ineffective because of insufficient antibiotic concentration at the site of infection because of deficient vascularization (Lipsky and Hoey, 2009). Hence, this pathogen affects the provision of empirical antibiotic therapy for DFU (Uçkay *et al.*, 2015; Lipsky *et al.*, 2004).

5.1.3 Phage therapy in DFU

Before the advent of antibiotics in the 1940s, phages were widely used in the USA and Europe. However, due to the development of many antimicrobials, lack of sufficient understanding about the biology of phages, inconsistent clinical trial results, and challenging regulatory environment (Kutter *et al.*, 2015; Summers, 2012; Monk *et al.*, 2010), phage treatment was excluded from western medicine, while it has continued to be practiced in eastern Europe for >100 years to treat bacterial infections (Summers, 2012; Sulakvelidze, 2005). Lytic phages kill their bacterial host by lysis (bursting the infected bacterial cell to release progeny phages) (Young, 1992a). The process of phage infection and subsequent self-replication in bacteria offers advantages over antibiotics: phages amplify themselves at the infection site provided there are sensitive bacterial hosts (Wittebole *et al.*, 2014). Phages are specific for the host bacteria species, an advantage over broadly active antimicrobials, as

phages are not expected to disrupt a patients' normal microflora. Phages lyse biofilm forms of bacteria, such as those typically found in infected DFU (Wittebole *et al.*, 2014; Donlan, 2009).

There is much evidence that phage use is safe (Brüssow, 2012; Pirnay *et al.*, 2011), and its extensive Russian and Georgian use has few adverse event reports (Brüssow, 2012). Phages are the most common biological entities in the biosphere, and in the gut, they outnumber bacteria (Reyes *et al.*, 2012; Edwards *et al.*, 2019). Phages are common in many environments; for instance, seawater has about 10⁷ phage particles/ml (Ortmann and Suttle, 2005). In 2006, the US Food and Drug Administration (FDA) gave GRAS ("Generally Regarded As Safe") status to food additive phage product Listex, against *Listeria monocytogenes* (Sharma, 2013); many phage products in this industry now have such status. There is no evidence in the literature that rapid bacterial lysis triggers harmful inflammatory responses to released bacterial contents.

The US Kutter group used Eliava Institute, Georgia, *S. aureus* phage Sb-1 to treat recalcitrant DFU infections (Fish *et al.*, 2016). In this 5-patient study, ulcers were all single-toe lesions, *S. aureus*-infected, unresponsive to standard antibiotics, with exposed, infected bone, and amputation was likely. Phages were applied for 2 days, then 5 days of standard dressings, repeated weekly. In each case, full wound closure was achieved, and the toes were saved. The product used was sequenced, purified, and approved for use in Georgia, but not made to cGMP quality, hindering the uptake of this product to western medicine. Further studies with a product that meets cGMP regulations would support the uptake of this treatment to broader western medicine.

Topical application of lytic phage treatment accompanied by appropriate wound dressing could effectively decrease bacterial load and improve wound healing in antibiotic-resistant infections (Rhoads *et al.*, 2009). To date, experimental evidence demonstrating the efficacy of phages in chronic DFU infections is scarce. This study sought to investigate the antibiotic activity and wound-healing potential of a topically administered phage cocktail. A cocktail of 3 *S. aureus* phages, designated AB-SA01, which has undergone two phase I trials and made to cGMP standard (Lehman *et al.*, 2019), was used in this study. The objective of this work was to determine the efficacy of this cGMP phage product in treating *S. aureus* infection in a diabetic mouse model.

5.1.4 Mouse skin wound model

To mimic the impaired wound healing in diabetes, animal models, especially mice, have been utilized (de Mayo *et al.*, 2017; Turner *et al.*, 2017). Mice wound models are most commonly used because of low cost, ease of management, and the ability to mimic key aspects of human wound healing (Park *et al.*, 2014). Excisional wound models in mice have been used to evaluate the efficacy of topically applied antibiotic agents (Davidson *et al.*, 2013). Unlike the re-epithelization and granulation process of human wound healing, wound healing in mice is through skin constriction (Dunn *et al.*, 2013). To mimic human wound healing in mice, a silicon sheet splint model that prevents excessive wound constriction was used (Park *et al.*, 2014; Lindblad, 2008). Due to the encouraging efficacy of AB-SA01 on single- and mixed-species planktonic cells and biofilms, as discussed in chapters 2 and 4 of this thesis, the diabetic mice splint wound infection treatment study reported herein was undertaken. In this chapter, the MDR *S. aureus* bacterial load reduction and wound healing in diabetic mice wound following AB-SA01 treatment is described.

5.2 Materials and methods

5.2.1 Bacterial isolates

Nine *S. aureus*, eight clinical isolates, and one laboratory strain, identified using standard microbiology methods and confirmed by MALDI-TOF MS, were examined as candidate organisms for this work. The antibiotic susceptibility profile of the isolates was determined using VITEK[®] 2. Furthermore, their susceptibility for phage cocktail AB-SA01, both in planktonic and biofilm states, was examined using methods detailed in chapter 2 of this thesis. All clinical isolates were MRSA, and four of them were also MDR. All isolates were susceptible to AB-SA01 and its three components phages in planktonic and AB-SA01 in biofilm cultures *in vitro*.

5.2.2 Phage cocktail

AmpliPhi Biosciences Corporation provided phage cocktail AB-SA01. AB-SA01 is a combination of *Myoviridae* phages designated J-Sa 36, Sa 83, and Sa 87. The mean titer was 9.3 log₁₀ (PFU/ml) for J-Sa 36 and Sa 83, 9 log₁₀ (PFU/ml) for Sa 87, and 9.1 log₁₀ (PFU/ml) for AB-SA01. The titer was evaluated on *S. aureus* laboratory strains RN4220 and SA6538 using plaque assay (Mirzaei and Nilsson, 2015; Merabishvili *et al.*, 2009). All component phages were fully sequenced by AmpliPhi Biosciences to confirm that they are free from antibiotic resistance or bacterial virulence genes (Lehman *et al.*, 2019).

5.2.3 Mice management

Female Balb/c mice were obtained from the Animal Resources Centre, Perth, Western Australia. Mice were free of obligate and opportunistic pathogens, including *S. aureus* <u>https://www.arc.wa.gov.au/wp-content/uploads/2019/07/BALBc-nude-stock-health-report-May-</u>

2019q.pdf. All experiments were approved by the Animal Welfare Committee, Flinders University/Southern Adelaide Local Health Network. Experimental design and reporting were carried out in compliance with the ARRIVE guidelines (Kilkenny *et al.*, 2010) and were consistent with the Australian Code for the Care and Use of Animals for Scientific Purposes, 8^{th} edition, 2013. Mice were kept at the College of Medicine and Public Health Animal Facility at 22 ± 3 °C and $55 \pm 5\%$ humidity under 12:12 hour light-dark cycle in Tecniplast GM500 Mouse IVC Greenline cages (Tecniplast Australia Pty Ltd, New South Wales, Australia) using corncob bedding. Mice were kept in *S. aureus* free housing conditions and provided water and a meat-free rodent maintenance diet (Glen Forrest Stockfeeders, Western Australia) *ad libitum*.

Mice were weighed 2-3 times each week and monitored for hunching, ruffled coat, lethargy, cold to touch, crinkling of skin, sunken eyes, and rapid or labored breathing at least once daily. Mice that showed over 15% weight loss or were critically ill were euthanized. Vancomycin (Sigma-Aldrich Corporation, New South Wales, Australia) was assessed for possible toxicity on six mice in a separate pilot study at 150 mg/kg dose rate, twice daily IP, for five consecutive days as described (Gibson *et al.*, 2007); these mice did not display any apparent adverse reaction. At the end of the experiment, mice were euthanized using 3% isoflurane, followed by cervical dislocation. Post-mortem examination of the external surfaces and visceral organs was conducted immediately following an established procedure (Parkinson *et al.*, 2011).

5.2.4 Induction of diabetes in mice

After a week of acclimatization, a total of 48 female 8-week-old Balb/c mice, housed in groups of 5, received STZ injection (Sigma-Aldrich Corporation, New South Wales, Australia) following an established protocol (Ito *et al.*, 2001; Paik *et al.*, 1980). STZ is a

naturally occurring alkylating antineoplastic agent that is toxic to pancreatic insulinproducing cells and which is used to treat pancreas islet cell carcinoma and to induce diabetes mellitus in laboratory rodents (Deeds *et al.*, 2011). After 4 hours of fasting, each mouse received an IP injection of 50 mg/kg freshly dissolved STZ in 0.05M citrate buffer pH 4.5 once daily for five consecutive days to avoid the acute toxicity of STZ. Balb/c mice are among the least susceptible to STZ toxicity (Hayashi *et al.*, 2006; Le May *et al.*, 2006; Bock *et al.*, 2005) and STZ-diabetic induced nephropathy (Hayashi *et al.*, 2006; Paik *et al.*, 1980).

Non-fasting blood glucose levels (BGLs) were checked every 2 - 3 days by tail vein bleeding after applying 3% lidocaine/prilocaine anesthetic cream. BGLs were measured using an Accu-Chek Performa (Roche Diabetes Care Australia, New South Wales, Australia) blood glucose meter. Mice were restrained using a modified tube, and the tail was cleaned with three alternating scrubs of 70% alcohol. Following a recommended procedure (Yano *et al.*, 2012), the distal end of the prominent tail vain was pricked using a 27-gauge needle, and blood was collected by gentle milking of the tail. The first drop of blood was touched by the tip of a test strip inserted into the digital blood glucose meter, and the reading was recorded. Blood flow was stopped by applying pressure on the pricked vein with sterile gauze. Blood glucose measurement using the blood glucose meter is a quick method to assess non-fasting BGL and needs only a drop of blood compared to the clinical chemistry method (Grant *et al.*, 2012).

Mice with non-fasting BGL < 13.9, between 13.9 and 22.2, and \geq 22.2 mmol/L on at least two different days were categorized as normal glycaemic, moderately hyperglycaemic, or severely hyperglycaemic status, respectively (Grant *et al.*, 2012). Mice were supplied with wet food and water *ad libitum*. Mice with no clinical signs were transferred to a group maintenance monitoring record and monitored for 4 weeks. Mice that showed any clinical signs were monitored twice daily. Diabetic mice were randomly assigned to one of the following groups: uninfected phage-, infected saline-, infected phage-, and infected vancomycin-treated.

5.2.5 Insulin treatment of diabetic mice

Diabetic mice were treated with 1.0 international unit (IU) NovoMix[®] 30 (Novo Nordisk, New South Wales, Australia) insulin daily subcutaneously to ameliorate the hyperglycaemic effects of diabetes and maintain body weight (Guo *et al.*, 2015). NovoMix[®] 30 is a pre-mixed neutral suspension consisting of rapid-acting (30%) and longer-acting (70%) protamine insulin (Thorisdottir *et al.*, 2009). The insulin treatment was administered between 9:00 am – 10:00 am daily.

5.2.6 Selection of S. aureus isolates for mice skin wound infections

In a pilot experiment, eight clinical and one laboratory *S. aureus* isolates were examined for their capability to establish skin infection on mice. Bilateral excisional wounds at the back of the mice 30-50 mm from the base of the skull and 10 mm from the midline were inflicted on each mouse as described (Holtfreter *et al.*, 2013; Dunn *et al.*, 2013; Mendes *et al.*, 2013). This area was chosen to hinder the grooming of the wound by the mouse itself. Each isolate was inoculated into two mice, with two PBS-inoculated mice used as negative controls. The results of the pilot experiment are below.

5.2.7 Preparation of bacteria for infection

An isolated colony of each *S. aureus* isolate was taken from 16 hours culture on MSA plates and grown in 3 ml TSB (Thermo Fisher Scientific, South Australia, Australia) overnight. One milliliter of overnight broth culture was adjusted to OD_{600} of 0.7 using an SP-830+ Metertech spectrophotometer (Adelab, South Australia, Australia) that corresponds to 8.3 log₁₀ (CFU/ml), through dilution by sterile physiological saline. Then, the broth culture was centrifuged at 3000 rpm, and pelleted cells were washed twice and resuspended with 1 ml PBS and ready for inoculation into the wound.

Bacterial density determination, in CFU/ml and then transformed to log_{10} , was carried out according to an established protocol (O'Toole, 2016). Fifty microliters of bacterial suspension were serially diluted into 450 µl of TSB. One hundred microliters bacterial suspension was mixed with 3 ml tryptic soya soft agar, spread on pre-warmed at 37 °C MSA plates in triplicates, and incubated for at least 24 hours. Colony count was performed on a plate containing 30-300 colonies. The total bacterial population was calculated by using the formula B = N/d where B = number of bacteria; N = average number of colonies counted on three plates; d = dilution factor (Mendes *et al.*, 2013).

5.2.8 Infection of mouse wound

Fifty microliters of bacterial suspension in PBS, containing an average 7.1 \log_{10} CFU, was applied to each mouse wound immediately after wound infliction. An equal volume of sterile PBS was inoculated into wounds of control mice. The wounds were then immediately overlaid with Opsite (Independence Australia, South Australia), which is a semi-occlusive waterproof film that adheres to the mouse skin. The infection was left for 3 days. On the 3rd and 10th days of infection, swab samples were taken by scrubbing the surface of each wound using sterile swabs to assess the establishment of infection. Each swab was rotated three times clockwise with enough pressure to produce a small amount of exudate from the wound and inserted into a separate tube of 3 ml TSB.

5.2.9 Quantification of bacteria in wound swabs

Tubes containing swabs were vortexed for 5 seconds, and a 100 μ l aliquot was used for serial dilutions. Quantification was performed using the 10-fold serial dilution method. One hundred microliters of each dilution were mixed into 3 ml tryptic soya soft agar warmed at 42 $^{\circ}$ C and immediately plated onto prewarmed at 37 $^{\circ}$ C MSA plates in triplicate. Colony count was performed following the standard procedure after 24 hours incubation (O'Toole and Wong, 2016). To evaluate the bacterial load of the wound, the quantity of CFU was determined by taking the average of colony counted as described earlier (Mendes *et al.*, 2013; Capparelli *et al.*, 2007) and a wound with more than 5.0 log₁₀ CFU on any of the days was considered critically infected (Levin and Antia, 2001).

5.2.10 Excisional wound infliction

Diabetic mice were conditioned with lemon-flavored paracetamol in drinking water at 1.34 mg/ml 3 days before wound infliction and provided for the entire experimental period. An established rodent wound infection model (Dunn *et al.*, 2013) was used. Mice were anesthetized using 3% isoflurane inhalation and maintained on 1.5% during surgery. Hair was removed on the dorsal skin using electric clippers and depilatory cream, and skin quickly sterilized using 70% ethanol. Mice were given a single 0.05 mg/kg buprenorphine injection subcutaneously just before wound infliction. Mice were injected with subcutaneous 170 μ l PBS immediately before the surgery to prevent dehydration. Lubricating eye drops were applied during surgery to protect eyes.

A skin wound extending through the *Panniculus carnosus* muscle was inflicted, using a 6 mm sterile biopsy punch and fine scissors, as indicated above. The bilateral wounds were inflicted at 10 mm on either side of the midline and 30 mm from the base of the skull. Swab

samples were collected from each wound to assess any pre-existing *S. aureus*. A sterile 1 mm thick silicone splint with a 7 mm diameter circular hole at the center was applied over the wound and sutured onto the skin concentric with the wound using 5-0 nylon suture. Cyanoacrylate glue was used to fix the silicone sheet splint to the skin before suturing. The purpose of using silicone splint was to mimic the human wound healing process through avoiding contraction type of mice wound healing. All surgical procedures were performed in a clean surgery room using sterile instruments.

5.2.11 Preparation of an isolate for mouse wound infection and treatment experiment

Among the eight *S. aureus* clinical isolates that established infection, MDR *S. aureus* 63-2498 isolate, which was susceptible to AB-SA01 and vancomycin, and showed higher CFU than the other MDR isolates, was selected for diabetic mice excisional splint wound infection treatment experiment. It was identified using MALDI-TOF MS. MDR *S. aureus* 63-2498 was tested using VITEK[®] 2 system and demonstrated resistance to multiple antibiotics, including benzylpenicillin, oxacillin, ciprofloxacin, erythromycin, and clindamycin and susceptibility to vancomycin. A single colony of MDR *S. aureus* 63-2498 from MSA plate was taken and grown in 3 ml TSB overnight. One milliliter from the TSB overnight broth culture was taken and adjusted to an optical density of 0.7 at OD₆₀₀ using an SP-830+ Metertech spectrophotometer that corresponds to 8.3 log₁₀ (CFU/ml) through dilution with sterile PBS. The adjusted broth culture was centrifuged at 3000 rpm, and pelleted cells were washed twice and resuspended with 1 ml PBS. This suspension was used for inoculation into the wound.

Mice were randomly assigned into *S. aureus*-inoculated phage-treated (n = 8), *S. aureus*inoculated vancomycin-treated (n = 6), *S. aureus*-inoculated PBS-treated (n = 7), and PBSinoculated phage-treated (n = 8) groups. Each *S. aureus*-inoculated mice wounds were infected with 50 µl bacterial suspension containing 6.7 \log_{10} CFU of MDR *S. aureus* 63-2498
isolate prepared as under preparation of bacteria for mice wound infection study using the established procedure (Mendes *et al.*, 2013). PBS-inoculated mice wounds received 50 μ l of PBS. Suspensions were applied directly into wounds using sterile pipette tips, overlaid with gauze, and covered with Opsite. On days 3, 5, and 7 post-infection, the Opsite was removed, and swab samples were taken. Treatments were administered immediately post-sample collection.

5.2.12 Vancomycin treatment

A pilot experiment was carried out to evaluate the toxicity of vancomycin treatment on mice. Six mice without wound infection were injected vancomycin (Sigma-Aldrich Corporation, New South Wales, Australia) for five consecutive days through IP at 150 mg/kg twice a day. Mice were checked for 10 days for any clinical signs. After 10 days post-treatment, mice were euthanized, and gross pathological lesion examination was conducted. During the wound infection treatment, vancomycin was administered IP at 150 mg/kg twice daily for five consecutive days, from day 3 to day seven post-infection, as described (Gibson *et al.*, 2007). This dosage resembles human dosing (Takigawa *et al.*, 2017; Domínguez-Herrera *et al.*, 2016; Docobo-Pérez *et al.*, 2012; Wold and Turnipseed, 1981).

5.2.13 Phage treatment

Treatments were administered immediately after sample collection commencing on day 3. Gentle debridement was applied before administering the treatments, as described previously (Seth *et al.*, 2013). Gauze (10 x 10 mm) was soaked with 70 μ l AB-SA01, equivalent to 7.9 log₁₀ PFU or with 70 μ l PBS solutions for control and vancomycin-treated mice, were applied to wounds and covered with Opsite, on days 3, 5, and 7.

5.2.14 Assessment of treatment effect

Wound size was measured in duplicate from different directions using a digital Vernier calliper by tracing the leading edge of epithelium within the wound (Shah *et al.*, 2013). The wound size on the excision day was defined as the original wound size. To assess the bacterial load, 100 μ l of 10-fold serially diluted swab sample suspensions were mixed with 3 ml trypticase soy soft agar and cultured on MSA. After 24 hours incubation at 37 °C aerobically, colony count was performed on plates with 30 - 300 colonies as recommended (O'Toole, 2016). The bacterial population was calculated as above.

5.2.15 Data management and statistical analysis

Data were managed using Microsoft Excel. SPSS software (IBM SPSS Statistics version 25) was used for statistical analysis. Data were reported as the mean \pm standard error of the mean and were compiled from at least three separate *in vitro* experiments or six mice per treatment group. A comparison of experimental groups was performed using one-way analysis of variance (two-tailed) or Student t-test. The association or independence of categorical variables was compared using Pearson's chi-square test, p < 0.05 values were accepted as statistically significant.

5.3 Results

5.3.1 Maintenance and clinical monitoring of mice

Mice were assessed for behavioral and physical abnormalities. Presence or absence of hunching and ruffled coat, lack of movement on stimulation, coldness to touch, crinkling of skin, sunken eyes, rapid and labored breathing, deterioration of nest quality, and wound infection were checked during clinical monitoring. None of the mice manifested any of these signs during the infection and treatment period, except one mouse from the infected-untreated group that manifested hunching and ruffled coat for 3 days. In the first 2 weeks post-STZ injection, most of the mice became diabetics, manifested sustained hyperglycemia, and continued weight loss. Subjective observation of cage litter showed increased urine production requiring daily rather than every third-day litter changes, consistent with polyuria. Furthermore, water intake by diabetic mice was estimated to be double their intake pre-STZ injection intake, indicative of polydipsia.

5.3.2 Infection establishment of S. aureus isolates on mice wound

Mice were monitored twice daily for signs of fatigue, stress, aggressiveness, weight loss, and any form of shock syndrome. No apparent clinical sign of systemic infection was observed for the 10-days follow-up time. From 18 infected wounds during the pilot *S. aureus* infection establishment study, 17 developed purulent exudate and necrosis. The remaining wound was relatively clean. All infected wounds presented signs of inflammation, including erythema, warmth, swelling, and tenderness at the earlier times of the experiment. At the later stage, wounds showed purulent exudate, discolored granulation tissue, and foul odor. All the saline inoculated control wounds were clean. At the end of the experiment, the mean wound diameter of infected mice was 7.7 mm, whilst the uninfected control wounds were 3 mm.

5.3.3 Evaluation of bacterial load during infection establishment study

All swab samples of infected wounds yielded *S. aureus* isolates on MSA. The bacterial load of these samples ranged from 7.9 to 9.9 \log_{10} (CFU/swab), as shown in Table 5.1. None of the control mice wounds yielded *S. aureus*.

ID	Antibiotic susceptibility	Log ₁₀ (CFU/swab)
63-5897	MRSA	9.1
63-2498	MRSA, MDR	9.4
63-7148	MRSA	9.7
60271003	MRSA	9.9
63-169	MRSA, MDR	8.5
63-6758	MRSA	9.6
63-7647	MRSA, MDR	8.9
46834854	MRSA, MDR	7.9
SA6538	Susceptible to all	8.6
Mean <u>+</u> err		9.1 <u>+</u> 0.7

Table 5. 1 Bacterial cell count of S. aureus from mice wound infection

5.3.4 Assessment of vancomycin toxicity in mice

During the vancomycin toxicity test, mice presented no visible clinical signs, and postmortem examination did not show any abnormalities.

5.3.5 Induction of diabetes

Of 48 female Balb/c mice that received STZ, 28 (58.3%) and 12 (25%) developed non-fasting BGL \geq 13.9 mmol/L within the first and second weeks of post-STZ administration, respectively. Of the remaining mice, three (6.3%) failed to develop diabetes, and five (10.4%) died before their diabetes status was determined. Of the diabetic mice, 21 (52.5%) developed severe hyperglycaemia, as defined above. While the mean body weight of mice before STZ administration was 18.9 \pm 0 gram, the mean body weight of normal glycaemic, moderately

hyperglycaemic and severely hyperglycaemic mice groups were 19.3 ± 1.0 , 17.8 ± 1.4 , and 16.7 ± 1.4 gram, respectively, at the end of the diabetes maintenance period.

5.3.6 Effect of insulin treatment on hyperglycaemic condition and body weight of mice

Following the increased number of severely hyperglycaemic mice and continued weight loss, 1 IU NovoMix[®] 30 was administered subcutaneously for diabetic mice daily. There was no significant difference between BGL records of 1 week pre- and post-insulin treatment in both hyperglycaemic groups. After 2 weeks on insulin treatment, the moderately hyperglycaemic mice showed 8.7% mean BGL decrease while severely hyperglycaemic mice showed a 4.5% mean BGL increase. The mean BGL measured on random days showed no significant difference from the record before the insulin treatment was started for both diabetic mice groups (p > 0.05; 20.5 vs 21.1 mmol/L for severely hyperglycaemic and 17.3 vs 17.5 mmol/L for moderately hyperglycaemic groups), as shown in Table 5.2. In general, the effect of insulin treatment in ameliorating the higher BGL condition was not pronounced in any of the diabetic groups.

To assess the effect of insulin treatment on body weight gain, mice were weighed twice a week at 9:00 am in their holding house. In general, there was a steady average body weight increment in diabetic mice, similar to the non-diabetic mice. No statistically significant difference was observed in body weight gain between insulin-treated moderately and severely hyperglycaemic groups (p > 0. 05; Table 5.2). Mice that failed to maintain body weight, before the commencement of the wound infection treatment study, were euthanised.

	Glycaemic status											
		Severely	hyperglycae	emic	Moderately hyperglycaemic				Normal glycaemic			
Insulin	No of		Body	Died/	No of		Body	Died/	No of		Body	Died/
injection	mice	BGL	Weight	Euth.	mice	BGL	Weight	Euth.	mice	BGL	Weight	Euth.
Before	21	21.1 ± 4.0	15.2 ± 1.6	0	19	17.3 ± 3.1	16.0 ± 1.7	0	3	8.3 ± 0.9	18.2 ± 1.0	0
1 week after	21	23.9 ± 3.6	15.4 ± 1.7	0	19	18.0 ± 3.3	16.4 ± 1.8	0	3	7.7 ± 1.0	18.6 ± 0.9	0
2 weeks after	21	22.6 ± 4.2	16.1 ± 2.0	0	19	15.8 ± 4.6	17.1 ± 1.7	0	3	7.9 ± 1.0	19.0 ± 0.7	0
3 weeks after	21		17.5 ± 2.2	0	17		18.6 ± 1.6	2	3		19.7 ± 1.0	0
4 weeks after	19		18.5 ± 2.2	2	15		19.5 ± 1.7	2	3		20.5 ± 1.5	0
5 weeks after	18		19.1 ± 2.3	1	15		19.8 ± 1.6	0	3		20.6 ± 1.4	0
6 weeks after	18			0	15			0	3			0
7 weeks after	16			2	13			2	3			0
8 weeks after	15			1	12			1	3			0
Random days	15	20.5		0	12	17.5 ± 5.3		0	3	7.1 ± 0.4		0
At the end of												
the experiment	15			6	12			7	3			0

Table 5. 2 Effect of insulin on BGL, body weight, and mortality of mice. BGL is in mmol/L; mouse weights are in grams.

Euth. represents euthanised.

5.3.7 Result of splint excisional wound infliction

Twenty-nine diabetic mice in good health and body condition were selected for infection treatment using the splint excisional wound model experiment. A pair of splint excisional wounds were inflicted on each mouse, as illustrated in Figure 5.1, and mice were categorized under four treatment groups. Most of the silicone sheet splint stayed fixed on the skin of mice, but a few needed re-suturing. Both the adhesive and anchoring sutures kept the splint on the skin.



Figure 5. 1 Splint excisional wound infliction and wound splint silicone sheet application: shaved wound area and 6 mm diameter fresh wound (first figure), silicone sheet splint applied (second figure) and gauze and Opsite applied after infection is inoculated (third figure).

5.3.8 Result of inoculation of MDR S. aureus into diabetic mice splint wound

The establishment of MDR *S. aureus* in the diabetic mice splint wound was examined on day 3 of inoculation using swab samples. The colony count result demonstrated that all infected wounds yielded *S. aureus*. The mean bacterial cell count ranged from $7.0 - 9.0 \log_{10}$ (CFU/swab). The non-infected control mice's wounds did not produce significant numbers of *S. aureus*.

5.3.9 Effect of AB-SA01 on MDR S. aureus infection in diabetic mice wounds

The bacteria count on days 3, 5, and 8 post-treatment showed a decrease in all phage- and vancomycin-treated mice. No *S. aureus* was detected from more than half (n = 5/8) of phage-treated wounds, and the mean bacterial load from this group was 1.1 log₁₀ (CFU/swab) on day 8 of the treatment period. In contrast, 4.8 log₁₀ (CFU/swab) mean *S. aureus* cells were detected in vancomycin-treated mice on day 8 of the treatment period, as shown in Figure 5.2 and Table 5.3.



Figure 5. 2 Effect of phage treatment on MDR *S. aureus* bacterial load compared to control and IP vancomycin treatment. Saline-treated mice (control, red); AB-SA01 phage-treated mice (green); vancomycin-treated mice (blue). The arrows on the horizontal axis indicate the infection date (first arrow, day 0) and treatment start date (second arrow, day 3), respectively. Treatments were applied from day 3 to day 7, as detailed in the methodology part of this chapter. The arrow on the vertical axis indicates the infection dose.

A statistically significant bacterial load decrease (p < 0.05) was observed on day 3 of the phage cocktail and vancomycin treatment. A more pronounced bacterial cell population reduction (p < 0.001; Figure 5.2 and Table 5.2) was observed on day 5 of the treatment period for both phage-treated and vancomycin-treated groups. On day 8 of treatment, the bacterial load reduction due to phage cocktail treatment was statistically different from vancomycin treatment (p < 0.05; 1.1 vs 4.8 log₁₀ (CFU/ml)). There was no significant number of *S. aureus* detected from uninfected phage-treated mice wounds throughout the experimental period.

Mouse ID	Treatment group	Day 0	Day 3	Day 5	Day 7	Day 10
4-NEM		0	8.99	8.36	8.51	8.57
4-1L		0	7.72	8.11	8.43	7.91
4-2R		0	8.34	8.85	8.38	8.18
5-NEM		0	7.00	8.34	8.41	7.62
5-1L	PRS	0	8.76	8.70	8.20	8.11
5-2R	r DS	0	6.97	7.73	8.51	8.53
10-RL		0	7.00	8.38	8.26	7.95
Mean <u>+</u> err		0	8.42 <u>+</u> 0.9	8.47 ± 0.4	8.39 <u>+</u> 0.1	8.23 <u>+</u> 0.3
7-1R		0	8.40	6.43	2.26	0.00
7-1L		0	8.40	6.18	5.72	2.00
7-RL		0	7.88	6.45	6.30	2.28
7-2R	Dhaga	0	8.56	6.26	6.63	0.00
9-NEM		1.1	8.38	6.28	2.48	4.34
9-1R	I hage	1.5	8.81	6.45	5.15	0.00
9-1L		0	8.41	5.80	5.72	0.00
9-RL		0	6.66	5.43	2.32	0.00
Mean <u>+</u> err		0.3	8.19 <u>+</u> 0.7	6.16 <u>+</u> 0.4	4.57 <u>+</u> 1.9	1.08 ± 1.6
6-1R		0	8.64	6.18	5.15	7.04
6-RL		0	7.46	4.70	3.96	5.86
6-2R		0	8.04	7.99	4.75	6.75
8-NEM	V	0	7.84	6.49	5.58	1.04
8-1L	v anconnychi	0	7.80	6.52	4.28	6.04
8-2R		0	7.72	7.92	4.96	2.11
Mean <u>+</u> err		0	7.92 <u>+</u> 0.4	6.63 <u>+</u> 1.2	4.78 <u>+</u> 0.6	4.81 <u>+</u> 2.6

Table 5. 3 Bacterial loads (in log_{10} CFU/swab) in diabetic mice wounds infected with MDR *S. aureus* isolate SA63-2498 and treated with phage cocktail AB-SA01 or controls.

5.3.10 Effect of AB-SA01 on diabetic mice wound healing infected with MDR S. aureus

Infected phage- and vancomycin-treated mice wound showed a decrease in size during and after the course of treatment, leading to complete wound healing. All uninfected phage-treated wounds also decreased in size through time and had complete healing at the end of the experiment. In contrast, infected PBS-treated wounds increased in size, with examples illustrated in Figures 5.3 and 5.4. The mean wound diameters at the end of the experiment were 0.0, 0.2, 0.3, and 7.8 mm for infected vancomycin-, infected phage-, uninfected phage-, and infected PBS-treated groups, respectively.



Figure 5. 3 Diabetic mice wound appearance at day 10: A infected PBS-treated, showing lack of healing and expansion of wounds; **B** uninfected phage-treated, **C** infected phage-treated, and **D** infected vancomycin-treated. Wounds in **B**, **C**, and **D** groups showed similar complete healing.

All infected PBS-treated wounds manifested non-healing ulcers characterised by purulent exudate, discoloured granulation tissue, and foul odour. Compared to the infected PBS-treated mice wound size, a statistically significant wound size decrease (p < 0.001) was found at the end of the experiment for infected phage- (7.8 vs 0.2 mm), uninfected phage- (7.8 vs 0.3 mm), and infected vancomycin-treated (7.8 vs 0.0 mm) mice. In contrast, the infected PBS-treated mice showed a statistically significant wound size increase at the end of the experiment compared to the original wound size (p < 0.05; 7.8 vs 6.0 mm). There was no statistically significant difference in wound healing progression among uninfected phage-, infected phage-, and infected vancomycin-treated groups (p > 0.05; Figure 5.4). Phage treatment caused no apparent adverse effects in mice in the absence or presence of its bacterial host.



Figure 5. 4 Effect of phage AB-SA01 on wound healing. The arrows indicate the infection date (day 0), and the date treatment started (day 3). No healing was seen in infected PBS-treated mice (red). In contrast, complete wound healing was seen in uninfected phage-treated wounds (blue), and infected wounds treated with phage (green), and in the control group with infected vancomycin-treated wounds (yellow).

5.3.10 Clinical assessment of diabetic mice infected with MDR S. aureus and treated

Hunching, ruffled coat, lethargy, cold to touch, crinkling of skin, sunken eyes, and rapid or labored breathing were criteria used to check the health of mice. No mice manifested these signs of systemic infection during the wound infection and treatment periods. Non-healing wounds with purulent exudate, discolored granulation tissue, and foul odor were noted in all infected wounds before treatment for all treatment groups, and the entire period of the experiment in the infected untreated groups. No mortality was recorded associated with infection.

5.3.11 Mice mortality and post-mortem findings

The mortality rate of mice during the entire experimental period was 33.3% (n = 16/48). Euthanasia based on animal welfare contributed to half of the mortality records. Manifesting two or more clinical signs that are indicated in the maintenance and clinical monitoring sheets and body weight loss were reasons for euthanasia. Many of the mice were euthanized on the ground of unacceptable weight loss that is greater than 15% of their initial body weight. The mortality rate was similar among severely hyperglycaemic, moderately hyperglycaemic, and normal glycaemic mice. The highest mortality rate (68.8%, n = 11/16) was recorded between week 5 and 7 post-STZ injection. No mortality was observed during the excisional wound infection and treatment period. A post-mortem examination was conducted only on euthanized mice because of most mouse deaths overnight, and typically they were cannibalized. Post-mortem examinations did not show any evident pathological lesions other than the skin wounds. There was no food in the gastrointestinal tract, and little to no abdominal fat was observed.

5.4 Discussion

An earlier study demonstrated that *S. aureus* establishes infection on mice skin within 4 hours of inoculation (Kugelberg *et al.*, 2005). The present study found that MRSA and MDR *S. aureus* isolates were successful in establishing infection on mice skin in 3 days. Clinical signs observed during infection establishment experiments of this study include inflammation, suppurative exudate, and widened and deepened wound, similar to lesions due to *S. aureus* infections reported earlier (Malachowa *et al.*, 2013). The mean bacterial count results, 9.1 log₁₀ (CFU/swab) showed the successful establishment of infection in mice skin by the clinical isolates of *S. aureus* collected from DFU patients, in agreement with earlier studies (Malachowa *et al.*, 2013; Kugelberg *et al.*, 2005).

Vancomycin is effective against most isolates of MRSA (Takigawa *et al.*, 2017). Vancomycin in mice has been non-toxic under standard dose, regimen, and duration (Domínguez-Herrera *et al.*, 2016; Gibson *et al.*, 2007). In this study, no evidence of toxicity was observed during the entire follow-up period, nor was any gross pathological lesion seen in any of the mice, consistent with similar studies (Domínguez-Herrera *et al.*, 2016; Docobo-Pérez *et al.*, 2012; Wold and Turnipseed, 1981).

Following the demonstration of the *in vitro* efficacy of phage cocktail AB-SA01 and its components on *S. aureus* isolates collected from DFU patients, as detailed in Chapter 2, the present study was initiated to examine the efficacy of AB-SA01 using a diabetic mouse wound infection model. The nongenetic diabetic mouse model was preferred because it mimics the late 139

stage of type 2 diabetes with a relative deficiency of insulin and a tendency for infections (Strowski *et al.*, 2004). Balb/c mice are among the least susceptible to STZ toxicity because they have relatively high pancreatic β -cell mass due to their large number of islets (Hayashi *et al.*, 2006; Bock *et al.*, 2005; Cardinal *et al.*, 1998). Moreover, female mice are relatively resistant to STZ toxicity compared to males because sex steroids may, to an extent, protect them from β -cell injury (Le May *et al.*, 2006). Besides, STZ-diabetic nephropathy is more pronounced in male mice compared to females (Hayashi *et al.*, 2006; Paik *et al.*, 1980). Hence, female Balb/c mice were selected for the current experiment. The induction of diabetes was started on 8-weeks-old mice because this age group is commonly considered appropriate in previous studies to the STZ-induced diabetes model (Dekel *et al.*, 2009; Ito *et al.*, 2001).

STZ is a broad-spectrum antibiotic extracted from *Streptomyces achromogenes* bacteria for the treatment of bacterial infections and carcinoma of pancreas islet cells (Deeds *et al.*, 2011). STZ has been commonly used for the induction of diabetes mellitus in laboratory rodents as it is toxic to β-cells of pancreatic islets responsible for insulin production. Injection of a low dose of STZ for five consecutive days has been recommended to develop progressive hyperglycaemia (Deeds *et al.*, 2011; Kume *et al.*, 2004; Paik *et al.*, 1980). The Animal Models of Diabetes Complications Consortium (AMDCC, <u>http://www.amdcc.org</u>) also recommends a low-dose model for STZ-induced diabetes. The result of this study agrees with these reports in terms of producing progressive hyperglycaemia and a large percentage of diabetic mice. This study found no significant difference between severely and moderately hyperglycaemic mice in terms of glycaemic control, weight gain, or survival rate through time.

The severe hyperglycaemia and substantial body weight loss in many mice observed in this study are consistent with a study that used a similar treatment regimen (Hayashi *et al.*, 2006). In an animal model of diabetes, it is essential to monitor the relationship between diabetes progression and loss of body weight to reduce excessive body weight loss (>10%) as well as mortality of experimental animals. In this study, about 33% of mice mortality was observed, which is higher than previous reports (Deeds *et al.*, 2011; Kume *et al.*, 2004; Paik *et al.*, 1980). Half of the loss of mice was due to STZ treatment-related morbidity and severe weight loss. No mortality occurred during the excisional wound infection and treatment period.

Phage cocktail AB-SA01 was well tolerated by the mice, as shown by the lack of clinical abnormalities such as a change in the well-being of mice or evidence of anaphylactic reactions due to harmful response to phages. This finding agrees with previous study results on the safety of phage therapy (Drilling *et al.*, 2017; Speck and Smithyman, 2016; Rhoads *et al.*, 2009; Capparelli *et al.*, 2007). Potential adverse effects such as toxic shock because of the rapid and considerable numbers of bacteria lysis and release of large quantities of toxins (Capparelli *et al.*, 2007) were not observed. The absence of adverse effects was expected as the AB-SA01 material used in this experiment is a well-characterized phage cocktail, produced under cGMP standards, and approved by that complies with the U.S. Food and Drug Administration (FDA) and the Australia's Therapeutic Goods Administration (TGA) to conduct clinical phase I trials and single-patient emergency treatment (Lehman *et al.*, 2019).

Skin wound healing in rodents involves skin contraction, a healing mechanism not seen in human wound healing, which occurs by secondary intention, granulation, and re-epithelialization (Fukui *et al.*, 2017; Park *et al.*, 2014; Mendes *et al.*, 2013). In this study, a splint wound model was employed to mimic human skin wound healing through avoiding contraction type of mice wound healing, as described (Dunn *et al.*, 2013).

In the current study, the therapeutic potential of AB-SA01, a cocktail of 3 *S. aureus Myoviridae* phages, was evaluated on diabetic mice to treat wounds infected with an MDR *S. aureus* clinical isolate resistant to multiple antibiotics, including those most commonly used to treat staphylococcal infections such as oxacillin, clindamycin, and trimethoprim-sulfamethoxazole. This phage application methodology is supported by data from a previous report (Seth *et al.*, 2013). AB-SA01 reduced bacterial load to low levels while the control mice were still infected with a high bacterial load. This finding is in line with studies that showed bacterial load reduction in mice infected with *S. aureus* (Capparelli *et al.*, 2007) and Wistar rats with *S. aureus* and *P. aeruginosa* (Mendes *et al.*, 2013). A significant bacterial load decrease was observed on day 3 post-AB-SA01 administration (p < 0.05), and it continued to decrease throughout the treatment period. The infection process in phage- and vancomycin-treated groups was controlled well by the end of the experiment compared to untreated control mice (p < 0.001).

The bacterial load in the vancomycin-treated group decreased at a similar trajectory to the phagetreated group during the treatment period but remained constant from the day treatment was completed onwards. The bacterial load reduction at the end of the experiment in the vancomycintreated mice was not as great, and significantly different from that in phage-treated mice (p < 0.05). The continued bacterial load reduction in phage-treated mice after the treatment was stopped demonstrates an advantage of phages over antibiotic treatment, because phages replicate in the presence of their bacterial host (Fu *et al.*, 2010; Payne *et al.*, 2000). Of 8 infected phage-treated mice, *S. aureus* had been eliminated from the wounds of 5 mice by the end of the experiment. In the three remaining mice with a detectable bacterial load, the bacterial population was minimal compared to that in PBS-treated mice wounds. In contrast to phage-treated mice, at the end of the experiment, *S. aureus* was still detected in all vancomycin-treated mice with a mean bacterial load of 4.8 log_{10} (CFU/swab). Four of the six vancomycin-treated mice had a residual bacterial burden of $\geq 5.9 log_{10}$ (CFU/swab) at the end of the experiment.

From the findings of this study, it is possible to conclude that phage treatment resulted in superior or equivalent efficacy to vancomycin. The data suggest that phages may be valuable antibiotic treatment adjuncts in this setting because they: a) may be used topically, b) show apparent excellent efficacy with a low number of treatments, which is compatible with standard wound care, c) show no local inflammatory reaction, d) may continue to work for an extended period post-administration, and e) have fewer documented adverse effects as compared to antibiotics such as vancomycin (Speck and Smithyman, 2016; Rhoads *et al.*, 2009).

The use of phage cocktails (instead of single phage) broadens the spectrum of activity of phage formulations and reduces the likelihood of development of phage-resistant bacteria (Chan *et al.*, 2013; Cairns and Payne, 2008). This could be because component phages complement each

other, possess different infection mechanisms, and may identify different receptors (Gu *et al.*, 2012). The three phage components in AB-SA01 were shown to broaden the spectrum of activity, complement each other, and reduce the frequency of bacterial resistance when combined (Lehman *et al.*, 2019).

Topical administration of phage cocktail AB-SA01 exhibited an effect similar to that observed with vancomycin in facilitating wound healing. The decrease in wound size associated well with the decrease in bacterial load. Wound size measurement showed a statistically significant difference between the treatment and control groups (p < 0.001). These observations are consistent with a previous report (Chhibber *et al.*, 2013). More than 85% of uninfected phage-, infected phage-, and infected vancomycin-treated mice wounds were completely closed, and hair regrowth was observed on some of the wounds. No statistically significant difference was observed in wound closure between these three treatment groups (p > 0.05).

In conclusion, the treatment of diabetic mice wounds infection caused by MDR *S. aureus* using topical application of phage cocktail AB-SA01 is effective with efficacy similar or superior to vancomycin treatment, as shown by bacterial load reduction and wound closure. The findings of this study show that phages represent a potentially effective topical treatment for diabetic ulcers infected with antibiotic-resistant pathogens.

Chapter 6 General Discussion

Chapter 6: General Discussion

6.1 Discussion

Diabetes mellitus is becoming a global epidemic. An estimated 463 million people are living with diabetes mellitus in 2019, which is expected to increase to 578 and 700 million people by 2030 and 2045, respectively (Saeedi *et al.*, 2019). DFU infection is a common, severe, costly, and disabling complication of diabetes mellitus and often associated with antibiotic-resistant polymicrobial infections (Alexiadou and Doupis, 2012). These infections are dominated by *Staphylococcus aureus* and *Pseudomonas aeruginosa* (Lee *et al.*, 2017; Malik *et al.*, 2013; Ramakant *et al.*, 2011).

S. aureus and *P. aeruginosa* are frequently isolated together in DFU infections, with joint occurrence causing more severe disease than infection with each species alone (Bessa *et al.*, 2015; Trivedi *et al.*, 2014; Pastar *et al.*, 2013; Körber *et al.*, 2010; Hendricks *et al.*, 2001). Together, their biofilm-forming capability contributes to their resistance against antibiotics and the immune system (Steenackers *et al.*, 2016; Ramakant *et al.*, 2011; Dowd *et al.*, 2008; Gilbert *et al.*, 2002). The objective of this work was to evaluate the treatment efficacy of two phage cocktails and their components, AB-SA01 against *S. aureus* and AB-PA01 against *P. aeruginosa*, respectively. The efficacy evaluation was conducted using *in vitro* and *in vivo* experiments.

Infections from multi-, extensive-, and pan-drug resistant strains are common in various groups of bacteria (Courvalin, 2016). The increasing failure of antibiotics prompts the development of phages as alternative anti-infective agents. This study was conducted on laboratory strains, and clinical isolates of *S. aureus* and *P. aeruginosa* sampled during the study from DFU infection patients at diabetes clinics in Adelaide, South Australia. The two bacterial species were selected because they are the most common infectious agents in DFU patients based on epidemiological studies and treatment failure due to their high antibiotic-resistance characteristics (Saltoglu *et al.*, 2018; Kateel *et al.*, 2018) and because phages against these species are available. Chapters 2 and 3 of this thesis reported the results of identification and antibiotic susceptibility profiles of *S. aureus* and *P. aeruginosa* clinical isolates. These chapters also examined the lytic efficacy and host range of phage cocktails AB-SA01 and AB-PA01 and their components against laboratory and clinical isolate hosts in planktonic and biofilm phases.

Phages are specific antibacterial agents that are often effective against certain strains of a bacterial species (Abatangelo *et al.*, 2017; Pincus *et al.*, 2015; Kwiatek *et al.*, 2015; Balarjishvili *et al.*, 2015; Collins, 1955). Hence, developing phage formulations with broad-spectrum efficacy on most strains of significant DFU infection bacterial species is a challenge. It has been shown that for some bacterial species, phage cocktails can be formulated to lyse most strains of the target bacterial species (Lehman *et al.*, 2019; Fong *et al.*, 2017; Bernasconi *et al.*, 2017; Chan *et al.*, 2013). In this study, two phage cocktails: AB-SA01 against *S. aureus*, and AB-PA01 against *P. aeruginosa* are used. AB-PA01 contained four phage components: two *Myoviridae* and two *Podoviridae*. AB-SA01 contained two *Myoviridae* phages. An important aspect of bringing 147

phages to the clinic is that they are made to high standards. All component phages were exclusively lytic and did not encode bacterial virulence factor genes (Lehman *et al.*, 2019; Fong *et al.*, 2017).

AB-SA01 and AB-PA01, and their component phages were tested on 53 *S. aureus* and 41 *P. aeruginosa* isolates, respectively. The findings in the present study show strong infectivity and broad host range characteristics of all phage preparations. These findings are consistent to earlier studies findings that used similar test protocols, phages used in this study, and different phages (Lehman *et al.*, 2019; Zhang *et al.*, 2018; Fong *et al.*, 2017; Leskinen *et al.*, 2017; Hall *et al.*, 2012; Mendes *et al.*, 2014; Fu *et al.*, 2010; O'Flynn *et al.*, 2004).

Chapter 2 examined the lytic effect and host range of the phage cocktail AB-SA01 and its components. AB-SA01 is a mixture of J-Sa 36, Sa 83, and Sa 87 phages. AB-SA01 lysed about 89% of the isolates tested regardless of the antibiotic susceptibility profile of the isolates. The host range of phage cocktails could potentially be improved by incorporating other phage types, although it has been suggested that too many phage types in the therapeutic mixture could bring a negative impact on non-target bacteria (Alves *et al.*, 2014; Chan *et al.*, 2013). It was observed that AB-SA01 produced a strong lytic effect, exhibited by complete and clear lysis, on over 95% of the susceptible isolates than any of the component phages alone. This finding shows that resistance strain to one component phage could be treated by the other in the mixture as reported earlier (Lehman *et al.*, 2019; Mendes *et al.*, 2014).

Studies suggest the use of phage cocktails over single phage minimizes the emergence of resistant mutants (Gu *et al.*, 2012; Tanji *et al.*, 2005). In the present study, it was also found that at least 75.5% of *S. aureus* clinical isolates, including the MDR and MRSA, were sensitive to each phage-type of the phage cocktail. This finding shows that AB-SA01 and its components were effective irrespective of antibiotic susceptibility of the isolates. These findings are in line with previous studies conducted on the efficacy of phage cocktails and component phages of *S. aureus* (Leskinen *et al.*, 2017; Mendes *et al.*, 2014; Kelly *et al.*, 2011) and other bacterial species (Manohar *et al.*, 2019; Gundogdu *et al.*, 2016; Ozkan *et al.*, 2016; Gu *et al.*, 2012; Hooton *et al.*, 2011; Fu *et al.*, 2010; O'Flynn *et al.*, 2004).

The broad host range and strong lytic efficacy of phage cocktail AB-SA01 and its components led to the evaluation of the phage cocktail on biofilms in this study. During the evaluation, significant biofilm biomass reduction was observed on more than 79% of the isolates due to the lytic efficacy of AB-SA01. Not surprisingly, most resistant and intermediately susceptible *S. aureus* isolates to AB-SA01 during spot tests were also resistant in their biofilm form. This resistance was not unexpected because bacteria in the biofilm state develop more resistance to phages than in a planktonic state (Fu *et al.*, 2010). The resistance might be because bacterial cells in biofilms are metabolically less active and are less accessible for phage infection than bacterial cells in planktonic forms (Gutierrez *et al.*, 2015; Sutherland *et al.*, 2004). The effect of AB-SA01 on *S. aureus* isolates biofilm biomass reduction is consistent with earlier reports (Alves et al., 2014, Kelly et al., 2011). In chapter 2, it was shown that the use of phage cocktail AB-SA01

could provide an alternative to antibiotics in combating antibiotic-resistance, including MRSA and MDR *S. aureus*, infection in planktonic and biofilm states.

In chapter 3 of this thesis, the lytic efficacy of phage cocktail AB-PA01 and its component phages on P. aeruginosa laboratory strains and clinical isolates was examined. The study was conducted using spot tests and biofilms. The results show that each component phage and their mixture displayed strong lytic efficacy. Moreover, AB-PA01 demonstrated stronger lytic efficacy and a broader host range compared with each phage type during spot test. AB-PA01 produced a lytic effect on more than 93% of isolates, while the host range of the component phages varied within 65% - 88% of the isolates. It is expected this effect to increase with high MOI because high MOI reduces biofilms and prevent biofilm regrowth (Alves et al., 2016; Abedon, 2009; Kasman et al., 2002). Similar findings were reported previously (Leskinen et al., 2017; Mendes et al., 2014; Chan et al., 2013; Hall et al., 2012). AB-PA01 reduced biofilm biomass from more than 82% of P. aeruginosa isolates biofilms within a 12-hour treatment period. This biofilm biomass reduction indicates the lytic efficacy of AB-PA01 against its targets in planktonic and biofilm phases, irrespective of the antibiotic susceptibility profile of the isolates. Studies on various phage cocktails have also found similar results (Alves et al., 2016; Gutierrez et al., 2015; Fu et al., 2010).

In chapters 2 and 3, it was observed that both AB-SA01 and AB-PA01 phage cocktails significantly reduce the biofilm biomass produced by their hosts in single-species biofilms. These results were obtained using a static biofilm model system on a polystyrene surface for 48

hours and applying phage cocktail treatment on the preformed biofilms for 12 hours. The effect of the phage cocktails was measured using crystal violet staining. Because the probability of occurrence of mutation is in the order of 1 x 10^{-8} strains (Matic *et al.*, 1997), a bacterium may need to undergo multiple mutations to achieve resistance.

The probability of resistance to the phage cocktail is lower than to single phage. If resistance occurs during phage cocktail therapy, it can be improved by including additional phage types to the therapeutic mixture. In the case of an *in vivo* situation, reducing the bacterial load could be enough to bring the bacterial density down to a level that can be controlled by the immune system or antibiotic treatment (Kirby *et al.*, 2014). Several studies have demonstrated that the use of a phage cocktail brings not only broad host range coverage but also prevents the development of phage resistant clones and failure of phage treatment (Alves *et al.*, 2016; Alves *et al.*, 2014; Gu *et al.*, 2012). Limited studies conducted on the effect of phage cocktails of *S. aureus* (Drilling *et al.*, 2014; Kelly *et al.*, 2012; Cerca *et al.*, 2007) and *P. aeruginosa* (Fong *et al.*, 2017; Pires *et al.*, 2011) showed a substantial reduction or elimination of biofilms produced by those species compared to PBS treated controls.

Following the treatment effect of phage cocktails, AB-SA01 and AB-PA01, on single species plate cultures and biofilms, the effect of these phage cocktails in mixed-species planktonic cultures and biofilm forms was investigated using fluorescently labelled laboratory strains and selected clinical isolates using bacterial cell count as a measurement of efficacy as discussed in chapter 4. Moreover, in chapter 4, the possibility of monitoring the effect of various phage

cocktails treatment effect and appearance of resistant mutants in real-time monitoring using red fluorescent protein labelled *S. aureus*, and green fluorescent protein labelled *P. aeruginosa* is demonstrated. Loss or significant decrease of mCherry or GFP fluorescence from the mixed-species bacterial populations of planktonic cultures was considered an indicator of the death of bacterial cells labelled with these fluorescence proteins. It is reported earlier that the decrease in fluorescence is an early and sensitive indicator of cell death (Steff *et al.*, 2001; Webb *et al.*, 2001). The decrease in fluorescence was correlated with a decrease in bacterial cell counts. Phage cocktails were as effective as gentamicin in killing bacteria in planktonic form, regardless of the antibiotic sensitivity profile of the bacterial hosts.

A high fluorescence signal in terms of relative fluorescence unit was observed in non-treated controls or bacterial species unaffected by the treatment. A decrease or increase in fluorescence intensity was associated with a decrease or increase in CFU, respectively. This observation was supported by reports that show the magnitude of fluorescence or absorbance, and colony count results are supplementary in showing bacterial density (Kong *et al.*, 2016; Russo *et al.*, 2015). The findings of experiments in chapter 4 show the feasibility of real-time monitoring of the efficacy of phage treatment using fluorescently labelled bacteria in mixed-species planktonic cultures. This assay combines rapidity and ease of management. Moreover, it is amenable to high-throughput screening setups for phage cocktail efficacy or antimicrobial agent evaluation.

Phage cocktail application on clinical isolates in mixed-species biofilms also resulted in bacterial cell reduction, suggesting that phage cocktails can control host cells in mixed-species biofilms.

The effect of phage cocktails in mixed-species biofilms followed similar patterns to those observed in single-species biofilm infections. The efficacy of phage cocktails in this study is consistent with the view that microheterogeneity of biofilms, with variable distribution of cells and presence of non-susceptible bacteria, does not prevent phages from reaching their hosts (Gutierrez *et al.*, 2015; Briandet *et al.*, 2008). This experiment demonstrates that phage cocktails can effectively kill target hosts within a mixed-species biofilm, in line with previous findings (Gonzalez *et al.*, 2017; Gutierrez *et al.*, 2015; Liao *et al.*, 2012; Sillankorva *et al.*, 2010; Briandet *et al.*, 2008; Harcombe and Bull, 2005).

In this study, it was seen that phages work more quickly on planktonic cultures than on biofilms, perhaps because bacterial cells in biofilms are in a lower metabolic activity state; hence phages cannot proliferate as efficiently and quickly compared to in free and actively growing cells in a planktonic state (Cerca *et al.*, 2007). Phage cocktails AB-SA01 and AB-PA01 are shown to be capable of killing their hosts in planktonic and biofilm phases in mixed-species cultures, as demonstrated in chapter 4. The findings in this chapter demonstrate that phage cocktails, alone or in combination, can kill their hosts in the presence of non-susceptible hosts in planktonic and biofilm phases. The data presented in this chapter strongly support the potential of the lytic phage cocktails in the treatment of antibiotic-resistant hosts in mixed-species infection both in planktonic and biofilm phases.

It is essential to corroborate the *in vitro* efficacy of phages using appropriate animal infection models. Studies indicated that *in vitro* efficacy of phages may not always correlate with the *in vivo* efficacy because the *in vivo* situation is more complicated as the physiological state of the 153

bacteria, including phage receptors expression, could be different (Tsonos *et al.*, 2014; Henry *et al.*, 2013). A mouse excisional wound model modified using silicone sheet splint to mimic the re-epithelization and granulation type of wound healing in humans is advised to evaluate the outcome of new treatment agents and wound healing *in vivo* (Davidson *et al.*, 2013; Dunn *et al.*, 2013).

Although rendering mice diabetic is challenging, the injection of STZ is reported to be effective and rapid in inducing type 2 diabetes mellitus (Deeds *et al.*, 2011). Following the established procedure, IP STZ injection was used as detailed in chapter 5 of this thesis. This experiment was successful in terms of producing a high proportion of diabetic mice, similar to the results of recent studies (Deeds *et al.*, 2011; Hayashi *et al.*, 2006). The excisional splint wound model employed in the present study facilitated re-epithelization and granulation wound healing similar to that in humans. Diabetic mice skin excisional wounds were infected with MDR *S. aureus* and treated with phage cocktail AB-SA01 topically in multiple doses at high phage titer (9.0 log₁₀ (PFU/ml)). Phage treatment studies on chicken skin (Goode *et al.*, 2003) and Balb/c mice burn wounds (Kumari *et al.*, 2010) show that low-titter phage treatment is unlikely to be successful. The mice tolerated phage cocktail AB-SA01 used in this study well since no adverse event or clinically manifested abnormality was observed in uninfected phage-treated wounds. In infected and treated wounds, adverse effects due to rapid bacterial lysis (Capparelli *et al.*, 2007; Dixon, 2004) were not observed as well.

Chapter 5 showed bacterial load reduction to the complete arrest of infection due to phage cocktail AB-SA01 treatment, and AB-SA01 efficacy was at least similar to vancomycin. Most of the phage cocktail treated wounds did not produce a detectable amount of *S. aureus* at the end of the experiment. The result of this multiple-dose scheme was similar to studies on *Salmonella* phages (Bardina *et al.*, 2012) and *E. coli* phages (Huff *et al.*, 2003) efficacy using chicken infection models that reported better results of repeated doses than the single dose. As opposed, a study that used only one phage on *ex vivo* human skin indicated that multiple doses do not improve the treatment outcome compared with a single dose treatment (Vieira *et al.*, 2012). In terms of wound healing, the topical administration of AB-SA01 produced a similar result to vancomycin IP treatment in hastening wound healing. This finding agrees with previous reports (Chhibber *et al.*, 2013; Mendes *et al.*, 2013). Most of the wounds from infected phage- and vancomycin-treated mice were completely closed, and hair regrowth was observed on some of the closed wounds. The wound size measurement scores also strongly associated with the decreased bacterial load.

Appropriate debridement is the standard procedure in wound management, especially before the application of topical antimicrobial (Nusbaum *et al.*, 2012), and phage treatment is not the exception (Mendes *et al.*, 2012a). Reports show that phage therapy will be better effective if applied after debridement (Jault *et al.*, 2019; Seth *et al.*, 2013; Mendes *et al.*, 2012a). In this study, the efficacy of topical phage cocktail treatment with gentle debridement was evaluated. The results on the efficacy of topical phage cocktail in diabetic mice wound infection in terms of bacterial load reduction and wound closure obtained in the current study represent the basis for 155

the initiation of human clinical trials to explore the potential of topical phage cocktail in the treatment of DFU and other chronic skin wound infections. The present wound infection treatment model: debridement, topical application of the phage cocktail, and coverage with a light dressing such as gauze and Opsite, may represent an appropriate blueprint for better chronic wound care.

6.2 Directions for future research

Wound healing outcomes depend on load and virulence of the infecting bacterial species, presence of antibiotic resistance factors including biofilms, immune status of the patient, and the efficacy of therapeutic agents. This study examined the lytic efficacy of phages on *in vitro* planktonic and biofilm phases and animal models using bacterial cell and biofilm biomass reduction as efficacy measurements. There are limited reports that phage-resistant mutants are less virulent (Hall *et al.*, 2012; Kutter, 2009; Capparelli *et al.*, 2007). A detailed investigation is needed to examine the phage resistance character and pathogenicity of surviving bacteria.

The present study did not evaluate the lytic effect of phage cocktails in multiple bacterial species infected wounds. This assessment is essential because most of the DFU infections are caused by more than one bacterial species (Mottola *et al.*, 2016; Ramakant *et al.*, 2011). The effect of phage therapy on healing of wounds infected by multiple bacterial species cannot be easily projected because of complex interactions among phages, infecting microbial species, and infected patient immune system.

This study also did not assess the effect of phage treatment on immune modulation and microflora of the skin of mice in the treatment area. A report on the effect of topical application of phages indicated that phage neutralization might not be a significant problem as the kinetics of phages are much rapid than antibodies production (Sulakvelidze *et al.*, 2001). Hence, research should be conducted to clarify these doubts. Examination of swab and skin samples collected throughout an experiment will be necessary for the proper assessment of the effect of phage 157

therapy on wound microbiome and examine the next dominant bacteria after the phage therapy. This microbiome assessment before and after phage treatment will help to examine the off-target effects of phages. Studies that assess the benefit of multispecies phages cocktails to reduce or eliminate multispecies biofilms should also be conducted.

Recent studies suggest that phages can evade bacterial host adaptive immunity, mainly CRISPR-Cas and restriction-modification systems, through protecting their genome using different proteins (Mendoza *et al.*, 2020; Malone *et al.*, 2020). These specific protein-based defence strategies should be investigated thoroughly. Moreover, a study indicated that phages could modulate human or animal immune responses through inducing anti- and pro-inflammatory responses (Van Belleghem *et al.*, 2017). A further in-depth study is recommended to examine the interaction of phages with the immune system and other possible side effects.

Studies on the cytotoxicity of phages are limited, particularly in wound infection models (Rose *et al.*, 2014; Mendes *et al.*, 2013). The findings of mice wound infection models in the present study show that phage therapy resulted in wound healing and significant bacterial load reduction. However, this study did not specifically evaluate the quality of wound healing. The clinical application of topical phage cocktail therapy may be more appropriate in the context of severe DFU infections that involve bones. So, future investigations of this practice on animal models of osteomyelitis are recommended. Formulations of phages for clinical application, including ointments or droplets, phage-impregnated wound dressings, and nanoparticle-based phage preparations, should be studied and developed.

In studies that involve laboratory animal infection models, the dose of phage cocktails needed to eliminate or reduce the infection to the level that can be controlled by the immune system or antibiotic treatment must be determined. The dose determination study should be conducted considering the "active" phage therapy principle that relies on self-amplifying and self-limiting character of phages to meet the necessary dose (Loc-Carrillo and Abedon, 2011).

In this study, the efficacy of a phage cocktail alone was evaluated against MDR *S. aureus* on the infected mice wound model. Studies coupling the use of phage cocktails with antibiotics on several clinical isolates should be future research focus. Collaboration research is required to take the findings of this study further to toxicological and clinical trials level.

The main goal of most of *in vitro* and pre-clinical experiments in antimicrobial therapy is to evaluate the efficacy of antimicrobial agents to pave the way to clinical trials. In line with this, the results of this study are important and need to be translated into further preclinical and clinical trials. Hence, the next phase of this study that will start within a few months to evaluate the application of the phage treatment using phage fixed on the wound dressing and gel form on animal models. Based on the results of the animal experiments, the next level study including clinical trial phase I/II will be recommended.

The treatment failure due to increasing resistance of bacterial infections to antibiotics forced scientists to search natural alternatives for treatment including phages. Phage libraries that identify and preserve mono- or cocktail-based phage therapy using the full spectrum of bacterial strains available in the clinical microbiology laboratories of a specific region or country is 159

needed. Hence, the shelf-ready phage strains or cocktails can reasonably cover antibioticresistant bacteria treatment in that region or country. Modern phage therapy has been suffering from insufficient credibility, patient and physician unfamiliarity and limited product availability. Currently, there is a limited supply of phages and phage cocktails as few phage companies or research institutions are working on phages. The results of this study are a good encouragement for phage industries. Therefore, collaborative efforts should be made among industries, governmental institutions, and charities to establish phage libraries and encourage phage commercialization. Governments and the World Health Organization should formulate appropriate regulations and intellectual property right that consider the unique characteristics of phages.

7. Appendices

The following publications and presentations have been produced from the chapters of this thesis.

Publications

Legesse Geredew Kifelew, James G. Mitchell & Peter Speck. Mini review: efficacy of lytic bacteriophages on multispecies biofilms. *Biofouling*, 2019, 1-10.

Legesse Garedew Kifelew, Morgyn S. Warner, Sandra Morales, Nicky Thomas, David L. Gordon, James G. Mitchell and Peter G. Speck. Efficacy of Lytic Phage Cocktails on *Staphylococcus aureus* and *Pseudomonas aeruginosa* in Mixed-Species Planktonic Cultures and Biofilms. *Viruses*, 2020, 12 (559), 1 - 16.

Legesse Garedew Kifelew, Morgyn S. Warner, Sandra Morales, Lewis Vaughan, Richard Woodman, Robert Fitridge, James G. Mitchel, and Peter Speck. Efficacy of Phage Cocktail AB-SA01 Therapy in Diabetic Mouse Wound Infections Caused by Multidrug-Resistant *Staphylococcus aureus*. BMC Microbiology (in press), 2020, 1 - 16.

Presentations

Legesse G. Kifelew, Morgyn Warner, Sandra Morales, Lewis Vaughan, James G. Mitchell, and Peter Speck. Efficacy of Phage AB-SA01 Therapy in Diabetic Mice Skin Wound Infections Caused by Multidrug Resistant *Staphylococcus aureus*. *Australasian Virology Society Conference*, Queenstown, New Zealand, 2 - 5 December 2019.

Legesse G. Kifelew, Morgyn Warner, Sandra Morales, Lewis Vaughan, James G. Mitchell, and Peter Speck. Efficacy of Phage AB-SA01 Therapy in Diabetic Mice Skin Wound Infections Caused by Multidrug Resistant *Staphylococcus aureus*. *Evergreen Phage Meeting*, Olympia, WA, USA, 4-9 August 2019, and *Diabetic Foot Australia Conference*, Brisbane, Queensland, AU, 8-10 September 2019.

Legesse G. Kifelew, Morgyn Warner, Sandra Morales, Nicky Thomas, David Gordon, Robert Fitridge, Peter Speck (2018). Bacteriophage efficacy against *Staphylococcus aureus* and *Pseudomonas aeruginosa ex vivo* isolates from diabetic foot ulcer (DFU) patients. *Oxford Bacteriophage Conference – Phages 2018*, Oxford, UK, 11-12 September 2018.


Check for updates

Mini-review: efficacy of lytic bacteriophages on multispecies biofilms

Legesse Geredew Kifelew*, James G. Mitchell and Peter Speck

Flinders University, College of Science and Engineering, Bedford Park, South Australia

ABSTRACT

There is potential for phages to prevent and control bacterial biofilms, but few studies have examined the effect of phages on the multispecies biofilms that characterize most bacterial infections. This paper reviews the mechanism of action of phages, the evidence supporting the view that phage therapy will be effective against bacterial targets and the opposite viewpoint, phage application approaches, and the comparative advantage of phage therapy in multispecies biofilms. The few reports measuring the actions of lytic phages against multispecies biofilms are also reviewed. The authors are cautiously optimistic about the application of phages against their targets when in multispecies biofilms because some lysis mechanisms do not require species specificity.

ARTICLE HISTORY Received 8 February 2019 Accepted 27 April 2019

KEYWORDS Bacteria; biofilm; efficacy; multispecies; phage; susceptible

Introduction

Human lifespan and quality of life are being compromised by the substantial rise in antimicrobial-resistant bacterial pathogens (Smith et al. 2015; Speck 2013). Unlike previous decades, there are few antimicrobials in development to counter the resistant microbes (WHO, 2014), perhaps in part because large pharmaceutical companies are ceasing to invest in this area (Projan 2003), despite microorganisms continuing to evolve and major challenges such as implant biofilms being unresolved (Kaur et al. 2016; Yilmaz et al. 2013). These microorganisms share a common milieu and coexist by forming surface-associated multispecies biofilms (Woods et al. 2012; Thein et al. 2007). Twothirds to three-quarters of human infections are thought to be biofilm-related (Seth et al. 2013). Moreover, most of these infections involve multispecies biofilms (Peters et al. 2012). Multispecies biofilms in a single community can provide benefits to the species involved, in the form of genetic material exchange, metabolic cooperation, quorum sensing system sharing, development of passive resistance, niche optimization, host immune modulation, virulence induction, and by-product influences that give the involved species competitive advantages (Shettigar et al. 2016; Tay et al. 2016; Wolcott et al. 2013). Bacterial biofilms are more resistant to biocides than planktonic cells due to the extracellular matrix that comprises most of the biofilm, and the bacterial cells in the biofilms that become metabolically inactive (Harper et al. 2014). This may complicate the infection and treatment outcome (Pastar et al. 2013; Dalton et al. 2011).

Bacteriophages ("phages", viruses that infect and may kill bacteria) are proposed as a means of ameliorating the antibiotic resistance crisis (Bhattacharjee et al. 2015; Yilmaz et al. 2013, Seth et al. 2013). Although the use of phages as antibacterial agents was started 100 years ago, humankind has not benefitted significantly from phage therapy to date. This is partly because of the introduction of antibiotics a half a century ago (Wei 2015; Golkar et al. 2014), scientific uncertainties about the co-evolutionary aspect of the phage-bacterium couplet (Pirnay et al. 2015; Høyland-Kroghsbo et al. 2013; Maura and Debarbieux 2012), poor understanding of what phages were or how they functioned (Storms and Sauvageau 2015; Pirnay et al. 2011; Kutter et al. 2010; Young 1992), scepticism toward efficacy of phage therapy (Young and Gill 2015; Henry et al. 2013; Krüger and Bickle 1983), absence of compatible regulatory frameworks for phage therapy (Verbeken et al. 2012), the difficulty of patenting phages (Kutter et al. 2015; Pirnay et al. 2012), and the lack of standardized criteria for phage preparation, purity, and potency (Merabishvili et al. 2009; Housby and Mann 2009; McVay et al. 2007; Marza et al. 2006; Wills et al. 2005).

High efficacy, bacterial specificity, quick onset of action, and low risk of resistance, if used in multi-

CONTACT Legesse Geredew Kifelew 🔊 legesse.kifelew@flinders.edu.au *Present address: St. Paul's Hospital Millennium Medical College, Addis Ababa, Ethiopia. © 2019 Informa UK Limited, trading as Taylor & Francis Group

2 🕢 L. G. KIFELEW ET AL.

phage cocktails, now make phage therapy a feasible option for use in humans (Pastagia et al. 2011). While lytic phages appear to be efficacious against bacteria within single-species biofilms, little is known about the lytic efficacy of phages when their target is contained within multispecies biofilms (Alves et al. 2016; Drilling et al. 2014; Carson et al. 2010). This paper will outline pertinent biofilm and phage characteristics, and the evidence for when phages will or will not maintain lytic efficacy in multispecies biofilms.

Bacterial biofilms

A bacterial biofilm is a group of bacteria attached to each other and usually to a surface (Allewell 2016; O'Toole et al. 2000; Skillman et al. 1998). The bacterial community in the biofilm is encased in an extracellular matrix; bacteria in this form are found ubiquitously in the environment (Teh et al. 2014; Kostakioti et al. 2013). This matrix is mainly composed of polysaccharides, proteins, lipids and DNA, and is often referred to as extracellular polymeric substance (EPS) (Teh et al. 2014; Karunakaran et al. 2011). Studies of EPS show that it reduces microbial mobility and limits the transfer of nutrients and oxygen (Steenackers et al. 2016). In addition, the matrix contributes to the development of an anaerobic environment and structural heterogeneity in biofilms (Lacroix-Gueu et al. 2005). Inside the matrix, the biofilm may consist of densely populated single or multiple species of bacteria (De Beer and Stoodley 2006). Scanning electron microscopy and related techniques reveal the micro-heterogeneity of biofilms, with variable distribution of cells, matrix, and fluid-filled channels and pores (Wood et al. 2000). Natural biofilms are a mixture of micro-organisms sharing a common milieu and coexisting in niches by forming multispecies biofilms (Thein et al. 2007). Natural biofilms are usually polymicrobial, and bacteria occupy 5%-30% of the volume of the biofilm (Zhao et al. 2013).

Of all human infections, 65–80% are biofilmrelated, and biofilm formation is often considered the underlying reason for antibiotic treatment failure (Coenye and Nelis 2010). Bacteria residing in biofilms are more tolerant to physical and chemical disruptions than their planktonic counterparts (Steenackers et al. 2016; Gilbert et al. 2002). Studies have also shown that bacteria in biofilms are 100-1000 times more resistant to antibiotics than in the planktonic phase (Burmolle et al. 2006; Ceri et al. 1999). Bacterial resistance to antibiotics in biofilms is mainly due to the complex nature of the EPS that give mechanical stability to the biofilm and serves as an external digestive system that keeps extracellular enzymes near the bacteria (Flemming and Wingender 2010). The extracellular matrix also protects bacteria since it reduces the access of solutes to the bacteria through the combination of ionic interaction and molecular sieving (Allison 2003). The presence of metabolically inactive persister cells also contributes to the antibiotic resistance of bacteria in biofilms (Harper et al. 2014).

Phages and their mechanism of action in biofilms

The mechanism by which phage spreads through biofilms and how they kill their hosts appears to be that lytic phages encode or stimulate their hosts to produce EPS-degrading enzymes, which facilitate their movement through biofilms (Sutherland et al. 2004). Then, phages move through biofilms, proliferate within their host bacteria, and finally eliminate their hosts via lytic activity (Zhang et al. 2014; Gilbert et al. 2002) as illustrated in Figure 1. Some phages carry genes to encode production of enzymes by the host bacteria that degrade bacterial capsules and other EPS; for example *Bacillus subtilis* phage Φ NIT1 encodes a monomeric 25-kDa degradation enzyme (designated y-PGA hydrolase, PghP) to break down poly- γ -glutamate (γ -PGA) - a host capsular polypeptide of glutamate with a γ -linkage, so that phage progenies easily infect encapsulated cells (Kimura and Itoh 2003). However, phage adsorption and proliferation does depend on the growth phase of the bacterial host. Phage entry and multiplication will be enhanced when phages are applied during an exponential growth phase of the bacteria. But dormant and dead host cells will make it difficult for the phages to migrate through and multiply in the biofilm that may lead to abortive infection. This difficulty will be exacerbated if these dead and dormant host cells are present in large numbers (Hu et al. 2012).

In multispecies biofilms, various structures of biofilm including EPS and heterologous microbial cells may impede phage access to the host cell surface (Sutherland 1999; Sutherland and Wilkinson 1965). The physiological state of the host cells such as stationary and declining growth phases, strong co-aggregation with other bacteria, tight cell-cell binding, and unavailability of phage receptors on their surfaces will affect the outcome of phage attack (Rickard et al. 2003). Some phages possess an enzyme, produced in the bacterial host due to their encoding effect, in their



Figure 1. Mechanism by which phages disrupt biofilms (taken from Gutierrez et al. 2016).

spikes (Pelkonen et al. 1992; Kwiatkowski et al. 1982; Rieger-Hug and Stirm 1981; Saxelin et al. 1979; Bessler et al. 1975). Once the phage and biofilm make contact, the biofilm matrix is thought to be degraded by these phage-associated hydrolytic enzymes termed polysaccharide depolymerases (Yan et al. 2014; Carson et al. 2010; Kimura and Itoh 2003). Most double-stranded phages encode hydrolytic enzymes and lyse the host bacteria at the last step of the lytic infection cycle (Gutierrez et al. 2016; Hughes et al. 1998). Phages are released as they achieve local lysis of susceptible cells and as associated enzymes degrade EPS within the biofilm. Some phage-associated depolymerases are highly specific for host-derived EPS and may have limited effects on multispecies biofilms (Chan and Abedon 2015; Nelson et al. 2001). There are also some capable of degrading the EPS of many genera (Deveau et al. 2002; Sutherland et al. 2004; Yoong et al. 2004; Pei and Lamas-Samanamud 2014).

Phages with depolymerase enzyme can readily reach the host cell surface by digesting their way as described by earlier studies (Gutierrez et al. 2016; Yan et al. 2014; Sutherland et al. 2004; Hughes et al. 1998). Phages penetrate the bacterial cell wall using their enzyme or by contracting a sheath and injecting their genome into the cytoplasm of the host bacterium (Figure 1, step 2). Inside the bacterium, the phage replicates its genome and synthesizes its enzymes and structural components using the host replication and metabolic pieces of machinery, respectively. Phage structures synthesis and replication can be achieved through the expression of phage early genes to regulate these machineries (Figure 1, step 3). By the expression of the phage late genes, assembly of the phage structures, packaging of the genome, and maturation will be carried out to produce phage particles (Figure 1, step 4) (Gutierrez et al. 2016). Finally, phage coded lysozyme breaks down the peptidoglycan layer of the host bacterium causing lysis and release of the new phage progeny ready to infect other similar bacterial cells (Figure 1, step 5) and remove the biofilm through the action of polysaccharide depolymerases (Gutierrez et al. 2016; Harper et al. 2014; Hughes et al. 1998).

Evidence of lytic efficacy of phages in multispecies biofilms

Although many studies have examined the efficacy of phages on planktonic bacteria and single-species biofilms, few have examined the efficacy of phages against multispecies biofilms. One such study (Kay et al. 2011) assessed the effects of λ W60 (*Escherichia coli* phage) and PB-1 (*Pseudomonas aeruginosa* phage) on mono- and dual-species planktonic and biofilm cultures of *E. coli* and *P. aeruginosa*. This study found that phages achieved a higher titre and depleted their bacterial host population more effectively within multispecies biofilms compared to planktonic or single species biofilm treatment. This could be evidence that shows the efficacy of phage against bacteria in the multispecies biofilms (Kay et al. 2011).

The importance of phage dependent depolymerase to degrade bacterial EPS has been known for over 60 years (Sutherland and Wilkinson 1965). Reports in the late 1990s indicated that the disruption of a biofilm by phages is a combination of EPS degradation by the depolymerases and infection and subsequent cell lysis by the phages (Hughes et al. 1998). Some phages can induce their host to produce and release

4 🕳 L. G. KIFELEW ET AL.

depolymerases (Yan et al. 2014; Hanlon et al. 2001; Hughes et al. 1998; Sutherland and Wilkinson 1965; Adams and Park 1956). Some phages can also contain depolymerases by their particles or spikes (Pei and Lamas-Samanamud 2014; Son et al. 2010; Sutherland et al. 2004; Pelkonen et al. 1992; Kwiatkowski et al. 1983). These reports showed that depolymeraseloaded or -inducing phages can more strongly disrupt multispecies biofilms through enzymatic degradation of bacterial EPS than non-depolymerase-possessing or non-depolymerase-inducing phages. This appears to be because enzymes released from the surface of the phages or the bacterial host cells during lysis can rapidly break down biofilms (Pei and Lamas-Samanamud 2014). Thus, phages that contain or induce depolymerase production by their host could access the host bacteria more readily in the biofilms than phages without depolymerases (Harper et al. 2014).

A study finding on a phage cocktail comprising ØIBB-PF7A (a phage against Pseudomonas fluores-ØIBB-SL58B (a phage cens) and against Staphylococcus lentus) on mono- and dual-species biofilms indicated that the phage cocktail effectively removed both bacterial hosts from all strata of the biofilm (Sillankorva et al. 2010). The finding also showed that ØIBB-SL58B and ØIBB-PF7A were able to multiply in the dual-species biofilms; ØIBB-PF7A reached its host when the host population was present in the minority and the majority in the dual-species biofilms. Moreover, the finding demonstrated that the phage cocktail caused a significant decrease in the number of bacteria released from the biofilms into the planktonic phase, and phage ØIBB-PF7A produced significant damage to the dual-species biofilm. It also indicated that the damage to the biofilm due to ØIBB-PF7A facilitated the release of non-susceptible S. lentus to the environment (Sillankorva et al. 2010). Non-susceptible bacteria and their extracellular matrix do not prevent phages from reaching their hosts (Briandet et al. 2008). A similar study that evaluated the efficacy of 5 phages on dual-species biofilms of Enterobacter cloacae NCTC 5920 and Enterobacter agglomerans also demonstrated that application of a cocktail of phages against each bacterial species could eliminate both bacterial hosts from dual-species biofilms (Tait et al. 2002).

In an investigation using two-photon excitation *in* situ fluorescence correlation spectroscopy, the activity of lactococcal c2 phage against biofilms formed by c2non-susceptible *Stenotrophomonas maltophilia* strain 114N-Sm, c2-susceptible *Lactococcus lactis* subsp. cremoris OSM31, and c2-resistant *Lactococcus lactis* subsp. *lactis* IL-1403 was examined. The lactococcal c2 phages penetrated and diffused through multispecies biofilms without being affected by EPS or non-susceptible hosts. In addition, the authors explained that the microheterogeneity of biofilms with a variable distribution of cells, EPS, and water-filled channels did not affect phage access into the interior of the biofilms (Briandet et al. 2008). A similar study conducted using *P. aeruginosa* phage Φ E2005-A on dual-species biofilm formed by susceptible *P. aeruginosa* and non-susceptible *E. coli* found that application of the phage Φ E2005-A on *P. aeruginosa* in the presence of non-susceptible *E. coli* species (Liao et al. 2012).

The lytic efficacy of S. lentus phage fIBB-SL58B and P. fluorescens phage fIBB-PF7A against their hosts in the dual-species biofilm that consists of P. fluorescens and S. lentus also support the research findings that indicated phages can effectively remove their hosts in the biofilms formed by their hosts and other bacteria (Sillankorva et al. 2010). Evaluation of E. coli-specific phages (T7, T5 or both) for their effect against E. coli on E. coli-Salmonella Typhimurium dual-species biofilms showed that these phages successfully reduced the E. coli population. The evaluation result also indicated that application of S. Typhimurium phage SP6 on E. coli-S. Typhimurium dual-species biofilms led to a decreased S. Typhimurium population in the presence or absence of E. coli. Consequently, it concluded that the capability of phages to reach their hosts could not be compromised by the presence or absence of nonsusceptible bacteria in multispecies biofilm (Harcombe and Bull 2005). Gutiérrez et al. (2015) also reported at least a 2-log unit reduction in the population of adhered Staphylococcus aureus by phiIPLA-RODI phage and Staphylococcus epidermidis by phiIPLA-C1C phage from S. aureus-S. epidermis dual-species biofilms. The authors suggested that this finding complemented previous research that documented the efficacy of phages in removing their hosts from multispecies biofilms and concluded that lytic phages are effective biofilm disrupting agents, including for multispecies biofilms in the presence of nonsusceptible hosts (Gutierrez et al. 2015).

The efficacy of *S. aureus* phage phiIPLA-RODI on biofilms formed by *S. aureus* in combination with *Lactobacillus plantarum*, *Enterococcus faecium*, or *Lactobacillus pentosus* was reported and the level of effectiveness of phage treatment on multispecies biofilms varies depending on the bacterial species accompanying the host bacteria (González et al. 2016). The propagation of phage phiIPLA-RODI was higher in biofilms formed by *S. aureus-L. pentosus* than in biofilms formed by *S. aureus* with the remaining two bacterial species. Scanning electron and confocal laser scanning microscopy also showed that phiIPLA-RODI treatment made the biofilms flatter and less organized and decreased intact *S. aureus* cells compared with untreated samples.

The search for effective use of phages in the treatment of biofilms continues to advance. Use of quorum-quenching phages that can lyse host bacteria and express quorum-quenching enzymes to affect diverse bacteria in multispecies biofilms could be one potential area to make the best out of phage therapy (Pei and Lamas-Samanamud 2014). Pei and Lamas-Samanamud (2014) reported the lytic efficacy of engineered phage T7 which was encoded with the acyl homoserine lactones (AHL) lactonase AiiA gene to produce lactonase. They further detailed in the report that phages encoding lactonase have the capacity of quenching quorum sensing through degrading AHLs that are necessary for biofilm formation by facilitating bacterial cells communication. The report was the result of a study performed on a mixed biofilm of E. coli and P. aeruginosa after proving the existence of synergism in biofilm formation between the two species and knowing that E. coli (host for the phage) cannot synthesize AHLs but can respond to AHL released by other bacteria and P. aeruginosa can synthesize and respond to AHLs. According to the report, the addition of lactonase producing phage T7 to multispecies biofilms containing P. aeruginosa and E. coli can inhibit biofilm formation and disrupt biofilms by affecting both bacterial species. It further suggested that phages that lyse host bacteria and express quorum-quenching enzymes to affect diverse bacteria in multispecies biofilm communities will be a potential alternative treatment against multispecies biofilm infections in the future.

Polyvalent phages (broad-host-range phages) represent a potentially effective strategy against multispecies biofilms (Kim et al. 2012; Lin et al. 2012; Sillankorva et al. 2010; O'Flaherty et al. 2005; Jensen et al. 1998). For instance, the polyvalent PA1Ø phage is an efficient lysing agent for mixed-biofilm of P. aeruginosa, S. aureus, S. epidermidis, and Staphylococcus hominis. Consequently, PA1Ø is a useful alternative antimicrobial agent in the treatment of mixed biofilm-related infections, particularly of P. aeruginosa and S. aureus (Kim et al. 2012). These reports conclude that phages can control their hosts in most circumstances (Sillankorva et al. 2010). The

capacity of phages to infect multiple bacterial host species can maximize opportunities for effective phage proliferation in multispecies biofilms since phages that are capable of interspecies or intergeneric infection will have increased chances to encounter suitable prey and replicate (Kim et al. 2012; Jensen et al. 1998). The ability of lytic phages to degrade multispecies biofilms, replicate within their host, and lyse the host even when applied at lower doses is documented in recent studies (Gutierrez et al. 2015; Chhibber et al. 2015; Sillankorva et al. 2010).

Findings of reduced lytic efficacy of phages in multispecies biofilms

Studies showing the reduced lytic efficacy of phages in multispecies biofilms are fewer in numbers compared to the number of studies showing good lytic efficacy of phages in multispecies biofilms. One possible reason for the reduced efficacy of phages in multispecies biofilms could be the limitation of phage particle migration by the presence of the EPS (Hughes et al. 1998). The diversity of EPS and the heterogeneous distribution of biofilm-forming bacteria and their EPS throughout a biofilm would hinder the action of phage depolymerase in multispecies biofilms (Neu and Lawrence 1997). This may be because some of the phage depolymerases are highly specific to targetting EPS chemistry (Hanlon et al. 2001). An experiment conducted on the activity of PB-1 and λ W60 phages on dual-species biofilm formed by P. aeruginosa strain PAO1 and E. coli strain MG1655 using transmission electron microscopy also showed that phages can be trapped in the EPS component of multispecies biofilms. The result of the experiment indicate that infectivity of PB-1 and $\lambda W60$ in multispecies biofilm was similar to that seen in planktonic cells when applied with EPS dissolving 0.5% Tween-20 but weaker when these phages are applied on multispecies biofilm in the absence of 0.5% Tween-20 (Kay et al. 2011).

Multispecies biofilm formed by *Microbacterium phyllosphaerae*, *Shewanella japonica*, *Dokdonia dong*-*haensis*, and *Acinetobacter lwoffii* showed an increase in biomass of >167% compared to single species biofilm biomass due to the synergetic interaction of participating bacterial species. Furthermore, multispecies biofilms formed by these bacteria displayed higher resistance to other invading bacteria and antibacterial proteins than biofilms formed by any of these bacterial species alone (Burmolle et al. 2006). The resistance observed in multispecies biofilms against bactericidal

6 🍙 L. G. KIFELEW ET AL.

proteins may hold true for phage activity against multispecies biofilms since the large biofilm biomass or the complex EPS may hinder phage penetration of the biofilm or entrap the phages (Kay et al. 2011) or it may deter the activity of phage depolymerase (Hughes et al. 1998). The complex EPS may reduce the efficacy of phages because of entrapment of phage particles in the biofilm matrix (Kay et al. 2011), reduce multiplication of phages due to the large proportion of metabolically inactive host cells (Harper et al. 2014), or shedding of phage receptors from the host bacteria (Teplitski and Ritchie 2009).

The reduced lytic efficacy of phages on multispecies biofilms was also reported by Tait et al. (2002) who evaluated the efficacy of 5 phages of E. cloacae NCTC 5920 and E. agglomerans on single- and dualspecies biofilms. These authors reported that E. cloacae NCTC 5920 could be removed by only 1 of the 5 phages; whereas E. agglomerans could be removed by 4 of the 5 phages used during the treatment of the dual-species biofilms. Hence, not all phages remove a susceptible bacterial host from multispecies biofilms. Since the complete elimination of E. cloacae NCTC 5920 was not achieved during the study, structural heterogeneity of multispecies biofilms through producing pockets of unattainable phage-susceptible bacteria was suggested as a possible reason. Based on these findings, it possible to conclude that the presence of non-susceptible bacterial populations could protect phage-susceptible bacteria from phage attack through creating 'spatial refugees' restricted in space and protected, within the biofilm (Tait et al. 2002). The common phage resistance mechanisms of bacteria including preventing phage adsorption, blocking entry or cutting phage nucleic acids, and abortive infection systems, can be used by the bacteria persisting inside the multispecies biofilms (Labrie et al. 2010).

Approaches to improve phage efficacy against multispecies biofilms

While it is broadly accepted that phages typically target a single bacterial species, this view might be biased by the methods used to discover phages, which usually involve the appearance of plaques on lawns of purified cultures of bacteria. A report describing phages isolated from Lake Michigan, USA by Malki et al. (2015), with a host range spanning several phyla of bacteria, shows that individual phage species may have a much broader target range than had been considered possible. The use of phages with a broad host range that can attack various bacterial species in multispecies biofilm would help in targeting such biofilms. It is possible to optimize the benefit of phages in controlling multispecies biofilms through facilitating their penetration. Li et al. (2017) found that use of polyvalent phages conjugated with magnetic colloidal nanoparticle clusters, in the presence of a magnetic field, achieved an enhanced penetration into biofilms.

Phages could be engineered for the extension of their host range in many ways, including homologous recombination of phages with plasmids that can alter phage tail fibre proteins (Lin et al. 2012) and successive passage of phages through the less sensitive target hosts (O'Flaherty et al. 2005). Genetically engineered phages could potentially be developed with enhanced capabilities for extracellular matrix dispersion, biofilm dissolution, and obstruction of bacteria-bacteria cell communication (Pei and Lamas-Samanamud 2014; Lu and Collins 2007). Pei and Lamas-Samanamud (2014) engineered T4 phage to encode a lactonase (a metalloenzyme, produced by certain species of bacteria and which inactivates AHLs) with broad-range activity for quenching of quorum sensing. Their work showed that genetic modifications of phages could enable phages to synthesise quorum quenching enzymes for higher efficacy against multispecies biofilms (Pei and Lamas-Samanamud 2014).

Phage cocktails have also been found to exploit the synergistic effect of component phages which could facilitate their adsorption and diffusion in multispecies biofilms. For instance, depolymerase produced by one phage can hydrolyse the EPS structure of the multispecies biofilm contributed from its host and closely related species. This may enhance the ability of the other phage to reach its host (Chhibber et al. 2015; Sillankorva et al. 2010). More effective removal of P. aeruginosa and Klebsiella pneumoniae from the mixed-species biofilm of the two bacterial species was observed when treated with K. pneumoniae-specific phage KPO1K2 and P. aeruginosa-specific phage Pa29 cocktail than treated by the respective phages alone. This could be attributed to the depolymerase coded by phage KPO1K2 that hydrolyzed the top layer of K. pneumoniae and P. aeruginosa from the multispecies biofilm and facilitated the entry of the two phages to reach the bacterial cell part of the biofilm (Chhibber et al. 2015). Similarly, the exposure of dual-species biofilms formed by P. fluorescens and S. lentus to a phage cocktail containing P. fluorescens phage fIBB-PF7A and S. lentus phage SL58 effectively removed the biofilm and killed their hosts (Sillankorva et al. 2010).

The clearance of multispecies biofilms through the application of phages is feasible but remains a challenging proposition. The available options to ensure that phages maintain lytic efficacy against multispecies biofilms include the formulation of phage cocktails that include phages against multiple species, cultivation of phages with broad host range, an extension of host range of phages, application of phages in combination with EPS-degrading enzymes, genetic modification of phages, or a combination of these strategies. By implementing these strategies, phage therapy could become a new alternative for the prevention and control of drug-resistant multispecies biofilms.

Disclosure statement

No potential conflict of interest was reported by the authors.

References

- Adams MH & Park BH. 1956. An enzyme produced by a phage-host cell system. II. The properties of the polysaccharide depolymerase. Virology. 2:719–736.
- Allewell NM. 2016. Introduction to biofilms. Thematic Minireview Series. J Biol Chem. 291:12527–12528. doi: 10.1074/jbc.R116.734103
- Allison DG. 2003. The biofilm matrix. Biofouling. 19: 139–150. doi:10.1080/0892701031000072190
- Alves DR, Perez-Esteban P, Kot W, Bean JE, Arnot T, Hansen LH, Enright MC & Jenkins AT. 2016. A novel bacteriophage cocktail reduces and disperses *Pseudomonas aeruginosa* biofilms under static and flow conditions. Microb Biotechnol. 9:61–74. doi:10.1111/ 1751-7915.12316
- Bessler W, Fehmel F, Freund-Molbert E, Knufermann H & Stirm S. 1975. *Escherichia coli* capsule bacteriophages. IV. Free capsule depolymerase 29. J Virol. 15:976–984.
- Bhattacharjee AS, Choi J, Motlagh AM, Mukherji ST & Goel R. 2015. Bacteriophage therapy for membrane biofouling in membrane bioreactors and antibiotic-resistant bacterial biofilms. Biotechnol Bioeng. 112:1644–1654. doi:10.1002/bit.25574
- Briandet R, Lacroix-Gueu P, Renault M, Lecart S, Meylheuc T, Bidnenko E, Steenkeste K, Bellon-Fontaine M.-N. & Fontaine-Aupart M.-P. 2008. Fluorescence correlation spectroscopy to study diffusion and reaction of bacteriophages inside biofilms. Appl Environ Microbiol. 74: 2135–2143. doi:10.1128/AEM.02304-07
- Burmolle M, Webb JS, Rao D, Hansen LH, Sorensen SJ & Kjelleberg S. 2006. Enhanced biofilm formation and increased resistance to antimicrobial agents and bacterial invasion are caused by synergistic interactions in multispecies biofilms. Appl Environ Microbiol. 72:3916–3923. doi:10.1128/AEM.03022-05

- Carson L, Gorman SP & Gilmore BF. 2010. The use of lytic bacteriophages in the prevention and eradication of biofilms of *Proteus mirabilis* and *Escherichia coli*. FEMS Immunol Med Microbiol. 59:447–455. doi:10.1111/j.1574-695X.2010.00696.x
- Ceri H, Olson M, Stremick C, Read R, Morck D & Buret A. 1999. The Calgary Biofilm Device: new technology for rapid determination of antibiotic susceptibilities of bacterial biofilms. J Clin Microbiol. 37:1771–1776.
- Chan BK & Abedon ST. 2015. Bacteriophages and their enzymes in biofilm control. Curr Pharm Des. 21:85–99.
- Chhibber S, Bansal S & Kaur S. 2015. Disrupting the mixed-species biofilm of *Klebsiella pneumoniae* B5055 and *Pseudomonas aeruginosa* PAO using bacteriophages alone or in combination with xylitol. Microbiology. 161: 1369–1377. doi:10.1099/mic.0.000104
- Coenye T & Nelis HJ. 2010. In vitro and in vivo model systems to study microbial biofilm formation. J Microbiol Methods. 83:89–105. doi:10.1016/j.mimet.2010.08.018
- Dalton T, Dowd SE, Wolcott RD, Sun Y, Watters C, Griswold JA & Rumbaugh KP. 2011. An in vivo polymicrobial biofilm wound infection model to study interspecies interactions. PloS one. 6:e27317. doi:10.1371/ journal.pone.0027317
- de Beer D & Stoodley P. 2006. Microbial biofilms. The prokaryotes. Springer.
- Deveau H, van Calsteren M-R & Moineau S. 2002. Effect of exopolysaccharides on phage-host interactions in *Lactococcus lactis*. Appl Environ Microbiol. 68: 4364–4369. doi:10.1128/AEM.68.9.4364-4369.2002
- Drilling A, Morales S, Jardeleza C, Vreugde S, Speck P & Wormald P-J. 2014. Bacteriophage reduces biofilm of *Staphylococcus aureus* ex vivo isolates from chronic rhinosinusitis patients. Am J Rhinol Allergy. 28:3–11. doi: 10.2500/ajra.2014.28.4001
- Flemming H-C. & Wingender J. 2010. The biofilm matrix. Nat Rev Microbiol. 8:623–633. doi:10.1038/nrmicro2415
- Gilbert P, Allison D & Mcbain A. 2002. Biofilms in vitro and in vivo: do singular mechanisms imply cross-resistance? J Appl Microbiol. 92.
- Golkar Z, Bagasra O & Pace D. 2014. Bacteriophage therapy: a potential solution for the antibiotic resistance crisis. J Infect Dev Ctries. 8:129–136.
- González S, Fernández L, Campelo A, Gutiérrez D, Martínez B, Rodríguez A & García P. 2016. The behavior of *Staphylococcus aureus* dual-species biofilms treated with bacteriophage phiIPLA-RODI depends on the accompanying microorganism. Appl Environ Microbiol. AEM. 02821-16.
- Gutierrez D, Rodriguez-Rubio L, Martinez B, Rodriguez A & Garcia P. 2016. Bacteriophages as weapons against bacterial biofilms in the food industry. Front Microbiol. 7:825.
- Gutierrez D, Vandenheuvel D, Martinez B, Rodriguez A, Lavigne R & Garcia P. 2015. Two phages, phiIPLA-RODI and phiIPLA-C1C, lyse mono- and dual-species staphylococcal biofilms. Appl Environ Microbiol. 81: 3336–3348. doi:10.1128/AEM.03560-14
- Hanlon G, Denyer S, Olliff C & Ibrahim L. 2001. Reduction in exopolysaccharide viscosity as an aid to bacteriophage penetration through *Pseudomonas*

8 🕳 L. G. KIFELEW ET AL.

aeruginosa biofilms. Appl Environ Microbiol. 67: 2746–2753. doi:10.1128/AEM.67.6.2746-2753.2001

- Harcombe W & Bull J. 2005. Impact of phages on two-species bacterial communities. Appl Environ Microbiol. 71: 5254–5259. doi:10.1128/AEM.71.9.5254-5259.2005
- Harper D, Parracho H, Walker J, Sharp R, Hughes G, Werthén M, Lehman S & Morales S. 2014. Bacteriophages and biofilms. Antibiotics. 3:270–284. doi: 10.3390/antibiotics3030270
- Henry M, Lavigne R & Debarbieux L. 2013. Predicting in vivo efficacy of therapeutic bacteriophages used to treat pulmonary infections. Antimicrob Agents Chemother. 57:5961–5968. doi:10.1128/AAC.01596-13
- Housby J & Mann N. 2009. Phage therapy. Drug Discov Today. 14:536–540. doi:10.1016/j.drudis.2009.03.006
- Høyland-Kroghsbo N, Mærkedahl R & Svenningsen S. 2013. A quorum-sensing-induced bacteriophage defense mechanism. MBio. 4:e00362–12.
- Hu J, Miyanaga K & Tanji Y. 2012. Diffusion of bacteriophages through artificial biofilm models. Biotechnol Prog. 28:319–26. doi:10.1002/btpr.742
- Hughes K, Sutherland I & Jones M. 1998. Biofilm susceptibility to bacteriophage attack: the role of phage-borne polysaccharide depolymerase. Microbiology. 144: 3039–3047. doi:10.1099/00221287-144-11-3039
- Jensen E, Schrader H, Rieland B, Thompson T, Lee K, Nickerson K & Kokjohn T. 1998. Prevalence of broadhost-range lytic bacteriophages of *Sphaerotilus natans*, *Escherichia coli*, and *Pseudomonas aeruginosa*. Appl Environ Microbiol. 64:575–580.
- Karunakaran E, Mukherjee J, Ramalingam B & Biggs C. 2011. "Biofilmology": a multidisciplinary review of the study of microbial biofilms. Appl Microbiol Biotechnol. 90:1869–1881. doi:10.1007/s00253-011-3293-4
- Kaur S, Harjai K & Chhibber S. 2016. In vivo assessment of phage and linezolid based implant coatings for treatment of methicillin resistant *S. aureus* (MRSA) mediated orthopaedic device related infections. PloS one. 11: e0157626. doi:10.1371/journal.pone.0157626
- Kay M, Erwin T, Mclean R & Aron G. 2011. Bacteriophage ecology in *Escherichia coli* and *Pseudomonas aeruginosa* mixed-biofilm communities. Appl Environ Microbiol. 77: 821–829. doi:10.1128/AEM.01797-10
- Kim S, Rahman M, Seol S, Yoon S & Kim J. 2012. *Pseudomonas aeruginosa* bacteriophage PA1O requires type IV pili for infection and shows broad bactericidal and biofilm removal activities. Appl Environ Microbiol. 78:6380–6385. doi:10.1128/AEM.00648-12
- Kimura K & Itoh Y. 2003. Characterization of poly-γ-glutamate hydrolase encoded by a bacteriophage genome: possible role in phage infection of *Bacillus subtilis* encapsulated with poly-γ-glutamate. Appl Environ Microbiol. 69:2491–2497. doi:10.1128/AEM.69.5.2491-2497.2003
- Kostakioti M, Hadjifrangiskou M & Hultgren S. 2013. Bacterial biofilms: development, dispersal, and therapeutic strategies in the dawn of the postantibiotic era. Cold Spring Harb Perspect Med. 3:a010306.
- Krüger D & Bickle T. 1983. Bacteriophage survival: multiple mechanisms for avoiding the deoxyribonucleic acid restriction systems of their hosts. Microbiol Rev. 47:345.
- Kutter E, de Vos D, Gvasalia G, Alavidze Z, Gogokhia L, Kuhl S & Abedon S. 2010. Phage therapy in clinical

practice: treatment of human infections. Curr Pharmaceut Biotehnol. 11:69–86. doi:10.2174/ 138920110790725401

- Kutter E, Kuhl S & Abedon S. 2015. Re-establishing a place for phage therapy in western medicine. Fut Microbiol. 10:685–688. doi:10.2217/fmb.15.28
- Kwiatkowski B, Boschek B, Thiele H & Stirm S. 1982. Endo-N-acetylneuraminidase associated with bacteriophage particles. J Virol. 43:697–704.
- Kwiatkowski B, Boschek B, Thiele H & Stirm S. 1983. Substrate specificity of two bacteriophage-associated endo-N-acetylneuraminidases. J Virol. 45:367–374.
- Labrie S, Samson J & Moineau S. 2010. Bacteriophage resistance mechanisms. Nat Rev Microbiol. 8:317–327. doi:10.1038/nrmicro2315
- Lacroix-Gueu P, Briandet R, Leveque-Fort S, Bellon-Fontaine M & Fontaine-Aupart M. 2005. In situ measurements of viral particles diffusion inside mucoid biofilms. C R Biol. 328:1065–1072. doi:10.1016/ j.crvi.2005.09.010
- Li L.-L., Yu P, Wang X, Yu S.-S., Mathieu J, Yu H.-Q. & Alvarez P. J. J. E. S. N. 2017. Enhanced biofilm penetration for microbial control by polyvalent phages conjugated with magnetic colloidal nanoparticle clusters (CNCs). 4:1817–1826.
- Liao K, Lehman S, Tweardy D, Donlan R & Trautner B. 2012. Bacteriophages are synergistic with bacterial interference for the prevention of *Pseudomonas aeruginosa* biofilm formation on urinary catheters. J Appl Microbiol. 113:1530–1539. doi:10.1111/j.1365-2672.2012.05432.x
- Lin T, Lo Y, Tseng P, Chang S, Lin Y & Chen T. 2012. A T3 and T7 recombinant phage acquires efficient adsorption and a broader host range. PLoS One. 7:e30954. doi: 10.1371/journal.pone.0030954
- Lu T & Collins J. 2007. Dispersing biofilms with engineered enzymatic bacteriophage. Proc Natl Acad Sci USA. 104: 11197–202. doi:10.1073/pnas.0704624104
- Malki K, Kula A, Bruder K, Sible E, Hatzopoulos T, Steidel S, Watkins S & Putonti C. 2015. Bacteriophages isolated from Lake Michigan demonstrate broad host-range across several bacterial phyla. Virol J. 12.
- Marza J, Soothill J, Boydell P & Collyns T. 2006. Multiplication of therapeutically administered bacteriophages in *Pseudomonas aeruginosa* infected patients. Burns. 32:644–646. doi:10.1016/j.burns.2006.02.012
- Maura D & Debarbieux L. 2012. On the interactions between virulent bacteriophages and bacteria in the gut. Bacteriophage. 2:229–233.
- Mcvay C, Velásquez M & Fralick J. 2007. Phage therapy of *Pseudomonas aeruginosa* infection in a mouse burn wound model. Antimicrob Agents Chemother. 51: 1934–1938. doi:10.1128/AAC.01028-06
- Merabishvili M, Pirnay J, Verbeken G, Chanishvili N, Tediashvili M, Lashkhi N, Glonti T, Krylov V, Mast J, van Parys L et al. 2009. Quality-controlled small-scale production of a well-defined bacteriophage cocktail for use in human clinical trials. PLoS One. 4:e4944. doi: 10.1371/journal.pone.0004944
- Nelson D, Loomis L & Fischetti V. 2001. Prevention and elimination of upper respiratory colonization of mice by group A streptococci by using a bacteriophage lytic

enzyme. Proc Natl Acad Sci USA. 98:4107-4112. doi: 10.1073/pnas.061038398

- Neu T & Lawrence J. 1997. Development and structure of microbial biofilms in river water studied by confocal laser scanning microscopy. FEMS Microbiol Ecol. 24: 11–25. doi:10.1111/j.1574-6941.1997.tb00419.x
- O'Flaherty S, Ross R, Meaney W, Fitzgerald G, Elbreki M & Coffey A. 2005. Potential of the polyvalent anti-*Staphylococcus* bacteriophage K for control of antibioticresistant staphylococci from hospitals. Appl Environ Microbiol. 71:1836–1842. doi:10.1128/AEM.71.4.1836-1842.2005
- O'Toole G, Kaplan H & Kolter R. 2000. Biofilm formation as microbial development. Annu Rev Microbiol. 54: 49–79. doi:10.1146/annurev.micro.54.1.49
- Pastagia M, Euler C, Chahales P, Fuentes-Duculan J, Krueger J & Fischetti V. 2011. A novel chimeric lysin shows superiority to mupirocin for skin decolonization of methicillin-resistant and -sensitive *Staphylococcus aureus* strains. Antimicrob Agents Chemother. 55:738–744. doi:10.1128/AAC.00890-10
- Pastar I, Nusbaum A, Gil J, Patel S, Chen J, Valdes J, Stojadinovic O, Plano L, Tomic-Canic M & Davis S. 2013. Interactions of methicillin resistant *Staphylococcus aureus* USA300 and *Pseudomonas aeruginosa* in polymicrobial wound infection. PloS one. 8:e56846. doi:10.1371/ journal.pone.0056846
- Pei R & Lamas-Samanamud G. 2014. Inhibition of biofilm formation by T7 bacteriophages producing quorumquenching enzymes. Appl Environ Microbiol. 80: 5340–5348. doi:10.1128/AEM.01434-14
- Pelkonen S, Aalto J & Finne J. 1992. Differential activities of bacteriophage depolymerase on bacterial polysaccharide: binding is essential but degradation is inhibitory in phage infection of K1-defective *Escherichia coli*. J Bacteriol. 174:7757–7761. doi:10.1128/jb.174.23.7757-7761.1992
- Peters B, Jabra-Rizk M, Graeme A, Costerton J & Shirtliff M. 2012. Polymicrobial interactions: impact on pathogenesis and human disease. Clin Microbiol Rev. 25: 193–213. doi:10.1128/CMR.00013-11
- Pirnay J.-P., Blasdel B, Bretaudeau L, Buckling A, Chanishvili N, Clark J, Corte-Real S, Debarbieux L, Dublanchet A & de Vos D. 2015. Quality and safety requirements for sustainable phage therapy products. Pharm Res. 32:2173–2179. doi:10.1007/s11095-014-1617-7
- Pirnay J.-P., Verbeken G, Rose T, Jennes S, Zizi M, Huys I, Lavigne R, Merabishvili M, Vaneechoutte M & Buckling A. 2012. Introducing yesterday's phage therapy in today's medicine. Fut Virol. 7:379–390. doi:10.2217/fvl.12.24
- Pirnay J, de Vos D, Verbeken G, Merabishvili M, Chanishvili N, Vaneechoutte M, Zizi M, Laire G, Lavigne R, Huys I, et al. 2011. The phage therapy paradigm: preta-porter or sur-mesure? Pharm Res. 28:934–937. doi: 10.1007/s11095-010-0313-5
- Projan S. 2003. Why is big pharma getting out of antibacterial drug discovery? Curr Opin Microbiol. 6:427–430. doi:10.1016/j.mib.2003.08.003
- Rickard A, Gilbert P, High N, Kolenbrander P & Handley P. 2003. Bacterial coaggregation: an integral process in the development of multi-species biofilms. Trends Microbiol. 11:94–100. doi:10.1016/S0966-842X(02)00034-3

- Rieger-Hug D & Stirm S. 1981. Comparative study of host capsule depolymerases associated with *Klebsiella* bacteriophages. Virology. 113:363–378. doi:10.1016/0042-6822(81)90162-8
- Saxelin M, Nurmiaho E, Korhola M & Sundman V. 1979. Partial characterization of a new C3-type capsule-dissolving phage of *Streptococcus cremoris*. Can J Microbiol. 25: 1182–1187. doi:10.1139/m79-183
- Seth A, Geringer M, Nguyen K, Agnew S, Dumanian Z, Galiano R, Leung K, Mustoe T & Hong S. 2013. Bacteriophage therapy for *Staphylococcus aureus* biofilminfected wounds: a new approach to chronic wound care. Plast Reconstr Surg. 131:225–234. doi:10.1097/ PRS.0b013e31827e47cd
- Shettigar K, Jain S, Bhat D, Acharya R, Ramachandra L, Satyamoorthy K & Murali T. 2016. Virulence determinants in clinical *Staphylococcus aureus* from monomicrobial and polymicrobial infections of diabetic foot ulcers. J Med Microbiol. 65:1392–1404. doi:10.1099/jmm.0.000370
- Sillankorva S, Neubauer P & Azeredo J. 2010. Phage control of dual species biofilms of *Pseudomonas fluorescens* and *Staphylococcus lentus*. Biofouling. 26:567–575. doi: 10.1080/08927014.2010.494251
- Skillman L, Sutherland I & Jones M. 1998. The role of exopolysaccharides in dual species biofilm development. J Appl Microbiol. 85.
- Smith R, M'Ikanatha N M. & Read A. 2015. Antibiotic resistance: a primer and call to action. Health Commun. 30:309–314. doi:10.1080/10410236.2014.943634
- Son J.-S., Lee S.-J., Jun S, Yoon S, Kang S, Paik H, Kang J & Choi Y.-J. 2010. Antibacterial and biofilm removal activity of a podoviridae *Staphylococcus aureus* bacteriophage SAP-2 and a derived recombinant cell-walldegrading enzyme. Appl Microbiol Biotechnol. 86: 1439–1449. doi:10.1007/s00253-009-2386-9
- Speck P. 2013. Antibiotics: avert an impending crisis. Nature. 496:169. doi:10.1038/496169a
- Steenackers H, Parijs I, Foster K & Vanderleyden J. 2016. Experimental evolution in biofilm populations. FEMS Microbiol Rev. 40:373–397. doi:10.1093/femsre/fuw002
- Storms Z & Sauvageau D. 2015. Modeling tailed bacteriophage adsorption: Insight into mechanisms. Virology. 485:355–362. doi:10.1016/j.virol.2015.08.007
- Sutherland I & Wilkinson J. 1965. Depolymerases for bacterial exopolysaccharides obtained from phage-infected bacteria. Microbiology. 39:373–383.
- Sutherland I. 1999. Polysaccharases for microbial exopolysaccharides. Carbohydrate Polymers. 38:319–328. doi: 10.1016/S0144-8617(98)00114-3
- Sutherland I, Hughes K, Skillman L & Tait K. 2004. The interaction of phage and biofilms. FEMS Microbiol Lett. 232:1–6. doi:10.1016/S0378-1097(04)00041-2
- Tait K, Skillman L & Sutherland I. 2002. The efficacy of bacteriophage as a method of biofilm eradication. Biofouling, 18:305–311. doi:10.1080/0892701021000034418
- Tay W, Chong K & Kline K. 2016. Polymicrobial-Host Interactions during Infection. J Mol Biol. 428:3355–3371. doi:10.1016/j.jmb.2016.05.006
- Teh K, Flint S, Palmer J, Andrewes P, Bremer P & Lindsay D. 2014. Biofilm an unrecognised source of spoilage enzymes in dairy products? Int Dairy J. 34:32–40. doi: 10.1016/j.idairyj.2013.07.002

10 👄 L. G. KIFELEW ET AL.

- Teplitski M & Ritchie K. 2009. How feasible is the biological control of coral diseases? Trends Ecol Evol. 24: 378–385. doi:10.1016/j.tree.2009.02.008
- Thein Z, Samaranayake Y & Samaranayake L. 2007. Dietary sugars, serum and the biocide chlorhexidine digluconate modify the population and structural dynamics of mixed *Candida albicans* and *Escherichia coli* biofilms. APMIS. 115:1241–1251. doi:10.1111/j.1600-0643.2007.00735.x
- Verbeken G, Pirnay J, de Vos D, Jennes S, Zizi M, Lavigne R, Casteels M & Huys I. 2012. Optimizing the European regulatory framework for sustainable bacteriophage therapy in human medicine. Arch Immunol Ther Exp (Warsz). 60:161–172. doi:10.1007/s00005-012-0175-0
- Wei H. 2015. Bacteriophages, revitalized after 100 years in the shadow of antibiotics. Virol Sin. 30:1–2. doi:10.1007/ s12250-014-3562-y
- WHO 2014. Antimicrobial resistance global report on surveillance: 2014 summary.
- Wills Q, Kerrigan C & Soothill J. 2005. Experimental bacteriophage protection against Staphylococcus aureus abscesses in a rabbit model. Antimicrob Agents Chemother. 49:1220–1221. doi:10.1128/AAC.49.3.1220-1221.2005
- Wolcott R, Costerton J, Raoult D & Cutler S. 2013. The polymicrobial nature of biofilm infection. Clin Microbiol Infect. 19:107–112. doi:10.1111/j.1469-0691.2012.04001.x
- Wood S, Kirkham J, Marsh P, Shore R, Nattress B & Robinson C. 2000. Architecture of intact natural human plaque biofilms studied by confocal laser scanning microscopy. J Dent Res. 79:21–27. doi:10.1177/ 00220345000790010201

- Woods J, Boegli L, Kirker K, Agostinho A, Durch A, Delancey Pulcini E, Stewart P & James G. 2012. Development and application of a polymicrobial, in vitro, wound biofilm model. J Appl Microbiol. 112:998–1006. doi:10.1111/j.1365-2672.2012.05264.x
- Yan J, Mao J & Xie J. 2014. Bacteriophage polysaccharide depolymerases and biomedical applications. BioDrugs. 28:265–274. doi:10.1007/s40259-013-0081-y
- Yilmaz C, Colak M, Yilmaz B, Ersoz G, Kutateladze M & Gozlugol M. 2013. Bacteriophage therapy in implantrelated infections: an experimental study. J Bone Joint Surg Am. 95:117–125. doi:10.2106/JBJS.K.01135
- Yoong P, Schuch R, Nelson D & Fischetti V. 2004. Identification of a broadly active phage lytic enzyme with lethal activity against antibiotic-resistant *Enterococcus faecalis* and Enterococcus faecium. J Bacteriol. 186: 4808–4812. doi:10.1128/JB.186.14.4808-4812.2004
- Young R. 1992. Bacteriophage lysis: mechanism and regulation. Microbiol Rev. 56:430–481.
- Young R & Gill J. 2015. MICROBIOLOGY. Phage therapy redux–What is to be done? Science. 350:1163–1164. doi: 10.1126/science.aad6791
- Zhang J, Ormala-Odegrip A, Mappes J & Laakso J. 2014. Top-down effects of a lytic bacteriophage and protozoa on bacteria in aqueous and biofilm phases. Ecol Evol. 4: 4444–4453. doi:10.1002/ece3.1302
- Zhao G, Usui M, Lippman S, James G, Stewart P, Fleckman P & Olerud J. 2013. Biofilms and Inflammation in chronic wounds. Adv Wound Care (New Rochelle). 2: 389–399.



Article

Efficacy of Lytic Phage Cocktails on *Staphylococcus aureus* and *Pseudomonas aeruginosa* in Mixed-Species Planktonic Cultures and Biofilms

Legesse Garedew Kifelew ^{1,2,*}, Morgyn S. Warner ^{3,4}, Sandra Morales ^{5,†}, Nicky Thomas ⁶, David L. Gordon ⁷, James G. Mitchell ¹ and Peter G. Speck ¹

- ¹ College of Science and Engineering, Flinders University, Bedford Park, SA 5042, Australia; jim.mitchell@flinders.edu.au (J.G.M.); peter.speck@flinders.edu.au (P.G.S.)
- ² St Paul's Hospital Millennium Medical College, Addis Ababa 1271, Ethiopia
- ³ Infectious Diseases Unit, Queen Elizabeth Hospital, Woodville, SA 5011, Australia; morgyn.warner@sa.gov.au
- ⁴ Discipline of Medicine, University of Adelaide, Adelaide, SA 5005, Australia
- ⁵ AmpliPhi Australia Pty Ltd., Brookvale, NSW 2100, Australia; morales.sandra@gmail.com
 ⁶ Basil Hotzel Institute for Translational Health Research, School of Pharmacy and Medical Science
- ⁶ Basil Hetzel Institute for Translational Health Research, School of Pharmacy and Medical Sciences, University of South Australia, Adelaide, SA 5001, Australia; nicky.thomas@unisa.edu.au
- ⁷ Department of Microbiology and Infectious Diseases, College of Medicine and Public Health, Flinders University, Bedford Park, SA 5042, Australia; d.gordon@flinders.edu.au
- * Correspondence: legesse.kifelew@flinders.edu.au
- + Present Address: Phage Consulting, Sydney, NSW 2095, Australia.

Received: 24 April 2020; Accepted: 13 May 2020; Published: 18 May 2020



MDPI

Abstract: The efficacy of phages in multispecies infections has been poorly examined. The in vitro lytic efficacies of phage cocktails AB-SA01, AB-PA01, which target Staphylococcus aureus and Pseudomonas aeruginosa, respectively, and their combination against their hosts were evaluated in S. aureus and P. aeruginosa mixed-species planktonic and biofilm cultures. Green fluorescent protein (GFP)-labelled P. aeruginosa PAO1 and mCherry-labelled S. aureus KUB7 laboratory strains and clinical isolates were used as target bacteria. During real-time monitoring using fluorescence spectrophotometry, the density of mCherry S. aureus KUB7 and GFP P. aeruginosa PAO1 significantly decreased when treated by their respective phage cocktail, a mixture of phage cocktails, and gentamicin. The decrease in bacterial density measured by relative fluorescence strongly associated with the decline in bacterial cell counts. This microplate-based mixed-species culture treatment monitoring through spectrophotometry combine reproducibility, rapidity, and ease of management. It is amenable to high-throughput screening for phage cocktail efficacy evaluation. Each phage cocktail, the combination of the two phage cocktails, and tetracycline produced significant biofilm biomass reduction in mixed-species biofilms. This study result shows that these phage cocktails lyse their hosts in the presence of non-susceptible bacteria. These data support the use of phage cocktails therapy in infections with multiple bacterial species.

Keywords: phage cocktail therapy; biofilm; planktonic culture; fluorescence; efficacy; mixed-species culture; *Staphylococcus aureus*; *Pseudomonas aeruginosa*

1. Introduction

Wound infections, particularly chronic wounds such as occur in diabetic foot ulcers (DFUs), are often polymicrobial and frequently involve multidrug-resistant (MDR) bacterial pathogens [1,2]. A recent study found that 50% of the bacterial isolates recovered from DFU infections were MDR [3].

Viruses 2020, 12, 559; doi:10.3390/v12050559

www.mdpi.com/journal/viruses

Polymicrobial wound infections have reportedly increased in the last decade and include MDR *S. aureus* and *P. aeruginosa*, which are frequently isolated and therapeutically challenging bacteria [4,5]. Both *S. aureus* and *P. aeruginosa* are associated with severe wound infections, including orthopedic infections [4,6], are often isolated together [5], and may occupy different parts of wounds [7]. Co-infections with *S. aureus* and *P. aeruginosa* usually result in worse patient outcomes than infections due to either pathogen alone [8,9]. In addition to their high level of antibiotic resistance, *S. aureus* and *P. aeruginosa* form *in vivo* biofilms that typically result in increased tolerance to antibiotics and contribute to bacterial virulence [8] and immune evasion [10].

Most chronic wound infections involve biofilms, which render antibiotic treatment less effective. Moreover, bacteria within biofilms exhibit altered metabolic properties as compared to planktonic bacteria, which reduces the efficacy of antibiotics in this setting [1,9,11,12]. Phages represent a potential alternative or adjunct therapy for infections with antibiotic-resistant bacteria [13]. However, the efficacy of phages in mixed-species bacterial infections has been incompletely examined, and the limited literature contains conflicting reports [14]. Lytic phages kill their bacterial host by lysis (bursting the infected bacterial cell to release progeny phages) [15]. The process of phage infection and subsequent self-replication in bacteria offers advantages over antibiotics: phages are highly specific to the bacterial species they infect, an advantage over broadly active antimicrobials, as phages are not expected to disrupt a patients' normal microflora. Phages are able to lyse biofilm forms of their host bacteria, such as those typically found in infected DFUs [16,17]. Some studies also suggest that both of these antimicrobial agents in combination are more effective in controlling pathogenic bacteria than either alone [18,19].

Difficulties in growing different bacterial species together *in vitro*, as in the case of *S. aureus* and *P. aeruginosa*, make the study of bacterial interactions and efficacy of antibiotic agents complicated [20,21]. *P. aeruginosa* mostly kills or outcompetes *S. aureus* in *in vitro* co-cultures [22,23]. Medium containing bovine serum albumin (BSA) is recommended to allow better growth of *S. aureus* in the presence of *P. aeruginosa* [24,25]. In this study, we used fluorescence spectrophotometry to evaluate the effectiveness of phage cocktails of *S. aureus* and *P. aeruginosa* under mixed-species planktonic cultures. The use of different fluorescent proteins-labelled bacterial species in mixed-species culture allows monitoring, in real-time, of phage treatment effects based on the detection of a decrease in fluorescence relative to the untreated controls. Fluorescence spectrophotometry is easy to handle, reproducible, and a rapid technique to evaluate treatment efficacy [26,27]. To confirm the spectrophotometry results, we carried out bacterial counts post-treatment, using selective agars. The efficacy of these phage cocktails was also examined in mixed-species biofilms using bacterial counts.

2. Materials and Methods

2.1. Bacterial Species

Clinical isolates of *S. aureus* (n = 4) and *P. aeruginosa* (n = 4) were randomly selected from isolates obtained from South Australia Pathology. These were among the isolates that showed strong susceptibility in spot test and 73–88% biofilm biomass reduction on single-species biofilm experiments because of the phage cocktail and its components treatment (data provided as Supplementary Table S1). Laboratory strains mCherry-labelled *S. aureus* KUB7 [28] and GFP-labelled *P. aeruginosa* PAO1 were generously donated by A/Prof. Heather Jordan of Mississippi State University, USA, and Dr. Nicky Thomas of University of South Australia, Australia, respectively. The fluorescent proteins in the laboratory strains are driven by constitutive promoters and do not need antibiotics to maintain fluorescence expression. Both strains were susceptible to all antibiotics they were exposed during the VITEK[®] 2 test.

2.2. Phage Cocktails

The phage cocktails AB-SA01 and AB-PA01 were provided by AmpliPhi Biosciences Corporation (now Armata Pharmaceuticals, Inc.) (Los Angeles, CA, USA). AB-SA01 is a combination of three *Myoviridae* staphylococcal phages designated J-Sa-36, Sa-83, and Sa-87. The mean titer, presented in plaque-forming unit/millilitre—PFU/mL, was 9.3 log₁₀ (PFU/mL) for J-Sa-36 and Sa-83, 9.0 log₁₀ (PFU/mL) for Sa-87, and 9.1 log₁₀ (PFU/mL) for the combined product AB-SA01 on *S. aureus* laboratory strains RN4220 and SA6538. AB-PA01 is a combination of Pa-193 and Pa-204 from *Myoviridae*, and Pa-222 and Pa-223 from *Podoviridae*. The titer was 10.5 log₁₀ (PFU/mL) for Pa-193, Pa-204, and Pa-222, 9.5 log₁₀ (PFU/mL) for Pa-223, and 10.3 log₁₀ (PFU/mL) for AB-PA01 on *P. aeruginosa* laboratory strains PAO1 and PA10145. The phage titre was determined using plaque assay, as described [29,30]. None of these phage components encode any known bacterial virulence or antibiotic resistance genes, and all phages were considered to be strictly lytic [29,30]. The phages were produced following the current good manufacturing practice standard (cGMP) and approved by the US Food and Drug Administration as investigational new drugs [30,31].

2.3. Bacterial Identification

The isolates were identified using standard microbiology methods and confirmed by matrix-assisted laser desorption/ionization-time of flight mass spectrometry (MALDI-TOF MS) biotyping (BRUKER Pty. LTD., Melbourne, Victoria, Australia). Antibiotic susceptibility patterns were determined by VITEK[®] 2 (bioMérieux Australia Pty Ltd., Sydney, New South wales, Australia).

2.4. Mixed-Species Planktonic Cultures and Phage Cocktail Treatment

Mixed-species planktonic cultures were performed following an established protocol [32] with few modifications indicated as follows. In the case of fluorescently-labelled laboratory strains, 2-3 colonies from 18 h selective agar culture plates were suspended in sterile PBS and adjusted to 1.0 McFarland equivalence turbidity standard, containing approximately 8.5 log₁₀ (CFU/mL). Each bacterial species suspension was diluted at 1:100 v/v with nutrient broth (Sigma-Aldrich, Sydney, New South wales, Australia) supplemented with 5% BSA (Sigma-Aldrich, Sydney, New South wales, Australia) and incubated for 2 h at 37 °C. The two bacterial suspensions were then mixed at the ratio of 1:3 v/v of GFP *P. aeruginosa* PAO1 to mCherry *S. aureus* KUB7 in a sterile 10 mL test tube. Two hundred microlitre triplicates of each mixture were transferred to a clear 96-well flat-bottom Greiner CELLSTAR[®] polystyrene tissue culture plate (Sigma-Aldrich, Sydney, New South wales, Australia), and respective treatments were applied.

Mixed-species planktonic cultures were treated with one of (i) *S. aureus* phage cocktail AB-SA01; (ii) *P. aeruginosa* phage cocktail AB-PA01; (iii) a mixture of the two phage cocktails, AB-SA01+AB-PA01; (iv) gentamicin (positive control); or (v) PBS (negative control). The effect of each phage cocktail and the combination of the two phage cocktails was compared to the gentamicin- and PBS-treated groups.

AB-SA01, AB-PA01, or AB-SA01+AB-PA01 phage cocktails were applied at a multiplicity of infection (MOI) of one to fluorescently labelled *S. aureus-P. aeruginosa* mixed-species culture. Gentamicin was used as a positive control at 16 µg/mL as the minimum inhibitory concentration (MIC) to these isolates was ≤ 8 µg/mL. An equal volume of PBS to phage solutions was applied to negative control groups. A plate cover was applied, and the plate was wrapped with aluminium foil from the top and sides. The plate was incubated in a CLARIOstar Omega plate reader (BMG LABTECH Pty. Ltd., Melbourne, Victoria, Australia) for 24 h at 37 °C with 100 rpm constant double orbital shaking between measurements as described [33] for fluorescent protein-labelled strains. Fluorescence of mCherry and GFP was measured (in relative fluorescence unit, RFU) every 30 min in each well. The excitation and emission wavelengths were set at 570-15 and 620-20 nm for mCherry, and 470-15 and 515-20 nm for GFP detection, respectively. Signals from triplicate wells were averaged and corrected for blank wells containing only nutrient broth. After 24 h incubation, bacterial colony counts were performed

using serial dilution on selective agars for each species, vancomycin-supplemented MacConkey for *P. aeruginosa*, and mannitol salt agar (MSA) for *S. aureus*. The experiment was repeated three times with the same protocol on different days. A similar protocol was followed for clinical isolates, except that incubation was in a standard incubator.

2.5. In Vitro Mixed-Species Biofilm Development and Phage Cocktail Treatment

Mixed-species biofilm development and treatment was conducted using a previously described procedure [34,35] with few modifications. Briefly, 2-3 colonies of 18 h culture of each isolate were independently suspended in sterile PBS and adjusted to 1.0 McFarland turbidity standard. These suspensions were pooled at 1:3 v/v ratio of *P. aeruginosa* to *S. aureus*, and 100 µL of the mixed suspension was transferred to 10 mL 5% BSA-nutrient broth. The final suspension was supplemented with 1% sterile glucose to facilitate biofilm development. Two hundred microlitres of the suspension was transferred in triplicate into a tissue culture plate and incubated for 48 h at 37 °C with 70 rpm agitation.

The treatment categories of mixed-species biofilms were (i) *S. aureus* phage cocktail AB-SA01; (ii) *P. aeruginosa* phage cocktail AB-PA01; (iii) a mixture of the two phage cocktails, AB-SA01+AB-PA01; (iv) tetracycline (positive control); and (v) PBS (negative control). The effect of each phage cocktail and a combination of the two phage cocktails was compared to the tetracycline- and PBS-treated groups. Tetracycline (Sigma-Aldrich Corporation, Sydney, New South wales, Australia) was used as a positive control in mixed-species biofilm treatment as it was strongly effective (p < 0.001), compared with PBS treatment, in biofilm biomass reduction on single-species biofilm treatment of both *S. aureus* and *P. aeruginosa* isolates. However, gentamicin did not produce significant biofilm biomass reduction (p > 0.05), compared with PBS treatment (Unpublished data). Tetracycline was not used as a positive control in mixed-species planktonic culture treatment experiments to avoid exaggerated fluorescence detection because of its color and fluorescent nature [36].

Next, the liquid culture was removed, and plates were washed gently twice using sterile deionized water. Then, 225 μ L of AB-SA01, AB-PA01, or AB-SA01+AB-PA01 in the nutrient broth was applied to the respective treatment group biofilms. An equivalent volume of tetracycline and PBS to phage solution in nutrient broth were also applied as controls. The concentration of tetracycline was 128 μ g/mL because *P. aeruginosa* isolates are susceptible to a higher concentration of tetracycline [37,38]. The MIC of tetracycline for *S. aureus* was $\leq 8 \mu$ g/mL during VITEK[®] 2 antimicrobial susceptibility test. Treated biofilms were incubated for 12 h at 37 °C with no agitation. The biofilm was washed twice using 250 μ L sterile PBS through careful pipetting. The biofilm-associated cells attached to the well surface were collected with 225 μ L nutrient broth by pipetting after scraping the surface with a loop as described earlier [39,40]. After homogenization with a vortex mixer, the cell suspension was serially diluted, $10^{-1}-10^{-8}$, in filter-sterilized 10 mM ferrous ammonium sulphate (FAS) supplemented nutrient broth to inactivate free phage [41] and incubated at room temperature for 15 min.

2.6. Viable Bacterial Cell Count

One hundred microlitres of the bacterial suspension from each serial dilution was then mixed with 3 mL nutrient soft agar warmed at 42 °C and dispensed over 37 °C pre-warmed MSA and vancomycin-supplemented MacConkey agar in triplicate and incubated at 37 °C for 24 h. Plates with approximately 30–300 colonies were taken from one of the dilutions, and colony count was carried out as previously described [42]. The bacterial cell count was calculated using the formula B = N/d where B = number of bacteria; N = average number of colonies counted on three plates; d = dilution factor as described earlier [43]. The results are expressed as logarithm-transformed values (log (CFU/mL)).

2.7. Statistical Analysis

STATA version 16 software was used for statistical analysis. Data are reported in terms of the mean. A comparison of experimental groups was performed using a one-way analysis of variance (two-tailed) or paired '*t-test*'. A p < 0.05 value was considered statistically significant.

3. Results

3.1. Effect of Phage Cocktails on Fluorescently Labelled Mixed-Species Planktonic Cultures

The fluorescence of each species alone or together was confirmed under a confocal microscope, as shown in Figure 1.



Figure 1. Fluorescence strain under confocal microscopy: (**A**) mCherry-labelled *S. aureus* KUB7, (**B**) GFP-labelled *P. aeruginosa* PAO1, and (**C**) mCherry-labelled *S. aureus* KUB7 mixed with GFP-labelled *P. aeruginosa* PAO1.

The growth of fluorescently-labelled *S. aureus* and *P. aeruginosa* in single- and mixed-species planktonic cultures in the presence and absence of phages is shown in Figure 2A–G. Compared to the phosphate-buffered saline (PBS) treatment, AB-SA01 and AB-PA01 treatments effectively halted the growth of their host throughout the 24 h follow-up period. In the single-species cultures treated with PBS, there was a marked increase in fluorescence of mCherry *S. aureus* KUB7 and GFP *P. aeruginosa* PAO1, as shown in Figure 2A,B, indicating bacterial growth. In the mixed-species cultures without phages, the magnitude of fluorescence slowly increased with time (Figure 2C). However, the maximum red and green fluorescence obtained was much lower than the fluorescence detected during single-species PBS-treated cultures (Figure 2A,B), suggesting that the mixed-species exhibited competition for nutrients or co-inhibitory effects. In mixed-species cultures treated with a single phage cocktail, the fluorescence of the target host was almost eliminated, while the non-target host was unaffected, as shown in Figure 2D,E, with fluorescence similar to PBS-treated single-species control.



Figure 2. Cont.



Figure 2. Cont.



Figure 2. Cont.



Figure 2. A–G: effect of phage cocktails on mCherry *S. aureus* KUB7 (red data points) and GFP *P. aeruginosa* PAO1 (green data points) single- and mixed-species planktonic cultures. RFU represents relative fluorescence unit. (A) PBS-treated mCherry *S. aureus* KUB7 (in red) single-species culture; the green graph is due to background detection since the machine was set for mCherry and GFP detection. (B) PBS-treated GFP *P. aeruginosa* PAO1 (in green) single-species culture; the red graph is due to background detection since the machine was set for mCherry and GFP detection. (C) PBS-treated mCherry *S. aureus* KUB7 and GFP *P. aeruginosa* PAO1 mixed-species culture. (E) AB-PA01-treated mCherry *S. aureus* KUB7 and GFP *P. aeruginosa* PAO1 mixed-species culture. (F) AB-PA01+AB-SA01-treated mCherry *S. aureus* KUB7 and GFP *P. aeruginosa* PAO1 mixed-species culture. (G) Gentamicin-treated mCherry *S. aureus* KUB7 and GFP *P. aeruginosa* PAO1 mixed-species culture.

When both phage cocktails were added to the mixed-species cultures, there was low fluorescence of both bacterial species, as shown in Figure 2F, similar to the inhibitory effect of gentamicin (Figure 2G), indicating that phage efficacy is not affected by the presence of non-host bacteria or other phages. The highest magnitude of red fluorescence in mCherry *S. aureus* KUB7 was detected from the untreated single-species culture (Figure 2A). In the case of GFP *P. aeruginosa* PAO1, the highest green fluorescence was detected from mixed-species culture treated with *S. aureus* phage cocktail, AB-SA01 (Figure 2D). The lowest magnitude of fluorescence from the target host in the mixed-species culture was observed when treated with each phage cocktail, as shown in Figure 2D,E. As expected, AB-SA01 and AB-PA01 exhibited no lytic effect on non-susceptible hosts (Figure 2D,E). While the fluorescence detected from a non-susceptible host showed an increase through time, the fluorescence obtained from the susceptible host remained low.

The decreases in fluorescence from each phage cocktail-, combinations of the two phage cocktails-, and gentamicin-treated groups were significantly lower compared to the PBS-treated group. The corresponding colony count results for each treatment group after 24 h are shown in Table 1 and confirm the results obtained with the fluorescence detection methodology.

Inclator		Bacterial Cell Counts after Treatment					
Combination	Evaluated Isolate	PBS	AB-SA01	AB-PA01	AB-SA01+ AB-PA01	Gentamicin	
S. aureus KUB7 ^S	S. aureus KUB7 ^S	5.5	3.6	8.7	3.0	0	
and GFP PAO1 ^P	GFP PAO1 ^P	7.9	8.9	0	3.7	0	
63-6538 ^S and	63-6538 ^S	5.6	1.5	7.9	0	0	
63-6598 ^P	63-6598 ^P	5.3	6.0	0	0	0	
63-2498 ^S and	63-2498 ^S	6.2	0	6.6	0	0	
63-5497 ^P	63-5497 ^P	7.5	8.0	3.5	4.0	0	
63-5656 ^S and	63-5656 ^S	4.8	3.6	6.1	0	0	
63-6036 ^P	63-6036 ^P	6.6	8.1	3.3	4.2	0	
Summarized treatment	nt effect						
	S. aureus mean	5.5	2.2	7.3	0.8	0	
	S. aureus reduction		3.3	+1.8	4.7	5.5	
	<i>P. aeruginosa</i> mean	6.8	7.8	1.7	3.0	0	
	<i>P. aeruginosa</i> reduction	_	+1.0	5.1	3.8	6.8	

Table 1. Bacteria cell count (log₁₀ (CFU/mL)) of *S. aureus* and *P. aeruginosa* in mixed-species planktonic culture after 24 h phage cocktails, gentamicin, or PBS treatment.

^S *S. aureus* isolates, ^P *P. aeruginosa* isolates, + indicates an increase in bacterial count compared to PBS treatment.

3.2. Efficacy of Phage Cocktails on Laboratory and Clinical Isolates Mixed-Species Planktonic Cultures

The population of each bacterial species in mixed-species planktonic cultures at the end of 24 h of treatment was assessed. Compared to PBS-treated samples, AB-SA01- and AB-PA01-treated samples produced 3.3 log₁₀ (CFU/mL) and 5.1 log₁₀ (CFU/mL) reduction on their hosts, respectively, as shown in Table 1. These reductions in bacterial cell count were associated with the susceptibility of each bacterial isolates to the specific phage cocktail and its component phages during spot test (see complementary data, Table S1). When the same samples were treated with the combination of the two phage cocktails, AB-SA01+AB-PA01, the mean cell count of *S. aureus* and *P. aeruginosa* reduced by 4.7 log₁₀ (CFU/mL) and 3.8 log₁₀ (CFU/mL), respectively. The cell counts of one bacterial species showed an increase when the culture was treated with only a phage cocktail of the other species in the mixed-species culture. All planktonic cultures treated with gentamicin yielded no viable bacterial cells.

3.3. Effect of Phage Cocktails Treatment on Mixed-Species Biofilms

The findings of this study demonstrate that phage cocktails AB-SA01, AB-PA01, and a mixture of AB-SA01 and AB-PA01 successfully lysed their hosts in the presence of biofilms of non-susceptible species. These phage cocktails applied to *S. aureus* and *P. aeruginosa* mixed-species biofilms caused a statistically significant (p < 0.05; Table 2) reduction in the host cell population compared to the PBS-treated group, as shown in Table 2. However, the reduction of the cell population in *S. aureus* and *P. aeruginosa* was less than half of the decrease observed in planktonic culture treatment. Most of the tetracycline-treated cultures produced no or the lowest number of bacterial cells.

Compared to PBS treatment, the application of AB-SA01 or AB-PA01 alone did not produce a statistically significant effect on the cell count of the non-host bacterial species population (p > 0.05; 6.9 vs. 6.8 for AB-SA01 and 6.2 vs. 5.8 for AB-PA01, Table 2). The mean bacterial cell population of each species remained unaffected when treated with the other species' phage cocktail alone. Treatment of mixed-species biofilms using the mixture of the two phage cocktails, AB-SA01+AB-PA01, produced similar cell reduction on both *S. aureus* and *P. aeruginosa* isolates as each phage cocktail treatment. The effect of tetracycline treatment caused a significant reduction in the population of both bacterial species.

Tecleter		Bacterial Cell Counts after Treatment				
Combination	Evaluated Isolate	PBS	AB-SA01	AB-PA01	AB-SA01+ AB-PA01	Tetracycline
S. aureus KUB7 ^S	S. aureus KUB7 ^S	6.2	4.5	7.4	5.5	3.8
and PAO1 GFP ^P	PAO1 GFP ^P	6.4	6.3	3.8	4.0	3.9
63-6538 ^S and	63-6538 ^S	5.2	4.4	4.7	3.6	3.0
63-6598 P	63-6598 ^P	5.5	5.6	3.6	3.7	0
63-2498 ^S and	63-2498 ^S	6.3	4.4	5.2	5.5	0
63-5497 ^P	63-5497 ^P	7.1	6.2	5.5	5.2	0
63-5656 ^S and	63-5656 ^S	7	4.9	5.9	5.3	0
63-6036 ^P	63-6036 ^P	8.5	8.9	4.7	6.3	0
Summarized treatme	ent effect					
	<i>S. aureus</i> mean	6.2	4.6	5.8	5.0	1.7
	S. aureus reduction		1.6	0.4	1.2	4.5
	P. aeruginosa mean	6.9	6.8	4.4	4.8	1.0
	<i>P. aeruginosa</i> reduction		0.1	2.5	2.1	5.9

Table 2. Bacteria count (log₁₀ (CFU/mL)) of *S. aureus* and *P. aeruginosa* in mixed-species biofilms after 24 h phage cocktails, tetracycline, and PBS treatment.

^S S. aureus isolates, ^P P. aeruginosa isolates.

4. Discussion

The rationale to examine the effect of phage treatment on mixed-species planktonic and biofilm cultures was that many wound infections are polymicrobial and contain bacteria in biofilm forms [44] and that there is a paucity of data on the action of phages under such circumstances. In polymicrobial infections, there exist interspecies interactions, ranging from antagonism to cooperation, that can significantly impact the pathogenicity of microbes and clinical outcomes of infections [45]. Examination of the bacterial population using fluorescence detection and CFU count results suggest that phage cocktail treatment is effective both in planktonic and biofilm states of the host bacteria under mixed-species cultures.

Multiple fluorescent proteins can be simultaneously applied to examine different microbial populations in real-time [46,47]. The use of mCherry in combination with GFP is suitable as the excitation and emission spectra of these proteins are well separated [46,48]. The advantages of using mCherry and GFP as markers include ease of detection, no exogenous substrate is required that may perturb biological samples, no need for cell processing for visualization, and they are suitable for real-time monitoring of cells in mixed cultures [49,50]. In this study, the high magnitude of fluorescence detected and the bacterial cell population obtained during single species culture without treatment show bacterial growth and fitness capability for the model while fluorescing [26,49–51]. Hence, we used mCherry *S. aureus* KUB7 and GFP *P. aeruginosa* PAO1 to distinguish them in mixed-species culture. In the current study, loss of mCherry or GFP fluorescence was considered an indication of bacterial death due to phage-induced lysis, which was supported by decreased or no CFU. Previous studies also showed that the decrease in fluorescence is due to cell death, suggesting it to be an early and sensitive marker of viability [26,52].

There was a strong association between the final bacterial density reading through fluorescence detection and bacterial cell population data as measured by colony count. For both mCherry *S. aureus* KUB7 and GFP *P. aeruginosa* PAO1, the fluorescence signal increased for PBS-treated or bacterial species unaffected by the treatment. The higher green fluorescence detected in AB-SA01-treated mixed-species culture than in GFP P. *aeruginosa* PAO1 single-species PBS-treated culture might be attributed to more pronounced *P. aeruginosa* population growth in the absence of competing organism or enhanced growth because of accessibility to more nutrients such as iron from the dead *S. aureus* cells [53]. The magnitude of fluorescence of mCherry or GFP obtained was minimal when the mixed-species

bacterial cultures were treated with each phage cocktail alone. The fluorescence magnitude and bacterial cell population obtained during AB-SA01+AB-PA01 and gentamicin treatments were similar; the fluorescence records are low; the fluorescence graphs are overlapped and remained at their lower levels throughout the experiment period, and bacterial population recovered at the end of the experiment is minimal or none. This finding shows that treatment with each phage cocktails separately or in combination produced a similar effect to gentamicin treatment in planktonic mixed-species culture. Our experiments also demonstrated that bacterial density could be estimated, and the treatment effect evaluated in mixed-species cultures through fluorescence spectrophotometry using different fluorescent proteins-labelled bacterial species.

The magnitude of fluorescence detected in the untreated mixed-species culture was lower compared to untreated single-species cultures, which might be attributable to competition between the two bacterial species [21–23]. All these decreased or increased fluorescence detected were strongly associated with the decreased or increased colony count results, respectively. Our observations support previous reports that show the magnitude of fluorescence, absorbance, and colony count results are supplementary to one another in characterizing bacterial growth [54,55].

The magnitudes of fluorescence obtained from the three replica wells across the three independent experiments on different days with the same protocol were similar, demonstrating reliability of the method and reproducibility of results. The decrease in fluorescence of treated samples is consistent with efficient bacterial cell lysis by the host-specific phage cocktail as measured through colony count. This result agrees with a study that compared the fluorescence with a CFU count method in single-species culture [26]. In the mixed-species planktonic cultures of clinical isolates, treatment with each phage cocktail alone caused a significant (p < 0.001) decline of the target bacterial host population. Treatment with a combination of AB-SA01 and AB-PA01 also resulted in a significant (p < 0.05) decrease in cell density of both bacterial species. Our findings are similar to a study that showed planktonic *E. coli* grown in co-culture with *Salmonella* enterica did not survive attack from *E. coli* specific phages [56].

Mixed-species biofilms are complex communities that affect the physiological state of host bacterial cells and the availability of phage receptors, possibly due to competition with other bacterial species [14,39]. Our findings of mixed-species biofilm treatment show that the effect of both phage cocktails separately and in combination, AB-SA01+AB-PA01, significantly ($p \le 0.05$) reduced the target bacterial host population. The effect of these phage cocktails was lower in biofilm than in planktonic states of their hosts, which agrees with previous reports [35,39,57–59]. Possible explanations include: the complex extracellular biofilm matrix may reduce the efficacy of phages because of entrapment of phage particles [57], reduced multiplication of phages due to the large proportion of metabolically inactive host cells in biofilms [60], or shedding of phage receptors from the host bacteria in biofilms [61]. It might also be partly explained by the fact that dead cells resulting from phage attack might support surviving bacteria through serving as a nutrient reservoir and providing a shield from phage attack by phage binding to receptors on dead bacteria [62]. Tetracycline treatment showed superior bacterial population reduction, on both bacterial species, compared to the phage cocktail formulations used in this study. Our finding is consistent with a previous report that showed the application of phiIPLA-RODI, phiIPLA-C1C, and the combination of the two phages is more efficient in the planktonic phase than that in biofilms phase during S. aureus IPLA16 and S. epidermidis LO5081 mixed-species cultures [39].

In this study, we noted that phage cocktails applied to mixed-species biofilms could effectively reduce bacterial host populations. We did not observe the protection of susceptible bacterial hosts from phage attack in mixed-species biofilms by non-susceptible bacteria. This observation is consistent with some studies [34,40]. However, it is in contrast with a previous phage treatment study that concluded the structural heterogeneity of the biofilm from mixed-species produced pockets of unreachable susceptible bacteria [63]. Because we found significant bacterial host cell reduction during each

phage cocktail treatment in mixed-species biofilm, protection from lysis for the target bacteria by the non-susceptible bacterial host cannot be assumed [40].

Our bacterial count data are in line with previous findings of biofilms infected with phages for 24 h and more extended periods [63,64]. Overall, our results confirm that lytic phages can be efficient in mixed-species planktonic and biofilm states. This study also demonstrates the feasibility of *in vitro* real-time monitoring of the efficacy of phage treatments using fluorescently labelled bacteria in mixed-species cultures through spectrophotometry, which is a simple, rapid, and reliable procedure.

5. Conclusions

Our findings suggest that the use of phage cocktails in mixed-species planktonic or biofilm state could provide practical alternatives to antibiotics in combating antibiotic-resistant infections. The association between the decrease or loss of fluorescence during real-time monitoring of the effect of phage cocktails with the decrease in numbers of bacterial cells as measured by colony counts, across the three replicas and three experiments in different days with the same protocol, shows the effectiveness of the phage cocktail treatments *in vitro* and repeatability of the results. The present study findings show that phages can reduce the host bacterial cell population significantly from planktonic and biofilm states. The lytic efficacy of AB-SA01, AB-PA01, and their combination on antibiotic-resistant bacterial hosts in mixed-species planktonic and biofilm phases is a clear indication of the potential of phages to mitigate antibiotic-resistant infections.

Supplementary Materials: The following are available online at http://www.mdpi.com/1999-4915/12/5/559/s1, Table S1: Phage cocktail and component phages susceptibility and biofilm reduction (in percentage) after phage cocktail treatment of the selected bacterial isolates

Author Contributions: L.G.K. participated in the conception of the research idea and design of the methodology, carried out laboratory experiment and data collection, and prepared the manuscript. P.G.S. and J.G.M. participated in the conception of the research idea and design of the methodology. P.G.S. and L.G.K. performed data analysis. M.S.W., S.M., N.T., D.L.G., J.G.M., and P.G.S. edited the manuscript. All researchers approved the final manuscript. All authors have read and agreed to the published version of the manuscript.

Funding: This research is funded by Diabetes Australia Research Program.

Acknowledgments: The generous support of a grant from Diabetes Australia is acknowledged. We are grateful to South Australia Pathology, A Heather Jordan of Mississippi State University, USA, and Nicky Thomas of South Australia University, Australia, for providing the bacterial strains and to AmpliPhi Biosciences Corporation, (now Armata Pharmaceuticals, Inc.), Los Angeles, CA, USA, for supplying AB-SA01 and AB-PA01 phage cocktails. We also acknowledge Richard Woodman for his statistical analysis support.

Conflicts of Interest: S.M was an AmpliPhi Biosciences employee at the time of performing the experiments and contributed to the scientific discussion. All other authors declare no conflict of interest.

References

- Bowler, P.G.; Duerden, B.I.; Armstrong, D.G. Wound microbiology and associated approaches to wound management. *Clin. Microbiol. Rev.* 2001, 14, 244–269. [CrossRef]
- Citron, D.M.; Goldstein, E.J.; Merriam, C.V.; Lipsky, B.A.; Abramson, M.A. Bacteriology of moderate-to-severe diabetic foot infections and in vitro activity of antimicrobial agents. *J. Clin. Microbiol.* 2007, 45, 2819–2828. [CrossRef] [PubMed]
- Noor, S.; Borse, A.G.; Ozair, M.; Raghav, A.; Parwez, I.; Ahmad, J. Inflammatory markers as risk factors for infection with multidrug-resistant microbes in diabetic foot subjects. *Foot* 2017, 32, 44–48. [CrossRef] [PubMed]
- Gjødsbøl, K.; Christensen, J.J.; Karlsmark, T.; Jørgensen, B.; Klein, B.M.; Krogfelt, K.A. Multiple bacterial species reside in chronic wounds: A longitudinal study. *Int. Wound J.* 2006, *3*, 225–231. [CrossRef] [PubMed]
- Trivedi, U.; Parameswaran, S.; Armstrong, A.; Burgueno-Vega, D.; Griswold, J.; Dissanaike, S.; Rumbaugh, K.P. Prevalence of multiple antibiotic resistant infections in diabetic versus nondiabetic wounds. *J. Pathog.* 2014, 2014, 173053. [CrossRef] [PubMed]
- 6. Madsen, S.M.; Westh, H.; Danielsen, L.; Rosdahl, V.T. Bacterial colonization and healing of venous leg ulcers. *Apmis* **1996**, *104*, 895–899. [CrossRef] [PubMed]

- Fazli, M.; Bjarnsholt, T.; Kirketerp-Moller, K.; Jorgensen, B.; Andersen, A.S.; Krogfelt, K.A.; Givskov, M.; Tolker-Nielsen, T. Nonrandom distribution of Pseudomonas aeruginosa and Staphylococcus aureus in chronic wounds. J. Clin. Microbiol. 2009, 47, 4084–4089. [CrossRef]
- DeLeon, S.; Clinton, A.; Fowler, H.; Everett, J.; Horswill, A.R.; Rumbaugh, K.P. Synergistic interactions of Pseudomonas aeruginosa and Staphylococcus aureus in an in vitro wound model. *Infect. Immun.* 2014, 82, 4718–4728. [CrossRef]
- Pastar, I.; Nusbaum, A.G.; Gil, J.; Patel, S.B.; Chen, J.; Valdes, J.; Stojadinovic, O.; Plano, L.R.; Tomic-Canic, M.; Davis, S.C. Interactions of methicillin resistant Staphylococcus aureus USA300 and Pseudomonas aeruginosa in polymicrobial wound infection. *PLoS ONE* 2013, *8*, e56846. [CrossRef]
- Tay, W.H.; Chong, K.K.; Kline, K.A. Polymicrobial-Host Interactions during Infection. J. Mol. Biol. 2016, 428, 3355–3371. [CrossRef]
- Dalton, T.; Dowd, S.E.; Wolcott, R.D.; Sun, Y.; Watters, C.; Griswold, J.A.; Rumbaugh, K.P. An in vivo polymicrobial biofilm wound infection model to study interspecies interactions. *PLoS ONE* 2011, *6*, e27317. [CrossRef] [PubMed]
- 12. Whiteley, M.; Bangera, M.G.; Bumgarner, R.E.; Parsek, M.R.; Teitzel, G.M.; Lory, S.; Greenberg, E.P. Gene expression in Pseudomonas aeruginosa biofilms. *Nature* **2001**, *413*, 860–864. [CrossRef] [PubMed]
- Kutter, E.; De Vos, D.; Gvasalia, G.; Alavidze, Z.; Gogokhia, L.; Kuhl, S.; Abedon, S.T. Phage therapy in clinical practice: Treatment of human infections. *Curr. Pharm. Biotechnol.* 2010, *11*, 69–86. [CrossRef] [PubMed]
- Geredew Kifelew, L.; Mitchell, J.G.; Speck, P. Mini-review: Efficacy of lytic bacteriophages on multispecies biofilms. *Biofouling* 2019, 35, 472–481. [CrossRef] [PubMed]
- Young, R. Bacteriophage lysis: Mechanism and regulation. *Microbiol. Rev.* 1992, 56, 430–481. [CrossRef] [PubMed]
- 16. Wittebole, X.; De Roock, S.; Opal, S.M. A historical overview of bacteriophage therapy as an alternative to antibiotics for the treatment of bacterial pathogens. *Virulence* **2014**, *5*, 226–235. [CrossRef]
- Donlan, R.M. Preventing biofilms of clinically relevant organisms using bacteriophage. *Trends Microbiol.* 2009, 17, 66–72. [CrossRef]
- Torres-Barcelo, C.; Hochberg, M.E. Evolutionary Rationale for Phages as Complements of Antibiotics. *Trends* Microbiol. 2016, 24, 249–256. [CrossRef]
- 19. Kirby, A.E. Synergistic action of gentamicin and bacteriophage in a continuous culture population of Staphylococcus aureus. *PLoS ONE* **2012**, *7*, e51017. [CrossRef]
- Woods, P.W.; Haynes, Z.M.; Mina, E.G.; Marques, C.N.H. Maintenance of S. aureus in Co-culture With P. aeruginosa While Growing as Biofilms. *Front. Microbiol.* 2018, 9, 3291. [CrossRef] [PubMed]
- Michelsen, C.F.; Christensen, A.M.; Bojer, M.S.; Hoiby, N.; Ingmer, H.; Jelsbak, L. Staphylococcus aureus alters growth activity, autolysis, and antibiotic tolerance in a human host-adapted Pseudomonas aeruginosa lineage. J. Bacteriol. 2014, 196, 3903–3911. [CrossRef] [PubMed]
- Armbruster, C.R.; Wolter, D.J.; Mishra, M.; Hayden, H.S.; Radey, M.C.; Merrihew, G.; MacCoss, M.J.; Burns, J.; Wozniak, D.J.; Parsek, M.R. Staphylococcus aureus protein A mediates interspecies interactions at the cell surface of Pseudomonas aeruginosa. *MBio* 2016, 7, e00538-16. [CrossRef]
- Baldan, R.; Cigana, C.; Testa, F.; Bianconi, I.; De Simone, M.; Pellin, D.; Di Serio, C.; Bragonzi, A.; Cirillo, D.M. Adaptation of Pseudomonas aeruginosa in Cystic Fibrosis airways influences virulence of Staphylococcus aureus in vitro and murine models of co-infection. *PLoS ONE* 2014, 9, e89614. [CrossRef] [PubMed]
- Tavernier, S.; Crabbe, A.; Hacioglu, M.; Stuer, L.; Henry, S.; Rigole, P.; Dhondt, I.; Coenye, T. Community Composition Determines Activity of Antibiotics against Multispecies Biofilms. *Antimicrob. Agents Chemother.* 2017, 61. [CrossRef] [PubMed]
- Radlinski, L.; Rowe, S.E.; Kartchner, L.B.; Maile, R.; Cairns, B.A.; Vitko, N.P.; Gode, C.J.; Lachiewicz, A.M.; Wolfgang, M.C.; Conlon, B.P. Pseudomonas aeruginosa exoproducts determine antibiotic efficacy against Staphylococcus aureus. *PLoS Biol.* 2017, *15*, e2003981. [CrossRef]
- Webb, J.S.; Barratt, S.R.; Sabev, H.; Nixon, M.; Eastwood, I.M.; Greenhalgh, M.; Handley, P.S.; Robson, G.D. Green fluorescent protein as a novel indicator of antimicrobial susceptibility in Aureobasidium pullulans. *Appl. Environ. Microbiol.* 2001, 67, 5614–5620. [CrossRef]
- Steffy, K.; Shanthi, G.; Maroky, A.S.; Selvakumar, S. Synthesis and characterization of ZnO phytonanocomposite using Strychnos nux-vomica L. (Loganiaceae) and antimicrobial activity against multidrug-resistant bacterial strains from diabetic foot ulcer. J. Adv. Res. 2018, 9, 69–77. [CrossRef] [PubMed]

- Burcham, Z.M.; Hood, J.A.; Pechal, J.L.; Krausz, K.L.; Bose, J.L.; Schmidt, C.J.; Benbow, M.E.; Jordan, H.R. Fluorescently labeled bacteria provide insight on post-mortem microbial transmigration. *Forensic Sci. Int.* 2016, 264, 63–69. [CrossRef]
- Fong, S.A.; Drilling, A.; Morales, S.; Cornet, M.E.; Woodworth, B.A.; Fokkens, W.J.; Psaltis, A.J.; Vreugde, S.; Wormald, P.J. Activity of Bacteriophages in Removing Biofilms of Pseudomonas aeruginosa Isolates from Chronic Rhinosinusitis Patients. *Front. Cell. Infect. Microbiol.* 2017, 7, 418. [CrossRef]
- Lehman, S.M.; Mearns, G.; Rankin, D.; Cole, R.A.; Smrekar, F.; Branston, S.D.; Morales, S. Design and Preclinical Development of a Phage Product for the Treatment of Antibiotic-Resistant Staphylococcus aureus Infections. *Viruses* 2019, 11, 88. [CrossRef]
- Law, N.; Logan, C.; Yung, G.; Furr, C.L.; Lehman, S.M.; Morales, S.; Rosas, F.; Gaidamaka, A.; Bilinsky, I.; Grint, P.; et al. Successful adjunctive use of bacteriophage therapy for treatment of multidrug-resistant Pseudomonas aeruginosa infection in a cystic fibrosis patient. *Infection* 2019, 47, 665–668. [CrossRef] [PubMed]
- 32. Kumar, A.; Ting, Y.P. Presence of Pseudomonas aeruginosa influences biofilm formation and surface protein expression of Staphylococcus aureus. *Environ. Microbiol.* **2015**, *17*, 4459–4468. [CrossRef] [PubMed]
- Alves, D.R.; Perez-Esteban, P.; Kot, W.; Bean, J.E.; Arnot, T.; Hansen, L.H.; Enright, M.C.; Jenkins, A.T. A novel bacteriophage cocktail reduces and disperses Pseudomonas aeruginosa biofilms under static and flow conditions. *Microb. Biotechnol.* 2016, *9*, 61–74. [CrossRef] [PubMed]
- Gonzalez, S.; Fernandez, L.; Campelo, A.B.; Gutierrez, D.; Martinez, B.; Rodriguez, A.; Garcia, P. The Behavior of Staphylococcus aureus Dual-Species Biofilms Treated with Bacteriophage phiIPLA-RODI Depends on the Accompanying Microorganism. *Appl. Environ. Microbiol.* 2017, 83. [CrossRef]
- Mendes, J.J.; Leandro, C.; Mottola, C.; Barbosa, R.; Silva, F.A.; Oliveira, M.; Vilela, C.L.; Melo-Cristino, J.; Gorski, A.; Pimentel, M.; et al. In vitro design of a novel lytic bacteriophage cocktail with therapeutic potential against organisms causing diabetic foot infections. J. Med Microbiol. 2014, 63, 1055–1065. [CrossRef]
- Ni, Y.; Deng, N.; Kokot, S. A simple kinetic spectrophotometric method for simultaneous determination of tetracyclines by use of chemometrics. *Anal. Methods* 2010, *2*, 1302–1309. [CrossRef]
- Zheng, Z.; Tharmalingam, N.; Liu, Q.; Jayamani, E.; Kim, W.; Fuchs, B.B.; Zhang, R.; Vilcinskas, A.; Mylonakis, E. Synergistic efficacy of Aedes aegypti antimicrobial peptide cecropin A2 and tetracycline against Pseudomonas aeruginosa. *Antimicrob. Agents Chemother.* 2017, *61*, e00686-17. [CrossRef]
- Pai, H.; Kim, J.-W.; Kim, J.; Lee, J.H.; Choe, K.W.; Gotoh, N. Carbapenem resistance mechanisms in Pseudomonas aeruginosa clinical isolates. *Antimicrob. Agents Chemother.* 2001, 45, 480–484. [CrossRef]
- Gutierrez, D.; Vandenheuvel, D.; Martinez, B.; Rodriguez, A.; Lavigne, R.; Garcia, P. Two Phages, phiIPLA-RODI and phiIPLA-C1C, Lyse Mono- and Dual-Species Staphylococcal Biofilms. *Appl. Environ. Microbiol.* 2015, *81*, 3336–3348. [CrossRef]
- Sillankorva, S.; Neubauer, P.; Azeredo, J. Phage control of dual species biofilms of Pseudomonas fluorescens and Staphylococcus lentus. *Biofouling* 2010, 26, 567–575. [CrossRef]
- McNerney, R.; Wilson, S.; Sidhu, A.; Harley, V.; Al Suwaidi, Z.; Nye, P.; Parish, T.; Stoker, N. Inactivation of mycobacteriophage D29 using ferrous ammonium sulphate as a tool for the detection of viable Mycobacterium smegmatis and M. tuberculosis. *Res. Microbiol.* **1998**, *149*, 487–495. [CrossRef]
- 42. O'Toole, G.A. Classic Spotlight: Plate Counting You Can Count On. J. Bacteriol. 2016, 198, 3127. [CrossRef] [PubMed]
- Mendes, J.J.; Leandro, C.; Corte-Real, S.; Barbosa, R.; Cavaco-Silva, P.; Melo-Cristino, J.; Gorski, A.; Garcia, M. Wound healing potential of topical bacteriophage therapy on diabetic cutaneous wounds. *Wound Repair Regen* 2013, 21, 595–603. [CrossRef] [PubMed]
- 44. Wolcott, R.; Costerton, J.W.; Raoult, D.; Cutler, S.J. The polymicrobial nature of biofilm infection. *Clin. Microbiol. Infect. Off. Publ. Eur. Soc. Clin. Microbiol. Infect. Dis.* **2013**, *19*, 107–112. [CrossRef]
- Antonic, V.; Stojadinovic, A.; Zhang, B.; Izadjoo, M.J.; Alavi, M. Pseudomonas aeruginosa induces pigment production and enhances virulence in a white phenotypic variant of Staphylococcus aureus. *Infect. Drug Resist.* 2013, 6, 175–186. [CrossRef]
- Lagendijk, E.L.; Validov, S.; Lamers, G.E.; de Weert, S.; Bloemberg, G.V. Genetic tools for tagging Gram-negative bacteria with mCherry for visualization in vitro and in natural habitats, biofilm and pathogenicity studies. *FEMS Microbiol. Lett.* 2010, 305, 81–90. [CrossRef]

- Chudakov, D.M.; Lukyanov, S.; Lukyanov, K.A. Fluorescent proteins as a toolkit for in vivo imaging. *Trends Biotechnol.* 2005, 23, 605–613. [CrossRef]
- Malone, C.L.; Boles, B.R.; Lauderdale, K.J.; Thoendel, M.; Kavanaugh, J.S.; Horswill, A.R. Fluorescent reporters for Staphylococcus aureus. J. Microbiol. Methods 2009, 77, 251–260. [CrossRef]
- Vickerman, M.M.; Mansfield, J.M.; Zhu, M.; Walters, K.S.; Banas, J.A. Codon-optimized fluorescent mTFP and mCherry for microscopic visualization and genetic counterselection of streptococci and enterococci. J. Microbiol. Methods 2015, 116, 15–22. [CrossRef]
- Pereira, P.M.; Veiga, H.; Jorge, A.M.; Pinho, M.G. Fluorescent reporters for studies of cellular localization of proteins in Staphylococcus aureus. *Appl. Environ. Microbiol.* 2010, 76, 4346–4353. [CrossRef]
- Maksimow, M.; Hakkila, K.; Karp, M.; Virta, M. Simultaneous detection of bacteria expressing GFP and DsRed genes with a flow cytometer. *Cytometry* 2002, 47, 243–247. [CrossRef] [PubMed]
- Steff, A.M.; Fortin, M.; Arguin, C.; Hugo, P. Detection of a decrease in green fluorescent protein fluorescence for the monitoring of cell death: An assay amenable to high-throughput screening technologies. *Cytometry* 2001, 45, 237–243. [CrossRef]
- 53. Mashburn, L.M.; Jett, A.M.; Akins, D.R.; Whiteley, M. Staphylococcus aureus serves as an iron source for Pseudomonas aeruginosa during in vivo coculture. *J. Bacteriol.* **2005**, *187*, 554–566. [CrossRef] [PubMed]
- Kong, Y.; Yang, D.; Cirillo, S.L.; Li, S.; Akin, A.; Francis, K.P.; Maloney, T.; Cirillo, J.D. Application of Fluorescent Protein Expressing Strains to Evaluation of Anti-Tuberculosis Therapeutic Efficacy In Vitro and In Vivo. *PLoS ONE* 2016, *11*, e0149972. [CrossRef] [PubMed]
- Russo, P.; Iturria, I.; Mohedano, M.L.; Caggianiello, G.; Rainieri, S.; Fiocco, D.; Angel Pardo, M.; Lopez, P.; Spano, G. Zebrafish gut colonization by mCherry-labelled lactic acid bacteria. *Appl. Microbiol. Biotechnol.* 2015, 99, 3479–3490. [CrossRef] [PubMed]
- Harcombe, W.; Bull, J. Impact of phages on two-species bacterial communities. *Appl. Environ. Microbiol.* 2005, 71, 5254–5259. [CrossRef]
- Kay, M.K.; Erwin, T.C.; McLean, R.J.; Aron, G.M. Bacteriophage ecology in Escherichia coli and Pseudomonas aeruginosa mixed-biofilm communities. *Appl. Environ. Microbiol.* 2011, 77, 821–829. [CrossRef]
- Cerca, N.; Oliveira, R.; Azeredo, J. Susceptibility of Staphylococcus epidermidis planktonic cells and biofilms to the lytic action of staphylococcus bacteriophage K. *Lett. Appl. Microbiol.* 2007, 45, 313–317. [CrossRef]
- Dvorackova, M.; Ruzicka, F.; Benesik, M.; Pantucek, R.; Dvorakova-Heroldova, M. Antimicrobial effect of commercial phage preparation Stafal(R) on biofilm and planktonic forms of methicillin-resistant Staphylococcus aureus. *Folia Microbiol.* 2019, 64, 121–126. [CrossRef]
- Harper, D.; Parracho, H.; Walker, J.; Sharp, R.; Hughes, G.; Werthén, M.; Lehman, S.; Morales, S. Bacteriophages and Biofilms. *Antibiotics* 2014, 3, 270–284. [CrossRef]
- 61. Labrie, S.J.; Samson, J.E.; Moineau, S. Bacteriophage resistance mechanisms. *Nat. Rev. Microbiol.* **2010**, *8*, 317–327. [CrossRef] [PubMed]
- Alves, D.R.; Gaudion, A.; Bean, J.; Esteban, P.P.; Arnot, T.; Harper, D.; Kot, W.; Hansen, L.H.; Enright, M.; Jenkins, A.T.A. Combined use of bacteriophage K and a novel bacteriophage to reduce Staphylococcus aureus biofilm formation. *Appl. Environ. Microbiol.* 2014, *80*, 6694–6703. [CrossRef] [PubMed]
- 63. Tait, K.; Skillman, L.C.; Sutherland, I.W. The efficacy of bacteriophage as a method of biofilm eradication. *Biofouling* **2002**, *18*, 305–311. [CrossRef]
- 64. Sharma, M.; Ryu, J.H.; Beuchat, L.R. Inactivation of Escherichia coli O157:H7 in biofilm on stainless steel by treatment with an alkaline cleaner and a bacteriophage. *J. Appl. Microbiol.* **2005**, *99*, 449–459. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/bv/4.0/).

				Phage	S	
N <u>o</u> .	Isolates ID	Pa 193	Pa 204	Pa 222	Pa 223	AB-PA01
1.	62-8241A	S	R	Ι	R	Ι
2.	48489280B	S	S	S	S	S
3.	49540983	S	S	S	S	S
4.	49669841A	S	Ι	S	S	S
5.	63-8601	S	S	S	S	S
6.	63-5497	S	S	S	S	S
7.	63-8199	S	Ι	S	S	S
8.	P04-5393	S	S	S	S	S
9.	63-7648	S	R	S	R	S
10.	63-6608	R	Ι	S	Ι	S
11.	62-8028A	Ι	R	Ι	R	Ι
12.	62-8001	Ι	R	S	S	S
13.	63-3521	S	Ι	Ι	Ι	S
14.	63-7889	S	S	S	S	S
15.	63-3624	R	R	R	R	R
16.	63-7651	S	S	S	Ι	S
17.	63-7780	S	S	S	S	S
18.	63-5405	S	R	R	R	Ι
19.	49669854	S	S	S	S	S
20.	63-7871	S	R	S	Ι	S
21.	63-7996	R	R	S	R	Ι
22.	63-8795	R	S	R	S	S
23.	63-8788	S	R	R	S	R
24.	63-9029	R	Ι	S	S	S
25.	63-2220	S	Ι	S	R	S

 Table 1: Susceptibility of P. aeruginosa isolates to phage cocktail AB-PA01 and its components

 Phages

26.	S-63-5289	S	S	Ι	S	S
27.	62-5656	Ι	Ι	S	Ι	S
28.	63-6728	S	S	S	Ι	S
29.	63-5713	S	S	S	S	S
30.	63-296P	S	R	S	R	S
31.	04-5755P	S	R	Ι	R	S
32.	63-9513	S	S	S	S	S
33.	63-6036	S	S	S	S	S
34.	8882-P	S	Ι	R	Ι	Ι
35.	8782-P	S	S	S	S	S
36.	63-6598	S	R	S	R	S
37.	P63-7786	S	R	S	R	S
38.	63-6299	S	S	S	S	S
39.	PA01GFP	S	S	S	S	S
40.	PA10145 FU	S	S	S	S	S
41.	PA15692	S	S	S	S	S

Key: S = susceptible, R = resistance, and I = intermediate

			Ph	ages	
N <u>o</u>	Bacterial ID	J-Sa 36	Sa 83	Sa 87	Cocktail
1.	SA4220 mCherry	S	S	S	S
2.	SA6538	S	S	S	S
3.	04-5755 Staph	R	Ι	S	R
4.	63-6354	S	Ι	S	S
5.	46834854	Ι	Ι	Ι	R
6.	63-2498	S	S	S	S
7.	63-6758	S	S	S	S
8.	63-6126	S	S	S	S
9.	63-169	S	S	S	S
10.	63-7647	R	R	S	S
11.	63-7148	S	S	S	S
12.	48556348	S	S	S	S
13.	62-9282	R	R	R	R
14.	60271003	S	S	S	S
15.	63-5879	S	S	S	S
16.	63-2749	Ι	S	S	S
17.	49639617	R	R	Ι	R
18.	49029122	S	S	S	S
19.	48561341	R	R	S	S
20.	60262911	S	S	S	S
21.	63-145	R	R	R	R
22.	63-2599	S	S	S	S
23.	63-3395	R	S	S	S
24.	63-5402	R	Ι	Ι	Ι
25.	63-5273	S	S	Ι	S
26.	63-5656	S	S	S	S
27.	O4-5089	R	R	S	Ι
28.	63-7334	S	S	S	S
29.	41535062	R	S	S	S
30.	90271071	S	S	S	S
31.	48392361	Ι	Ι	S	S
32.	62-8600	Ι	R	R	S
33.	62-9187	R	R	Ι	R

Table 2: Susceptibility of S. aureus isolates to phage cocktail AB-SA01 and its components

34.	63-2482	S	S	S	S	
35.	63-3485	Ι	S	S	S	
36.	63-5612	S	S	S	S	
37.	63-6352	S	S	S	S	
38.	63-6832	S	S	S	S	
39.	O4-5216	S	S	S	S	
40.	63-8782S	S	Ι	S	S	
41.	47597332	S	S	S	S	
42.	99279922	R	Ι	Ι	S	
43.	93952969	Ι	Ι	S	S	
44.	62-8701	S	S	S	S	
45.	63-2548	S	S	S	S	
46.	63-3534	S	S	S	S	
47.	63-5339	R	R	S	S	
48.	63-5222	S	S	S	S	
49.	63-5853	S	S	S	S	
50.	63-6378	S	S	S	S	
51.	63-7019	S	S	S	S	
52.	63-7252	S	S	S	S	
53.	63-6036	S	S	S	S	

Key: S = susceptible, R = resistance, and I = intermediate

			OD reading	after treatment	Difference between phage and
N <u>o</u> .	Isolates ID P		Phage treated	PBS treated	PBS treated (percentage)
	1.	SA4220mCh	0.14	0.29	51.72%
	2.	SA6538	0.08	0.96	91.66%
	3.	63-5897	0.27	0.54	50%
	4.	48561341	0.31	0.46	32.60%
	5.	63-6378	0.03	0.16	81.25%
	6.	63-7148	0.19	0.29	34.48%
	7.	63-2498	0.04	0.2	80.00%
	8.	48556348	0.06	0.13	53.85%
	9.	48556348	0.04	0.06	33.33%
	10.	63-6758	0.04	0.15	73.33%
	11.	60271003	0.1	0.19	47.36%
	12.	63-7647	0.26	0.35	25.71%
	13.	63-2482	0.08	0.21	61.90%
	14.	41535062	0.07	0.16	56.25%
	15.	63-2599Staph	0.21	0.95	77.89%
	16.	63-6036	0.07	0.25	72.00%
	17.	63-5656	0.09	0.37	75.68%
	18.	63952969	0.06	0.22	72.72%
	19.	63-3534	0.06	0.19	68.42%
	20.	47597332	0.06	0.19	68.50%
	21.	62-8600	0.06	0.27	77.78%
	22.	48392361	0.73	0.96	23.96%
	23.	62-8701	0.08	0.26	69.23%
	24.	63-5339	0.06	0.2	70%
	25.	62-5222	0.05	0.22	77.27%
					102

 Table 3: Effect of phage cocktail AB-SA01 on S. aureus biofilm biomass

26.	63-2548	0.06	0.19	68.42%
27.	63-5612	0.06	0.21	71.43%
28.	63-7019	0.06	0.26	76.92%
29.	63-5853	0.05	0.21	76.19%
30.	63-6832	0.16	0.47	65.96%
31.	63-5273	0.74	1.82	59.34%
32.	63-169	0.13	0.34	61.76%
33.	60262911	0.09	0.45	80.00%
34.	99279922	0.22	0.26	15.38%
35.	Sa04-5089	0.15	0.19	21.05%
36.	90271071	0.09	0.13	30.77%
37.	63-7334	0.06	0.13	53.84%
38.	63-5402	0.16	0.18	11.11%
39.	63-145	0.13	0.14	7.14%
40.	49639617	0.18	0.2	10%
41.	62-9187	0.14	0.14	0.00%
42.	Sa04-5216	0.07	0.24	70.83%
43.	63-3395	0.07	0.24	70.83%
44.	63-3485	0.1	0.2	50.00%
45.	63-6352	0.99	0.15	P84.85%
46.	63-8782S	0.13	0.23	43.48%
47.	63-7252	0.07	0.28	75%
48.	63-2749	0.15	0.21	28.57%
49.	62-9282	0.22	0.25	12%
50.	46834854	0.2	0.22	9.09%
51.	Sa04-5755	0.16	0.16	0.00%
52.	63-6354	0.31	0.39	+20.5%
53.	63-6126	0.39	0.48	+18.75%

N <u>o</u> .	ID	Phage Treated	Not Phage treated	Difference in %
1.	63-5405	1.24	1.29	3.88%
2.	63-8601	0.16	0.17	5.88%
3.	62-2220	1.1	1.29	14.73%
4.	63-8788	2.29	2.75	16.73%
5.	63-9029	0.9	1.18	23.73%
6.	63-7651	1.45	1.99	27.14%
7.	48489280B	1.2	1.7	29.41%
8.	62-8241A	0.91	1.42	35.92%
9.	63-8795	1.98	3.14	36.94%
10.	63-7786	1.44	2.34	38.46%
11.	63-7871	2.26	3.7	38.92%
12.	63-5656	1.13	2.14	47.20%
13.	49540983	0.85	1.67	49.10%
14.	63-6608	1.15	2.36	51.27%
15.	PA15692	0.16	0.33	51.51%
16.	49669854	0.93	1.92	51.56%
17.	62-8001	0.68	1.42	52.11%
18.	63-7996	0.86	1.83	53.01%
19.	PAO1 GFP	0.24	0.55	56.36%
20.	63-6299	0.97	2.36	58.90%
21.	63-6728	0.96	2.41	60.02%
22.	63-3521	0.33	0.85	61.18%
23.	PA04-5755	0.72	2.1	65.71%
24.	PA04-5393	0.63	2	68.50%
25.	63-5713	0.52	1.81	71.27%
26.	PA10145FU	0.44	1.58	72.15%

Table 4: Effect of phage cocktail AB-PA01 on P. aeruginosa biofilm biomass

27.	63-296 P	0.5	1.83	72.68%
28.	63-6036	0.49	1.81	72.78%
29.	63-9513	0.36	1.33	72.93%
30.	63-8199	0.33	1.29	74.42%
31.	49669841A	0.22	0.99	77.78%
32.	63-7889	0.17	0.78	78.21%
33.	63-5497	0.34	1.76	80.68%
34.	62-8028A	0.15	0.81	81.48%
35.	63-8782P	0.33	1.83	81.97%
36.	63-7780	0.04	0.3	86.87%
37.	63-6598	0.12	1.02	88.24%
38.	63-3624	1.56	1.41	+10.64%
39.	63-8882P	1.29	1.06	+21.70%
40.	63-7648	1.68	1.33	+26.32%

	OD m	easurement a	fter:	Biofilm biomass reduction %	
	Phage	PBS	TTC		
Isolates ID	treatment	treatment	Treatment	Phage Vs PBS	TTC Vs PBS
PA10145FU	1.02	1.46	0.56	30.14	61.64
49669841A	0.03	0.39	0.03	92.27	91.50
63-8199	0.14	1.88	0.27	92.55	85.63
63-7780	0.03	0.11	0.06	72.72	45.45
49669854	0.08	0.71	0.14	88.73	80.28
63-5713	0.17	1.13	0.35	84.96	69.02
63-6299	0.09	0.45	0.14	80	31.11
63-6598	0.078	0.37	0.1047	78.38	72.97
49540983	0.219	0.8	0.26667	72.50	66.25

Table 5: Comparison of effect of phage and tetracycline (TTC) treatment on *P. aeruginosa*

 biofilm reduction

Table 6: Comparison of effect of phage and tetracycline (TTC) treatment on *S. aureus* biofilm reduction

	OD measurement after:			Biofilm biomass	reduction %
Isolates	Phage	PBS	TTC		
ID	treatment	treatment	Treatment	Phage Vs PBS	TTC Vs PBS
SA6538	0.38	0.84	0.06	54.76	78
49029122	0.04	0.1	0.08	60	20
63-7334	0.08	0.22	0.3	63.63	+36.36
41535062	0.03	0.15	0.04	80	73.33
63-3534	0.0373	0.11	0.053	63.63	54.54
60262911	0.04467	0.15	0.0627	73.33	60
47597332	0.071	0.24	0.044	70.83	83.33

Treatment group	Mice ID	Day 0	Day 3	Day 5	Day 7	Day 10
Not-infected-phage treated	3-1R	6	5.7	5.35	4.15	1.5
	3-NEM	6	5.8	4.6	4.05	0
	3-RL	6	5.7	5.65	4.4	0
	3-L	6	5.5	5.2	3.95	0
	1-1R	6	5.7	4.9	4.95	0
	1-RL	6	5.45	4.95	3.3	1.2
	2-RL	6	5.5	5	4.45	0
	1-2R	6	5.4	5.6	5.5	0
	Mean <u>+</u> err	6 <u>+</u> 0.0	5.59 <u>+</u> 0.15	5.16 <u>+</u> 0.36	4.34 <u>+</u> 0.66	0.34 <u>+</u> 0.63
Infected-PBS treated	4-NEM	6	5.85	5.65	6.75	6.2
	4-1L	6	5.8	5.75	6.3	6.1
	4-2R	6	5.6	5.7	6.15	7.15
	5-NEM	6	6	6	6.5	8.6
	5-1L	6	5.9	5.95	6.55	8.95
	5-2R	6	5.8	5.9	6.4	8.75
	10-RL	6	5.75	5.85	6.6	8.5
	Mean <u>+</u> err	6 <u>+</u> 0.0	5.81 <u>+</u> 0.12	5.83 <u>+</u> 0.13	6.46 <u>+</u> 0.2	7.75 <u>+</u> 1.23
Infected-phage treated	7-1R	6	5.95	5	4.45	0
	7-1L	6	5.95	5.5	4.9	0
	7-RL	6	5.9	4.85	4.6	0
	7-2R	6	6	5.85	4.6	0
	9-NEM	6	5.9	4.85	5.05	1.55
	9-1R	6	5.9	5.7	5.35	0
	9-1L	6	5.85	4.85	4	0
	9-RL	6	6.05	5.65	4.7	0
	Mean <u>+</u> err	6 <u>+</u> 0.0	5.94 <u>+</u> 0.06	5.28 <u>+</u> 0.43	4.71 <u>+</u> 0.41	0.19 <u>+</u> 0.55
Infected-vancomycin treated	6-1R	6	6.7	6.1	4.45	0
	6-RL	6	7	6	5.05	0
	6-2R	6	6.95	6.45	3.6	0
	8-NEM	6	7	6.65	4.2	0
	8-1L	6	7	6.85	1.8	0
	8-2R	6	7	6.4	4.7	0
	Mean <u>+</u> err	6 <u>+</u> 0.0	6.94 <u>+</u> 0.12	6.41 <u>+</u> 0.32	3.97 <u>+</u> 1.17	0 <u>+</u> 0.0

Table 7: Wound size of excisional wound of diabetic mice infected with MDR *S. aureus* isolate

 SA63-2498 and treated with phage cocktail AB-SA01 or controls.
Treatment group	Mice ID	Day 0	Day 3	Day 5	Day 7	Day 10
	1-1R	19.9	19.9	19.5	19.6	19.6
	1-RL	19.3	19.3	19.3	19.2	19.6
	1-2R	17.2	17.3	17	16.2	16.3
Not-infected-	2-RL	18.6	18	18	18.3	18.1
phage treated	3-NEM	21	20.9	20.4	20.3	19.9
	3-1R	20.4	20.3	19.9	20	19
	3-L	19.5	19.7	19.6	19.4	19.4
	3-RL	17.6	17.9	17.5	16.8	16.2
	Mean + err	19.19 <u>+</u> 1.32	19.16 <u>+</u> 1.29	18.9 <u>+</u> 1.23	18.73 <u>+</u> 1.5	18.52 <u>+</u> 1.5
	4-NEM	20.6	19.7	18.8	19.2	18.4
	4-1L	18.5	17.3	16.1	16	16.5
	4-2R	20.1	18.9	17.3	18	18.4
Infected-PBS	5-NEM	19	16.9	16.2	16.1	15.5
treated	5-1L	20.3	19.7	18.2	17.5	17.6
	5-2R	18.7	17	15.4	15.1	N/A
	10-RL	16.5	16.3	16.3	15.1	15.3
	Mean + err	19.1 <u>+</u> 1.41	17.97 <u>+</u> 1.42	16.9 <u>+</u> 1.24	16.71 <u>+</u> 1.56	16.95 <u>+</u> 1.39
	7-1R	20.6	17.7	17.6	17.5	18.4
	7-1L	22.4	19.7	19.3	19.8	19.3
	7-RL	19.3	17.7	17.2	17.4	17.3
Infected-phage	7-2R	22	20.3	19.2	18.2	17.9
treated	9-NEM	18.9	16.8	16.3	15.9	16.7
	9-1R	22.2	19.8	19.6	18.9	20
	9-1L	22	21	20.3	19.9	20.4
	9-RL	21	20.9	19.2	19.3	19.8
	Mean + err	21.05 <u>+</u> 1.35	19.24 <u>+</u> 1.61	18.59 <u>+</u> 1.38	18.36 <u>+</u> 1.38	18.73 <u>+</u> 1.35
	6-1R	21.6	19.7	18.2	17.7	18.7
	6-RL	20	18.2	17.2	17.6	17.5
Infected-	6-2R	21	20.6	19.3	18.4	18.7
vancomycin					160	16.0
	8-NEM	18.9	17.6	16.6	16.2	16.9
treated	8-NEM 8-1L	18.9 19.8	17.6 19.1	16.6 17.3	16.2 17.9	16.9
treated	8-NEM 8-1L 8-2R	18.9 19.8 18.7	17.6 19.1 16.4	16.6 17.3 16.4	16.2 17.9 16.7	16.9 17.5 16.7

Table 8: Bodyweight of excisional wound experiment mice

	Dates and BGL									
Mice ID	7-8/8/18	9-10/8/18	13-14/8/18	20-21/8/18	27-28/8/18					
1-NEM	7.5	6.2	8.8	8.7	8.9					
1-R	9.8	8.9	17.8	18.9	14.7					
1 - L	8.4	8.4	11.5	11	12.5					
1-RL	7.4	6.8	12.4	14.3	9.8					
1-2R	14.2	18.5	18.5	24	26.1					
2-NEM	18.2	20.6	25	19.6	25.4					
2-R	9.2	8.8	16.6	24.1	14.1					
2-L	8.2	11.5	Died							
2-RL	9.2	20.6	21	21.4	23.8					
2-2R	14	21.5	21	21.9	26.6					
3-NEM	7	9.5	21.6	22.6	20.5					
3-R	7.3	13.3	18	22.1	17.5					
3-L	10.5	12.5	18	22	19.3					
3-RL	8.4	15.3	19.9	21.5	22.1					
3-2R	5.8	6.6	8.8	6.8	7					
4-NEM	11.3	15.4	19.9	19.8	20					
4-R	8.5	11.6	Died							
4-L	10.4	15.1	22	22.3	20.4					
4-RL	7.2	7.2	11.5	11.8	Died					
4-2R	12.5	17.2	17.6	25.7	18.9					
5-NEM	12.4	13.4	20.3	20.9	19.5					
5-2R	10.9	13.8	11.6	14.1	8.2					
5-L	8	8.4	18	17.8	14.9					
5-RL	10.9	12	18.6	20.9	19.4					

Table 9: Fasting blood glucose level (BGL, in mmol/L) of mice from a week of post-STZ

 injection

5-2R	7	13.3	20.8	23.5	19.6
6-NEM	17.8	14.7	11.1	11.1	10.8
6-R	9.6	18.1	21.9	26.4	19.3
6-L	19.5	22.8	25.3	24.6	23.9
6-RL	18.1	20.2	11.1	19	24.1
6-2R	14.8	23.6	24.2	22.1	21.9
7-NEM	12.8	16.2	18	17.1	17.3
7-R	10.2	16	20	18.7	18.4
7-L	12.1	18.1	21.9	21.4	24.1
7-RL	12.7	19.5	19.1	20.6	20.9
7-2R	12.3	16.9	23	22.9	18
8-NEM	12.2	28.1	19.5	21	21.7
8-R	11.2	17	16.1	16.8	16.4
8-L	9.1	14	18.6	22.6	18.6
8-RL	7	8.2	7.2	7.6	7.9
8-2R	12.1	17.8	18.9	20.4	19.9
9-NEM	22.1	18	17.9	24.6	23.3
9-R	14.2	14.2	14	16.1	11.9
9-L	12.1	15.3	14.8	11.9	7.4
9-RL	9.8	16.5	21.5	17	11
10-NEM	12.8	13.6	22.6	26.6	23.9
10-R	Died				
10-L	10.1	24.4	31.4	31.9	Hi
10-RL	13.2	15.1	22.4	Hi	27.5

	Mea	an OD after tr	eatment by:	% MBR	because of:		
Bacterial ID	PBS	AB-SA01	2.5% CPC	AB-SA01	2.5% CPC		
63-5897	0.6	0.3	0.8	-50.0	+33.3		
48561341	0.5	0.3	0.7	-40.0	+40.0		
63-6378	0.2	0.1	0.3	-50.0	+50.0		
	PBS	AB-SA01	Triton x-100	AB-SA01	Triton x-100		
48556348	0.1	0.04	0.4	-60.0	+30.0		
63-6758	0.4	0.1	0.6	-75.0	-40.0		
63-2498	0.2	0.04	0.7	-80.0	+250.0		
	PBS	AB-SA01	Tetracycline	AB-SA01	Tetracycline		
63-3534	0.1	0.04	0.05	-60.0	-50.0		
60262911	0.2	0.04	0.05	-80.0	-75.0		
41535062	0.2	0.05	0.06	-75.0	-70.0		

 Table 10: Efficacy of disinfectants and tetracycline on S. aureus biofilms.

Key: - and + indicate the biofilm biomass decrease and increase, respectively, because of the treatment compared to the PBS treatment.

		% biofilm reduction										
N <u>o</u> .	Bacterial isolate	J-Sa-36	Sa-83	Sa-87	AB-SA01	Pa-93	Pa-204	Pa-222	Pa-223	AB-PA01	AB-SA01	AB-PA01
1.	S. aureus KUB7	S	S	S	S	n/a	n/a	n/a	n/a	n/a	72.84	n/a
2.	63-6538	S	S	S	S	n/a	n/a	n/a	n/a	n/a	87.66	n/a
3.	63-2498	S	S	S	S	n/a	n/a	n/a	n/a	n/a	80.00	n/a
4.	63-6565	S	S	S	S	n/a	n/a	n/a	n/a	n/a	75.68	n/a
5.	GFP PAO1	n/a	n/a	n/a	n/a	S	S	S	S	S	n/a	73.36
6.	63-6598	n/a	n/a	n/a	n/a	S	Ι	S	S	S	n/a	88.24
7.	63-5497	n/a	n/a	n/a	n/a	S	S	S	S	S	n/a	80.68
8.	63-6036	n/a	n/a	n/a	n/a	Ι	S	S	S	S	n/a	72.78

Table 3: Phage cocktail and component phages susceptibility and biofilm reduction (in percentage - %) after phage cocktail treatment of the selected bacterial isolates

Key: S = Susceptible, I = Intermediate, n/a = Not applicable

Table 12: Source and antimicrobial susceptibility of S. aureus

				Antimicrob	ial Suscep	ptibility T	est Result	t of Clinical and	Lab Strains	of S. aureus												
ID	Date collected	Site	Cefoxitin	Benyzlepeni	Oxacillin	Gentamy	Ciproflox	Induced Clindan	Erythromyci	Clindamycir	Linezolid	Daptomycin	Teicoplanin	Vancomyc	Tetracyclin	Nitrofurant	Fusidic Ad	ci Mupirocin	Rifampicin	TSM	MRSA	MDR
SA4220		Lab strain	N	S	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
SA6538		Lab strain	N	S	S	S	S	N	S	S	S	S	S	S	S	S	S		S	S	N	N
48556348	7/12/2016	RAH	N	R	S	S	S	N	S	S	S	S	s	S	s	S	S		S	s	N	N
63-7148	21/03/2017	QEH	N	R	S	S	S	N	S	S	S	S	s	S	s	S	S		S	s	N	N
63-169	8/01/2017	RAH	Р	R	R	S	R	N	R	R	S	S	S	S	R	S	S		S	S	Y	Y
63-2498	1/02/2017	GP	Р	R	R	S	R	Р	R	R	S	S	S	S	S	S	S		S	S	Y	Y
63-7647	21/03/2017	RAH	Р	R	R	S	S	Р	R	R	S	S	S	S	S	S	S		S	s	Y	Y
63-5897	5/03/2017	GP	N	R	S	S	S	N	R	R	S	S	S	S	S	S	S	S	S	s	N	N
63-6758	14/03/2017	GP	N	S	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
60271003	17/02/2017	FMC	Р	R	R	S	S	N	S	I	S	S	S	S	S	S	S	S	S	s	Y	Y
63-6354	7/03/2017	GP	Р	R	R	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	Y	Y
63-6126	5/03/2017	LMH	N	R	R	S	R	N	R	R	R	R	R	I	S	S	S	S	S	s	Y	Y
S04-5089	14/03/2017	LMH	N	S	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
04-5755S	18/04/2017	QEH	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
62-9282	8/01/2017	RAH	Р	R	R	S	S	N	R	R	S	S	S	S	S	S	S	S	S	R	Y	Y
63-2749	27/02/2017	RAH	Р	R	R	S	R	N	S	S	S	S	S	S	S	S	S	S	S	s	Y	Y
63-5612	27/02/2017	LMH	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
63-5273	27/02/2017	Country	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
63-7019	14/03/2017	RAH	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
46834854	7/12/2016	RAH	Р	R	R	S	S	Р	R	R	S	S	S	S	S	S	S	S	S	S	N	Y
63-6124	11/12/2016	RAH	N	R	S	S	S	N	R	R	S	S	S	S	S	S	S	S	S	s	N	Y
48392361	7/12/2016	GP	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
63-5853	5/03/2017	LMH	N	R	S	S	S	N	S	S	S	R	R	R	S	S	S	S	S	s	N	Y
63-8600A	8/01/2017	GP	N	S	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
90271071	7/12/2016	RAH	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
63-8782S	30/03/2017	LMH	Р	R	R	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	Y	Y
63-5339	27/02/2017	Country	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
63-5402	27/02/2017	GP	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
63-7252	21/03/2017	GP	N	R	S	S	R	N	R	S	S	S	S	S	S	S	S	S	S	S	N	Y
63-3395	9/02/2017	GP	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
62-9187	8/01/2017	GP	N	S	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
63-2548	1/02/2017	RAH	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
99279922	7/12/2016	LMH	N	R	S	S	S	P	R	R	S	S	S	S	S	S	S	S	S	S	N	Y
47597332	30/11/2016	FMC	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
63-3534	5/02/2017	RAH	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
63-2482	1/02/2017	Country	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
63-145	8/01/2017	GP	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	S	N	N
S04-5216	21/03/2017	RAH	N	S	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
60262911	12/02/2017	FMC	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
63-6832	14/03/2017	MOD	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
62-8701	8/01/2017	LMH	N	R	S	S	S	N	S	S	S	R	R	R	S	S	S	S	S	s	N	Y
93952969	12/02/2017	FMC	N	S	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
63-2599	1/02/2017	GP	N	R	S	S	S	N	S	s	S	S	s	s	s	s	s	s	S	s	N	N
63-5656	5/03/2017	GP	N	R	S	S	S	N	S	S	S	S	S	S	S	S	S	S	S	s	N	N
63-6352	//03/2017	RAH	IN N	R	5	5	5	N	5	5	5	5	S	5	S	5	5	8	5	5	IN N	IN N
63-3485	4/02/2017	GP	IN N	R	5	5	5	N	5	5	5	5	S	5	S	5	5	8	5	5	IN N	IN N
48561341	7/12/2016	GP	IN N	R	5	5	5	N	5	5	5	5	S	5	S	5	5	8	5	5	IN N	IN N
49639617	30/11/2016	FMC	IN N	5	5	5	5	N	5	5	5	5	S	5	S	5	5	8	5	5	IN N	IN N
63-5222	27/02/2017	QEH	IN N	5	5	5	5	IN N	5	5	5	5	5	5	5	5	5	5	5	5	IN N	IN
03-7334	21/03/2017	GP	IN N	5	5	5	5	IN N	5	5	5	5	5	5	5	5	5	5	5	5	IN N	IN
41535062	50/11/2016	FMC	IN N	ĸ	5	5	5	IN N	5	5	5	5	5	5	5	5	5	5	5	5	IN N	IN
03-6036	5/03/2017	LMH	IN N	ĸ	5	5	5	IN N	ĸ	ĸ	5	5	5	5	5	5	5	5	5	5	IN N	IN
49029122	7/12/2016	CD	IN N	5 D	3 6	5	5	IN N	5	5	5	3 C	3 C	5	5	5 6	5	5	3	5	IN N	- IN N
63-63/8	7/03/2017	GP GD	IN N	R. C.	5	5	0	IN N	5	5	5	3	3	5	3	3	0	3	3	5	1N N	- IN N
03-5289	27/02/2017	ur	1N	3	3	э	3	1N	3	3	3	3	3	3	3	3	3	3	3	3	1N	11

			Antimicrobial Su	sceptibility Test R	Result of Clin	ical and L	ab Strains of	P. aerugin	osa			
ID	Date collected	Site	Ticarcillin/Clavu	Piperacillin/Tazo	Ceftazidime	Cefepime	Meropenem	Amikacin	Gentamyci	Tobramycin	Ciprofloxa	Norfloxacin
PAO1 GFP		Lab strain	S	S	S	S	S	S	R	S	S	S
PA15692		Lab strain	R	R	S	S	S	S	S	S	S	S
PAO10145 FL	J	Lab strain	R	S	S	S	S	S	S	S	S	S
62-8001	11/12/2016	QEH	S	S	S	S	I	S	S	S	R	I
63-5405	27/02/2017	Modbury	R	R	S	S	S	S	S	S	S	S
63-6598	9/03/2017	QEH	S	S	S	S	S	S	S	S	S	S
63-7648	21/03/2017	RAH	R	S	S	S	S	S	S	S	S	S
63-7786	21/03/2017	RAH	S	S	S	S	S	S	S	S	S	S
63-7996	28/03/2017	RAH	S	S	S	S	S	S	S	S	R	S
63-9029	4/03/2017	GP	S	S	S	S	S	S	S	S	S	S
63-9513	21/03/2017	RAH	S	S	S	S	S	S	S	S	S	S
04-5755P	18/04/2017	QEH	S	S	S	S	S	S	S	S	S	S
62-8028A	11/12/2016	QEH	R	S	S	S	S	S	S	S	S	S
62-8241A	11/12/2016	RAH	S	S	S	S	S	S	S	S	S	S
63-8882P	30/03/2017	GP	S	S	S	S	S	S	S	S	S	S
63-7651	28/03/2017	QEH	R	R	R	R	S	S	S	S	S	S
63-8795	30/03/2017	RAH	R	R	R	R	S	S	S	S	S	S
63-8782P	30/03/2017	LMH	S	S	S	S	S	S	S	S	S	S
63-296P	18/04/2017	RAH	R	S	S	S	S	S	S	S	S	S
49669841A	9/02/2017	FMC	R	S	S	S	S	S	S	S	S	S
48489280B	7/12/2016	RAH	R	S	S	S	S	S	S	S	S	S
62-2220	1/02/2017	GP	S	S	S	S	S	S	S	S	S	S
63-5713	7/03/2017	LMH	S	S	S	S	S	S	S	S	S	S
63-6728	14/03/2017	RAH	R	S	S	S	S	S	S	S	S	S
63-7889	28/03/2017	RAH	R	S	S	S	S	S	S	S	S	S
63-3521	9/02/2017	LMH	R	S	S	S	S	S	S	S	S	S
63-5497	27/02/2017	RAH	S	S	S	S	S	S	S	S	S	S
63-6299	7/03/2017	RAH	S	S	S	S	S	S	S	S	S	S
63-3624	6/02/2017	GP	S	S	S	S	S	S	S	S	S	S
63-6608	14/03/2017	RAH	R	S	S	S	S	S	S	S	S	S
63-7780	21/03/2017	RAH	R	S	S	S	I	I	S	S	R	R
63-7871	28/03/2017	GP	S	S	S	S	S	S	S	S	S	S
63-8199	28/03/2017	LMH	S	S	S	S	S	S	S	S	S	S
63-8601	28/03/2017	GP	S	S	S	S	S	S	S	S	S	S
63-8788	30/03/2017	LMH	R	S	S	S	S	S	S	S	S	S
49540983	7/12/2016	RAH	R	R	R	R	S	S	S	S	S	S
49669854	17/02/2017	FMC	R	S	S	S	S	S	S	S	S	S
P04-5393	30/03/2017	RAH	R	R	R	R	S	S	S	S	R	I
63-5656	5/03/2017	GP	S	S	S	S	I	s	S	S	S	S

Table 12: Source and antimicrobial susceptibility of P. aeruginosa

Key: RAH - Royal Adelaide Hospital, QEH - Queen Elizabeth Hospital, FMC - Flinders Medical Centre, GP, LMH, S - susceptible, I - intermediate, R - resistant, N -negative/not, Y- yes, MRSA - methicillin resistant S. aureus, MDR - multidrug resistant

8. References < Chan-2013-Phage cocktails and the future of p2.pdf>.

- Abatangelo, V., Peressutti Bacci, N., Boncompain, C. A., Amadio, A. F., Carrasco, S., Suarez, C. A. & Morbidoni, H. R. 2017. Broad-range lytic bacteriophages that kill Staphylococcus aureus local field strains. *PLoS One*, 12, e0181671.
- Abdulamir, A. S., Jassim, S. A., Hafidh, R. R. & Bakar, F. A. 2015. The potential of bacteriophage cocktail in eliminating Methicillin-resistant Staphylococcus aureus biofilms in terms of different extracellular matrices expressed by PIA, ciaA-D and FnBPA genes. *Ann Clin Microbiol Antimicrob*, 14, 49.
- Abedon, S. T. 1990. Selection for lysis inhibition in bacteriophage. *Journal of theoretical biology*, 146, 501-511.
- Abedon, S. T. 1992. Lysis of lysis-inhibited bacteriophage T4-infected cells. *Journal of bacteriology*, 174, 8073-8080.
- Abedon, S. T. 2009. Kinetics of phage-mediated biocontrol of bacteria. *Foodborne Pathog Dis,* 6, 807-15.
- Abedon, S. T. 2012. Bacterial 'immunity' against bacteriophages. Bacteriophage, 2, 50-54.
- Abendroth, A., Kinchington, P. R. & Slobedman, B. 2010. Varicella zoster virus immune evasion strategies. *Curr Top Microbiol Immunol*, 342, 155-71.
- Abouhmad, A., Mamo, G., Dishisha, T., Amin, M. A. & Hatti-Kaul, R. 2016. T4 lysozyme fused with cellulose-binding module for antimicrobial cellulosic wound dressing materials. *J Appl Microbiol*, 121, 115-25.
- Ackermann, H.-W. & Prangishvili, D. 2012. Prokaryote viruses studied by electron microscopy. *Archives* of virology, 157, 1843-1849.
- Ackermann, H. W. 2009. Phage classification and characterization. *Methods Mol Biol*, 501, 127-40.
- Al Ayed, M. Y., Ababneh, M., Alwin Robert, A., Alzaid, A., Ahmed, R. A., Salman, A., Musallam, M. A. & Al Dawish, M. A. 2018. Common Pathogens and Antibiotic Sensitivity Profiles of Infected Diabetic Foot Ulcers in Saudi Arabia. *Int J Low Extrem Wounds*, 1534734618793557.
- Alexiadou, K. & Doupis, J. 2012. Management of diabetic foot ulcers. *Diabetes Therapy*, 3, 4.
- Alisky, J., Iczkowski, K., Rapoport, A. & Troitsky, N. 1998. Bacteriophages show promise as antimicrobial agents. *J Infect*, 36, 5-15.
- Allison, D. G. 2003. The biofilm matrix. *Biofouling*, 19, 139-150.
- Alves, D. R., Gaudion, A., Bean, J., Esteban, P. P., Arnot, T., Harper, D., Kot, W., Hansen, L. H., Enright, M. & Jenkins, A. T. A. 2014. Combined use of bacteriophage K and a novel bacteriophage to reduce Staphylococcus aureus biofilm formation. *Applied and environmental microbiology*, 80, 6694-6703.
- Alves, D. R., Perez-Esteban, P., Kot, W., Bean, J. E., Arnot, T., Hansen, L. H., Enright, M. C. & Jenkins, A. T.
 2016. A novel bacteriophage cocktail reduces and disperses Pseudomonas aeruginosa biofilms under static and flow conditions. *Microb Biotechnol*, 9, 61-74.
- Angeletti, S. 2017. Matrix assisted laser desorption time of flight mass spectrometry (MALDI-TOF MS) in clinical microbiology. *Journal of microbiological methods*, 138, 20-29.
- Antonic, V., Stojadinovic, A., Zhang, B., Izadjoo, M. J. & Alavi, M. 2013. Pseudomonas aeruginosa induces pigment production and enhances virulence in a white phenotypic variant of Staphylococcus aureus. *Infect Drug Resist*, 6, 175-86.

- Anzai, Y., Kim, H., Park, J.-Y., Wakabayashi, H. & Oyaizu, H. 2000. Phylogenetic affiliation of the pseudomonads based on 16S rRNA sequence. *International journal of systematic and evolutionary microbiology*, 50, 1563-1589.
- Armbruster, C. R., Wolter, D. J., Mishra, M., Hayden, H. S., Radey, M. C., Merrihew, G., Maccoss, M. J., Burns, J., Wozniak, D. J. & Parsek, M. R. 2016. Staphylococcus aureus protein A mediates interspecies interactions at the cell surface of Pseudomonas aeruginosa. *MBio*, 7, e00538-16.
- Armstrong, D. G., Boulton, A. J. & Bus, S. A. 2017. Diabetic foot ulcers and their recurrence. *New England Journal of Medicine*, 376, 2367-2375.
- Armstrong, D. G., Hurwitz, B. L. & Lipsky, B. A. 2015. Set Phages to Stun: Reducing the Virulence of Staphylococcus aureus in Diabetic Foot Ulcers. *Diabetes*, 64, 2701-3.
- Armstrong, D. G., Wrobel, J. & Robbins, J. M. 2007. Guest Editorial: are diabetes-related wounds and amputations worse than cancer? *Int Wound J*, 4, 286-7.
- Armstrong, D. W. & He, L. 2001. Determination of cell viability in single or mixed samples using capillary electrophoresis laser-induced fluorescence microfluidic systems. *Analytical Chemistry*, 73, 4551-4557.
- Ashong, C. N., Raheem, S. A., Hunter, A. S., Mindru, C. & Barshes, N. R. 2017. Methicillin-Resistant Staphylococcus aureus in Foot Osteomyelitis. *Surg Infect (Larchmt)*, 18, 143-148.
- Avila-Novoa, M.-G., Iñíguez-Moreno, M., Solís-Velázquez, O.-A., González-Gómez, J.-P., Guerrero-Medina, P.-J. & Gutiérrez-Lomelí, M. 2018. Biofilm Formation by Staphylococcus aureus Isolated from Food Contact Surfaces in the Dairy Industry of Jalisco, Mexico. *Journal of Food Quality*, 2018, 1-8.
- Aysert Yildiz, P., Ozdil, T., Dizbay, M., Guzel Tunccan, O. & Hizel, K. 2018. Peripheral arterial disease increases the risk of multidrug-resistant bacteria and amputation in diabetic foot infections. *Turk J Med Sci*, 48, 845-850.
- Bader, M. S. 2008. Diabetic foot infection. *Am Fam Physician*, 78, 71-79.
- Bakker, K., Apelqvist, J., Lipsky, B., Van Netten, J., Schaper, N. & Foot, I. W. G. O. T. D. 2016. The 2015 IWGDF guidance documents on prevention and management of foot problems in diabetes: development of an evidence-based global consensus. *Diabetes/metabolism research and reviews*, 32, 2-6.
- Balarjishvili, N. S., Kvachadze, L. I., Kutateladze, M. I., Meskhi, T. S., Pataridze, T. K., Berishvili, T. A. & Tevdoradze, E. S. 2015. New virulent bacteriophages active against multiresistant Pseudomonas aeruginosa strains. *Applied Biochemistry and Microbiology*, 51, 674-682.
- Baldan, R., Cigana, C., Testa, F., Bianconi, I., De Simone, M., Pellin, D., Di Serio, C., Bragonzi, A. & Cirillo, D. M. 2014. Adaptation of Pseudomonas aeruginosa in cystic fibrosis airways influences virulence of Staphylococcus aureus in vitro and murine models of co-infection. *PLoS One*, 9, e89614.
- Barbier, M. & Damron, F. H. J. P. O. 2016. Rainbow vectors for broad-range bacterial fluorescence labeling. 11, e0146827.
- Bardina, C., Spricigo, D. A., Cortés, P. & Llagostera, M. 2012. Significance of the bacteriophage treatment schedule in reducing Salmonella colonization of poultry. *Appl. Environ. Microbiol.*, 78, 6600-6607.

- Barnes, R. J., Leung, K. T., Schraft, H. & Ulanova, M. 2008. Chromosomal gfp labelling of Pseudomonas aeruginosa using a mini-Tn7 transposon: application for studies of bacteria-host interactions. *Can J Microbiol*, 54, 48-57.
- Barnini, S., Ghelardi, E., Brucculeri, V., Morici, P. & Lupetti, A. 2015. Rapid and reliable identification of Gram-negative bacteria and Gram-positive cocci by deposition of bacteria harvested from blood cultures onto the MALDI-TOF plate. *BMC microbiology*, 15, 124.
- Barreiro, J. R., Goncalves, J. L., Braga, P. a. C., Dibbern, A. G., Eberlin, M. N. & Veiga Dos Santos, M. 2017. Non-culture-based identification of mastitis-causing bacteria by MALDI-TOF mass spectrometry. *J Dairy Sci*, 100, 2928-2934.
- Barwell, N. D., Devers, M. C., Kennon, B., Hopkinson, H. E., Mcdougall, C., Young, M. J., Robertson, H. M.
 A., Stang, D., Dancer, S. J., Seaton, A. & Leese, G. P. 2017. Diabetic foot infection: Antibiotic therapy and good practice recommendations. *Int J Clin Pract*, 71.
- Bernardo, K., Pakulat, N., Macht, M., Krut, O., Seifert, H., Fleer, S., Hunger, F. & Kronke, M. 2002. Identification and discrimination of Staphylococcus aureus strains using matrix-assisted laser desorption/ionization-time of flight mass spectrometry. *Proteomics*, 2, 747-53.
- Bernasconi, O. J., Dona, V., Tinguely, R. & Endimiani, A. 2017. In vitro activity of three commercial bacteriophage cocktails against multidrug-resistant Escherichia coli and Proteus spp. strains of human and non-human origin. J Glob Antimicrob Resist, 8, 179-185.
- Berney, M., Hammes, F., Bosshard, F., Weilenmann, H. U. & Egli, T. 2007. Assessment and interpretation of bacterial viability by using the LIVE/DEAD BacLight Kit in combination with flow cytometry. *Appl Environ Microbiol*, 73, 3283-90.
- Bertesteanu, S., Triaridis, S., Stankovic, M., Lazar, V., Chifiriuc, M. C., Vlad, M. & Grigore, R. 2014. Polymicrobial wound infections: pathophysiology and current therapeutic approaches. *Int J Pharm*, 463, 119-26.
- Bessa, L. J., Fazii, P., Di Giulio, M. & Cellini, L. 2015. Bacterial isolates from infected wounds and their antibiotic susceptibility pattern: some remarks about wound infection. *International wound journal*, 12, 47-52.
- Biswas, L., Biswas, R., Schlag, M., Bertram, R. & Götz, F. 2009. Small-colony variant selection as a survival strategy for Staphylococcus aureus in the presence of Pseudomonas aeruginosa. *Applied and environmental microbiology*, 75, 6910-6912.
- Bjarnsholt, T., Ciofu, O., Molin, S., Givskov, M. & Høiby, N. 2013. Applying insights from biofilm biology to drug development [mdash] can a new approach be developed? *Nature Reviews Drug Discovery*, 12, 791-808.
- Bjarnsholt, T., Kirketerp-Møller, K., Jensen, P. Ø., Madsen, K. G., Phipps, R., Krogfelt, K., Høiby, N. & Givskov, M. 2008. Why chronic wounds will not heal: a novel hypothesis. *Wound repair and regeneration*, 16, 2-10.
- Blair, E., Emerson, J. & Tull, A. 1967. A new medium, salt mannitol plasma agar, for the isolation of Staphylococcus aureus. *American Journal of Clinical Pathology*, 47, 30-39.
- Blair, J. M., Webber, M. A., Baylay, A. J., Ogbolu, D. O. & Piddock, L. J. 2015. Molecular mechanisms of antibiotic resistance. *Nat Rev Microbiol*, 13, 42-51.
- Bobenchik, A. M., Deak, E., Hindler, J. A., Charlton, C. L. & Humphries, R. M. 2015. Performance of Vitek 2 for antimicrobial susceptibility testing of Enterobacteriaceae with Vitek 2 (2009 FDA) and 2014 CLSI breakpoints. *J Clin Microbiol*, 53, 816-23.

- Bobenchik, A. M., Deak, E., Hindler, J. A., Charlton, C. L. & Humphries, R. M. 2017. Performance of Vitek 2 for Antimicrobial Susceptibility Testing of Acinetobacter baumannii, Pseudomonas aeruginosa, and Stenotrophomonas maltophilia with Vitek 2 (2009 FDA) and CLSI M100S 26th Edition Breakpoints. J Clin Microbiol, 55, 450-456.
- Bobenchik, A. M., Hindler, J. A., Giltner, C. L., Saeki, S. & Humphries, R. M. 2014. Performance of Vitek 2 for antimicrobial susceptibility testing of Staphylococcus spp. and Enterococcus spp. *J Clin Microbiol*, 52, 392-7.
- Bock, T., Pakkenberg, B. & Buschard, K. 2005. Genetic background determines the size and structure of the endocrine pancreas. *Diabetes*, 54, 133-7.
- Born, Y., Fieseler, L., Marazzi, J., Lurz, R., Duffy, B. & Loessner, M. J. 2011. Novel virulent and broad-hostrange Erwinia amylovora bacteriophages reveal a high degree of mosaicism and a relationship to Enterobacteriaceae phages. *Applied and environmental microbiology*, **77**, 5945-5954.
- Boulton, A. J. 2008. The diabetic foot: grand overview, epidemiology and pathogenesis. *Diabetes/metabolism research and reviews*, 24.
- Boulton, A. J., Vileikyte, L., Ragnarson-Tennvall, G. & Apelqvist, J. 2005. The global burden of diabetic foot disease. *The Lancet*, 366, 1719-1724.
- Bowler, P. G., Duerden, B. I. & Armstrong, D. G. 2001. Wound microbiology and associated approaches to wound management. *Clin Microbiol Rev*, 14, 244-69.
- Boyd, E. F. & Brüssow, H. 2002. Common themes among bacteriophage-encoded virulence factors and diversity among the bacteriophages involved. *TRENDS in Microbiology*, 10, 521-529.
- Bradbury, A., Ruckley, C., Fowkes, F., Forbes, J., Gillespie, I. & Adam, D. 2005. Bypass versus angioplasty in severe ischaemia of the leg (BASIL): multicentre, randomised controlled trial. *Lancet*, 366, 1925-1934.
- Breidenstein, E. B., De La Fuente-Núñez, C. & Hancock, R. E. 2011. Pseudomonas aeruginosa: all roads lead to resistance. *Trends in microbiology*, 19, 419-426.
- Breitbart, M., Hewson, I., Felts, B., Mahaffy, J. M., Nulton, J., Salamon, P. & Rohwer, F. 2003. Metagenomic analyses of an uncultured viral community from human feces. *J Bacteriol*, 185, 6220-3.
- Brem, H., Sheehan, P. & Boulton, A. J. J. T. a. J. O. S. 2004. Protocol for treatment of diabetic foot ulcers. 187, S1-S10.
- Briandet, R., Lacroix-Gueu, P., Renault, M., Lecart, S., Meylheuc, T., Bidnenko, E., Steenkeste, K., Bellon-Fontaine, M.-N. & Fontaine-Aupart, M.-P. 2008. Fluorescence correlation spectroscopy to study diffusion and reaction of bacteriophages inside biofilms. *Applied and environmental microbiology*, 74, 2135-2143.
- Brodsky, M. H. & Nixon, M. C. 1973. Rapid method for detection of Pseudomonas aeruginosa on MacConkey agar under ultraviolet light. *Appl Microbiol*, 26, 219-20.
- Brown, D. F. & Walpole, E. 2003. Evaluation of selective and enrichment media for isolation of glycopeptide-resistant enterococci from faecal specimens. *J Antimicrob Chemother*, 51, 289-96.
- Bruins, M. J., Bloembergen, P., Ruijs, G. J. & Wolfhagen, M. J. 2004. Identification and susceptibility testing of Enterobacteriaceae and Pseudomonas aeruginosa by direct inoculation from positive BACTEC blood culture bottles into Vitek 2. *Journal of clinical microbiology*, 42, 7-11.
- Brüssow, H. 2012. What is needed for phage therapy to become a reality in Western medicine? *Virology*, 434, 138-142.

- Burcham, Z. M., Hood, J. A., Pechal, J. L., Krausz, K. L., Bose, J. L., Schmidt, C. J., Benbow, M. E. & Jordan,
 H. R. 2016. Fluorescently labeled bacteria provide insight on post-mortem microbial transmigration. *Forensic Sci Int*, 264, 63-9.
- Burmolle, M., Webb, J. S., Rao, D., Hansen, L. H., Sorensen, S. J. & Kjelleberg, S. 2006. Enhanced biofilm formation and increased resistance to antimicrobial agents and bacterial invasion are caused by synergistic interactions in multispecies biofilms. *Appl Environ Microbiol*, 72, 3916-23.
- Cairns, B. J. & Payne, R. J. 2008. Bacteriophage therapy and the mutant selection window. *Antimicrobial agents and chemotherapy*, 52, 4344-4350.
- Cairns, B. J., Timms, A. R., Jansen, V. A., Connerton, I. F. & Payne, R. J. 2009. Quantitative models of in vitro bacteriophage–host dynamics and their application to phage therapy. *PLoS Pathogens*, 5, e1000253.
- Camoez, M., Sierra, J. M., Dominguez, M. A., Ferrer-Navarro, M., Vila, J. & Roca, I. 2016. Automated categorization of methicillin-resistant Staphylococcus aureus clinical isolates into different clonal complexes by MALDI-TOF mass spectrometry. *Clin Microbiol Infect*, 22, 161.e1-161.e7.
- Canchaya, C., Fournous, G., Chibani-Chennoufi, S., Dillmann, M. L. & Brussow, H. 2003. Phage as agents of lateral gene transfer. *Curr Opin Microbiol*, 6, 417-24.
- Capparelli, R., Parlato, M., Borriello, G., Salvatore, P. & Iannelli, D. 2007. Experimental phage therapy against Staphylococcus aureus in mice. *Antimicrob Agents Chemother*, 51, 2765-73.
- Cardinal, J. W., Allan, D. J. & Cameron, D. P. 1998. Differential metabolite accumulation may be the cause of strain differences in sensitivity to streptozotocin-induced beta cell death in inbred mice. *Endocrinology*, 139, 2885-91.
- Carlton, R. M. 1999. Phage therapy: past history and future prospects. Arch Immunol Ther Exp (Warsz), 47, 267-74.
- Carnicero, A., Mansito, T. B., Roldan, J. M. & Falcon, M. A. 1990. Staphylolytic enzyme from Pseudomonas aeruginosa: characterization and immunocytochemical localization. *Arch Microbiol*, 154, 37-41.
- Cartwright, E. J., Paterson, G. K., Raven, K. E., Harrison, E. M., Gouliouris, T., Kearns, A., Pichon, B., Edwards, G., Skov, R. L., Larsen, A. R., Holmes, M. A., Parkhill, J., Peacock, S. J. & Torok, M. E. 2013. Use of Vitek 2 antimicrobial susceptibility profile to identify mecC in methicillin-resistant Staphylococcus aureus. J Clin Microbiol, 51, 2732-4.
- Casey, E., Van Sinderen, D. & Mahony, J. 2018. In Vitro Characteristics of Phages to Guide 'Real Life' Phage Therapy Suitability. *Viruses*, 10.
- Cerca, N., Oliveira, R. & Azeredo, J. 2007. Susceptibility of Staphylococcus epidermidis planktonic cells and biofilms to the lytic action of staphylococcus bacteriophage K. *Lett Appl Microbiol*, 45, 313-7.
- Ceri, H., Olson, M., Stremick, C., Read, R., Morck, D. & Buret, A. 1999. The Calgary Biofilm Device: new technology for rapid determination of antibiotic susceptibilities of bacterial biofilms. *Journal of clinical microbiology*, 37, 1771-1776.
- Cervantes-Garcia, E., Garcia-Gonzalez, R., Reyes-Torres, A., Resendiz-Albor, A. A. & Salazar-Schettino, P.
 M. 2015. Staphylococcus aureus small colony variants in diabetic foot infections. *Diabet Foot Ankle*, 6, 26431.
- Ceyssens, P. J., Noben, J. P., Ackermann, H. W., Verhaegen, J., De Vos, D., Pirnay, J. P., Merabishvili, M., Vaneechoutte, M., Chibeu, A., Volckaert, G. & Lavigne, R. 2009. Survey of Pseudomonas

aeruginosa and its phages: de novo peptide sequencing as a novel tool to assess the diversity of worldwide collected viruses. *Environ Microbiol,* 11, 1303-13.

- Chan, B. K., Abedon, S. T. & Loc-Carrillo, C. 2013. Phage cocktails and the future of phage therapy. *Future microbiology*, 8, 769-783.
- Chanishvili, N., Tediashvili, M. & Chanishvili, T. Phages and experience for their application in the former Soviet Union. IUMS Congress (Paris), 2002.
- Chapman, G. H. 1945. The significance of sodium chloride in studies of staphylococci. *Journal of bacteriology*, 50, 201.
- Chatterjee, M., Anju, C. P., Biswas, L., Anil Kumar, V., Gopi Mohan, C. & Biswas, R. 2016. Antibiotic resistance in Pseudomonas aeruginosa and alternative therapeutic options. *Int J Med Microbiol*, 306, 48-58.
- Chen, H., Zheng, C., Zhang, X., Li, J., Li, J., Zheng, L. & Huang, K. 2011. Apelin alleviates diabetesassociated endoplasmic reticulum stress in the pancreas of Akita mice. *Peptides*, 32, 1634-1639.
- Chen, S., Li, R., Cheng, C., Xu, J. Y., Jin, C., Gao, F., Wang, J., Zhang, J., Zhang, J., Wang, H., Lu, L., Xu, G. T.
 & Tian, H. 2018. Pseudomonas aeruginosa infection alters the macrophage phenotype switching process during wound healing in diabetic mice. *Cell Biol Int*, 42, 877-889.
- Chhibber, S., Bansal, S. & Kaur, S. 2015. Disrupting the mixed-species biofilm of Klebsiella pneumoniae B5055 and Pseudomonas aeruginosa PAO using bacteriophages alone or in combination with xylitol. *Microbiology*, 161, 1369-1377.
- Chhibber, S., Kaur, T. & Sandeep, K. 2013. Co-therapy using lytic bacteriophage and linezolid: effective treatment in eliminating methicillin resistant Staphylococcus aureus (MRSA) from diabetic foot infections. *PLoS One*, *8*, e56022.
- Chopin, M.-C., Chopin, A. & Bidnenko, E. 2005. Phage abortive infection in lactococci: variations on a theme. *Current opinion in microbiology*, *8*, 473-479.
- Chudakov, D. M., Lukyanov, S. & Lukyanov, K. A. 2005. Fluorescent proteins as a toolkit for in vivo imaging. *Trends Biotechnol*, 23, 605-13.
- Coenye, T. & Nelis, H. J. 2010. In vitro and in vivo model systems to study microbial biofilm formation. *Journal of Microbiological Methods*, 83, 89-105.
- Collins, E. 1955. Action of Bacteriophage on Mixed Strain Cultures: III. Strain Dominance Due to the Action of Bacteriophage and Variations in the Acid Production of Secondary Growth Bacteria. *Applied microbiology*, **3**, 137.
- Commons, R. J., Robinson, C. H., Gawler, D., Davis, J. S. & Price, R. N. 2015. High burden of diabetic foot infections in the top end of Australia: An emerging health crisis (DEFINE study). *Diabetes Res Clin Pract*, 110, 147-57.
- Cornelissen, A., Ceyssens, P. J., T'syen, J., Van Praet, H., Noben, J. P., Shaburova, O. V., Krylov, V. N., Volckaert, G. & Lavigne, R. 2011. The T7-related Pseudomonas putida phage phi15 displays virion-associated biofilm degradation properties. *PLoS One*, 6, e18597.
- Cortright, R. N., Collins, H. L., Chandler, M. P., Lemon, P. W. & Dicarlo, S. E. 1996. Diabetes reduces growth and body composition more in male than in female rats. *Physiology & behavior*, 60, 1233-1238.
- Cosgrove, S. E., Sakoulas, G., Perencevich, E. N., Schwaber, M. J., Karchmer, A. W. & Carmeli, Y. 2003. Comparison of mortality associated with methicillin-resistant and methicillin-susceptible Staphylococcus aureus bacteremia: a meta-analysis. *Clin Infect Dis*, 36, 53-9.

Courvalin, P. 2016. Why is antibiotic resistance a deadly emerging disease? *Clin Microbiol Infect*, 22, 405-7.

Cucarella, C., Solano, C., Valle, J., Amorena, B., Lasa, Í. & Penadés, J. R. 2001. Bap, a Staphylococcus aureus surface protein involved in biofilm formation. *Journal of bacteriology*, 183, 2888-2896.

D'hérelle, F. & Smith, G. H. 1922. The bacteriophage, its role in immunity, Williams & Wilkins.

- D'herelle, F. 1917. Sur un microbe invisible antagoniste des bacilles dysentériques. CR Acad. Sci. Paris, 165, 373-375.
- Dalton, T., Dowd, S. E., Wolcott, R. D., Sun, Y., Watters, C., Griswold, J. A. & Rumbaugh, K. P. 2011. An in vivo polymicrobial biofilm wound infection model to study interspecies interactions. *PloS one*, 6, e27317.
- Davidson, J. M. 1998. Animal models for wound repair. Arch Dermatol Res, 290 Suppl, S1-11.
- Davidson, J. M., Yu, F. & Opalenik, S. R. 2013. Splinting strategies to overcome confounding wound contraction in experimental animal models. *Advances in wound care*, 2, 142-148.
- Davies, E. V., James, C. E., Kukavica-Ibrulj, I., Levesque, R. C., Brockhurst, M. A. & Winstanley, C. 2016. Temperate phages enhance pathogen fitness in chronic lung infection. *The ISME journal*, 10, 2553.
- Davis, S. C., Martinez, L. & Kirsner, R. 2006. The diabetic foot: the importance of biofilms and wound bed preparation. *Current diabetes reports*, 6, 439-445.
- De Jong, N. W., Van Der Horst, T., Van Strijp, J. A. & Nijland, R. 2017. Fluorescent reporters for markerless genomic integration in Staphylococcus aureus. *Scientific reports*, **7**, 43889.
- De Mayo, T., Conget, P., Becerra-Bayona, S., Sossa, C. L., Galvis, V. & Arango-Rodriguez, M. L. 2017. The role of bone marrow mesenchymal stromal cell derivatives in skin wound healing in diabetic mice. *PLoS One*, 12, e0177533.
- De Paepe, M., Leclerc, M., Tinsley, C. R. & Petit, M. A. 2014. Bacteriophages: an underestimated role in human and animal health? *Front Cell Infect Microbiol*, 4, 39.
- Deeds, M. C., Anderson, J. M., Armstrong, A. S., Gastineau, D. A., Hiddinga, H. J., Jahangir, A., Eberhardt, N. L. & Kudva, Y. C. 2011. Single dose streptozotocin-induced diabetes: considerations for study design in islet transplantation models. *Lab Anim*, 45, 131-40.
- Dekel, Y., Glucksam, Y., Elron-Gross, I. & Margalit, R. 2009. Insights into modeling streptozotocininduced diabetes in ICR mice. *Lab Anim (NY)*, 38, 55-60.
- Deleon, S., Clinton, A., Fowler, H., Everett, J., Horswill, A. R. & Rumbaugh, K. P. 2014. Synergistic interactions of Pseudomonas aeruginosa and Staphylococcus aureus in an in vitro wound model. *Infection and immunity*, 82, 4718-4728.
- Demling, R. H. & Waterhouse, B. 2007. The increasing problem of wound bacterial burden and infection in acute and chronic soft-tissue wounds caused by methicillin-resistant Staphylococcus aureus. *J Burns Wounds*, 7, e8.
- Diemond-Hernandez, B., Solorzano-Santos, F., Leanos-Miranda, B., Peregrino-Bejarano, L. & Miranda-Novales, G. 2010. Production of icaADBC-encoded polysaccharide intercellular adhesin and therapeutic failure in pediatric patients with Staphylococcal device-related infections. *BMC Infect Dis*, 10, 68.
- Dietrich, L. E., Price-Whelan, A., Petersen, A., Whiteley, M. & Newman, D. K. 2006. The phenazine pyocyanin is a terminal signalling factor in the quorum sensing network of Pseudomonas aeruginosa. *Mol Microbiol*, 61, 1308-21.

- Ding, Q., Li, D. Q., Wang, P. H., Chu, Y. J., Meng, S. Y. & Sun, Q. 2012. [Risk factors for infections of methicillin-resistant Staphylococci in diabetic foot patients]. *Zhonghua Yi Xue Za Zhi*, 92, 228-31.
- Dinges, M. M., Orwin, P. M. & Schlievert, P. M. J. C. M. R. 2000. Exotoxins of Staphylococcus aureus. 13, 16-34.
- Dingle, T. C. & Butler-Wu, S. M. 2013. Maldi-tof mass spectrometry for microorganism identification. *Clin Lab Med*, 33, 589-609.
- Dixon, B. 2004. New dawn for phage therapy. Lancet Infect Dis, 4, 186.
- Docobo-Pérez, F., López-Rojas, R., Domínguez-Herrera, J., Jiménez-Mejias, M. E., Pichardo, C., Ibáñez-Martínez, J. & Pachón, J. 2012. Efficacy of linezolid versus a pharmacodynamically optimized vancomycin therapy in an experimental pneumonia model caused by methicillin-resistant Staphylococcus aureus. *Journal of antimicrobial chemotherapy*, 67, 1961-1967.
- Doern, G. V., Brueggemann, A. B., Perla, R., Daly, J., Halkias, D., Jones, R. N. & Saubolle, M. A. 1997. Multicenter laboratory evaluation of the bioMérieux Vitek antimicrobial susceptibility testing system with 11 antimicrobial agents versus members of the family Enterobacteriaceae and Pseudomonas aeruginosa. *Journal of clinical microbiology*, 35, 2115-2119.
- Domínguez-Herrera, J., López-Rojas, R., Smani, Y., Labrador-Herrera, G. & Pachón, J. 2016. Efficacy of ceftaroline versus vancomycin in an experimental foreign-body and systemic infection model caused by biofilm-producing methicillin-resistant Staphylococcus epidermidis. *International journal of antimicrobial agents*, 48, 661-665.
- Donlan, R. M. 2009. Preventing biofilms of clinically relevant organisms using bacteriophage. *Trends Microbiol*, 17, 66-72.
- Doolittle, M. M., Cooney, J. J. & Caldwell, D. E. 1996. Tracing the interaction of bacteriophage with bacterial biofilms using fluorescent and chromogenic probes. *J Ind Microbiol*, 16, 331-41.
- Dowd, S. E., Wolcott, R. D., Sun, Y., Mckeehan, T., Smith, E. & Rhoads, D. 2008. Polymicrobial nature of chronic diabetic foot ulcer biofilm infections determined using bacterial tag encoded FLX amplicon pyrosequencing (bTEFAP). *PloS one*, **3**, e3326.
- Drenkard, E. & Ausubel, F. M. 2002. Pseudomonas biofilm formation and antibiotic resistance are linked to phenotypic variation. *Nature*, 416, 740.
- Drilling, A., Morales, S., Jardeleza, C., Vreugde, S., Speck, P. & Wormald, P.-J. 2014. Bacteriophage reduces biofilm of Staphylococcus aureus ex vivo isolates from chronic rhinosinusitis patients. *American journal of rhinology & allergy*, 28, 3-11.
- Drilling, A. J., Ooi, M. L., Miljkovic, D., James, C., Speck, P., Vreugde, S., Clark, J. & Wormald, P. J. 2017. Long-Term Safety of Topical Bacteriophage Application to the Frontal Sinus Region. *Front Cell Infect Microbiol*, 7, 49.
- Dunn, L., Prosser, H. C., Tan, J. T., Vanags, L. Z., Ng, M. K. & Bursill, C. a. J. J. O. V. E. J. 2013. Murine model of wound healing.
- Dunyach-Remy, C., Ngba Essebe, C., Sotto, A. & Lavigne, J. P. 2016. Staphylococcus aureus Toxins and Diabetic Foot Ulcers: Role in Pathogenesis and Interest in Diagnosis. *Toxins (Basel)*, 8.
- Duplantier, A. J. & Van Hoek, M. L. 2013. The Human Cathelicidin Antimicrobial Peptide LL-37 as a Potential Treatment for Polymicrobial Infected Wounds. *Front Immunol*, 4, 143.
- Dutilh, B. E., Cassman, N., Mcnair, K., Sanchez, S. E., Silva, G. G., Boling, L., Barr, J. J., Speth, D. R., Seguritan, V. & Aziz, R. K. 2014. A highly abundant bacteriophage discovered in the unknown sequences of human faecal metagenomes. *Nature communications*, 5.

- Eckmanns, T., Oppert, M., Martin, M., Amorosa, R., Zuschneid, I., Frei, U., Rüden, H. & Weist, K. 2008. An outbreak of hospital-acquired Pseudomonas aeruginosa infection caused by contaminated bottled water in intensive care units. *Clinical Microbiology and Infection*, 14, 454-458.
- Edwards-Jones, V., Claydon, M. A., Evason, D. J., Walker, J., Fox, A. J. & Gordon, D. B. 2000. Rapid discrimination between methicillin-sensitive and methicillin-resistant Staphylococcus aureus by intact cell mass spectrometry. *J Med Microbiol*, 49, 295-300.
- Edwards, R., Vega, A., Norman, H., Ohaeri, M. C., Levi, K., Dinsdale, E., Cinek, O., Aziz, R., Mcnair, K. & Barr, J. 2019. Global phylogeography and ancient evolution of the widespread human gut virus crAssphage. *bioRxiv*, 527796.
- El Didamony, G., Askora, A. & Shehata, A. A. 2015. Isolation and characterization of T7-like lytic bacteriophages infecting multidrug resistant Pseudomonas aeruginosa isolated from Egypt. *Curr Microbiol*, 70, 786-91.
- Elbehiry, A., Al-Dubaib, M., Marzouk, E., Osman, S. & Edrees, H. 2016. Performance of MALDI biotyper compared with Vitek() 2 compact system for fast identification and discrimination of Staphylococcus species isolated from bovine mastitis. *Microbiologyopen*, 5, 1061-1070.
- Eleftheriadou, I., Tentolouris, N., Argiana, V., Jude, E. & Boulton, A. J. 2010. Methicillin-resistant Staphylococcus aureus in diabetic foot infections. *Drugs*, 70, 1785-97.
- Erez, Z., Steinberger-Levy, I., Shamir, M., Doron, S., Stokar-Avihail, A., Peleg, Y., Melamed, S., Leavitt, A., Savidor, A. & Albeck, S. 2017. Communication between viruses guides lysis–lysogeny decisions. *Nature*, 541, 488-493.
- Ertugrul, B. M., Lipsky, B. A., Ture, M. & Sakarya, S. 2017. Risk Factors for Infection with Pseudomonas aeruginosa in Diabetic Foot Infections. *J Am Podiatr Med Assoc,* 107, 483-489.
- Ertugrul, B. M., Oryasin, E., Lipsky, B. A., Willke, A. & Bozdogan, B. 2018. Virulence genes fliC, toxA and phzS are common among Pseudomonas aeruginosa isolates from diabetic foot infections. *Infect Dis (Lond)*, 50, 273-279.
- Essoh, C., Blouin, Y., Loukou, G., Cablanmian, A., Lathro, S., Kutter, E., Thien, H. V., Vergnaud, G. & Pourcel, C. 2013. The susceptibility of Pseudomonas aeruginosa strains from cystic fibrosis patients to bacteriophages. *PLoS One*, *8*, e60575.
- Estrella, L. A., Quinones, J., Henry, M., Hannah, R. M., Pope, R. K., Hamilton, T., Teneza-Mora, N., Hall, E.
 & Biswajit, B. 2016. Characterization of novel Staphylococcus aureus lytic phage and defining their combinatorial virulence using the OmniLog[®] system. *Bacteriophage*, 6, e1219440.
- Falagas, M. E., Rafailidis, P. I., Matthaiou, D. K., Virtzili, S., Nikita, D. & Michalopoulos, A. 2008. Pandrugresistant Klebsiella pneumoniae, Pseudomonas aeruginosa and Acinetobacter baumannii infections: characteristics and outcome in a series of 28 patients. *International journal of antimicrobial agents*, 32, 450-454.
- Fazli, M., Bjarnsholt, T., Kirketerp-Møller, K., Jørgensen, B., Andersen, A. S., Krogfelt, K. A., Givskov, M. & Tolker-Nielsen, T. 2009. Nonrandom distribution of Pseudomonas aeruginosa and Staphylococcus aureus in chronic wounds. *Journal of clinical microbiology*, 47, 4084-4089.
- Fihman, V., Messika, J., Hajage, D., Tournier, V., Gaudry, S., Magdoud, F., Barnaud, G., Billard-Pomares, T., Branger, C. & Dreyfuss, D. 2015. Five-year trends for ventilator-associated pneumonia: correlation between microbiological findings and antimicrobial drug consumption. *International journal of antimicrobial agents*, 46, 518-525.

- Filkins, L. M., Graber, J. A., Olson, D. G., Dolben, E. L., Lynd, L. R., Bhuju, S. & O'toole, G. A. 2015. Coculture of Staphylococcus aureus with Pseudomonas aeruginosa Drives S. aureus towards Fermentative Metabolism and Reduced Viability in a Cystic Fibrosis Model. *J Bacteriol*, 197, 2252-64.
- Fish, R., Kutter, E., Wheat, G., Blasdel, B., Kutateladze, M. & Kuhl, S. 2016. Bacteriophage treatment of intransigent diabetic toe ulcers: a case series. *Journal of Wound Care*, 25, S27-S33.
- Flemming, H.-C. & Wingender, J. 2010. The biofilm matrix. Nature Reviews Microbiology, 8, 623-633.
- Fondrie, W. E., Liang, T., Oyler, B. L., Leung, L. M., Ernst, R. K., Strickland, D. K. & Goodlett, D. R. 2018. Pathogen Identification Direct From Polymicrobial Specimens Using Membrane Glycolipids. *Sci Rep*, 8, 15857.
- Fong, S. A., Drilling, A., Morales, S., Cornet, M. E., Woodworth, B. A., Fokkens, W. J., Psaltis, A. J., Vreugde, S. & Wormald, P. J. 2017. Activity of Bacteriophages in Removing Biofilms of Pseudomonas aeruginosa Isolates from Chronic Rhinosinusitis Patients. *Front Cell Infect Microbiol*, 7, 418.
- Fong, S. A., Drilling, A. J., Ooi, M. L., Paramasivan, S., Finnie, J. W., Morales, S., Psaltis, A. J., Vreugde, S. & Wormald, P.-J. 2019. Safety and efficacy of a bacteriophage cocktail in an in vivo model of Pseudomonas aeruginosa sinusitis. *Translational Research*, 206, 41-56.
- Fournier, J. M., Bouvet, A., Mathieu, D., Nato, F., Boutonnier, A., Gerbal, R., Brunengo, P., Saulnier, C., Sagot, N., Slizewicz, B. & Et Al. 1993. New latex reagent using monoclonal antibodies to capsular polysaccharide for reliable identification of both oxacillin-susceptible and oxacillin-resistant Staphylococcus aureus. J Clin Microbiol, 31, 1342-4.
- Frimmersdorf, E., Horatzek, S., Pelnikevich, A., Wiehlmann, L. & Schomburg, D. 2010. How Pseudomonas aeruginosa adapts to various environments: a metabolomic approach. *Environmental microbiology*, 12, 1734-1747.
- Fu, W., Forster, T., Mayer, O., Curtin, J. J., Lehman, S. M. & Donlan, R. M. 2010. Bacteriophage cocktail for the prevention of biofilm formation by Pseudomonas aeruginosa on catheters in an in vitro model system. *Antimicrob Agents Chemother*, 54, 397-404.
- Fukui, T., Kawaguchi, A. T., Takekoshi, S., Miyasaka, M., Sumiyoshi, H. & Tanaka, R. 2017. Liposome-Encapsulated Hemoglobin Accelerates Skin Wound Healing in Diabetic dB/dB Mice. Artif Organs, 41, 319-326.
- Gagnaire, J., Dauwalder, O., Boisset, S., Khau, D., Freydiere, A. M., Ader, F., Bes, M., Lina, G., Tristan, A., Reverdy, M. E., Marchand, A., Geissmann, T., Benito, Y., Durand, G., Charrier, J. P., Etienne, J., Welker, M., Van Belkum, A. & Vandenesch, F. 2012. Detection of Staphylococcus aureus deltatoxin production by whole-cell MALDI-TOF mass spectrometry. *PLoS One*, 7, e40660.
- Galiano, R. D., Michaels, J. T., Dobryansky, M., Levine, J. P. & Gurtner, G. C. 2004. Quantitative and reproducible murine model of excisional wound healing. *Wound Repair Regen*, **12**, 485-92.
- Galkowska, H., Podbielska, A., Olszewski, W. L., Stelmach, E., Luczak, M., Rosinski, G. & Karnafel, W.
 2009. Epidemiology and prevalence of methicillin-resistant Staphylococcus aureus and Staphylococcus epidermidis in patients with diabetic foot ulcers: focus on the differences between species isolated from individuals with ischemic vs. neuropathic foot ulcers. *Diabetes Res Clin Pract*, 84, 187-93.
- Gama, J. A., Reis, A. M., Domingues, I., Mendes-Soares, H., Matos, A. M. & Dionisio, F. 2013. Temperate bacterial viruses as double-edged swords in bacterial warfare. *PLoS One*, *8*, e59043.

- Gardiner, B. J., Grayson, M. L. & Wood, G. M. 2013. Inducible resistance to clindamycin in Staphylococcus aureus: validation of Vitek-2 against CLSI D-test. *Pathology*, 45, 181-4.
- Gautam, V., Sharma, M., Singhal, L., Kumar, S., Kaur, P., Tiwari, R. & Ray, P. 2017. MALDI-TOF mass spectrometry: An emerging tool for unequivocal identification of non-fermenting Gram-negative bacilli. *Indian J Med Res*, 145, 665-672.
- Geredew Kifelew, L., Mitchell, J. G. & Speck, P. 2019. Mini-review: efficacy of lytic bacteriophages on multispecies biofilms. *Biofouling*, 1-10.
- Gibson, G. W., Kreuser, S. C., Riley, J. M., Rosebury-Smith, W. S., Courtney, C. L., Juneau, P. L., Hollembaek, J. M., Zhu, T., Huband, M. D. & Brammer, D. W. 2007. Development of a mouse model of induced Staphylococcus aureus infective endocarditis. *Comparative medicine*, 57, 563-569.
- Gilbert, P., Allison, D. & Mcbain, A. 2002. Biofilms in vitro and in vivo: do singular mechanisms imply cross-resistance? *Journal of Applied Microbiology*, 92.
- Gill, J. J. & Hyman, P. 2010. Phage choice, isolation, and preparation for phage therapy. *Curr Pharm Biotechnol*, 11, 2-14.
- Gjødsbøl, K., Christensen, J. J., Karlsmark, T., Jørgensen, B., Klein, B. M. & Krogfelt, K. A. 2006. Multiple bacterial species reside in chronic wounds: a longitudinal study. *International wound journal*, 3, 225-231.
- Gnanamani, A., Hariharan, P. & Paul-Satyaseela, M. 2017. Staphylococcus aureus: Overview of bacteriology, clinical diseases, epidemiology, antibiotic resistance and therapeutic approach. *Frontiers in Staphylococcus aureus*, 4-28.
- Goldufsky, J., Wood, S. J., Jayaraman, V., Majdobeh, O., Chen, L., Qin, S., Zhang, C., Dipietro, L. A. & Shafikhani, S. H. 2015. Pseudomonas aeruginosa uses T3SS to inhibit diabetic wound healing. *Wound Repair and Regeneration*, 23, 557-564.
- Golkar, Z., Bagasra, O. & Pace, D. G. 2014. Bacteriophage therapy: a potential solution for the antibiotic resistance crisis. *Journal of Infection in Developing Countries*, 8, 129-136.
- Gonzalez, S., Fernandez, L., Campelo, A. B., Gutierrez, D., Martinez, B., Rodriguez, A. & Garcia, P. 2017. The Behavior of Staphylococcus aureus Dual-Species Biofilms Treated with Bacteriophage philPLA-RODI Depends on the Accompanying Microorganism. *Appl Environ Microbiol*, 83.
- Goode, D., Allen, V. M. & Barrow, P. A. 2003. Reduction of Experimental Salmonella and Campylobacter Contamination of Chicken Skin by Application of Lytic Bacteriophages. *Applied and Environmental Microbiology*, 69, 5032-5036.
- Gorski, A., Dabrowska, K., Switala-Jelen, K., Nowaczyk, M., Weber-Dabrowska, B., Boratynski, J., Wietrzyk, J. & Opolski, A. 2003. New insights into the possible role of bacteriophages in host defense and disease. *Med Immunol*, 2, 2.
- Górski, A. & Weber-Dabrowska, B. 2005. The potential role of endogenous bacteriophages in controlling invading pathogens. *Cellular and molecular life sciences*, 62, 511-519.
- Grant, C. W., Duclos, S. K., Moran-Paul, C. M., Yahalom, B., Tirabassi, R. S., Arreaza-Rubin, G., Spain, L. M.
 & Guberski, D. L. 2012. Development of standardized insulin treatment protocols for spontaneous rodent models of type 1 diabetes. *Comparative medicine*, 62, 381-390.
- Gu, J., Liu, X., Li, Y., Han, W., Lei, L., Yang, Y., Zhao, H., Gao, Y., Song, J. & Lu, R. 2012. A method for generation phage cocktail with great therapeutic potential. *PLoS One*, *7*, e31698.

- Gu, J., Xu, W., Lei, L., Huang, J., Feng, X., Sun, C., Du, C., Zuo, J., Li, Y., Du, T., Li, L. & Han, W. 2011.
 LysGH15, a novel bacteriophage lysin, protects a murine bacteremia model efficiently against lethal methicillin-resistant Staphylococcus aureus infection. J Clin Microbiol, 49, 111-7.
- Gundogdu, A., Bolkvadze, D. & Kilic, H. 2016. In vitro Effectiveness of Commercial Bacteriophage Cocktails on Diverse Extended-Spectrum Beta-Lactamase Producing Escherichia coli Strains. *Front Microbiol*, 7, 1761.
- Guo, A., Daniels, N. A., Thuma, J., Mccall, K. D., Malgor, R. & Schwartz, F. L. 2015. Diet is critical for prolonged glycemic control after short-term insulin treatment in high-fat diet-induced type 2 diabetic male mice. *PLoS One*, 10, e0117556.
- Gupta, R. & Prasad, Y. 2011. Efficacy of polyvalent bacteriophage P-27/HP to control multidrug resistant Staphylococcus aureus associated with human infections. *Curr Microbiol*, 62, 255-60.
- Gutierrez, D., Delgado, S., Vazquez-Sanchez, D., Martinez, B., Cabo, M. L., Rodriguez, A., Herrera, J. J. & Garcia, P. 2012. Incidence of Staphylococcus aureus and analysis of associated bacterial communities on food industry surfaces. *Appl Environ Microbiol*, **78**, 8547-54.
- Gutierrez, D., Fernandez, L., Rodriguez, A. & Garcia, P. 2018. Are Phage Lytic Proteins the Secret Weapon To Kill Staphylococcus aureus? *MBio*, 9.
- Gutierrez, D., Rodriguez-Rubio, L., Martinez, B., Rodriguez, A. & Garcia, P. 2016. Bacteriophages as Weapons Against Bacterial Biofilms in the Food Industry. *Front Microbiol*, **7**, 825.
- Gutierrez, D., Vandenheuvel, D., Martinez, B., Rodriguez, A., Lavigne, R. & Garcia, P. 2015. Two Phages, phiIPLA-RODI and phiIPLA-C1C, Lyse Mono- and Dual-Species Staphylococcal Biofilms. *Appl Environ Microbiol*, 81, 3336-48.
- Ha, J., Hong, S. K., Han, G. H., Kim, M., Yong, D. & Lee, K. 2018. Same-Day Identification and Antimicrobial Susceptibility Testing of Bacteria in Positive Blood Culture Broths Using Short-Term Incubation on Solid Medium with the MicroFlex LT, Vitek-MS, and Vitek2 Systems. Annals of laboratory medicine, 38, 235-241.
- Haddad Kashani, H., Schmelcher, M., Sabzalipoor, H., Seyed Hosseini, E. & Moniri, R. 2018. Recombinant Endolysins as Potential Therapeutics against Antibiotic-Resistant Staphylococcus aureus: Current Status of Research and Novel Delivery Strategies. *Clin Microbiol Rev*, 31.
- Hall, A. R., De Vos, D., Friman, V. P., Pirnay, J. P. & Buckling, A. 2012. Effects of sequential and simultaneous applications of bacteriophages on populations of Pseudomonas aeruginosa in vitro and in wax moth larvae. *Appl Environ Microbiol*, 78, 5646-52.
- Hancock, R. E. 1998. Resistance mechanisms in Pseudomonas aeruginosa and other nonfermentative gram-negative bacteria. *Clinical Infectious Diseases*, 27, S93-S99.
- Hannigan, G. D., Meisel, J. S., Tyldsley, A. S., Zheng, Q., Hodkinson, B. P., Sanmiguel, A. J., Minot, S., Bushman, F. D. & Grice, E. A. 2015. The human skin double-stranded DNA virome: topographical and temporal diversity, genetic enrichment, and dynamic associations with the host microbiome. *MBio*, 6, e01578-15.
- Haq, I. U., Chaudhry, W. N., Akhtar, M. N., Andleeb, S. & Qadri, I. 2012. Bacteriophages and their implications on future biotechnology: a review. *Virol J*, 9, 9.
- Harbarth, S., Balkhy, H. H., Goossens, H., Jarlier, V., Kluytmans, J., Laxminarayan, R., Saam, M., Van Belkum, A. & Pittet, D. 2015. Antimicrobial resistance: one world, one fight! Antimicrobial Resistance and Infection Control, 4, 1.

- Harcombe, W. & Bull, J. 2005. Impact of phages on two-species bacterial communities. *Applied and environmental microbiology*, 71, 5254-5259.
- Harper, D., Parracho, H., Walker, J., Sharp, R., Hughes, G., Werthén, M., Lehman, S. & Morales, S. 2014. Bacteriophages and Biofilms. *Antibiotics*, **3**, 270-284.
- Harper, D. R. & Enright, M. C. 2011. Bacteriophages for the treatment of Pseudomonas aeruginosa infections. *J Appl Microbiol*, 111, 1-7.
- Harris, L. G., El-Bouri, K., Johnston, S., Rees, E., Frommelt, L., Siemssen, N., Christner, M., Davies, A. P., Rohde, H. & Mack, D. 2010. Rapid identification of staphylococci from prosthetic joint infections using MALDI-TOF mass-spectrometry. *Int J Artif Organs*, 33, 568-74.
- Hatipoglu, M., Mutluoglu, M., Turhan, V., Uzun, G., Lipsky, B. A., Sevim, E., Demiraslan, H., Eryilmaz, E.,
 Ozuguz, C. & Memis, A. 2016. Causative pathogens and antibiotic resistance in diabetic foot infections: A prospective multi-center study. *Journal of Diabetes and its Complications*, 30, 910-916.
- Hatipoglu, M., Mutluoglu, M., Uzun, G., Karabacak, E., Turhan, V. & Lipsky, B. A. 2014. The microbiologic profile of diabetic foot infections in Turkey: a 20-year systematic review: diabetic foot infections in Turkey. *Eur J Clin Microbiol Infect Dis,* 33, 871-8.
- Hayashi, K., Kojima, R. & Ito, M. 2006. Strain differences in the diabetogenic activity of streptozotocin in mice. *Biological and Pharmaceutical Bulletin*, 29, 1110-1119.
- Heilmann, S., Sneppen, K. & Krishna, S. 2012. Coexistence of phage and bacteria on the boundary of selforganized refuges. *Proc Natl Acad Sci U S A*, 109, 12828-33.
- Hendricks, K. J., Burd, T. A., Anglen, J. O., Simpson, A. W., Christensen, G. D. & Gainor, B. J. 2001. Synergy between Staphylococcus aureus and Pseudomonas aeruginosa in a rat model of complex orthopaedic wounds. *J Bone Joint Surg Am*, 83, 855-861.
- Henry, M., Lavigne, R. & Debarbieux, L. 2013. Predicting in vivo efficacy of therapeutic bacteriophages used to treat pulmonary infections. *Antimicrob Agents Chemother*, 57, 5961-8.
- Hingorani, A., Lamuraglia, G. M., Henke, P., Meissner, M. H., Loretz, L., Zinszer, K. M., Driver, V. R., Frykberg, R., Carman, T. L. & Marston, W. 2016. The management of diabetic foot: a clinical practice guideline by the Society for Vascular Surgery in collaboration with the American Podiatric Medical Association and the Society for Vascular Medicine. *Journal of vascular surgery*, 63, 3S-21S.
- Hiramatsu, K., Katayama, Y., Matsuo, M., Sasaki, T., Morimoto, Y., Sekiguchi, A. & Baba, T. 2014. Multidrug-resistant Staphylococcus aureus and future chemotherapy. *J Infect Chemother*, 20, 593-601.
- Hoerr, V., Ziebuhr, W., Kozitskaya, S., Katzowitsch, E. & Holzgrabe, U. 2007. Laser-induced fluorescencecapillary electrophoresis and fluorescence microplate reader measurement: two methods to quantify the effect of antibiotics. *Anal Chem*, **79**, **7510-8**.
- Hoffman, L. R., Déziel, E., D'argenio, D. A., Lépine, F., Emerson, J., Mcnamara, S., Gibson, R. L., Ramsey,
 B. W. & Miller, S. I. 2006. Selection for Staphylococcus aureus small-colony variants due to growth in the presence of Pseudomonas aeruginosa. *Proceedings of the National Academy of Sciences*, 103, 19890-19895.
- Høgsberg, T., Bjarnsholt, T., Thomsen, J. S. & Kirketerp-Møller, K. 2011. Success rate of split-thickness skin grafting of chronic venous leg ulcers depends on the presence of Pseudomonas aeruginosa: a retrospective study. *PLoS One*, 6, e20492.

- Høiby, N., Bjarnsholt, T., Givskov, M., Molin, S. & Ciofu, O. 2010. Antibiotic resistance of bacterial biofilms. *International journal of antimicrobial agents*, 35, 322-332.
- Holguin, A. V., Rangel, G., Clavijo, V., Prada, C., Mantilla, M., Gomez, M. C., Kutter, E., Taylor, C., Fineran,
 P. C., Barrios, A. F. & Vives, M. J. 2015. Phage PhiPan70, a Putative Temperate Phage, Controls
 Pseudomonas aeruginosa in Planktonic, Biofilm and Burn Mouse Model Assays. *Viruses*, 7, 4602-23.
- Holtfreter, S., Radcliff, F. J., Grumann, D., Read, H., Johnson, S., Monecke, S., Ritchie, S., Clow, F., Goerke,
 C. & Bröker, B. M. 2013. Characterization of a mouse-adapted Staphylococcus aureus strain. *PloS one*, 8, e71142.
- Hooton, S. P., Atterbury, R. J. & Connerton, I. F. 2011. Application of a bacteriophage cocktail to reduce Salmonella Typhimurium U288 contamination on pig skin. *Int J Food Microbiol*, 151, 157-63.
- Horstkotte, M. A., Knobloch, J. K.-M., Rohde, H., Dobinsky, S. & Mack, D. 2002. Rapid detection of methicillin resistance in coagulase-negative staphylococci with the VITEK 2 system. *Journal of clinical microbiology*, 40, 3291-3295.
- Hoyos-Mallecot, Y., Cabrera-Alvargonzalez, J. J., Miranda-Casas, C., Rojo-Martin, M. D., Liebana-Martos,
 C. & Navarro-Mari, J. M. 2014. MALDI-TOF MS, a useful instrument for differentiating metallobeta-lactamases in Enterobacteriaceae and Pseudomonas spp. *Lett Appl Microbiol*, 58, 325-9.
- Huff, W., Huff, G., Rath, N., Balog, J. & Donoghue, A. 2003. Bacteriophage treatment of a severe Escherichia coli respiratory infection in broiler chickens. *Avian diseases*, 47, 1399-1405.
- Hughes, K. A., Sutherland, I. W. & Jones, M. V. 1998. Biofilm susceptibility to bacteriophage attack: the role of phage-borne polysaccharide depolymerase. *Microbiology*, 144, 3039-3047.
- Hyman, P. & Abedon, S. T. 2010. Bacteriophage host range and bacterial resistance. *Adv Appl Microbiol*, 70, 217-48.
- Ibberson, C. B., Stacy, A., Fleming, D., Dees, J. L., Rumbaugh, K., Gilmore, M. S. & Whiteley, M. 2017. Coinfecting microorganisms dramatically alter pathogen gene essentiality during polymicrobial infection. *Nat Microbiol*, 2, 17079.
- Illgner, U., Uekoetter, A., Runge, S. & Wetz, H. H. 2013. Infections with Pseudomonas aeruginosa in Charcot arthropathy of the foot. *Foot Ankle Int*, 34, 234-7.
- Ito, M., Kondo, Y., Nakatani, A., Hayashi, K. & Naruse, A. 2001. Characterization of low dose streptozotocin-induced progressive diabetes in mice. *Environmental toxicology and pharmacology*, 9, 71-78.
- Jaffar-Bandjee, M. C., Lazdunski, A., Bally, M., Carrère, J., Chazalette, J. P. & Galabert, C. 1995. Production of elastase, exotoxin A, and alkaline protease in sputa during pulmonary exacerbation of cystic fibrosis in patients chronically infected by Pseudomonas aeruginosa. *Journal of clinical microbiology*, 33, 924-929.
- James, G. A., Swogger, E., Wolcott, R., Pulcini, E. D., Secor, P., Sestrich, J., Costerton, J. W. & Stewart, P. S. 2008. Biofilms in chronic wounds. *Wound Repair and regeneration*, 16, 37-44.
- Jault, P., Leclerc, T., Jennes, S., Pirnay, J. P., Que, Y.-A., Resch, G., Rousseau, A. F., Ravat, F., Carsin, H. & Le Floch, R. 2019. Efficacy and tolerability of a cocktail of bacteriophages to treat burn wounds infected by Pseudomonas aeruginosa (PhagoBurn): a randomised, controlled, double-blind phase 1/2 trial. *The Lancet Infectious Diseases*, 19, 35-45.
- Jayaseelan, S., Ramaswamy, D. & Dharmaraj, S. 2014. Pyocyanin: production, applications, challenges and new insights. *World Journal of Microbiology and Biotechnology*, 30, 1159-1168.

- Jensen, E. C., Schrader, H. S., Rieland, B., Thompson, T. L., Lee, K. W., Nickerson, K. W. & Kokjohn, T. A. 1998. Prevalence of Broad-Host-Range Lytic Bacteriophages of Sphaerotilus natans, Escherichia coli, andPseudomonas aeruginosa. *Applied and Environmental Microbiology*, 64, 575-580.
- Jia, J., Alaoui-El-Azher, M., Chow, M., Chambers, T. C., Baker, H. & Jin, S. 2003. c-Jun NH2-terminal kinase-mediated signaling is essential for Pseudomonas aeruginosa ExoS-induced apoptosis. *Infect Immun*, 71, 3361-70.
- Johnson, T. J., Hildreth, M. B., Gu, L., Zhou, R. & Gibbons, W. R. 2015. Testing a dual-fluorescence assay to monitor the viability of filamentous cyanobacteria. *J Microbiol Methods*, 113, 57-64.
- Juretschko, S., Labombardi, V. J., Lerner, S. A., Schreckenberger, P. C. & Group, P. a. S. 2007. Accuracies of β-lactam susceptibility test results for Pseudomonas aeruginosa with four automated systems (BD Phoenix, MicroScan WalkAway, Vitek, and Vitek 2). *Journal of clinical microbiology*, 45, 1339-1342.
- Kacaniova, M., Kluga, A., Kantor, A., Medo, J., Ziarovska, J., Puchalski, C. & Terentjeva, M. 2019. Comparison of MALDI-TOF MS Biotyper and 16S rDNA sequencing for identification of Pseudomonas species isolated from fish. *Microb Pathog*.
- Kahl, B. C. 2017. Staphylococcus aureus and Pseudomonas aeruginosa Respiratory Tract Coinfection— What Can We Learn From Animal Models? : Oxford University Press US.
- Kart, D., Tavernier, S., Van Acker, H., Nelis, H. J. & Coenye, T. 2014. Activity of disinfectants against multispecies biofilms formed by Staphylococcus aureus, Candida albicans and Pseudomonas aeruginosa. *Biofouling*, 30, 377-83.
- Kasman, L. M., Kasman, A., Westwater, C., Dolan, J., Schmidt, M. G. & Norris, J. S. 2002. Overcoming the phage replication threshold: a mathematical model with implications for phage therapy. *Journal of virology*, 76, 5557-5564.
- Kateel, R., Augustine, A. J., Prabhu, S., Ullal, S., Pai, M. & Adhikari, P. 2018. Clinical and microbiological profile of diabetic foot ulcer patients in a tertiary care hospital. *Diabetes Metab Syndr*, 12, 27-30.
- Kay, M. K., Erwin, T. C., Mclean, R. J. & Aron, G. M. 2011. Bacteriophage ecology in Escherichia coli and Pseudomonas aeruginosa mixed-biofilm communities. *Appl Environ Microbiol*, **77**, 821-9.
- Kelly, D., Mcauliffe, O., Ross, R. P. & Coffey, A. 2012. Prevention of Staphylococcus aureus biofilm formation and reduction in established biofilm density using a combination of phage K and modified derivatives. *Lett Appl Microbiol*, 54, 286-91.
- Kelly, D., Mcauliffe, O., Ross, R. P., O'mahony, J. & Coffey, A. 2011. Development of a broad-host-range phage cocktail for biocontrol. *Bioeng Bugs*, 2, 31-7.
- Kenny, J. G., Leach, S., Ana, B., Venema, G., Kok, J., Fitzgerald, G. F., Nauta, A., Alonso, J. C. & Van Sinderen, D. 2006. Characterization of the lytic–lysogenic switch of the lactococcal bacteriophage Tuc2009. *Virology*, 347, 434-446.
- Keogh, D., Tay, W. H., Ho, Y. Y., Dale, J. L., Chen, S., Umashankar, S., Williams, R. B. H., Chen, S. L., Dunny, G. M. & Kline, K. A. 2016. Enterococcal Metabolite Cues Facilitate Interspecies Niche Modulation and Polymicrobial Infection. *Cell Host Microbe*, 20, 493-503.
- Kerr, M., Barron, E., Chadwick, P., Evans, T., Kong, W. M., Rayman, G., Sutton-Smith, M., Todd, G., Young, B. & Jeffcoate, W. J. 2019. The cost of diabetic foot ulcers and amputations to the National Health Service in England. *Diabetic Medicine*, 36, 995-1002.

- Kharel, M. K., Basnet, D. B., Lee, H. C., Liou, K., Woo, J. S., Kim, B. G. & Sohng, J. K. 2004. Isolation and characterization of the tobramycin biosynthetic gene cluster from Streptomyces tenebrarius. *FEMS Microbiol Lett*, 230, 185-90.
- Kilkenny, C., Browne, W. J., Cuthill, I. C., Emerson, M. & Altman, D. G. J. P. B. 2010. Improving bioscience research reporting: the ARRIVE guidelines for reporting animal research. 8, e1000412.
- Kim, J. C. & Jeon, B. 2016. Novel adjuvant strategy to potentiate bacitracin against MDR MRSA. J Antimicrob Chemother, 71, 1260-3.
- Kim, M. S., Park, E. J., Roh, S. W. & Bae, J. W. 2011. Diversity and abundance of single-stranded DNA viruses in human feces. *Appl Environ Microbiol*, 77, 8062-70.
- Kim, S., Rahman, M., Seol, S. Y., Yoon, S. S. & Kim, J. 2012. Pseudomonas aeruginosa bacteriophage PA1O requires type IV pili for infection and shows broad bactericidal and biofilm removal activities. *Appl Environ Microbiol*, 78, 6380-5.
- Kirby, D. T., Savage, J. M. & Plotkin, B. J. 2014. Menaquinone (Vitamin K2) Enhancement of <i>Staphylococcus aureus</i> Biofilm Formation. *Journal of Biosciences and Medicines*, 02, 26-32.
- Kirisits, M. J., Prost, L., Starkey, M. & Parsek, M. R. 2005. Characterization of colony morphology variants isolated from Pseudomonas aeruginosa biofilms. *Appl Environ Microbiol*, **71**, 4809-21.
- Klimenko, A. I., Matushkin, Y. G., Kolchanov, N. A. & Lashin, S. A. 2016. Bacteriophages affect evolution of bacterial communities in spatially distributed habitats: a simulation study. *BMC Microbiol,* 16 Suppl 1, 10.
- Kluytmans, J., Van Belkum, A. & Verbrugh, H. 1997. Nasal carriage of Staphylococcus aureus: epidemiology, underlying mechanisms, and associated risks. *Clin Microbiol Rev*, 10, 505-20.
- Kong, Y., Yang, D., Cirillo, S. L., Li, S., Akin, A., Francis, K. P., Maloney, T. & Cirillo, J. D. 2016. Application of Fluorescent Protein Expressing Strains to Evaluation of Anti-Tuberculosis Therapeutic Efficacy In Vitro and In Vivo. *PLoS One*, 11, e0149972.
- Körber, A., Schmid, E., Buer, J., Klode, J., Schadendorf, D. & Dissemond, J. 2010. Bacterial colonization of chronic leg ulcers: current results compared with data 5 years ago in a specialized dermatology department. *Journal of the European Academy of Dermatology and Venereology*, 24, 1017-1025.
- Korgaonkar, A., Trivedi, U., Rumbaugh, K. P. & Whiteley, M. 2013. Community surveillance enhances Pseudomonas aeruginosa virulence during polymicrobial infection. *Proceedings of the National Academy of Sciences*, 110, 1059-1064.
- Kostakioti, M., Hadjifrangiskou, M. & Hultgren, S. J. 2013. Bacterial biofilms: development, dispersal, and therapeutic strategies in the dawn of the postantibiotic era. *Cold Spring Harb Perspect Med*, 3, a010306.
- Kouidhi, B., Zmantar, T., Hentati, H. & Bakhrouf, A. 2010. Cell surface hydrophobicity, biofilm formation, adhesives properties and molecular detection of adhesins genes in Staphylococcus aureus associated to dental caries. *Microb Pathog*, 49, 14-22.
- Kristensen, D. M., Waller, A. S., Yamada, T., Bork, P., Mushegian, A. R. & Koonin, E. V. 2013. Orthologous gene clusters and taxon signature genes for viruses of prokaryotes. *J Bacteriol*, 195, 941-50.
- Krylov, V., Shaburova, O., Krylov, S. & Pleteneva, E. 2013. A genetic approach to the development of new therapeutic phages to fight pseudomonas aeruginosa in wound infections. *Viruses*, 5, 15-53.

- Kugelberg, E., Norstrom, T., Petersen, T. K., Duvold, T., Andersson, D. I. & Hughes, D. 2005. Establishment of a superficial skin infection model in mice by using Staphylococcus aureus and Streptococcus pyogenes. *Antimicrob Agents Chemother*, 49, 3435-41.
- Kumar, A. & Ting, Y. P. 2015. Presence of Pseudomonas aeruginosa influences biofilm formation and surface protein expression of Staphylococcus aureus. *Environmental microbiology*, 17, 4459-4468.
- Kumari, S., Harjai, K. & Chhibber, S. 2009. Bacteriophage treatment of burn wound infection caused by Pseudomonas aeruginosa PAO in BALB/c mice. *Am. J. Biomed. Sci,* **1**, 385-394.
- Kumari, S., Harjai, K. & Chhibber, S. 2010. Topical treatment of Klebsiella pneumoniae B5055 induced burn wound infection in mice using natural products. *J Infect Dev Ctries*, **4**, 367-77.
- Kume, E., Fujimura, H., Matsuki, N., Ito, M., Aruga, C., Toriumi, W., Kitamura, K. & Doi, K. 2004. Hepatic changes in the acute phase of streptozotocin (SZ)-induced diabetes in mice. *Exp Toxicol Pathol*, 55, 467-80.
- Kutateladze, M. & Adamia, R. 2010. Bacteriophages as potential new therapeutics to replace or supplement antibiotics. *Trends Biotechnol*, 28, 591-5.
- Kutter, E. 2009. Phage host range and efficiency of plating. *Bacteriophages*. Springer.
- Kutter, E., De Vos, D., Gvasalia, G., Alavidze, Z., Gogokhia, L., Kuhl, S. & Abedon, S. T. 2010. Phage therapy in clinical practice: treatment of human infections. *Current pharmaceutical biotechnology*, **11**, 69-86.
- Kutter, E. & Sulakvelidze, A. 2004. Bacteriophages: biology and applications, CRC Press.
- Kutter, E. M., Kuhl, S. J. & Abedon, S. T. 2015. Re-establishing a place for phage therapy in western medicine. *Future microbiology*, 10, 685-688.
- Kwiatek, M., Mizak, L., Parasion, S., Gryko, R., Olender, A. & Niemcewicz, M. 2015. Characterization of five newly isolated bacteriophages active against Pseudomonas aeruginosa clinical strains. *Folia Microbiol (Praha)*, 60, 7-14.
- Labrie, S. J., Samson, J. E. & Moineau, S. 2010. Bacteriophage resistance mechanisms. *Nat Rev Microbiol*, 8, 317-27.
- Lagendijk, E. L., Validov, S., Lamers, G. E., De Weert, S. & Bloemberg, G. V. 2010. Genetic tools for tagging Gram-negative bacteria with mCherry for visualization in vitro and in natural habitats, biofilm and pathogenicity studies. *FEMS Microbiol Lett*, 305, 81-90.
- Lan, C.-C. E., Wu, C.-S., Huang, S.-M., Wu, I.-H. & Chen, G.-S. 2013. High-glucose environment enhanced oxidative stress and increased interleukin-8 secretion from keratinocytes: new insights into impaired diabetic wound healing. *Diabetes*, 62, 2530-2538.
- Lasch, P., Fleige, C., Stämmler, M., Layer, F., Nübel, U., Witte, W. & Werner, G. 2014. Insufficient discriminatory power of MALDI-TOF mass spectrometry for typing of Enterococcus faecium and Staphylococcus aureus isolates. *Journal of microbiological methods*, 100, 58-69.
- Lau, G. W., Hassett, D. J., Ran, H. & Kong, F. 2004. The role of pyocyanin in Pseudomonas aeruginosa infection. *Trends in molecular medicine*, 10, 599-606.
- Lavallee, C., Rouleau, D., Gaudreau, C., Roger, M., Tsimiklis, C., Locas, M. C., Gagnon, S., Delorme, J. & Labbe, A. C. 2010. Performance of an agar dilution method and a Vitek 2 card for detection of inducible clindamycin resistance in Staphylococcus spp. J Clin Microbiol, 48, 1354-7.

- Lavery, L. A., Armstrong, D. G., Murdoch, D. P., Peters, E. J. & Lipsky, B. A. 2007. Validation of the Infectious Diseases Society of America's diabetic foot infection classification system. *Clinical infectious diseases*, 44, 562-565.
- Lavery, L. A., Armstrong, D. G., Wunderlich, R. P., Mohler, M. J., Wendel, C. S. & Lipsky, B. A. 2006. Risk factors for foot infections in individuals with diabetes. *Diabetes care*, 29, 1288-1293.
- Lavery, L. A., Armstrong, D. G., Wunderlich, R. P., Tredwell, J. & Boulton, A. J. 2003. Diabetic foot syndrome: evaluating the prevalence and incidence of foot pathology in Mexican Americans and non-Hispanic whites from a diabetes disease management cohort. *Diabetes care*, 26, 1435-1438.
- Lavery, L. A., Fontaine, J. L., Bhavan, K., Kim, P. J., Williams, J. R. & Hunt, N. A. 2014. Risk factors for methicillin-resistant Staphylococcus aureus in diabetic foot infections. *Diabet Foot Ankle*, 5.
- Law, N., Logan, C., Yung, G., Furr, C.-L. L., Lehman, S. M., Morales, S., Rosas, F., Gaidamaka, A., Bilinsky, I.
 & Grint, P. 2019. Successful adjunctive use of bacteriophage therapy for treatment of multidrug-resistant Pseudomonas aeruginosa infection in a cystic fibrosis patient. *Infection*, 1-4.
- Lay, J. O. 2001. MALDI-TOF mass spectrometry of bacteria. Mass spectrometry reviews, 20, 172-194.
- Le May, C., Chu, K., Hu, M., Ortega, C. S., Simpson, E. R., Korach, K. S., Tsai, M.-J. & Mauvais-Jarvis, F. 2006. Estrogens protect pancreatic β-cells from apoptosis and prevent insulin-deficient diabetes mellitus in mice. *Proceedings of the National Academy of Sciences*, 103, 9232-9237.
- Lebowitz, D., Gariani, K., Kressmann, B., Dach, E. V., Huttner, B., Bartolone, P., Le, N., Mohamad, M., Lipsky, B. A. & Uckay, I. 2017. Are antibiotic-resistant pathogens more common in subsequent episodes of diabetic foot infection? *Int J Infect Dis*, 59, 61-64.
- Lee, J. S., Son, S. T. & Han, S.-K. 2017. Risk Factors of Methicillin-Resistant Staphylococcus Aureus and Pseudomonas Infection in Diabetic Foot Ulcers in Korea. *Journal of Wound Management and Research*, 13, 29-34.
- Lee, M., Chung, H.-S., Moon, H.-W., Lee, S. H. & Lee, K. 2015. Comparative evaluation of two matrixassisted laser desorption ionization time-of-flight mass spectrometry (MALDI-TOF MS) systems, Vitek MS and Microflex LT, for the identification of Gram-positive cocci routinely isolated in clinical microbiology laboratories. *Journal of microbiological methods*, 113, 13-15.
- Lehman, S. M., Mearns, G., Rankin, D., Cole, R. A., Smrekar, F., Branston, S. D. & Morales, S. 2019. Design and Preclinical Development of a Phage Product for the Treatment of Antibiotic-Resistant Staphylococcus aureus Infections. *Viruses*, 11.
- Leiter, E. 1985. Differential susceptibility of BALB/c sublines to diabetes induction by multi-dose streptozotocin treatment. *The BALB/c Mouse.* Springer.
- Lenzen, S. 2008. The mechanisms of alloxan-and streptozotocin-induced diabetes. *Diabetologia*, 51, 216-226.
- Lepage, P., Leclerc, M. C., Joossens, M., Mondot, S., Blottiere, H. M., Raes, J., Ehrlich, D. & Dore, J. 2013. A metagenomic insight into our gut's microbiome. *Gut*, 62, 146-58.
- Leskinen, K., Tuomala, H., Wicklund, A., Horsma-Heikkinen, J., Kuusela, P., Skurnik, M. & Kiljunen, S. 2017. Characterization of vB_SauM-fRuSau02, a Twort-Like Bacteriophage Isolated from a Therapeutic Phage Cocktail. *Viruses*, 9.
- Levin, B. R. & Antia, R. 2001. Why we don't get sick: the within-host population dynamics of bacterial infections. *Science*, 292, 1112-5.
- Levin, B. R., Moineau, S., Bushman, M. & Barrangou, R. 2013. The population and evolutionary dynamics of phage and bacteria with CRISPR-mediated immunity. *PLoS Genet*, 9, e1003312.

- Li, X.-Z., Livermore, D. M. & Nikaido, H. 1994. Role of efflux pump (s) in intrinsic resistance of Pseudomonas aeruginosa: resistance to tetracycline, chloramphenicol, and norfloxacin. *Antimicrobial agents and chemotherapy*, 38, 1732-1741.
- Li, X.-Z., Zhang, L. & Poole, K. 2000. Interplay between the MexA-MexB-OprM multidrug efflux system and the outer membrane barrier in the multiple antibiotic resistance of Pseudomonas aeruginosa. *Journal of Antimicrobial Chemotherapy*, 45, 433-436.
- Liao, K. S., Lehman, S. M., Tweardy, D. J., Donlan, R. M. & Trautner, B. W. 2012. Bacteriophages are synergistic with bacterial interference for the prevention of Pseudomonas aeruginosa biofilm formation on urinary catheters. *J Appl Microbiol*, 113, 1530-9.
- Ligozzi, M., Bernini, C., Bonora, M. G., De Fatima, M., Zuliani, J. & Fontana, R. 2002. Evaluation of the VITEK 2 system for identification and antimicrobial susceptibility testing of medically relevant gram-positive cocci. *Journal of clinical microbiology*, 40, 1681-1686.
- Lin, T. Y., Lo, Y. H., Tseng, P. W., Chang, S. F., Lin, Y. T. & Chen, T. S. 2012. A T3 and T7 recombinant phage acquires efficient adsorption and a broader host range. *PLoS One*, *7*, e30954.
- Lindblad, W. J. 2008. Considerations for selecting the correct animal model for dermal wound-healing studies. *J Biomater Sci Polym Ed*, 19, 1087-96.
- Lipsky, B., Peters, E., Senneville, E., Berendt, A., Embil, J., Lavery, L., Urbančič-Rovan, V. & Jeffcoate, W. 2012a. Expert opinion on the management of infections in the diabetic foot. *Diabetes/metabolism research and reviews*, 28, 163-178.
- Lipsky, B. A. 1999. Evidence-based antibiotic therapy of diabetic foot infections. *FEMS immunology & medical microbiology*, 26, 267-276.
- Lipsky, B. A., Aragon-Sanchez, J., Diggle, M., Embil, J., Kono, S., Lavery, L., Senneville, E., Urbancic-Rovan,
 V., Van Asten, S. & Peters, E. J. 2016. IWGDF guidance on the diagnosis and management of foot infections in persons with diabetes. *Diabetes Metab Res Rev*, 32 Suppl 1, 45-74.
- Lipsky, B. A., Berendt, A. R., Cornia, P. B., Pile, J. C., Peters, E. J., Armstrong, D. G., Deery, H. G., Embil, J. M., Joseph, W. S., Karchmer, A. W., Pinzur, M. S. & Senneville, E. 2012b. 2012 Infectious Diseases Society of America clinical practice guideline for the diagnosis and treatment of diabetic foot infections. *Clin Infect Dis*, 54, e132-73.
- Lipsky, B. A., Berendt, A. R., Deery, H. G., Embil, J. M., Joseph, W. S., Karchmer, A. W., Lefrock, J. L., Lew, D. P., Mader, J. T. & Norden, C. 2004. Diagnosis and treatment of diabetic foot infections. *Clinical Infectious Diseases*, 39, 885-910.
- Lipsky, B. A. & Hoey, C. 2009. Topical antimicrobial therapy for treating chronic wounds. *Clin Infect Dis,* 49, 1541-9.
- Lipsky, B. A., Itani, K. M., Weigelt, J. A., Joseph, W., Paap, C. M., Reisman, A., Myers, D. E. & Huang, D. B. 2011. The role of diabetes mellitus in the treatment of skin and skin structure infections caused by methicillin-resistant Staphylococcus aureus: results from three randomized controlled trials. *Int J Infect Dis*, 15, e140-6.
- Lipsky, B. A., Tabak, Y., Johannes, R., Vo, L., Hyde, L. & Weigelt, J. 2010. Skin and soft tissue infections in hospitalised patients with diabetes: culture isolates and risk factors associated with mortality, length of stay and cost. *Diabetologia*, 53, 914-923.
- Lister, P. D., Wolter, D. J. & Hanson, N. D. 2009. Antibacterial-resistant Pseudomonas aeruginosa: clinical impact and complex regulation of chromosomally encoded resistance mechanisms. *Clinical microbiology reviews*, 22, 582-610.

Loc-Carrillo, C. & Abedon, S. T. 2011. Pros and cons of phage therapy. Bacteriophage, 1, 111-114.

- Loesche, M., Gardner, S. E., Kalan, L., Horwinski, J., Zheng, Q., Hodkinson, B. P., Tyldsley, A. S., Franciscus, C. L., Hillis, S. L., Mehta, S., Margolis, D. J. & Grice, E. A. 2017. Temporal Stability in Chronic Wound Microbiota Is Associated With Poor Healing. *J Invest Dermatol*, 137, 237-244.
- Lowy, F. D. 1998. Staphylococcus aureus infections. N Engl J Med, 339, 520-32.
- Ly-Chatain, M. H. 2014. The factors affecting effectiveness of treatment in phages therapy. *Front Microbiol*, 5, 51.
- Machan, Z. A., Pitt, T. L., White, W., Watson, D., Taylor, G. W., Cole, P. J. & Wilson, R. 1991. Interaction between Pseudomonas aeruginosa and Staphylococcus aureus: description of an anti-staphylococcal substance. *J Med Microbiol*, 34, 213-7.
- Macia, M. D., Rojo-Molinero, E. & Oliver, A. 2014. Antimicrobial susceptibility testing in biofilm-growing bacteria. *Clin Microbiol Infect*, 20, 981-90.
- Madsen, S. M., Westh, H., Danielsen, L. & Rosdahl, V. T. 1996. Bacterial colonization and healing of venous leg ulcers. *Apmis*, 104, 895-9.
- Mah, T.-F., Pitts, B., Pellock, B., Walker, G. C., Stewart, P. S. & O'toole, G. A. 2003. A genetic basis for Pseudomonas aeruginosa biofilm antibiotic resistance. *Nature*, 426, 306.
- Maksimow, M., Hakkila, K., Karp, M. & Virta, M. 2002. Simultaneous detection of bacteria expressing GFP and DsRed genes with a flow cytometer. *Cytometry*, 47, 243-7.
- Malachowa, N., Kobayashi, S. D., Braughton, K. R. & Deleo, F. R. 2013. Mouse model of Staphylococcus aureus skin infection. *Mouse Models of Innate Immunity*. Springer.
- Malik, A., Mohammad, Z. & Ahmad, J. 2013. The diabetic foot infections: biofilms and antimicrobial resistance. *Diabetes & Metabolic Syndrome: Clinical Research & Reviews*, 7, 101-107.
- Malki, K., Kula, A., Bruder, K., Sible, E., Hatzopoulos, T., Steidel, S., Watkins, S. C. & Putonti, C. 2015. Bacteriophages isolated from Lake Michigan demonstrate broad host-range across several bacterial phyla. *Virology Journal*, 12.
- Malone, C. L., Boles, B. R., Lauderdale, K. J., Thoendel, M., Kavanaugh, J. S. & Horswill, A. R. 2009. Fluorescent reporters for Staphylococcus aureus. *J Microbiol Methods*, **77**, 251-60.
- Malone, L. M., Warring, S. L., Jackson, S. A., Warnecke, C., Gardner, P. P., Gumy, L. F. & Fineran, P. C.
 2020. A jumbo phage that forms a nucleus-like structure evades CRISPR–Cas DNA targeting but is vulnerable to type III RNA-based immunity. *Nature Microbiology*, 5, 48-55.
- Manohar, P., Tamhankar, A. J., Lundborg, C. S. & Nachimuthu, R. 2019. Therapeutic Characterization and Efficacy of Bacteriophage Cocktails Infecting Escherichia coli, Klebsiella pneumoniae, and Enterobacter Species. *Front Microbiol*, 10, 574.
- Marko, D. C., Saffert, R. T., Cunningham, S. A., Hyman, J., Walsh, J., Arbefeville, S., Howard, W., Pruessner, J., Safwat, N. & Cockerill, F. R. 2012. Evaluation of the Bruker Biotyper and Vitek MS matrix-assisted laser desorption ionization—time of flight mass spectrometry systems for identification of nonfermenting Gram-negative bacilli isolated from cultures from cystic fibrosis patients. *Journal of clinical microbiology*, 50, 2034-2039.
- Mashburn, L. M., Jett, A. M., Akins, D. R. & Whiteley, M. 2005. Staphylococcus aureus serves as an iron source for Pseudomonas aeruginosa during in vivo coculture. *Journal of bacteriology*, 187, 554-566.

- Mastropaolo, M. D., Evans, N. P., Byrnes, M. K., Stevens, A. M., Robertson, J. L. & Melville, S. B. 2005. Synergy in polymicrobial infections in a mouse model of type 2 diabetes. *Infect Immun*, 73, 6055-63.
- Matic, I., Radman, M., Taddei, F., Picard, B., Doit, C., Bingen, E., Denamur, E. & Elion, J. 1997. Highly variable mutation rates in commensal and pathogenic Escherichia coli. *Science*, 277, 1833-4.
- Matz, M. V., Fradkov, A. F., Labas, Y. A., Savitsky, A. P., Zaraisky, A. G., Markelov, M. L. & Lukyanov, S. A. 1999. Fluorescent proteins from nonbioluminescent Anthozoa species. *Nat Biotechnol*, 17, 969-73.
- Maura, D. & Debarbieux, L. 2011. Bacteriophages as twenty-first century antibacterial tools for food and medicine. *Appl Microbiol Biotechnol*, 90, 851-9.
- Maura, D. & Debarbieux, L. 2012. On the interactions between virulent bacteriophages and bacteria in the gut. *Bacteriophage*, 2, 229-233.
- Mcnerney, R., Wilson, S., Sidhu, A., Harley, V., Al Suwaidi, Z., Nye, P., Parish, T. & Stoker, N. 1998. Inactivation of mycobacteriophage D29 using ferrous ammonium sulphate as a tool for the detection of viable Mycobacterium smegmatis and M. tuberculosis. *Research in microbiology*, 149, 487-495.
- Mcvay, C. S., Velasquez, M. & Fralick, J. A. 2007. Phage therapy of Pseudomonas aeruginosa infection in a mouse burn wound model. *Antimicrob Agents Chemother*, 51, 1934-8.
- Mendes, J. J., Leandro, C., Corte-Real, S., Barbosa, R., Cavaco-Silva, P., Melo-Cristino, J., Gorski, A. & Garcia, M. 2013. Wound healing potential of topical bacteriophage therapy on diabetic cutaneous wounds. *Wound Repair Regen*, 21, 595-603.
- Mendes, J. J., Leandro, C., Mottola, C., Barbosa, R., Silva, F. A., Oliveira, M., Vilela, C. L., Melo-Cristino, J., Gorski, A., Pimentel, M., Sao-Jose, C., Cavaco-Silva, P. & Garcia, M. 2014. In vitro design of a novel lytic bacteriophage cocktail with therapeutic potential against organisms causing diabetic foot infections. J Med Microbiol, 63, 1055-65.
- Mendes, J. J., Leandro, C. I., Bonaparte, D. P. & Pinto, A. L. 2012a. A rat model of diabetic wound infection for the evaluation of topical antimicrobial therapies. *Comp Med*, 62, 37-48.
- Mendes, J. J., Marques-Costa, A., Vilela, C., Neves, J., Candeias, N., Cavaco-Silva, P. & Melo-Cristino, J. 2012b. Clinical and bacteriological survey of diabetic foot infections in Lisbon. *Diabetes Res Clin Pract*, 95, 153-61.
- Mendoza, S. D., Nieweglowska, E. S., Govindarajan, S., Leon, L. M., Berry, J. D., Tiwari, A., Chaikeeratisak,
 V., Pogliano, J., Agard, D. A. & Bondy-Denomy, J. 2020. A bacteriophage nucleus-like compartment shields DNA from CRISPR nucleases. *Nature*, 577, 244-248.
- Merabishvili, M., Pirnay, J. P., Verbeken, G., Chanishvili, N., Tediashvili, M., Lashkhi, N., Glonti, T., Krylov, V., Mast, J., Van Parys, L., Lavigne, R., Volckaert, G., Mattheus, W., Verween, G., De Corte, P., Rose, T., Jennes, S., Zizi, M., De Vos, D. & Vaneechoutte, M. 2009. Quality-controlled small-scale production of a well-defined bacteriophage cocktail for use in human clinical trials. *PLoS One*, 4, e4944.
- Mesaros, N., Nordmann, P., Plesiat, P., Roussel-Delvallez, M., Van Eldere, J., Glupczynski, Y., Van Laethem, Y., Jacobs, F., Lebecque, P., Malfroot, A., Tulkens, P. M. & Van Bambeke, F. 2007. Pseudomonas aeruginosa: resistance and therapeutic options at the turn of the new millennium. *Clin Microbiol Infect*, 13, 560-78.

- Messad, N., Prajsnar, T. K., Lina, G., O'callaghan, D., Foster, S. J., Renshaw, S. A., Skaar, E. P., Bes, M., Dunyach-Remy, C., Vandenesch, F., Sotto, A. & Lavigne, J. P. 2015. Existence of a Colonizing Staphylococcus aureus Strain Isolated in Diabetic Foot Ulcers. *Diabetes*, 64, 2991-5.
- Michelsen, C. F., Christensen, A. M., Bojer, M. S., Hoiby, N., Ingmer, H. & Jelsbak, L. 2014. Staphylococcus aureus alters growth activity, autolysis, and antibiotic tolerance in a human host-adapted Pseudomonas aeruginosa lineage. *J Bacteriol*, 196, 3903-11.
- Mills, S., Shanahan, F., Stanton, C., Hill, C., Coffey, A. & Ross, R. P. 2013. Movers and shakers: influence of bacteriophages in shaping the mammalian gut microbiota. *Gut Microbes*, 4, 4-16.
- Minot, S., Sinha, R., Chen, J., Li, H., Keilbaugh, S. A., Wu, G. D., Lewis, J. D. & Bushman, F. D. 2011. The human gut virome: inter-individual variation and dynamic response to diet. *Genome Res*, 21, 1616-25.
- Mirzaei, M. K. & Nilsson, A. S. 2015. Isolation of phages for phage therapy: a comparison of spot tests and efficiency of plating analyses for determination of host range and efficacy. *PLoS One*, 10, e0118557.
- Mitchell, G., Séguin, D. L., Asselin, A.-E., Déziel, E., Cantin, A. M., Frost, E. H., Michaud, S. & Malouin, F. 2010. Staphylococcus aureus sigma B-dependent emergence of small-colony variants and biofilm production following exposure to Pseudomonas aeruginosa 4-hydroxy-2heptylquinoline-N-oxide. *BMC microbiology*, 10, 33.
- Moisan, H., Brouillette, E., Jacob, C. L., Langlois-Begin, P., Michaud, S. & Malouin, F. 2006. Transcription of virulence factors in Staphylococcus aureus small-colony variants isolated from cystic fibrosis patients is influenced by SigB. *J Bacteriol*, 188, 64-76.
- Monk, A., Rees, C., Barrow, P., Hagens, S. & Harper, D. 2010. Bacteriophage applications: where are we now? *Letters in Applied Microbiology*, **51**, 363-369.
- Morello, E., Saussereau, E., Maura, D., Huerre, M., Touqui, L. & Debarbieux, L. 2011. Pulmonary bacteriophage therapy on Pseudomonas aeruginosa cystic fibrosis strains: first steps towards treatment and prevention. *PLoS One*, *6*, e16963.
- Motlagh, A. M., Bhattacharjee, A. S. & Goel, R. 2016. Biofilm control with natural and geneticallymodified phages. *World J Microbiol Biotechnol*, 32, 67.
- Mottola, C., Mendes, J. J., Cristino, J. M., Cavaco-Silva, P., Tavares, L. & Oliveira, M. 2016. Polymicrobial biofilms by diabetic foot clinical isolates. *Folia Microbiol (Praha)*, 61, 35-43.
- Mulcahy, M. E. & Mcloughlin, R. M. 2016. Host-Bacterial Crosstalk Determines Staphylococcus aureus Nasal Colonization. *Trends Microbiol*, 24, 872-886.
- Murali, T. S., Kavitha, S., Spoorthi, J., Bhat, D. V., Prasad, A. S., Upton, Z., Ramachandra, L., Acharya, R. V.
 & Satyamoorthy, K. 2014. Characteristics of microbial drug resistance and its correlates in chronic diabetic foot ulcer infections. *J Med Microbiol*, 63, 1377-85.
- Murray, P. R., Rosenthal, K. S. & Pfaller, M. A. 2015. *Medical microbiology*, Elsevier Health Sciences.
- Mustafa, M. H., Chalhoub, H., Denis, O., Deplano, A., Vergison, A., Rodriguez-Villalobos, H., Tunney, M.
 M., Elborn, J. S., Kahl, B. C., Traore, H., Vanderbist, F., Tulkens, P. M. & Van Bambeke, F. 2016.
 Antimicrobial Susceptibility of Pseudomonas aeruginosa Isolated from Cystic Fibrosis Patients in Northern Europe. Antimicrob Agents Chemother, 60, 6735-6741.
- Myles, I. A., Reckhow, J. D., Williams, K. W., Sastalla, I., Frank, K. M. & Datta, S. K. 2016. A method for culturing Gram-negative skin microbiota. *BMC Microbiol*, 16, 60.

- Nebe-Von-Caron, G., Stephens, P., Hewitt, C., Powell, J. & Badley, R. 2000. Analysis of bacterial function by multi-colour fluorescence flow cytometry and single cell sorting. *Journal of microbiological methods*, 42, 97-114.
- Neyra, R. C., Frisancho, J. A., Rinsky, J. L., Resnick, C., Carroll, K. C., Rule, A. M., Ross, T., You, Y., Price, L. B. & Silbergeld, E. K. 2014. Multidrug-resistant and methicillin-resistant Staphylococcus aureus (MRSA) in hog slaughter and processing plant workers and their community in North Carolina (USA). *Environ Health Perspect*, 122, 471-7.
- Nhan, T.-X., Leclercq, R., Cattoir, V. J. E. J. O. C. M. & Diseases, I. 2011. Prevalence of toxin genes in consecutive clinical isolates of Staphylococcus aureus and clinical impact. 30, 719-725.
- Ni, Y., Deng, N. & Kokot, S. 2010. A simple kinetic spectrophotometric method for simultaneous determination of tetracyclines by use of chemometrics. *Analytical Methods*, 2, 1302-1309.
- Nicas, T. & Hancock, R. 1983. Pseudomonas aeruginosa outer membrane permeability: isolation of a porin protein F-deficient mutant. *Journal of bacteriology*, 153, 281-285.
- Nobrega, F. L., Vlot, M., De Jonge, P. A., Dreesens, L. L., Beaumont, H. J., Lavigne, R., Dutilh, B. E. & Brouns, S. J. 2018. Targeting mechanisms of tailed bacteriophages. *Nature Reviews Microbiology*, 16, 760-773.
- Nonhoff, C., Rottiers, S. & Struelens, M. J. 2005. Evaluation of the Vitek 2 system for identification and antimicrobial susceptibility testing of Staphylococcus spp. *Clin Microbiol Infect*, 11, 150-3.
- Nusbaum, A. G., Gil, J., Rippy, M. K., Warne, B., Valdes, J., Claro, A. & Davis, S. C. 2012. Effective method to remove wound bacteria: comparison of various debridement modalities in an in vivo porcine model. *Journal of Surgical Research*, 176, 701-707.
- O'flaherty, S., Ross, R., Meaney, W., Fitzgerald, G., Elbreki, M. & Coffey, A. 2005. Potential of the polyvalent anti-Staphylococcus bacteriophage K for control of antibiotic-resistant staphylococci from hospitals. *Applied and environmental microbiology*, **71**, 1836-1842.
- O'flaherty, S., Ross, R. P. & Coffey, A. 2009. Bacteriophage and their lysins for elimination of infectious bacteria. *FEMS Microbiol Rev*, 33, 801-19.
- O'flynn, G., Ross, R. P., Fitzgerald, G. F. & Coffey, A. 2004. Evaluation of a cocktail of three bacteriophages for biocontrol of Escherichia coli O157:H7. *Appl Environ Microbiol*, 70, 3417-24.
- O'toole, G., Kaplan, H. B. & Kolter, R. 2000. Biofilm formation as microbial development. *Annu Rev Microbiol*, 54, 49-79.
- O'toole, G. A. 2016. Classic Spotlight: Plate Counting You Can Count On. J Bacteriol, 198, 3127.
- O'toole, G. A. & Wong, G. C. 2016. Sensational biofilms: surface sensing in bacteria. *Curr Opin Microbiol,* 30, 139-46.
- Olszak, T., Zarnowiec, P., Kaca, W., Danis-Wlodarczyk, K., Augustyniak, D., Drevinek, P., De Soyza, A., Mcclean, S. & Drulis-Kawa, Z. 2015. In vitro and in vivo antibacterial activity of environmental bacteriophages against Pseudomonas aeruginosa strains from cystic fibrosis patients. *Appl Microbiol Biotechnol*, 99, 6021-33.
- Ooi, M. L., Drilling, A. J., Morales, S., Fong, S., Moraitis, S., Macias-Valle, L., Vreugde, S., Psaltis, A. J. & Wormald, P.-J. 2019. Safety and Tolerability of Bacteriophage Therapy for Chronic Rhinosinusitis Due to Staphylococcus aureus. *JAMA Otolaryngology–Head & Neck Surgery*.
- Oppenheim, A. B. & Adhya, S. L. 2007. A new look at bacteriophage λ genetic networks. *Journal of bacteriology*, 189, 298-304.

- Oppenheim, A. B., Kobiler, O., Stavans, J., Court, D. L. & Adhya, S. 2005. Switches in bacteriophage lambda development. *Annu. Rev. Genet.*, 39, 409-429.
- Orlandi, V. T., Bolognese, F., Chiodaroli, L., Tolker-Nielsen, T. & Barbieri, P. 2015. Pigments influence the tolerance of Pseudomonas aeruginosa PAO1 to photodynamically induced oxidative stress. *Microbiology*, 161, 2298-2309.
- Ortmann, A. C. & Suttle, C. A. 2005. High abundances of viruses in a deep-sea hydrothermal vent system indicates viral mediated microbial mortality. *Deep Sea Research Part I: Oceanographic Research Papers*, 52, 1515-1527.
- Otto-Karg, I., Jandl, S., Muller, T., Stirzel, B., Frosch, M., Hebestreit, H. & Abele-Horn, M. 2009. Validation of Vitek 2 nonfermenting gram-negative cards and Vitek 2 version 4.02 software for identification and antimicrobial susceptibility testing of nonfermenting gram-negative rods from patients with cystic fibrosis. *J Clin Microbiol*, 47, 3283-8.
- Ozkan, I., Akturk, E., Yeshenkulov, N., Atmaca, S., Rahmanov, N. & Atabay, H. I. 2016. Lytic Activity of Various Phage Cocktails on Multidrug-Resistant Bacteria. *Clin Invest Med*, 39, 27504.
- Pai, H., Kim, J.-W., Kim, J., Lee, J. H., Choe, K. W. & Gotoh, N. 2001. Carbapenem resistance mechanisms in Pseudomonas aeruginosa clinical isolates. *Antimicrobial agents and chemotherapy*, 45, 480-484.
- Paik, S. G., Fleischer, N. & Shin, S. I. 1980. Insulin-dependent diabetes mellitus induced by subdiabetogenic doses of streptozotocin: obligatory role of cell-mediated autoimmune processes. *Proc Natl Acad Sci U S A*, 77, 6129-33.
- Parasion, S., Kwiatek, M., Gryko, R., Mizak, L. & Malm, A. 2014. Bacteriophages as an alternative strategy for fighting biofilm development. *Pol J Microbiol*, 63, 137-45.
- Park, S. A., Raghunathan, V. K., Shah, N. M., Teixeira, L., Motta, M. J., Covert, J., Dubielzig, R., Schurr, M., Isseroff, R. R. & Abbott, N. L. 2014. PDGF-BB does not accelerate healing in diabetic mice with splinted skin wounds. *PloS one*, 9, e104447.
- Parkinson, C. M., O'brien, A., Albers, T. M., Simon, M. A., Clifford, C. B. & Pritchett-Corning, K. R. 2011. Diagnostic necropsy and selected tissue and sample collection in rats and mice. *J Vis Exp*.
- Pastagia, M., Euler, C., Chahales, P., Fuentes-Duculan, J., Krueger, J. G. & Fischetti, V. A. 2011. A novel chimeric lysin shows superiority to mupirocin for skin decolonization of methicillin-resistant and -sensitive Staphylococcus aureus strains. *Antimicrob Agents Chemother*, 55, 738-44.
- Pastar, I., Nusbaum, A. G., Gil, J., Patel, S. B., Chen, J., Valdes, J., Stojadinovic, O., Plano, L. R., Tomic-Canic, M. & Davis, S. C. 2013. Interactions of methicillin resistant Staphylococcus aureus USA300 and Pseudomonas aeruginosa in polymicrobial wound infection. *PloS one*, 8, e56846.
- Paul, V. D., Rajagopalan, S. S., Sundarrajan, S., George, S. E., Asrani, J. Y., Pillai, R., Chikkamadaiah, R., Durgaiah, M., Sriram, B. & Padmanabhan, S. 2011. A novel bacteriophage Tail-Associated Muralytic Enzyme (TAME) from Phage K and its development into a potent antistaphylococcal protein. *BMC Microbiol*, 11, 226.
- Payne, R. J., Phil, D. & Jansen, V. A. 2000. Phage therapy: the peculiar kinetics of self-replicating pharmaceuticals. *Clin Pharmacol Ther*, 68, 225-30.
- Pei, R. & Lamas-Samanamud, G. R. 2014. Inhibition of biofilm formation by T7 bacteriophages producing quorum-quenching enzymes. *Appl Environ Microbiol*, 80, 5340-8.
- Percival, S. L., Hill, K. E., Williams, D. W., Hooper, S. J., Thomas, D. W. & Costerton, J. W. 2012. A review of the scientific evidence for biofilms in wounds. *Wound Repair Regen*, 20, 647-57.

- Pereira, P. M., Veiga, H., Jorge, A. M. & Pinho, M. G. 2010. Fluorescent reporters for studies of cellular localization of proteins in Staphylococcus aureus. *Appl Environ Microbiol*, 76, 4346-53.
- Perez-Sancho, M., Vela, A. I., Horcajo, P., Ugarte-Ruiz, M., Dominguez, L., Fernandez-Garayzabal, J. F. & De La Fuente, R. 2018. Rapid differentiation of Staphylococcus aureus subspecies based on MALDI-TOF MS profiles. J Vet Diagn Invest, 30, 813-820.
- Pernet, E., Guillemot, L., Burgel, P.-R., Martin, C., Lambeau, G., Sermet-Gaudelus, I., Sands, D., Leduc, D., Morand, P. C. & Jeammet, L. 2014. Pseudomonas aeruginosa eradicates Staphylococcus aureus by manipulating the host immunity. *Nature communications*, 5, 5105.
- Petersen, P. J., Jacobus, N., Weiss, W., Sum, P. & Testa, R. 1999. In vitro and in vivo antibacterial activities of a novel glycylcycline, the 9-t-butylglycylamido derivative of minocycline (GAR-936). *Antimicrobial agents and chemotherapy*, 43, 738-744.
- Petrakis, I., Kyriopoulos, I. J., Ginis, A. & Athanasakis, K. 2017. Losing a foot versus losing a dollar; a systematic review of cost studies in diabetic foot complications. *Expert Rev Pharmacoecon Outcomes Res*, 17, 165-180.
- Pfaller, M. A., Diekema, D. J., Procop, G. W. & Wiederhold, N. P. 2014. Multicenter evaluation of the new Vitek 2 yeast susceptibility test using new CLSI clinical breakpoints for fluconazole. J Clin Microbiol, 52, 2126-30.
- Pfirman, K. S. & Haile, R. 2018. Intracardiac Abscess and Pacemaker Lead Infection Secondary to Hematogenous Dissemination of Methicillin-Sensitive Staphylococcus Aureus from a Prior Diabetic Foot Ulcer and Osteomyelitis. *Am J Case Rep,* 19, 224-228.
- Pickwell, K., Siersma, V., Kars, M., Apelqvist, J., Bakker, K., Edmonds, M., Holstein, P., Jirkovska, A., Jude,
 E., Mauricio, D., Piaggesi, A., Ragnarson Tennvall, G., Reike, H., Spraul, M., Uccioli, L., Urbancic,
 V., Van Acker, K., Van Baal, J. & Schaper, N. 2015. Predictors of lower-extremity amputation in patients with an infected diabetic foot ulcer. *Diabetes Care*, 38, 852-7.
- Pincus, N. B., Reckhow, J. D., Saleem, D., Jammeh, M. L., Datta, S. K. & Myles, I. A. 2015. Strain Specific Phage Treatment for Staphylococcus aureus Infection Is Influenced by Host Immunity and Site of Infection. *PLoS One*, 10, e0124280.
- Pires, D., Sillankorva, S., Faustino, A. & Azeredo, J. 2011. Use of newly isolated phages for control of Pseudomonas aeruginosa PAO1 and ATCC 10145 biofilms. *Res Microbiol*, 162, 798-806.
- Pirnay, J. P., Blasdel, B. G., Bretaudeau, L., Buckling, A., Chanishvili, N., Clark, J. R., Corte-Real, S., Debarbieux, L., Dublanchet, A., De Vos, D., Gabard, J., Garcia, M., Goderdzishvili, M., Gorski, A., Hardcastle, J., Huys, I., Kutter, E., Lavigne, R., Merabishvili, M., Olchawa, E., Parikka, K. J., Patey, O., Pouilot, F., Resch, G., Rohde, C., Scheres, J., Skurnik, M., Vaneechoutte, M., Van Parys, L., Verbeken, G., Zizi, M. & Van Den Eede, G. 2015. Quality and safety requirements for sustainable phage therapy products. *Pharm Res*, 32, 2173-9.
- Pirnay, J. P., De Vos, D., Verbeken, G., Merabishvili, M., Chanishvili, N., Vaneechoutte, M., Zizi, M., Laire, G., Lavigne, R., Huys, I., Van Den Mooter, G., Buckling, A., Debarbieux, L., Pouillot, F., Azeredo, J., Kutter, E., Dublanchet, A., Gorski, A. & Adamia, R. 2011. The phage therapy paradigm: pret-a-porter or sur-mesure? *Pharm Res*, 28, 934-7.
- Prevost, G., Couppie, P., Prevost, P., Gayet, S., Petiau, P., Cribier, B., Monteil, H. & Piemont, Y. 1995. Epidemiological data on Staphylococcus aureus strains producing synergohymenotropic toxins. *Journal of medical microbiology*, 42, 237-245.

- Pride, D. T., Salzman, J., Haynes, M., Rohwer, F., Davis-Long, C., White, R. A., 3rd, Loomer, P., Armitage, G. C. & Relman, D. A. 2012. Evidence of a robust resident bacteriophage population revealed through analysis of the human salivary virome. *Isme j*, 6, 915-26.
- Proctor, R. A., Von Eiff, C., Kahl, B. C., Becker, K., Mcnamara, P., Herrmann, M. & Peters, G. 2006. Small colony variants: a pathogenic form of bacteria that facilitates persistent and recurrent infections. *Nat Rev Microbiol*, *4*, 295-305.
- Prompers, L., Huijberts, M., Apelqvist, J., Jude, E., Piaggesi, A., Bakker, K., Edmonds, M., Holstein, P., Jirkovska, A., Mauricio, D., Ragnarson Tennvall, G., Reike, H., Spraul, M., Uccioli, L., Urbancic, V., Van Acker, K., Van Baal, J., Van Merode, F. & Schaper, N. 2007. High prevalence of ischaemia, infection and serious comorbidity in patients with diabetic foot disease in Europe. Baseline results from the Eurodiale study. *Diabetologia*, 50, 18-25.
- Qadri, S. M., Akhter, J. & Qadri, S. G. 1991. Latex agglutination and hemagglutination tests for the rapid identification of methicillin sensitive and methicillin resistant Staphylococcus aureus. J Hyg Epidemiol Microbiol Immunol, 35, 65-71.
- Qazi, S., Middleton, B., Muharram, S. H., Cockayne, A., Hill, P., O'shea, P., Chhabra, S. R., Camara, M. & Williams, P. 2006. N-acylhomoserine lactones antagonize virulence gene expression and quorum sensing in Staphylococcus aureus. *Infect Immun*, 74, 910-9.
- Qin, Z., Yang, L., Qu, D., Molin, S. & Tolker-Nielsen, T. 2009. Pseudomonas aeruginosa extracellular products inhibit staphylococcal growth, and disrupt established biofilms produced by Staphylococcus epidermidis. *Microbiology*, 155, 2148-56.
- Raahave, D., Friis-Moller, A., Bjerre-Jepsen, K., Thiis-Knudsen, J. & Rasmussen, L. B. 1986. The infective dose of aerobic and anaerobic bacteria in postoperative wound sepsis. *Arch Surg*, 121, 924-9.
- Radlinski, L., Rowe, S. E., Kartchner, L. B., Maile, R., Cairns, B. A., Vitko, N. P., Gode, C. J., Lachiewicz, A. M., Wolfgang, M. C. & Conlon, B. P. 2017. Pseudomonas aeruginosa exoproducts determine antibiotic efficacy against Staphylococcus aureus. *PLoS Biol*, 15, e2003981.
- Ramakant, P., Verma, A., Misra, R., Prasad, K., Chand, G., Mishra, A., Agarwal, G., Agarwal, A. & Mishra,
 S. 2011. Changing microbiological profile of pathogenic bacteria in diabetic foot infections: time for a rethink on which empirical therapy to choose? *Diabetologia*, 54, 58-64.
- Ramirez, K., Cazarez-Montoya, C., Lopez-Moreno, H. S. & Castro-Del Campo, N. 2018. Bacteriophage cocktail for biocontrol of Escherichia coli O157:H7: Stability and potential allergenicity study. *PLoS One*, 13, e0195023.
- Reveles, K. R., Duhon, B. M., Moore, R. J., Hand, E. O. & Howell, C. K. 2016. Epidemiology of methicillinresistant Staphylococcus aureus diabetic foot infections in a large academic hospital: implications for antimicrobial stewardship. *PloS one*, **11**, e0161658.
- Reyes, A., Semenkovich, N. P., Whiteson, K., Rohwer, F. & Gordon, J. I. 2012. Going viral: next-generation sequencing applied to phage populations in the human gut. *Nature Reviews Microbiology*, 10, 607-617.
- Rhoads, D., Wolcott, R., Kuskowski, M., Wolcott, B., Ward, L. & Sulakvelidze, A. 2009. Bacteriophage therapy of venous leg ulcers in humans: results of a phase I safety trial. *Journal of wound care*, 18, 237-243.
- Rice, J. B., Desai, U., Cummings, A. K. G., Birnbaum, H. G., Skornicki, M. & Parsons, N. B. 2013. Burden of Diabetic foot Ulcers for Medicare and Private Insurers. *Diabetes Care*, DC_132176.

- Rich, J. & Lee, J. C. 2005. The pathogenesis of Staphylococcus aureus infection in the diabetic NOD mouse. *Diabetes*, 54, 2904-2910.
- Richard, J.-L., Sotto, A. & Lavigne, J.-P. 2011. New insights in diabetic foot infection. *World journal of diabetes*, 2, 24.
- Richards, G. P. 2014. Bacteriophage remediation of bacterial pathogens in aquaculture: a review of the technology. *Bacteriophage*, 4, e975540.
- Rickard, A. H., Gilbert, P., High, N. J., Kolenbrander, P. E. & Handley, P. S. 2003. Bacterial coaggregation: an integral process in the development of multi-species biofilms. *Trends in microbiology*, **11**, 94-100.
- Riou, M., Carbonnelle, S., Avrain, L., Mesaros, N., Pirnay, J.-P., Bilocq, F., De Vos, D., Simon, A., Piérard, D. & Jacobs, F. 2010. In vivo development of antimicrobial resistance in Pseudomonas aeruginosa strains isolated from the lower respiratory tract of Intensive Care Unit patients with nosocomial pneumonia and receiving antipseudomonal therapy. *International journal of antimicrobial agents*, 36, 513-522.
- Robbins, J. M., Strauss, G., Aron, D., Long, J., Kuba, J. & Kaplan, Y. 2008. Mortality rates and diabetic foot ulcers: is it time to communicate mortality risk to patients with diabetic foot ulceration? *Journal of the American Podiatric Medical Association*, 98, 489-493.
- Robson, M. C. 1997. Wound infection. A failure of wound healing caused by an imbalance of bacteria. *Surg Clin North Am*, 77, 637-50.
- Robson, M. C. 1999. Award recipient address: Lessons gleaned from the sport of wound watching. *Wound Repair Regen*, 7, 2-6.
- Roca, I., Akova, M., Baquero, F., Carlet, J., Cavaleri, M., Coenen, S., Cohen, J., Findlay, D., Gyssens, I. & Heure, O. 2015. The global threat of antimicrobial resistance: science for intervention. New Microbes New Infect 6: 22–29.
- Rodríguez-Rojas, A., Mena, A., Martín, S., Borrell, N., Oliver, A. & Blazquez, J. 2009. Inactivation of the hmgA gene of Pseudomonas aeruginosa leads to pyomelanin hyperproduction, stress resistance and increased persistence in chronic lung infection. *Microbiology*, 155, 1050-1057.
- Rohde, H., Burandt, E. C., Siemssen, N., Frommelt, L., Burdelski, C., Wurster, S., Scherpe, S., Davies, A. P., Harris, L. G., Horstkotte, M. A., Knobloch, J. K., Ragunath, C., Kaplan, J. B. & Mack, D. 2007. Polysaccharide intercellular adhesin or protein factors in biofilm accumulation of Staphylococcus epidermidis and Staphylococcus aureus isolated from prosthetic hip and knee joint infections. *Biomaterials*, 28, 1711-20.
- Rohwer, F. & Edwards, R. 2002. The Phage Proteomic Tree: a genome-based taxonomy for phage. *Journal of bacteriology*, 184, 4529-4535.
- Ronan, E., Edjiu, N., Kroukamp, O., Wolfaardt, G. & Karshafian, R. 2016. USMB-induced synergistic enhancement of aminoglycoside antibiotics in biofilms. *Ultrasonics*, 69, 182-90.
- Rose, T., Verbeken, G., Vos, D. D., Merabishvili, M., Vaneechoutte, M., Lavigne, R., Jennes, S., Zizi, M. & Pirnay, J. P. 2014. Experimental phage therapy of burn wound infection: difficult first steps. *Int J Burns Trauma*, 4, 66-73.
- Russo, P., Iturria, I., Mohedano, M. L., Caggianiello, G., Rainieri, S., Fiocco, D., Angel Pardo, M., Lopez, P.
 & Spano, G. 2015. Zebrafish gut colonization by mCherry-labelled lactic acid bacteria. *Appl Microbiol Biotechnol*, 99, 3479-90.

- Ryan, E. M., Gorman, S. P., Donnelly, R. F. & Gilmore, B. F. 2011. Recent advances in bacteriophage therapy: how delivery routes, formulation, concentration and timing influence the success of phage therapy. *Journal of Pharmacy and Pharmacology*, 63, 1253-1264.
- Sader, H. S., Fritsche, T. R. & Jones, R. N. 2006. Accuracy of three automated systems (MicroScan WalkAway, VITEK, and VITEK 2) for susceptibility testing of Pseudomonas aeruginosa against five broad-spectrum beta-lactam agents. *Journal of clinical microbiology*, 44, 1101-1104.
- Saeedi, P., Petersohn, I., Salpea, P., Malanda, B., Karuranga, S., Unwin, N., Colagiuri, S., Guariguata, L., Motala, A. A. & Ogurtsova, K. 2019. Global and regional diabetes prevalence estimates for 2019 and projections for 2030 and 2045: Results from the International Diabetes Federation Diabetes Atlas. *Diabetes research and clinical practice*, 157, 107843.
- Saegeman, V., Huynen, P., Colaert, J., Melin, P. & Verhaegen, J. 2005. Susceptibility testing of Pseudomonas aeruginosa by the Vitek 2 system: a comparison with Etest results. *Acta Clin Belg*, 60, 3-9.
- Sagona, A. P., Grigonyte, A. M., Macdonald, P. R. & Jaramillo, A. 2016. Genetically modified bacteriophages. *Integr Biol (Camb)*, 8, 465-74.
- Saltoglu, N., Ergonul, O., Tulek, N., Yemisen, M., Kadanali, A., Karagoz, G., Batirel, A., Ak, O., Sonmezer, C., Eraksoy, H., Cagatay, A., Surme, S., Nemli, S. A., Demirdal, T., Coskun, O., Ozturk, D., Ceran, N., Pehlivanoglu, F., Sengoz, G., Aslan, T., Akkoyunlu, Y., Oncul, O., Ay, H., Mulazimoglu, L., Erturk, B., Yilmaz, F., Yoruk, G., Uzun, N., Simsek, F., Yildirmak, T., Yasar, K. K., Sonmezoglu, M., Kucukardali, Y., Tuna, N., Karabay, O., Ozgunes, N. & Sargin, F. 2018. Influence of multidrug resistant organisms on the outcome of diabetic foot infection. *Int J Infect Dis*, 70, 10-14.
- Sanders, C. C., Peyret, M., Moland, E. S., Shubert, C., Thomson, K. S., Boeufgras, J.-M. & Sanders, W. E. 2000. Ability of the VITEK 2 Advanced Expert System To Identify β-Lactam Phenotypes in Isolates of Enterobacteriaceae andPseudomonas aeruginosa. *Journal of clinical microbiology*, 38, 570-574.
- Saussereau, E., Vachier, I., Chiron, R., Godbert, B., Sermet, I., Dufour, N., Pirnay, J. P., De Vos, D., Carrié,
 F. & Molinari, N. 2014. Effectiveness of bacteriophages in the sputum of cystic fibrosis patients. *Clinical Microbiology and Infection*, 20, 0983-0990.
- Scalise, A., Bianchi, A., Tartaglione, C., Bolletta, E., Pierangeli, M., Torresetti, M., Marazzi, M. & Di Benedetto, G. 2015. Microenvironment and microbiology of skin wounds: the role of bacterial biofilms and related factors. *Semin Vasc Surg*, 28, 151-9.
- Schaumann, R., Knoop, N., Genzel, G. H., Losensky, K., Rosenkranz, C., Stingu, C. S., Schellenberger, W., Rodloff, A. C. & Eschrich, K. 2012. A step towards the discrimination of beta-lactamaseproducing clinical isolates of Enterobacteriaceae and Pseudomonas aeruginosa by MALDI-TOF mass spectrometry. *Med Sci Monit*, 18, Mt71-7.
- Schechter-Perkins, E. M., Mitchell, P. M., Murray, K. A., Rubin-Smith, J. E., Weir, S. & Gupta, K. 2011. Prevalence and predictors of nasal and extranasal staphylococcal colonization in patients presenting to the emergency department. *Ann Emerg Med*, 57, 492-9.
- Schmerer, M., Molineux, I. J. & Bull, J. J. 2014. Synergy as a rationale for phage therapy using phage cocktails. *PeerJ*, 2, e590.
- Schooley, R. T., Biswas, B., Gill, J. J., Hernandez-Morales, A., Lancaster, J., Lessor, L., Barr, J. J., Reed, S. L.,
 Rohwer, F., Benler, S., Segall, A. M., Taplitz, R., Smith, D. M., Kerr, K., Kumaraswamy, M., Nizet,
 V., Lin, L., Mccauley, M. D., Strathdee, S. A., Benson, C. A., Pope, R. K., Leroux, B. M., Picel, A. C.,

Mateczun, A. J., Cilwa, K. E., Regeimbal, J. M., Estrella, L. A., Wolfe, D. M., Henry, M. S., Quinones, J., Salka, S., Bishop-Lilly, K. A., Young, R. & Hamilton, T. 2017. Development and Use of Personalized Bacteriophage-Based Therapeutic Cocktails To Treat a Patient with a Disseminated Resistant Acinetobacter baumannii Infection. *Antimicrob Agents Chemother*, 61.

- Schumann, P. & Maier, T. 2014. MALDI-TOF mass spectrometry applied to classification and identification of bacteria. *Methods in Microbiology*. Elsevier.
- Sekhar, M. S., M, K. U., Rodrigues, G. S., Vyas, N. & Mukhopadhyay, C. 2018. Antimicrobial susceptibility pattern of aerobes in diabetic foot ulcers in a South-Indian tertiary care hospital. *Foot (Edinb)*, 37, 95-100.
- Sekhar, S., Vyas, N., Unnikrishnan, M., Rodrigues, G. & Mukhopadhyay, C. 2014. Antimicrobial susceptibility pattern in diabetic foot ulcer: a pilot study. *Ann Med Health Sci Res*, 4, 742-5.
- Seo, B. J., Song, E. T., Lee, K., Kim, J. W., Jeong, C. G., Moon, S. H., Son, J. S., Kang, S. H., Cho, H. S., Jung, B. Y. & Kim, W. I. 2018. Evaluation of the broad-spectrum lytic capability of bacteriophage cocktails against various Salmonella serovars and their effects on weaned pigs infected with Salmonella Typhimurium. J Vet Med Sci, 80, 851-860.
- Serra, R., Grande, R., Butrico, L., Rossi, A., Settimio, U. F., Caroleo, B., Amato, B., Gallelli, L. & De Franciscis, S. 2015. Chronic wound infections: the role of Pseudomonas aeruginosa and Staphylococcus aureus. *Expert review of anti-infective therapy*, 13, 605-613.
- Seth, A. K., Geringer, M. R., Nguyen, K. T., Agnew, S. P., Dumanian, Z., Galiano, R. D., Leung, K. P., Mustoe, T. A. & Hong, S. J. 2013. Bacteriophage therapy for Staphylococcus aureus biofilm– infected wounds: a new approach to chronic wound care. *Plastic and reconstructive surgery*, 131, 225-234.
- Shah, A., Wollak, C. & Shah, J. 2013. Wound measurement techniques: comparing the use of ruler method, 2D imaging and 3D scanner. *Journal of the American College of Clinical Wound Specialists*, 5, 52-57.
- Sharma, M. 2013. Lytic bacteriophages: Potential interventions against enteric bacterial pathogens on produce. *Bacteriophage*, 3, e25518.
- Shettigar, K., Jain, S., Bhat, D. V., Acharya, R., Ramachandra, L., Satyamoorthy, K. & Murali, T. S. 2016. Virulence determinants in clinical Staphylococcus aureus from monomicrobial and polymicrobial infections of diabetic foot ulcers. *J Med Microbiol*, 65, 1392-1404.
- Shivshetty, N., Hosamani, R., Ahmed, L., Oli, A. K., Sannauallah, S., Sharanbassappa, S., Patil, S. A. & Kelmani, C. R. 2014. Experimental protection of diabetic mice against Lethal P. aeruginosa infection by bacteriophage. *Biomed Res Int*, 2014, 793242.
- Sillankorva, S., Neubauer, P. & Azeredo, J. 2010. Phage control of dual species biofilms of Pseudomonas fluorescens and Staphylococcus lentus. *Biofouling*, 26, 567-75.
- Sivanmaliappan, T. S. & Sevanan, M. 2011. Antimicrobial susceptibility patterns of Pseudomonas aeruginosa from diabetes patients with foot ulcers. *International journal of microbiology*, 2011.
- Smole, S. C., Aronson, E., Durbin, A., Brecher, S. M. & Arbeit, R. D. 1998. Sensitivity and specificity of an improved rapid latex agglutination test for identification of methicillin-sensitive and-resistant Staphylococcus aureus isolates. *Journal of clinical microbiology*, 36, 1109-1112.
- Sneath, P. H., Mair, N. S., Sharpe, M. E. & Holt, J. G. 1986. Bergey's manual of systematic bacteriology. Volume 2, Williams & Wilkins.
- Soge, O. O., No, D., Michael, K. E., Dankoff, J., Lane, J., Vogel, K., Smedley, J. & Roberts, M. C. 2016. Transmission of MDR MRSA between primates, their environment and personnel at a United States primate centre. *J Antimicrob Chemother*, **71**, 2798-803.
- Sotto, A., Lina, G., Richard, J. L., Combescure, C., Bourg, G., Vidal, L., Jourdan, N., Etienne, J. & Lavigne, J.
 P. 2008. Virulence potential of Staphylococcus aureus strains isolated from diabetic foot ulcers: a new paradigm. *Diabetes Care*, 31, 2318-24.
- Sowole, L., Ming, D. K. & Davies, F. 2018. Multidrug-resistant bacteria. *Br J Hosp Med (Lond),* 79, C66-c69.
- Spanu, T., De Carolis, E., Fiori, B., Sanguinetti, M., D'inzeo, T., Fadda, G. & Posteraro, B. 2011. Evaluation of matrix-assisted laser desorption ionization-time-of-flight mass spectrometry in comparison to rpoB gene sequencing for species identification of bloodstream infection staphylococcal isolates. *Clin Microbiol Infect*, 17, 44-9.
- Speck, P. & Smithyman, A. 2016. Safety and efficacy of phage therapy via the intravenous route. *FEMS Microbiol Lett*, 363.
- Spoering, A. L. & Lewis, K. 2001. Biofilms and planktonic cells of Pseudomonas aeruginosa have similar resistance to killing by antimicrobials. *Journal of bacteriology*, 183, 6746-6751.
- Steenackers, H. P., Parijs, I., Foster, K. R. & Vanderleyden, J. 2016. Experimental evolution in biofilm populations. *FEMS Microbiol Rev*, 40, 373-97.
- Steff, A. M., Fortin, M., Arguin, C. & Hugo, P. 2001. Detection of a decrease in green fluorescent protein fluorescence for the monitoring of cell death: an assay amenable to high-throughput screening technologies. *Cytometry*, 45, 237-43.
- Stehling, E. G., Silveira, W. D. D. & Leite, D. D. S. 2008. Study of biological characteristics of Pseudomonas aeruginosa strains isolated from patients with cystic fibrosis and from patients with extra-pulmonary infections. *Brazilian Journal of Infectious Diseases*, **12**, 86-88.
- Stern, A. & Sorek, R. 2011. The phage-host arms race: shaping the evolution of microbes. *Bioessays*, 33, 43-51.
- Stewart, P. S. 2002. Mechanisms of antibiotic resistance in bacterial biofilms. *Int J Med Microbiol,* 292, 107-13.
- Stratton, C. W. 2003. Dead bugs don't mutate: susceptibility issues in the emergence of bacterial resistance. *Emerg Infect Dis*, 9, 10-6.
- Strowski, M. Z., Li, Z., Szalkowski, D., Shen, X., Guan, X. M., Juttner, S., Moller, D. E. & Zhang, B. B. 2004. Small-molecule insulin mimetic reduces hyperglycemia and obesity in a nongenetic mouse model of type 2 diabetes. *Endocrinology*, 145, 5259-68.
- Stryjewski, M. E. & Corey, G. R. 2014. Methicillin-resistant Staphylococcus aureus: an evolving pathogen. *Clinical Infectious Diseases*, 58, S10-S19.
- Sulakvelidze, A. 2013. Using lytic bacteriophages to eliminate or significantly reduce contamination of food by foodborne bacterial pathogens. *J Sci Food Agric*, 93, 3137-46.
- Sulakvelidze, A., Alavidze, Z. & Morris, J. G., Jr. 2001. Bacteriophage therapy. *Antimicrob Agents Chemother*, 45, 649-59.
- Sulakvelidze, A., And Kutter, E. 2005. Bacteriophage therapy in humans., Boca Raton, FL., CRC Press.
- Summers, W. C. 2012. The strange history of phage therapy. *Bacteriophage*, 2, 130-133.
- Sutherland, I. W., Hughes, K. A., Skillman, L. C. & Tait, K. 2004. The interaction of phage and biofilms. *FEMS microbiology letters*, 232, 1-6.

- Synnott, A. J., Kuang, Y., Kurimoto, M., Yamamichi, K., Iwano, H. & Tanji, Y. 2009. Isolation from sewage influent and characterization of novel Staphylococcus aureus bacteriophages with wide host ranges and potent lytic capabilities. *Appl Environ Microbiol*, **75**, 4483-90.
- Szczuka, E., Urbanska, K., Pietryka, M. & Kaznowski, A. 2013. Biofilm density and detection of biofilmproducing genes in methicillin-resistant Staphylococcus aureus strains. *Folia Microbiol (Praha)*, 58, 47-52.
- Szkudelski, T. 2001. The mechanism of alloxan and streptozotocin action in B cells of the rat pancreas. *Physiol Res,* 50, 537-46.
- Tait, K., Skillman, L. C. & Sutherland, I. W. 2002. The efficacy of bacteriophage as a method of biofilm eradication. *Biofouling*, 18, 305-311.
- Takigawa, M., Masutomi, H., Kishimoto, Y., Shimazaki, Y., Hamano, Y., Kondo, Y., Arai, T., Lee, J., Ishii, T.
 & Mori, Y. 2017. Time-Dependent Alterations of Vancomycin-Induced Nephrotoxicity in Mice. Biological and Pharmaceutical Bulletin, 40, 975-983.
- Tang, J., Chen, J., Li, H., Zeng, P. & Li, J. 2013. Characterization of adhesin genes, staphylococcal nuclease, hemolysis, and biofilm formation among Staphylococcus aureus strains isolated from different sources. *Foodborne pathogens and disease*, 10, 757-763.
- Tanji, Y., Shimada, T., Fukudomi, H., Miyanaga, K., Nakai, Y. & Unno, H. 2005. Therapeutic use of phage cocktail for controlling Escherichia coli O157: H7 in gastrointestinal tract of mice. *Journal of bioscience and bioengineering*, 100, 280-287.
- Tawil, N., Sacher, E., Mandeville, R. & Meunier, M. 2014. Bacteriophages: biosensing tools for multi-drug resistant pathogens. *Analyst*, 139, 1224-36.
- Tay, W. H., Chong, K. K. & Kline, K. A. 2016. Polymicrobial-Host Interactions during Infection. *J Mol Biol*, 428, 3355-71.
- Tentolouris, N., Petrikkos, G., Vallianou, N., Zachos, C., Daikos, G. L., Tsapogas, P., Markou, G. & Katsilambros, N. 2006. Prevalence of methicillin-resistant Staphylococcus aureus in infected and uninfected diabetic foot ulcers. *Clin Microbiol Infect*, 12, 186-9.
- Teplitski, M. & Ritchie, K. 2009. How feasible is the biological control of coral diseases? *Trends Ecol Evol,* 24, 378-85.
- Thein, Z., Samaranayake, Y. & Samaranayake, L. 2007. Dietary sugars, serum and the biocide chlorhexidine digluconate modify the population and structural dynamics of mixed Candida albicans and Escherichia coli biofilms. *Apmis*, 115, 1241-1251.
- Thomas, N., Dong, D., Richter, K., Ramezanpour, M., Vreugde, S., Thierry, B., Wormald, P.-J. & Prestidge, C. A. 2015. Quatsomes for the treatment of Staphylococcus aureus biofilm. *Journal of Materials Chemistry B*, 3, 2770-2777.
- Thorisdottir, R. L., Parkner, T., Chen, J. W., Ejskjær, N., Christiansen, J. S. J. B., Pharmacology, C. & Toxicology 2009. A comparison of pharmacokinetics and pharmacodynamics of biphasic insulin aspart 30, 50, 70 and pure insulin aspart: a randomized, quadruple crossover study. 104, 216-221.
- Thurber, E. G., Kisuule, F., Humbyrd, C. & Townsend, J. 2017. Inpatient Management of Diabetic Foot Infections: A Review of the Guidelines for Hospitalists. *J Hosp Med*, 12, 994-1000.
- Tong, S. Y., Davis, J. S., Eichenberger, E., Holland, T. L. & Fowler, V. G. 2015. Staphylococcus aureus infections: epidemiology, pathophysiology, clinical manifestations, and management. *Clinical microbiology reviews*, 28, 603-661.

- Tracz, D. M., Tober, A. D., Antonation, K. S. & Corbett, C. R. 2018. MALDI-TOF mass spectrometry and high-consequence bacteria: safety and stability of biothreat bacterial sample testing in clinical diagnostic laboratories. *Journal of medical microbiology*.
- Trivedi, U., Parameswaran, S., Armstrong, A., Burgueno-Vega, D., Griswold, J., Dissanaike, S. & Rumbaugh, K. P. 2014. Prevalence of multiple antibiotic resistant infections in diabetic versus nondiabetic wounds. *Journal of pathogens*, 2014.
- Tsonos, J., Vandenheuvel, D., Briers, Y., De Greve, H., Hernalsteens, J.-P. & Lavigne, R. 2014. Hurdles in bacteriophage therapy: deconstructing the parameters. *Veterinary microbiology*, 171, 460-469.
- Turan, Y., Ertugrul, B. M., Lipsky, B. A. & Bayraktar, K. 2015. Does physical therapy and rehabilitation improve outcomes for diabetic foot ulcers? *World journal of experimental medicine*, *5*, 130.
- Turner, C. T., Mcinnes, S. J., Melville, E., Cowin, A. J. & Voelcker, N. H. 2017. Delivery of Flightless I Neutralizing Antibody from Porous Silicon Nanoparticles Improves Wound Healing in Diabetic Mice. Adv Healthc Mater, 6.
- Twort, F. W. 1915. An investigation on the nature of ultra-microscopic viruses. *The Lancet,* 186, 1241-1243.
- Uçkay, I., Aragón-Sánchez, J., Lew, D. & Lipsky, B. A. 2015. Diabetic foot infections: what have we learned in the last 30 years? *International Journal of Infectious Diseases*, 40, 81-91.
- Uçkay, I., Gariani, K., Dubois-Ferrière, V., Suvà, D. & Lipsky, B. A. 2016. Diabetic foot infections: recent literature and cornerstones of management. *Current opinion in infectious diseases*, 29, 145-152.
- Van Belkum, A., Durand, G., Peyret, M., Chatellier, S., Zambardi, G., Schrenzel, J., Shortridge, D., Engelhardt, A. & Dunne, W. M., Jr. 2013. Rapid clinical bacteriology and its future impact. Ann Lab Med, 33, 14-27.
- Van Belleghem, J. D., Clement, F., Merabishvili, M., Lavigne, R. & Vaneechoutte, M. 2017. Pro-and antiinflammatory responses of peripheral blood mononuclear cells induced by Staphylococcus aureus and Pseudomonas aeruginosa phages. *Scientific reports*, 7, 1-13.
- Van Voorthuizen, E., Ashbolt, N. & Schäfer, A. 2001. Role of hydrophobic and electrostatic interactions for initial enteric virus retention by MF membranes. *Journal of Membrane Science*, 194, 69-79.
- Vandenesch, F., Lebeau, C., Bes, M., Mcdevitt, D., Greenland, T., Novick, R. & Etienne, J. 1994. Coagulase deficiency in clinical isolates of Staphylococcus aureus involves both transcriptional and post-transcriptional defects. *Journal of medical microbiology*, 40, 344-349.
- Vander Elst, N. & Meyer, E. 2018. Potential therapeutic application of bacteriophages and phage-derived endolysins as alternative treatment of bovine mastitis. *Vlaams Diergeneeskundig Tijdschrift*, 87, 181-187.
- Velander, P., Theopold, C., Hirsch, T., Bleiziffer, O., Zuhaili, B., Fossum, M., Hoeller, D., Gheerardyn, R., Chen, M., Visovatti, S., Svensson, H., Yao, F. & Eriksson, E. 2008. Impaired wound healing in an acute diabetic pig model and the effects of local hyperglycemia. *Wound Repair Regen*, 16, 288-93.
- Vickerman, M. M., Mansfield, J. M., Zhu, M., Walters, K. S. & Banas, J. A. 2015. Codon-optimized fluorescent mTFP and mCherry for microscopic visualization and genetic counterselection of streptococci and enterococci. *J Microbiol Methods*, 116, 15-22.
- Vidlak, D. & Kielian, T. 2016. Infectious Dose Dictates the Host Response during Staphylococcus aureus Orthopedic-Implant Biofilm Infection. *Infect Immun*, 84, 1957-1965.

- Vieira, A., Silva, Y. J., Cunha, A., Gomes, N. C., Ackermann, H. W. & Almeida, A. 2012. Phage therapy to control multidrug-resistant Pseudomonas aeruginosa skin infections: in vitro and ex vivo experiments. *Eur J Clin Microbiol Infect Dis*, 31, 3241-9.
- Viquez-Molina, G., Aragon-Sanchez, J., Perez-Corrales, C., Murillo-Vargas, C., Lopez-Valverde, M. E. & Lipsky, B. A. 2018. Virulence Factor Genes in Staphylococcus aureus Isolated From Diabetic Foot Soft Tissue and Bone Infections. *Int J Low Extrem Wounds*, 17, 36-41.
- Visca, P., Imperi, F. & Lamont, I. L. 2007. Pyoverdine siderophores: from biogenesis to biosignificance. *Trends in microbiology*, **15**, 22-30.
- Vu, B. G., Stach, C. S., Salgado-Pabón, W., Diekema, D. J., Gardner, S. E. & Schlievert, P. M. J. T. J. O. I. D. 2014. Superantigens of Staphylococcus aureus from patients with diabetic foot ulcers. 210, 1920-1927.
- Wang, T., Feng, Y., Sun, H., Zhang, L., Hao, L., Shi, C., Wang, J., Li, R., Ran, X., Su, Y. & Zou, Z. 2012. miR-21 regulates skin wound healing by targeting multiple aspects of the healing process. *Am J Pathol*, 181, 1911-20.
- Webb, J. S., Barratt, S. R., Sabev, H., Nixon, M., Eastwood, I. M., Greenhalgh, M., Handley, P. S. & Robson, G. D. 2001. Green fluorescent protein as a novel indicator of antimicrobial susceptibility in Aureobasidium pullulans. *Appl Environ Microbiol*, 67, 5614-20.
- Weber, R. E., Layer, F., Klare, I., Werner, G. & Strommenger, B. 2017. Comparative evaluation of VITEK(R)
 2 and three commercial gradient strip assays for daptomycin susceptibility testing of Staphylococcus aureus. J Antimicrob Chemother, 72, 3059-3062.
- Wertheim, H. F., Melles, D. C., Vos, M. C., Van Leeuwen, W., Van Belkum, A., Verbrugh, H. A. & Nouwen, J. L. 2005. The role of nasal carriage in Staphylococcus aureus infections. *Lancet Infect Dis,* 5, 751-62.
- Westwater, C., Kasman, L. M., Schofield, D. A., Werner, P. A., Dolan, J. W., Schmidt, M. G. & Norris, J. S. 2003. Use of genetically engineered phage to deliver antimicrobial agents to bacteria: an alternative therapy for treatment of bacterial infections. *Antimicrobial agents and chemotherapy*, 47, 1301-1307.
- Weyrich, L. S., Dixit, S., Farrer, A. G., Cooper, A. J. & Cooper, A. J. 2015. The skin microbiome: Associations between altered microbial communities and disease. *Australas J Dermatol*, 56, 268-74.
- Willner, D., Furlan, M., Haynes, M., Schmieder, R., Angly, F. E., Silva, J., Tammadoni, S., Nosrat, B., Conrad, D. & Rohwer, F. 2009. Metagenomic analysis of respiratory tract DNA viral communities in cystic fibrosis and non-cystic fibrosis individuals. *PLoS One*, 4, e7370.
- Wills, Q. F., Kerrigan, C. & Soothill, J. S. 2005. Experimental bacteriophage protection against Staphylococcus aureus abscesses in a rabbit model. *Antimicrob Agents Chemother*, 49, 1220-1.
- Wilson, L. G. 1987. The early recognition of streptococci as causes of disease. *Medical history*, 31, 403-414.
- Wittebole, X., De Roock, S. & Opal, S. M. 2014. A historical overview of bacteriophage therapy as an alternative to antibiotics for the treatment of bacterial pathogens. *Virulence*, *5*, 226-35.
- Wolcott, R., Costerton, J. W., Raoult, D. & Cutler, S. J. 2013. The polymicrobial nature of biofilm infection. *Clin Microbiol Infect*, 19, 107-12.
- Wold, J. S. & Turnipseed, S. A. 1981. Toxicology of vancomycin in laboratory animals. *Reviews of infectious diseases*, 3, S224-S229.

- Wong, V. W., Martindale, R. G., Longaker, M. T. & Gurtner, G. C. 2013. From germ theory to germ therapy: skin microbiota, chronic wounds, and probiotics. *Plastic and reconstructive surgery*, 132, 854e-861e.
- Wood, S., Kirkham, J., Marsh, P., Shore, R., Nattress, B. & Robinson, C. 2000. Architecture of intact natural human plaque biofilms studied by confocal laser scanning microscopy. *Journal of Dental Research*, 79, 21-27.
- Woods, P. W., Haynes, Z. M., Mina, E. G. & Marques, C. N. H. 2018. Maintenance of S. aureus in Coculture With P. aeruginosa While Growing as Biofilms. *Front Microbiol*, 9, 3291.
- Wu, W. X., Liu, D., Wang, Y. W., Wang, C., Yang, C., Liu, X. Z., Mai, L. F., Ren, M. & Yan, L. 2017. Empirical Antibiotic Treatment in Diabetic Foot Infection: A Study Focusing on the Culture and Antibiotic Sensitivity in a Population From Southern China. Int J Low Extrem Wounds, 16, 173-182.
- Wuthiekanun, V., Dance, D. A., Wattanagoon, Y., Supputtamongkol, Y., Chaowagul, W. & White, N. J. 1990. The use of selective media for the isolation of Pseudomonas pseudomallei in clinical practice. J Med Microbiol, 33, 121-6.
- Wylie, K. M., Mihindukulasuriya, K. A., Zhou, Y., Sodergren, E., Storch, G. A. & Weinstock, G. M. 2014. Metagenomic analysis of double-stranded DNA viruses in healthy adults. *BMC Biol*, 12, 71.
- Yang, L., Liu, Y., Markussen, T., Hoiby, N., Tolker-Nielsen, T. & Molin, S. 2011. Pattern differentiation in co-culture biofilms formed by Staphylococcus aureus and Pseudomonas aeruginosa. FEMS Immunol Med Microbiol, 62, 339-47.
- Yano, H., Kinoshita, M., Fujino, K., Nakashima, M., Yamamoto, Y., Miyazaki, H., Hamada, K., Ono, S., Iwaya, K., Saitoh, D., Seki, S. & Tanaka, Y. 2012. Insulin treatment directly restores neutrophil phagocytosis and bactericidal activity in diabetic mice and thereby improves surgical site Staphylococcus aureus infection. *Infect Immun*, 80, 4409-16.
- Yosef, I., Kiro, R., Molshanski-Mor, S., Edgar, R. & Qimron, U. 2014. Different approaches for using bacteriophages against antibiotic-resistant bacteria. *Bacteriophage*, 4, e28491.
- Young, R. 1992a. Bacteriophage lysis: mechanism and regulation. *Microbiological reviews*, 56, 430-481.
- Young, R. 1992b. Bacteriophage lysis: mechanism and regulation. *Microbiology and Molecular Biology Reviews*, 56, 430-481.
- Young, R. 2014. Phage lysis: three steps, three choices, one outcome. *Journal of microbiology*, 52, 243-258.
- Young, R., Wang, N. & Roof, W. D. 2000. Phages will out: strategies of host cell lysis. *Trends in microbiology*, 8, 120-128.
- Zaine, N. H., Burns, J., Vicaretti, M., Fletcher, J. P., Begg, L. & Hitos, K. 2014. Characteristics of diabetic foot ulcers in Western Sydney, Australia. *Journal of foot and ankle research*, **7**, 39.
- Zetola, N., Francis, J. S., Nuermberger, E. L. & Bishai, W. R. 2005. Community-acquired meticillinresistant Staphylococcus aureus: an emerging threat. *Lancet Infect Dis*, 5, 275-86.
- Zhang, G., Zhao, Y., Paramasivan, S., Richter, K., Morales, S., Wormald, P. J. & Vreugde, S. Bacteriophage effectively kills multidrug resistant Staphylococcus aureus clinical isolates from chronic rhinosinusitis patients. International forum of allergy & rhinology, 2018. Wiley Online Library, 406-414.
- Zhang, J., Chu, Y., Wang, P., Ji, X., Li, X., Wang, C. & Peng, Y. 2014a. Clinical outcomes of multidrug resistant pseudomonas aeruginosa infection and the relationship with type III secretion system in patients with diabetic foot. *Int J Low Extrem Wounds*, 13, 205-10.

- Zhang, J., Ormala-Odegrip, A. M., Mappes, J. & Laakso, J. 2014b. Top-down effects of a lytic bacteriophage and protozoa on bacteria in aqueous and biofilm phases. *Ecol Evol*, *4*, 4444-53.
- Zhao, G., Usui, M. L., Lippman, S. I., James, G. A., Stewart, P. S., Fleckman, P. & Olerud, J. E. 2013. Biofilms and Inflammation in Chronic Wounds. *Adv Wound Care (New Rochelle)*, **2**, 389-399.
- Zheng, Z., Tharmalingam, N., Liu, Q., Jayamani, E., Kim, W., Fuchs, B. B., Zhang, R., Vilcinskas, A. & Mylonakis, E. 2017. Synergistic efficacy of Aedes aegypti antimicrobial peptide cecropin A2 and tetracycline against Pseudomonas aeruginosa. *Antimicrobial agents and chemotherapy*, 61, e00686-17.